Some basic topics in topology and set theory

Stephen Semmes Rice University

Preface

These informal notes are intended to complement more detailed texts, some of which are mentioned in the bibliography. The reader is expected to be familiar with some basic notions in analysis, including metric spaces and countability. The brief introduction in [174] may be of interest in this regard.

Some references concerning related topics are mentioned in the bibliography too. This includes more involved areas connected to topology and set theory, as well as more on some of the sorts of questions discussed here.

In particular, some very nice surveys may be found in [20, 21, 27, 32, 41, 42, 44, 45, 89, 90, 134, 169, 171, 203, 238, 278, 312, 317, 321, 322, 340, 341, 366, 376, 377, 383, 395, 413, 440, 441, 443, 447, 448, 449, 450].

Some additional expositions that one may find to be quite informative include [15, 16, 23, 53, 55, 64, 65, 68, 73, 74, 136, 143, 149, 151, 161, 162, 163, 165, 219, 233, 245, 282, 339, 351, 407, 410, 415, 426].

Some more articles with some helpful background material are [6, 7, 12, 14, 22, 48, 51, 54, 56, 57, 58, 59, 60, 61, 62, 80, 84, 113, 115, 116, 122, 123, 144, 145, 155, 158, 159, 195, 206, 212, 217, 239, 246, 253, 279, 280, 288, 292, 293, 300, 305, 324, 327, 328, 336, 388, 409, 420, 435].

Of course, there are quite a few other papers in the bibliography related to topology and set theory. See [81, 82, 138, 229, 308, 337, 453, 454] for some more ways in which topological notions or properties may be used.

Although various aspects of history may be found in a number of the references, one may be particularly interested in the collections [24, 25, 26, 218].

Some interesting perspectives may be found in [4, 5, 63, 70, 83, 131, 137, 141, 142, 150, 194, 214, 386, 417, 421, 433, 438].

One may also enjoy the contributions [29, 104, 125, 140, 157, 164, 170, 172, 177, 196, 197, 198, 223, 224, 234, 259, 284, 285, 298, 332, 333, 334, 335, 343, 344, 345, 347, 348, 356, 371, 374, 397, 398, 402, 419, 437, 439, 458, 460] to the "Miscellanea" section of the American Mathematical Monthly, for instance. Some suggestions for questions to ask at the end of a lecture in mathematics may be found in [289].

I would like to dedicate these notes to the memory of Juha Heinonen, who was also interested in various connections between analysis, geometry, and topology. This is related to the notions of *Lipschitz* and *bilipschitz* mappings, as in [286, 287], and of *quasisymmetric embeddings*, as in [430, 432]. Some additional related notions may be found in [34]. Of course, there are many other types of

connections between these areas as well, some aspects of which may be found in some of the other references included here.

The reader may notice that there are a number of references to articles in the American Mathematical Monthly and Mathematics Magazine in the bibliography. These journals aim to have papers that are accessible to rather wide mathematical audiences, and in particular hopefully fit well with these notes. These journals also happen to have a lot of papers on topics related to those discussed here. However, there are surely many more papers with similar qualities that have not been listed here, in other publications as well.

It is well known that there are many connections between abstract algebra and each of topology and set theory. Although we do not really get into this here, readers familiar with some abstract algebra can find many related topics in the bibliography. A brief introduction to groups, rings, and fields may be found in [18]. Note that one may be interested in algebraic structures related to topological spaces, as in *algebraic topology*, as well as topological structures on algebraic objects, as in *topological algebra*.

Contents

1	Som		c notions in topology	1
	1.1	Some	definitions and examples	1
		1.1.1	Some basic examples	1
		1.1.2	Some basic separation conditions	2
	1.2	Semin	netrics and intervals	3
		1.2.1	Open balls and open sets	3
		1.2.2	Intervals in the set of extended real numbers	4
	1.3	Dense	sets and stronger topologies	5
		1.3.1	Stronger topologies	5
		1.3.2	The topologies τ_+ , τ on the real line	5
		1.3.3	Some properties of $\tau_+, \tau \dots \dots \dots \dots$	6
	1.4	Closed	l sets and limit points	6
		1.4.1	Limit points, adherent points, and closures of sets	7
		1.4.2	Interiors of sets	7
	1.5	Induce	ed topologies	7
		1.5.1	Closed sets with respect to the induced topology	8
		1.5.2	Induced topologies and separation conditions	8
		1.5.3	Induced topologies and semimetrics	9
	1.6	Conve	ergent sequences	9
		1.6.1	Some properties of convergent sequences	10
		1.6.2	Convergent sequences in R with respect to τ_+, τ	10
	1.7	Local	bases	10
		1.7.1	Countable local bases	11
		1.7.2	A criterion for the convergence of a sequence	11
		1.7.3	Local bases and induced topologies	12
		1.7.4	The first countability condition	12
	1.8	Direct	ed systems and nets	12
		1.8.1	Nets of elements of sets	13
		1.8.2	Convergence of nets	13
		1.8.3	Uniqueness of the limit in Hausdorff spaces	14
	1.9	More	on convergent nets	14
		1.9.1		14
		1.9.2	A criterion for the convergence of a net	15
	1.10	Regul		15

CONTENTS

		1.10.1 Regularity in the strong sense implies Hausdorff	16
		1.10.2 Regularity and semimetrics	16
		1.10.3 Regularity of R with respect to $\tau_+, \tau \dots \dots$	17
		1.10.4 Regularity and induced topologies	17
	1.11	Completely Hausdorff spaces	17
		1.11.1 The completely Hausdorff condition	18
		1.11.2 More on completely Hausdorff spaces	18
	1.12	Separated sets and connectedness	19
		1.12.1 Connectedness	20
	1.13		20
			20
			21
		v	21
	1.14		22
			23
	1.15		23
			24
2	Car	dinality and some more topology	2 6
	2.1	Cardinal equivalence	26
		2.1.1 The power set	27
			27
	2.2	Comparing cardinalities	28
			29
	2.3		29
			30
			30
	2.4		31
		2.4.1 Exponentials of products	31
	2.5	Strong limit points and separability	32
			33
	2.6		33
			34
			34
			34
			35
			35
			35
			36
	2.7		36
	•		36
		v c	37
			$\frac{37}{37}$
			$\frac{37}{37}$
			38
		*	38

vi *CONTENTS*

	2.7.7	Locally constant mappings and connectedness	 38
2.8	More o	on continuity	 39
	2.8.1	Homeomorphisms	 39
	2.8.2	Open mappings	 39
	2.8.3	Restrictions of continuous mappings	 39
	2.8.4	Continuity and connectedness	 40
	2.8.5	Path connectedness	 40
2.9	The pr	roduct topology	41
	2.9.1	Defining the product topology	41
	2.9.2	The strong product topology	42
	2.9.3	Some basic open sets	 42
	2.9.4	Using the discrete topology on the factors	 42
	2.9.5	The standard topology on \mathbb{R}^n	 42
2.10	Coord	inate mappings	43
		Some properties of p_l	43
		Continuity of the coordinate mappings	43
		Products of closed sets	44
		Convergence of nets and the product topology	44
		Closures of products of sets	44
		Some basic separation conditions for the product	45
		Regularity of the product	45
2.11		and finite products	45
		Countability of $\mathcal{B}(x)$	46
		Another local base for the product topology at $x cdot$	46
		Some bases for the product topology on X	47
2.12		and countable products	47
		A local base for the product topology at $x cdots cdots$	47
		Countability of $\mathcal{B}^n(x)$	48
		Another countable local base for X at $x cdots cdots cdots$	48
		Some bases for $X \dots \dots \dots \dots \dots$	48
		Countable bases for X	49
		Some local bases for the strong product topology	49
		Some bases for the strong product topology	49
2.13		products and semimetrics	50
		Open balls in X with respect to d	50
		Relation with the product topology on $X cdots cdots$	50
		Another semimetric on X	51
		An additional semimetric on X	51
2.14		ating semimetrics	51
		Open balls in X with respect to d_t	51
2.15		able products and semimetrics $\dots \dots \dots$	52
		A semimetric on X	52
		The corresponding open balls	52
		Relation between d and the product topology on X .	53

CONTENTEC	••
CONTENTS	V11

	apactness and related topics	5
3.1	Compact sets	5
	3.1.1 Some examples of compact sets	5
	3.1.2 Another class of examples	5
	3.1.3 Compactness and induced topologies	5
	$3.1.4$ Intersections of compact sets and closed sets $\dots \dots$	5
	3.1.5 An additional class of examples	5
3.2	Compactness in Hausdorff spaces	5
	3.2.1 Disjoint compact sets	5
	3.2.2 Disjoint compact and closed sets	5
3.3	Continuity and compactness	5
	3.3.1 The extreme value theorem \dots	5
	3.3.2 Compactness and homeomorphisms	5
3.4	The limit point property	5
	3.4.1 Compactness implies the strong limit point property	5
	3.4.2 A partial converse	5
	3.4.3 Some related results $\dots \dots \dots \dots \dots \dots$	6
3.5	Sequential compactness	6
	3.5.1 A consequence of sequential compactness	6
	3.5.2 A criterion for sequential compactness	6
	3.5.3 Sequential compactness and induced topologies	6
	3.5.4 Sequentially closed sets	6
	3.5.5 Sequential continuity	6
3.6	Countable compactness	6
	3.6.1 The Lindelöf property	6
	3.6.2 Lindelöf's theorem	6
	3.6.3 Some additional results	6
3.7	The finite intersection property	6
	3.7.1 A characterization of compactness	6
	3.7.2 A characterization of countable compactness	6
3.8	The strong limit point property	6
3.9	Compactness and bases	6
	3.9.1 A criterion for compactness	6
	3.9.2 The proof of the proposition	6
	3.9.3 Subbases	7
	3.9.4 Alexander's subbase theorem	7
3 10	Products of two compact sets	7
0.10	3.10.1 Some open coverings of K in Y	7
	3.10.2 An open covering of H in X	7
	3.10.3 Using these finite subcoverings	7
3 11	Products of more compact sets	7
5.11	3.11.1 An induction argument	7
	3.11.2 Some compact subsets of \mathbb{R}^n	7
	3.11.3 Compactness of arbitrary products of compact sets	7
2 10	Sequential compactness and finite products	7
		1

viii CONTENTS

		3.12.2	A subsequence of the first subsequence when $n \geq 2$. 74
		3.12.3	Repeating the process	. 74
	3.13	Sequer	ntial compactness in countable products	75
		3.13.1	A sequence of subsequences	. 75
		3.13.2	Another subsequence	. 76
		3.13.3	More on this subsequence	. 77
	3.14		actness and sequential compactness	
			er result of Tychonoff	
4	Som	ie mor	re set theory	80
_	4.1		lemma	
		4.1.1	The statement of Zorn's lemma	
		4.1.2	Countably-infinite sets A	
	4.2		orff's maximality principle	
		4.2.1	Maximal chains in countable sets	
		4.2.2	Zorn's lemma from maximal chains	
	4.3		nal chains from Zorn's lemma	
	1.0	4.3.1	An upper bound for \mathcal{E} in \mathcal{C}	
	4.4		xiom of choice	
		4.4.1	A partially-ordered set A	
		4.4.2	Upper bounds for arbitrary chains in A	
	4.5		ve mappings	
	1.0	4.5.1	A partially-ordered set \mathcal{A}	
		4.5.2	Upper bounds for arbitrary chains in \mathcal{A}	
	4.6		ordered sets	
	1.0	4.6.1	The well-ordering principle	
		4.6.2	Ideals and segments in partially-ordered sets	
		4.6.3	Ideals in well-ordered sets	
		4.6.4	Checking that a linear ordering is a well ordering	
	4.7	-	rell-ordering principle	
	2	4.7.1	The collection of well-ordered subsets of $A \dots \dots$	
		4.7.2	Upper bounded for chains of well-ordered subsets	
		4.7.3	Elements of C correspond to ideals in A_C	
		4.7.4	Checking that $A_{\mathcal{C}}$ is well ordered by $\leq_{\mathcal{C}}$	
	4.8		isomorphisms	
	1.0	4.8.1	Automorphisms of well-ordered sets	90
		4.8.2	Uniqueness of isomorphisms between well-ordered sets	
		4.8.3	Isomorphisms onto ideals in well-ordered sets	
		4.8.4	Uniqueness of isomorphic segments in well-ordered sets	
	4.9		isomorphisms, continued	
	2.0	4.9.1	A subset I_A of A	
		4.9.2	More on I_A	
		4.9.3	A subset I_B of B	
		4.9.4	Either $I_A = A$ or $I_B = B$	93
			<i>D</i>	

CONTENTS ix

			tional notions in topology	94
Ę	5.1		uous real-valued functions	94
		5.1.1	Sums and products	94
		5.1.2	Reciprocals of continuous functions	95
		5.1.3	Urysohn spaces	95
		5.1.4	Continuous functions from semimetrics	95
_		5.1.5	Maxima and minima of continuous real-valued functions . $% \left(\frac{1}{2}\right) =\frac{1}{2}\left(\frac{1}{2}\right)$	95
٤	5.2		in functions	96
		5.2.1	Urysohn's lemma	96
		5.2.2	Urysohn's metrization theorem	97
_	- 0	5.2.3	Some properties of Urysohn spaces	97
٤	5.3	_	ete regularity	97
		5.3.1	Relation with other separation conditions	98
		5.3.2	Relation with induced topologies	98
-	- 1	5.3.3	More on completely regular spaces	99
Ę	5.4		compactness and manifolds	99
		5.4.1	Locally compact Hausdorff spaces	
		5.4.2	Locally Euclidean of dimension $n cdots$	
		5.4.3	Topological manifolds	
_		5.4.4	Baire's theorem	
Ę	5.5		apactness	
		5.5.1	Some criteria for σ -compactness	
		5.5.2	Some bases from open coverings	
_	- 0	5.5.3	An example in \mathbf{R}	
٤	5.6		ent spaces	
_		5.6.1	The quotient or identification topology	
Ę	5.7		connectedness	
		5.7.1	Local path connectedness	
_	- 0	5.7.2	Some examples and basic properties	
Ę	5.8		cted and path-connected components	
_	- 0	5.8.1	Another equivalence relation on X	
Ę	5.9		pint compactifications	
		5.9.1	Checking that this defines a topology on X^*	
		5.9.2	Compactness of X^*	
-	- 10	5.9.3	When X^* is Hausdorff	
Ę	0.10		opic mappings	
			Additional conditions on $f, g \dots \dots \dots$	
_	- 11		Convex sets Y in \mathbb{R}^n	
Ę	0.11		able unions and intersections	
			Some subsets of semimetric spaces	
			Some remarks about inverse images	
			Perfectly normal spaces	
	- 10		Nowhere dense sets and category	
٤).12		ce functions	
			Continuity of distance functions	
		5.12.2	Separated sets and open sets	112

x CONTENTS

		5.12.3	Urysohn's lemma for semimetric spaces	 113
	5.13	Local	finiteness	 113
		5.13.1	Local finiteness and adherent points	 113
		5.13.2	Local finiteness and closures	 114
		5.13.3	Local finiteness and induced topologies	 114
		5.13.4	Local finiteness and compactness	 114
	5.14		of real-valued functions	
		5.14.1	Local finiteness and continuity	 115
		5.14.2	Some partitions of unity	 115
	5.15		compactness and σ -compactness	
		5.15.1	Some sequences of compact and open sets	 116
		5.15.2	A locally finite sequence of compact sets	 117
		5.15.3	A locally finite sequence of open sets	 117
6	Son	ne spac	ces of mappings	118
	6.1	Bound	led sets and mappings	 118
		6.1.1	Countable compactness implies boundedness	 118
		6.1.2	Bounded mappings	 118
		6.1.3	Some related real-valued functions	 118
		6.1.4	Supremum semimetrics	 119
		6.1.5	Some supremum semimetrics on $C(X,Y)$	 119
		6.1.6	Existence of the maximum	 119
		6.1.7	Bounded continuous mappings	
	6.2	Unifor	rm convergence	 120
		6.2.1	Relation with the product topology	
		6.2.2	Using a semimetric on Y	 120
		6.2.3	Some sets $N_{\epsilon}(f)$ and $\widetilde{N}_{\epsilon}(f)$	 120
		6.2.4	The topology of uniform convergence	 121
		6.2.5	More on $\widetilde{N}_{\eta}(f)$	 121
		6.2.6	Uniformly convergent nets	 122
		6.2.7	Relation with the supremum semimetric on $\mathcal{B}(X,Y)$.	 122
		6.2.8	Uniform convergence and continuity	
	6.3	Infinit	e series of functions	 122
		6.3.1	Some continuous functions defined using infinite series	 123
		6.3.2	Perfect normality implies complete normality	 123
	6.4	Tietze	's extension theorem	 124
		6.4.1	An initial approximation	 124
		6.4.2	Some more precise approximations	 124
		6.4.3	A refinement of this argument	
		6.4.4	Another version of Tietze's theorem	 125
	6.5	Unifor	rm convergence on compact sets	 125
		6.5.1	More on $N_{E,\epsilon}(f)$ and $\widetilde{N}_{E,\epsilon}(f)$	
		6.5.2	The topology of uniform convergence on compact sets	
		6.5.3	Some properties of this topology	
		6.5.4	Convergent nets with respect to this topology	 127
	6.6	The co	ompact-open topology	 127

CONTENTS xi

		6.6.1	Some properties of the compact-open topology	127
		6.6.2	The compact-open topology on $C(X,Y)$	128
		6.6.3	Using a semimetric on Y again	
		6.6.4	A property of compact subsets of Y	128
		6.6.5	Some remarks about continuous mappings and compact	
			sets	128
		6.6.6	More on continuous mappings and compact sets	129
		6.6.7	Finitely many smaller compact sets	129
		6.6.8	Using the compact-open topology	130
	6.7	Contin	nuity on compact sets	130
		6.7.1	Local compactness and continuity on compact sets	130
		6.7.2	Some remarks about sequential continuity	130
		6.7.3	Using the first countability condition	131
		6.7.4	Uniform convergence and continuity conditions	131
	6.8	Suppor	rts of real-valued functions	131
		6.8.1	Compactly-supported functions	132
		6.8.2	Another version of Urysohn's lemma	132
	6.9	Functi	ons with finite support	132
		6.9.1	Relation with Cartesian products	132
		6.9.2	Using positive weights on X	132
		6.9.3	Another metric associated to a	133
		6.9.4	The case where $X = \mathbf{Z}_+ \dots \dots \dots \dots$.	133
	6.10		subspaces of $c(X)$	133
		6.10.1	Relation with $c(E)$ and $c_{00}(E)$	134
		6.10.2	Relation with Cartesian products again	134
			Using positive weights again	
	6.11		mappings into products	
			Continuous mappings into Y	
			Mappings into $c_{00}(A)$	
			A local finiteness condition	135
		6.11.4	Continuity of ϕ	136
_	T3:14		1 1 01	40-
7			d ultrafilters	137
	7.1	Filters		
		7.1.1	Some basic examples of filters	
		7.1.2	Pre-filters or filter bases	
		7.1.3	Filter bases from nets	
		7.1.4		
		7.1.5	Uniqueness of the limit in Hausdorff spaces again	
		7.1.6	Adherent points and filters	139
	7.0	7.1.7	Filters and induced topologies	139
	7.2		nuity and refinements	139
		7.2.1	Pushing filters forward	140
		7.2.2	Refinements of filters	140
		7.2.3	Convergent refinements	140
		7.2.4	Filters and compact sets	140

xii CONTENTS

		7.2.5	Filter bases and the finite intersection property		141
	7.3	Ultrafi	lters		141
		7.3.1	Some ultrafilters		142
		7.3.2	Compactness and ultrafilters		142
		7.3.3	Some refinements of filters		142
		7.3.4	A characterization of ultrafilters		142
		7.3.5	Pushing ultrafilters forward		142
		7.3.6	Filters and the product topology		143
		7.3.7	Tychonoff's theorem using ultrafilters		143
	7.4	Filters	and relations		143
		7.4.1	An equivalence relation associated to the filter ${\mathcal F}$		143
		7.4.2	The corresponding quotient space		143
		7.4.3	The case where \mathcal{F} is an ultrafilter \ldots		144
		7.4.4	Some related partial orderings		144
0	т.	•1•	1		1.40
8			$egin{array}{lll} {f nd} & {f coverings} \ {f ngs} & {f and} & {f refinements} & {f .} & {\bf .} & {\bf$		146
	8.1	8.1.1	Refinements of families of sets		
		8.1.2	Refinements of refinements		
		-	Some subfamilies associated to refinements		
		8.1.3 8.1.4	Some common refinements		
	8.2		coverings		
	0.2	8.2.1	Another look at Lindelöf's theorem		
		8.2.2	Another look at compactness with respect to a base		
		8.2.3	A theorem of Lebesgue		
		8.2.4	The proof of Lebesgue's theorem		
	8.3		ompactness		
	0.5	8.3.1	Paracompactness in the strict sense		
		8.3.2	Paracompactness in the strict sense		
		8.3.3	Another version of paracompactness		
		8.3.4	A paracompactness property of closed sets in X		
		8.3.5	Paracompactness and induced topologies		
		8.3.6	Paracompactness in the strong sense implies regularity		
		8.3.7	Paracompactness and normality		
		8.3.8	Some criteria for paracompactness		
	8.4	0.0.0	refinements		
	0.4		A variant of paracompactness		
		8.4.2	Another variant of paracompactness		
		8.4.3	Using the first variant		
		8.4.4	Using the second variant		152
	8.5		efinements		152 153
	0.0	8.5.1	The star of x with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$		
		8.5.2	Star refinements of a family of subsets of X		
		8.5.3	Fully normal spaces in the strict sense		
		8.5.4	A criterion for full normality in the strict sense		
		0.0.4	11 officeron for full normains in the strict sense	•	194

CONTENTS xiii

	8.5.5	Checking that $W_1(x)$, $x \in X$, is a star refinement of	
		$\{U_{\alpha}\}_{\alpha\in A}$	
	8.5.6	Normality from full normality	
	8.5.7	Fully normal spaces in the strong sense	155
8.6	Using	another refinement	
	8.6.1	The star of E with respect to $\{W_{\gamma}\}_{\gamma inC}$	155
	8.6.2	The star of the star of $x \in X$ with respect to $\{W_{\gamma}\}_{{\gamma} \in C}$.	
	8.6.3	Star refinements of star refinements	155
	8.6.4	Using full normality in the strict sense	156
	8.6.5	The star of E_{β} with resect to $\{W_{\gamma}\}_{{\gamma}\in C}$	156
	8.6.6	Local finiteness of $\{Z_{\beta}\}_{{\beta}\in B}$	156
	8.6.7	Staring with an open covering of $X cdot . $	157
	8.6.8	Using this to get paracompactness in the strict sense	157
8.7	σ -Loca	al finiteness	157
	8.7.1	σ -Locally finite refinements	
	8.7.2	Using σ -locally finite refinements	158
	8.7.3	A subset E_{β} of V_{β}	158
	8.7.4	The E_{β} 's, $\beta \in B$, cover $X \dots \dots \dots \dots \dots \dots$	158
	8.7.5	Local finiteness of $\{E_{\beta}\}_{{\beta}\in B}$	159
8.8	Semim	netrics and full normality	159
	8.8.1	Some collections of open balls in $X cdots cdots cdots$	159
8.9	Discre	te families of sets	160
	8.9.1	Discrete families and induced topologies	160
	8.9.2	Discrete families and full normality	160
	8.9.3	σ -Discrete families of sets	161
8.10	Sequer	nces of star refinements	161
	8.10.1	The star of $E \in \mathcal{B}_j$ with respect to \mathcal{B}_j	161
	8.10.2	The star of the star of E with respect to \mathcal{B}_{j+1}	161
	8.10.3	Some more collections of subsets of X	162
8.11	σ -Disc	reteness and full normality	163
	8.11.1	Using a well-ordering on X to get some more sets in X .	163
8.12	Point f	${ m finiteness}$	164
	8.12.1	Metacompactness	164
	8.12.2	Using point finiteness	164
8.13	Partiti	ions of unity	166
8.14	Minim	al coverings	167
		Using metacompactness	
Bibliog	raphy		169
Index			196

xiv CONTENTS

Chapter 1

Some basic notions in topology

1.1 Some definitions and examples

Let X be a set.

Definition 1.1.1 A collection τ of subsets of X is said to define a topology on X if it satisfies the following three conditions. First,

$$(1.1.2) \emptyset, X \in \tau.$$

Second, if U_1, \ldots, U_n are finitely many elements of τ , then

$$(1.1.3) \qquad \bigcap_{j=1}^{n} U_j \in \tau.$$

Third, if A is a nonempty set, and $U_{\alpha} \in \tau$ for every $\alpha \in A$, then

$$(1.1.4) \qquad \qquad \bigcup_{\alpha \in A} U_{\alpha} \in \tau.$$

In this case, (X, τ) is said to be a topological space, and the elements of τ are called open sets in X.

Sometimes we may refer to a topological space X, in which case the topology τ is implicit.

1.1.1 Some basic examples

Definition 1.1.5 If X is any set, then the discrete topology on X is defined by taking τ to be the collection of all subsets of X. The indiscrete topology on X is defined by taking τ to be the collection consisting of only the empty set and X.

It is easy to see that the discrete and indiscrete topologies satisfy the requirements of a topology.

If d(x, y) is a metric on a set X, then the notion of an open subset of X with respect to $d(\cdot, \cdot)$ can be defined in a standard way. An extension of this will be mentioned in the next section. It is well known and not difficult to check that this defines a topology on X.

In particular, the discrete metric on a set X is defined by putting d(x, y) = 1 when x and y are distinct elements of X, and d(x, x) = 0 for every $x \in X$. This defines a metric on X, for which the corresponding topology is the discrete topology.

Let **R** be the real line, as usual. If $x \in \mathbf{R}$, then the absolute value of x is defined by |x| = x when $x \ge 0$, and |x| = -x when $x \le 0$. The standard Euclidean metric on **R** is defined by

$$(1.1.6) d(x,y) = |x - y|.$$

The standard topology on \mathbf{R} may be defined as the topology determined on \mathbf{R} by (1.1.6).

1.1.2 Some basic separation conditions

Let (X, τ) be a topological space.

Definition 1.1.7 We say that (X, τ) satisfies the first separation condition if for every pair x, y of distinct elements of X, there is an open subset U of X such that $x \in U$ and $y \notin U$. Equivalently, we say that X is a T_1 space in this case.

The first separation condition is supposed to be symmetric in x and y, so that there should also be an open set $V \subseteq X$ such that $y \in V$ and $x \notin V$.

Definition 1.1.8 We say that (X, τ) satisfies the zeroth separation condition if for every pair x, y of distinct elements of X there is an open set $W \subseteq X$ such that either $x \in W$ and $y \notin W$, or $y \in W$ and $x \notin W$. In this case, we may also say that (X, τ) is a T_0 space.

Thus the first separation condition implies the zeroth separation condition.

Definition 1.1.9 We say that (X, τ) satisfies the second separation condition if for every pair x, y of distinct elements of X there are disjoint open subsets U, V of X such that $x \in U$ and $y \in V$. We may also say that (X, τ) is a T_2 space in this situation, or equivalently that (X, τ) is Hausdorff.

Hausdorff spaces obviously satisfy the first separation condition. If X is a set with at least two elements equipped with the indiscrete topology, then X does not satisfy the zeroth separation condition. Any set equipped with the discrete topology is Hausdorff.

1.2 Semimetrics and intervals

Let X be a set.

Definition 1.2.1 A nonnegative real-valued function d(x,y) defined for x, y in X is said to be a semimetric or pseudometric on X if it satisfies the following three conditions. First,

$$(1.2.2) d(x,x) = 0$$

for every $x \in X$. Second, d(x,y) is symmetric in x and y, so that

$$(1.2.3) d(x,y) = d(y,x)$$

for every $x, y \in X$. Third, the triangle inequality

$$(1.2.4) d(x,z) < d(x,y) + d(y,z)$$

holds for every $x, y, z \in X$. If we also have that

$$(1.2.5) d(x,y) > 0$$

for every $x, y \in X$ with $x \neq y$, then $d(\cdot, \cdot)$ is said to be a metric on X.

1.2.1 Open balls and open sets

Let $d(\cdot, \cdot)$ be a semimetric on X. If $x \in X$ and r is a positive real number, then the *open ball* in X centered at x with radius r with respect to $d(\cdot, \cdot)$ is defined by

$$(1.2.6) B(x,r) = B_d(x,r) = \{ y \in X : d(x,y) < r \}.$$

A subset U of X is said to be an *open set* with respect to $d(\cdot, \cdot)$ if for every $x \in U$ there is an r > 0 such that

$$(1.2.7) B(x,r) \subseteq U.$$

It is well known and not difficult to show that this defines a topology on X. If d(x,y)=0 for every $x,y\in X$, then B(x,r)=X for every $x\in X$ and r>0, and the corresponding topology on X is the indiscrete topology.

If $w \in X$ and t > 0, then one can check that B(w,t) is an open set in X with respect to $d(\cdot,\cdot)$. More precisely, if $x \in B(w,t)$, then r = t - d(w,x) > 0, and one can verify that

$$(1.2.8) B(x,r) \subseteq B(w,t),$$

using the triangle inequality.

If $w_1, w_2 \in X$ satisfy $d(w_1, w_2) > 0$, then it is easy to see that

$$(1.2.9) B(w_1, d(w_1, w_2)/2) \cap B(w_2, d(w_1, w_2)/2) = \emptyset,$$

using the triangle inequality again. In particular, if $d(\cdot, \cdot)$ is a metric on X, then X is Hausdorff with respect to the topology determined by $d(\cdot, \cdot)$. If X satisfies the zeroth separation condition with respect to the topology determined by a semimetric $d(\cdot, \cdot)$, then $d(\cdot, \cdot)$ is a metric on X.

1.2.2 Intervals in the set of extended real numbers

Now let X be the set of extended real numbers, which consists of the real numbers together with $+\infty$ and $-\infty$, where

$$(1.2.10) -\infty < x < +\infty$$

for every $x \in \mathbf{R}$. If $a, b \in X$ and a < b, then the corresponding open interval is defined by

$$(1.2.11) (a,b) = \{x \in \mathbf{R} : a < x < b\}.$$

It is easy to see that this is an open set in \mathbf{R} with respect to the standard topology. If $a, b \in \mathbf{R}$, then this is the same as the open ball centered at the midpoint (a+b)/2 with radius (b-a)/2 with respect to the standard Euclidean metric on \mathbf{R} .

If $a, b \in X$ and $a \leq b$, then the corresponding closed interval is defined by

$$[a,b] = \{x \in X : a \le x \le b\}.$$

If a < b, then we get the half-open, half-closed intervals

$$(1.2.13) [a,b) = \{x \in X : a < x < b\}$$

and

$$(1.2.14) (a,b] = \{x \in X : a < x < b\}.$$

Let us say that $U \subseteq X$ is an open set if for each $x \in U$, one of the following three conditions holds. If $x \in \mathbf{R}$, then there are $a, b \in \mathbf{R}$ such that a < x < b and

$$(1.2.15) (a,b) \subseteq U.$$

If $x = +\infty$, then there is an $a \in \mathbf{R}$ such that

$$(1.2.16) (a, +\infty] \subseteq U.$$

If $x = -\infty$, then there is a $b \in \mathbf{R}$ such that

$$(1.2.17) [-\infty, b) \subseteq U.$$

One can check that this defines a topology on X, which we may refer to as the standard topology on the set of extended real numbers.

If $U \subseteq \mathbf{R}$, then U is an open set in X if and only if U is an open set in \mathbf{R} with respect to the standard topology. In particular, open intervals in \mathbf{R} are open sets in X with respect to this topology. Similarly, one can verify that $(a, +\infty]$ and $[-\infty, b)$ are open sets in X with respect to this topology for all $a, b \in \mathbf{R}$. Using this, it is easy to see that X is Hausdorff with respect to this topology.

1.3 Dense sets and stronger topologies

Let (X, τ) be a topological space.

Definition 1.3.1 A subset E of X is said to be dense in X with respect to τ if for every $x \in X$ and open set $U \subseteq X$ with $x \in U$ there is a $y \in E$ such that $y \in U$. Equivalently, this means that for every nonempty open set $U \subseteq X$, we have that

$$(1.3.2) E \cap U \neq \emptyset.$$

Of course, X is dense in itself. If X is any set equipped with the discrete topology, then X is the only dense subset of itself. If X is a nonempty set equipped with the indiscrete topology, then $E \subseteq X$ is dense in X if and only if $E \neq \emptyset$. It is well known that the set \mathbf{Q} of rational numbers is dense in the real line with respect to the standard topology.

1.3.1 Stronger topologies

Proposition 1.3.3 Let X be a set, and let τ , $\tilde{\tau}$ be topologies on X such that

In this case, we may say that $\tilde{\tau}$ is at least as strong as τ on X.

- (a) If $j \in \{0,1,2\}$ and (X,τ) satisfies the jth separation condition, then $(X,\tilde{\tau})$ satisfies the jth separation condition as well.
- (b) If $E \subseteq X$ is dense in X with respect to $\widetilde{\tau}$, then E is dense in X with respect to τ .

This can be verified directly from the definitions.

1.3.2 The topologies τ_+, τ_- on the real line

Let τ_+ be the collection of subsets U of the real line such that for every $x \in U$ there is a $b \in \mathbf{R}$ with x < b and

$$(1.3.5) [x,b) \subseteq U.$$

Similarly, let τ_- be the collection of subsets U of $\mathbf R$ such that for every $x \in U$ there is an $a \in \mathbf R$ with a < x and

$$(1.3.6) (a, x] \subseteq U.$$

One can check that τ_+ and τ_- define topologies on **R**. Every open set in **R** with respect to the standard topology is an open set with respect to τ_+ and τ_- . More precisely,

is the same as the standard topology on R.

1.3.3 Some properties of τ_+, τ_-

If $a \in \mathbf{R}$, $b \in \mathbf{R} \cup \{+\infty\}$, and a < b, then it is easy to see that

$$(1.3.8)$$
 $[a,b)$

is an open set with respect to τ_+ . Similarly, if $a \in \mathbf{R} \cup \{-\infty\}$, $b \in \mathbf{R}$, and a < b, then

$$(1.3.9) (a,b]$$

is an open set with respect to τ_- . Note that the real line is Hausdorff with respect to τ_+ and τ_- , by Proposition 1.3.3, and because **R** is Hausdorff with respect to the standard topology. If $E \subseteq \mathbf{R}$ is dense in **R** with respect to the standard topology, then one can check that E is dense in **R** with respect to τ_+ and τ_- . Of course, if E is dense in **R** with respect to τ_+ or τ_- , then E is dense in **R** with respect to the standard topology, as in Proposition 1.3.3.

1.4 Closed sets and limit points

Let (X, τ) be a topological space.

Definition 1.4.1 A subset E of X is said to be a closed set in X with respect to τ if the complement

$$(1.4.2) X \setminus E = \{x \in X : x \notin E\}$$

of E in X is an open set in X with respect to τ .

Of course,

$$(1.4.3) X \setminus (X \setminus W) = W$$

for every subset W of X. This implies that the complement of an open subset of X is a closed set.

Note that X and the empty set are closed subsets of X. If E_1, \ldots, E_n are finitely many closed subsets of X, then their union is a closed set as well. More precisely,

(1.4.4)
$$X \setminus \left(\bigcup_{j=1}^{n} E_{j}\right) = \bigcap_{j=1}^{n} (X \setminus E_{j})$$

is an open set in X, by the definition of a topology.

If A is a nonempty set, and E_{α} is a closed set in X for every $\alpha \in A$, then $\bigcap_{\alpha \in A} E_{\alpha}$ is a closed set in X too. Indeed,

(1.4.5)
$$X \setminus \left(\bigcap_{\alpha \in A} E_{\alpha}\right) = \bigcup_{\alpha \in A} (X \setminus E_{\alpha})$$

is an open set in X in this case, by the definition of a topological space.

1.4.1 Limit points, adherent points, and closures of sets

Definition 1.4.6 A point $p \in X$ is said to be a limit point of $E \subseteq X$ if for every open set $U \subseteq X$ with $p \in U$ there is a $q \in E \cap U$ such that $p \neq q$. We say that $p \in X$ is adherent to E if for every open set $U \subseteq X$ with $p \in U$, we have that $E \cap U \neq \emptyset$.

Thus every element of E and every limit point of E in X is adherent to E. If $p \in X$ is adherent to E and $p \notin E$, then p is a limit point of E in X.

Definition 1.4.7 The closure of a subset E of X is the set \overline{E} consisting of all points in X that are adherent to E. Equivalently, \overline{E} is the set of $p \in X$ such that $p \in E$, p is a limit point of E, or both.

Note that $E \subseteq X$ is dense in X if and only if $\overline{E} = X$.

Proposition 1.4.8 *If* $E \subseteq X$ *is a closed set, then* $\overline{E} = E$.

It suffices to check that $\overline{E} \subseteq E$. If E is a closed set and $p \in X \setminus E$, then it is easy to see that p is not adherent to E, because $X \setminus E$ is an open set. Thus $p \notin \overline{E}$, as desired.

1.4.2 Interiors of sets

Definition 1.4.9 The interior of a subset A of X is the set Int A of $p \in A$ for which there is an open set $U \subseteq X$ such that $p \in U$ and $U \subseteq A$. Equivalently, Int A is the union of all the open subsets of X that are contained in A.

Thus Int A is automatically an open subset of X, and Int A = A when A is an open set in X.

Proposition 1.4.10 If E is any subset of X, then

$$(1.4.11) X \setminus \overline{E} = \operatorname{Int}(X \setminus E).$$

In particular, \overline{E} is a closed set in X.

Indeed, $p \in X$ is not adherent to E exactly when there is an open set $U \subseteq X$ such that $p \in U$ and $U \cap E = \emptyset$. This is the same as saying that $p \in U$ and $U \subseteq X \setminus E$, which means that $p \in \operatorname{Int}(X \setminus E)$.

1.5 Induced topologies

Let (X, τ) be a topological space, and let Y be a subset of X.

Definition 1.5.1 A subset E of Y is said to be relatively open in Y if there is an open set $U \subseteq X$ such that

$$(1.5.2) E = U \cap Y.$$

Proposition 1.5.3 The collection of relatively open subsets of Y defines a topology on Y, which is called the induced topology on Y.

Clearly $Y = X \cap Y$ and $\emptyset = \emptyset \cap Y$ are relatively open in Y. If E_1, \ldots, E_n are finitely many relatively open subsets of Y, then there are open subsets U_1, \ldots, U_n of X such that $E_j = U_j \cap Y$ for each $j = 1, \ldots, n$. This implies that

$$(1.5.4) \qquad \bigcap_{j=1}^{n} E_j = \bigcap_{j=1}^{n} (U_j \cap Y) = \left(\bigcap_{j=1}^{n} U_j\right) \cap Y$$

is relatively open in Y.

Let A be a nonempty set, and suppose that $E_{\alpha} \subseteq Y$ is relatively open for every $\alpha \in A$. Thus, for each $\alpha \in A$, there is an open subset U_{α} of X such that $E_{\alpha} = U_{\alpha} \cap Y$. It follows that

(1.5.5)
$$\bigcup_{\alpha \in A} E_{\alpha} = \bigcup_{\alpha \in A} (U_{\alpha} \cap Y) = \Big(\bigcup_{\alpha \in A} U_{\alpha}\Big) \cap Y$$

is relatively open in Y, as desired.

1.5.1 Closed sets with respect to the induced topology

Proposition 1.5.6 A subset A of Y is a closed set with respect to the induced topology if and only if there is a closed set $A_1 \subseteq X$ such that

$$(1.5.7) A = A_1 \cap Y.$$

If A_1 is a closed set in X, then $X \setminus A_1$ is an open set in X, and hence $(X \setminus A_1) \cap Y$ is relatively open in Y. One can check that

$$(1.5.8) Y \setminus (A_1 \cap Y) = (X \setminus A_1) \cap Y,$$

so that $A_1 \cap Y$ is a closed set in Y with respect to the induced topology. Conversely, if $A \subseteq Y$ is a closed set with respect to the induced topology, then $Y \setminus A$ is relatively open in Y. This implies that there is an open subset U of X such that

$$(1.5.9) Y \setminus A = U \cap Y.$$

This means that $A_1 = X \setminus U$ is a closed set in X, and it is easy to see that (1.5.7) holds.

1.5.2 Induced topologies and separation conditions

Proposition 1.5.10 If $j \in \{0,1,2\}$ and X satisfies the jth separation condition, then Y satisfies the jth separation condition, with respect to the induced topology.

This can be verified directly from the definitions.

1.5.3 Induced topologies and semimetrics

Let $d(\cdot, \cdot)$ be a semimetric on X, and suppose now that X is equipped with the topology determined by $d(\cdot, \cdot)$. It is easy to see that the restriction of d(x, y) to $x, y \in Y$ defines a semimetric on Y.

Proposition 1.5.11 Under the conditions just mentioned, the induced topology on Y is the same as the topology determined on Y by the restriction of $d(\cdot, \cdot)$ to Y

If $x \in X$ and r > 0, then let $B_X(x,r)$ be the open ball in X centered at x with radius r with respect to $d(\cdot,\cdot)$. Similarly, if $x \in Y$, then let $B_Y(x,r)$ be the open ball in Y centered at x with radius r > 0 with respect to the restriction of $d(\cdot,\cdot)$ to Y. It is easy to see that

$$(1.5.12) B_Y(x,r) = B_X(x,r) \cap Y$$

for every $x \in Y$ and r > 0.

If E is a relatively open subset of Y, then there is an open subset U of X such that $E = U \cap Y$. One can check directly that E is an open set in Y with respect to the topology determined by the restriction of $d(\cdot, \cdot)$ to Y, using the analogous property of U in X.

To show the converse, observe first that (1.5.12) is relatively open in Y for every $x \in Y$ and r > 0, because open balls in X are open sets. If $E \subseteq Y$ is an open set with respect to the topology determined by the restriction of $d(\cdot, \cdot)$ to Y, then E can be expressed as a union of open balls in Y. This implies that E is relatively open in Y, because the union of any family of relatively open subsets of Y is relatively open in Y, as before.

1.6 Convergent sequences

Let (X, τ) be a topological space.

Definition 1.6.1 A sequence $\{x_j\}_{j=1}^{\infty}$ of elements of X is said to converge to an element x of X if for every open subset U of X with $x \in U$ there is a positive integer L such that

$$(1.6.2) x_j \in U$$

for every $j \geq L$.

If $x_j = x$ for all but finitely many j, then $\{x_j\}_{j=1}^{\infty}$ automatically converges to x. If X is equipped with the discrete topology, and $\{x_j\}_{j=1}^{\infty}$ converges to x, then $x_j = x$ for all but finitely many j. If X is equipped with the indiscrete topology, then every sequence of elements of X converges to every element of X.

1.6.1 Some properties of convergent sequences

Proposition 1.6.3 If (X, τ) is a Hausdorff topological space, then a sequence of elements of X can converge to at most one element of X.

This is not difficult to show, and an extension of this will be discussed soon.

Proposition 1.6.4 Let E be a subset of X, and let $\{x_j\}_{j=1}^{\infty}$ be a sequence of elements of E. If $\{x_j\}_{j=1}^{\infty}$ converges to an element x of X, then x is adherent to E. If, for each positive integer j, we also have that $x_j \neq x$, then x is a limit point of E in X.

This can be verified directly from the definitions.

Proposition 1.6.5 Let Y be a subset of X, let $\{x_j\}_{j=1}^{\infty}$ be a sequence of elements of Y, and let x be an element of Y. Under these conditions, $\{x_j\}_{j=1}^{\infty}$ converges to x in X if and only if $\{x_j\}_{j=1}^{\infty}$ converges to x in Y, with respect to the induced topology.

This can also be verified directly from the definitions.

1.6.2 Convergent sequences in R with respect to τ_+, τ_-

Let $\{x_j\}_{j=1}^{\infty}$ be a sequence of real numbers, and let x be a real number too. If $\{x_j\}_{j=1}^{\infty}$ converges to x with respect to either of the topologies τ_+ or τ_- defined in Subsection 1.3.2, then $\{x_j\}_{j=1}^{\infty}$ converges to x with respect to the standard topology on \mathbf{R} . More precisely, if $\{x_j\}_{j=1}^{\infty}$ converges to x with respect to τ_+ , then

(1.6.6)
$$x_j \ge x$$
 for all but finitely many j .

This can be seen by taking $U = [x, +\infty)$ in (1.6.2). Conversely, if $\{x_j\}_{j=1}^{\infty}$ satisfies (1.6.6), and $\{x_j\}_{j=1}^{\infty}$ converges to x with respect to the standard topology on \mathbf{R} , then $\{x_j\}_{j=1}^{\infty}$ converges to x with respect to τ_+ .

Similarly, if $\{x_j\}_{j=1}^{\infty}$ converges to x with respect to τ_- , then

(1.6.7)
$$x_j \le x$$
 for all but finitely many j .

Conversely, if $\{x_j\}_{j=1}^{\infty}$ satisfies (1.6.7), and if $\{x_j\}_{j=1}^{\infty}$ converges to x with respect to the standard topology on \mathbf{R} , then $\{x_j\}_{j=1}^{\infty}$ converges to x with respect to τ_- .

1.7 Local bases

Let (X, τ) be a topological space, and let x be an element of X.

Definition 1.7.1 A collection $\mathcal{B}(x)$ of open subsets of X is said to be a local base for the topology of X at x if it satisfies the following two conditions. First, for every $U \in \mathcal{B}(x)$, we have that $x \in U$. Second, if V is an open subset of X such that $x \in V$, then there is an element U of $\mathcal{B}(x)$ with

$$(1.7.2) U \subseteq V.$$

1.7. LOCAL BASES

11

The collection of all open subsets U of X with $x \in U$ is a local base for the topology of X at x. If X is equipped with the indiscrete topology, then X is the only element of this collection. If X is equipped with the discrete topology, then the collection consisting only of $\{x\}$ is a local base for the topology of X at x.

1.7.1 Countable local bases

We are often particularly interested in situations where there may be a local base for the topology of X at x with only finitely or countably many elements, which can therefore be listed in a sequence. Equivalently, we may wish to have a sequence $U_1(x), U_2(x), U_3(x), \ldots$ of open subsets of X such that the collection of the $U_j(x)$'s is a local base for the topology of X at x. This means that

$$(1.7.3) x \in U_j(x) for every j,$$

and that

(1.7.4) for every open set
$$V \subseteq X$$
 with $x \in V$ we have that $U_j(x) \subseteq V$ for some j .

In this situation, it is frequently helpful to also ask that

(1.7.5)
$$U_{j+1}(x) \subseteq U_j(x) \text{ for every } j.$$

This can always be arranged, by replacing $U_j(x)$ with $\bigcap_{l=1}^j U_l(x)$ for each j, if necessary.

If the topology on X is determined by a semimetric $d(\cdot, \cdot)$, then

$$(1.7.6) U_j(x) = B(x, 1/j)$$

satisfies the conditions just mentioned. If X is the real line equipped with the topology τ_+ defined in Subsection 1.3.2, then

$$(1.7.7) U_j(x) = [x, x + 1/j)$$

satisfies these conditions. Similarly, if $X = \mathbf{R}$ equipped with the topology τ_- , then

$$(1.7.8) U_j(x) = (x - 1/j, x]$$

satisfies these conditions.

1.7.2 A criterion for the convergence of a sequence

Let (X, τ) be any topological space again, and let $x \in X$ be given.

Proposition 1.7.9 Suppose that $\{U_j(x)\}_{j=1}^{\infty}$ is a sequence of open subsets of X that satisfies (1.7.3), (1.7.4), and (1.7.5). Let $\{x_j\}_{j=1}^{\infty}$ be a sequence of elements of X such that

$$(1.7.10) x_j \in U_j(x)$$

for every j. Under these conditions, $\{x_j\}_{j=1}^{\infty}$ converges to x in X.

This is easy to check, directly from the definitions.

Corollary 1.7.11 Suppose that there is a local base for the topology of X at x with only finitely or countably many elements. If x is adherent to $E \subseteq X$, then there is a sequence $\{x_j\}_{j=1}^{\infty}$ of elements of E that converges to x in X. If x is a limit point of E in X, then we may also ask that $x_j \neq x$ for each j.

The hypothesis of the corollary implies that there is a sequence $\{U_j(x)\}_{j=1}^{\infty}$ of open sets in X as in the previous proposition. If x is adherent to E, then we can choose

$$(1.7.12) x_j \in E \cap U_j(x)$$

for each j. If x is a limit point of E, then we can also choose x_j to be different from x for every j.

1.7.3 Local bases and induced topologies

Proposition 1.7.13 *Let* Y *be a subset of* X *such that* $x \in Y$. *If* $\mathcal{B}(x)$ *is a local base for the topology of* X *at* x, *then*

(1.7.14)
$$\mathcal{B}_{Y}(x) = \{ U \cap Y : U \in \mathcal{B}(x) \}$$

is a local base for the induced topology on Y at x.

This can be verified directly from the definitions. Note that if $\mathcal{B}(x)$ has only finitely or countably many elements, then $\mathcal{B}_Y(x)$ has only finitely or countably many elements.

1.7.4 The first countability condition

Definition 1.7.15 We say that (X, τ) satisfies the first countability condition if for every $x \in X$ there is a local base $\mathcal{B}(x)$ for the topology of X at x with only finitely or countably many elements.

If (X, τ) satisfies the first countability condition and $Y \subseteq X$, then Y satisfies the first countability condition with respect to the induced topology, by the previous remark.

1.8 Directed systems and nets

Let A be a set.

Definition 1.8.1 A binary relation \leq on A is said to be a partial ordering if it satisfies the following three conditions. First, \leq should be reflexive on A, in the sense that

$$(1.8.2) a \leq a for every a \in A.$$

Second,

(1.8.3) if
$$a, b \in A$$
 satisfy $a \leq b$ and $b \leq a$, then $a = b$.

Third, \prec should be transitive on A, in the sense that

(1.8.4) if
$$a, b, c \in A$$
 satisfy $a \leq b$ and $b \leq c$, then $a \leq c$.

If, in addition,

(1.8.5) for every
$$a, b \in A$$
, we have that $a \leq b$ or $b \leq a$,

then \leq is said to be a linear ordering or total ordering on A.

A binary relation on A that is reflexive and transitive may be called a *pre-order* on A.

Definition 1.8.6 We say that (A, \preceq) is a directed system if \preceq is a partial ordering on A, and if

(1.8.7) for every $a, b \in A$ there is a $c \in A$ such that $a \leq c$ and $b \leq c$.

Sometimes one also considers pre-orderings that satisfy (1.8.7), which can work just as well for some purposes. Note that linearly-ordered sets are directed systems.

1.8.1 Nets of elements of sets

Suppose that (A, \preceq) is a nonempty directed system, and let X be a set.

Definition 1.8.8 A net $\{x_a\}_{a\in A}$ of elements of X indexed by A assigns to each $a\in A$ an element x_a of X.

Thus this is a function defined on A with values in X, where A is also equipped with the partial ordering \leq . If A is the set \mathbf{Z}_+ of positive integers with the standard ordering, then this is the same as a sequence of elements of X.

1.8.2 Convergence of nets

Now let (X, τ) be a topological space.

Definition 1.8.9 A net $\{x_a\}_{a\in A}$ of elements of X indexed by A is said to converge to an element x of X if for every open subset U of X with $x\in U$ there is a $b\in A$ such that

$$(1.8.10) x_a \in U$$

for every $a \in A$ with $b \leq a$.

This reduces to the earlier definition of convergent sequences when $A = \mathbf{Z}_+$ with the standard ordering.

1.8.3 Uniqueness of the limit in Hausdorff spaces

Proposition 1.8.11 Let $\{x_a\}_{a\in A}$ be a net of elements of X that converges to elements x and x' of X. If (X, τ) is Hausdorff, then x = x'.

Suppose for the sake of a contradiction that $x \neq x'$. This implies that there are disjoint open subsets U and V of X such that $x \in U$ and $x' \in V$, because X is Hausdorff. Using the convergence of $\{x_a\}_{a \in A}$ to x, we get that there is a $b \in A$ such that (1.8.10) holds for every $a \in A$ with $b \leq a$. Similarly, there is a $c \in A$ such that

$$(1.8.12) x_a \in V$$

for every $a \in A$ with $c \leq a$, because $\{x_a\}_{a \in A}$ converges to x' in X.

There is an $a \in A$ such that $b \leq a$ and $c \leq a$, by the definition of a directed system. In this case, we get that $x_a \in U \cap V$, contradicting the disjointness of U and V.

Conversely, if (X, τ) is not Hausdorff, then one can find a net of elements of X indexed by some nonempty directed system that converges to two different elements of X. If (X, τ) also satisfies the first countability condition, then one can find a sequence of elements of X that converges to two different elements of X.

1.9 More on convergent nets

Let (A, \preceq) be a nonempty directed system, and let (X, τ) be a topological space.

Proposition 1.9.1 Let E be a subset of X, and let $\{x_a\}_{a\in A}$ be a net of elements of E indexed by A. If $\{x_a\}_{a\in A}$ converges to an element x of X, then x is adherent to E in X. If, for every $a \in A$, we also have that $x_a \neq x$, then x is a limit point of E in X.

This follows easily from the definitions, as in the case of sequences.

Proposition 1.9.2 Let Y be a subset of X, let $\{x_a\}_{a\in A}$ be a net of elements of Y indexed by A, and let x be an element of Y. Under these conditions, $\{x_a\}_{a\in A}$ converges to x in X if and only if $\{x_a\}_{a\in A}$ converges to x in Y, with respect to the induced topology.

This can also be verified directly from the definitions, as before.

1.9.1 Using a local base to get a directed system

Let $x \in X$ be given, and let $\mathcal{B}(x)$ be a local base for the topology of X at x. Let us define a binary relation \leq on $\mathcal{B}(x)$ by

(1.9.3)
$$U \leq V$$
 when $U, V \in \mathcal{B}(x)$ satisfy $V \subseteq U$.

Proposition 1.9.4 With respect to the binary relation \leq defined in (1.9.3), $\mathcal{B}(x)$ is a directed system.

It is easy to see that \leq is a partial ordering on $\mathcal{B}(x)$. To check that $\mathcal{B}(x)$ is a directed system with respect to \leq , let $U, V \in \mathcal{B}(x)$ be given. Thus U and V are open subsets of X that contain x, so that $U \cap V$ is an open set that contains x too. It follows that there is a $W \in \mathcal{B}(x)$ such that

$$(1.9.5) W \subseteq U \cap V,$$

by definition of a local base. This implies that $U \leq W$ and $V \leq W$, as desired.

1.9.2 A criterion for the convergence of a net

Proposition 1.9.6 If $\{x_U\}_{U\in\mathcal{B}(x)}$ is a net of elements of X indexed by $\mathcal{B}(x)$ such that

$$(1.9.7) x_U \in U$$

for every $U \in \mathcal{B}(x)$, then $\{x_U\}_{U \in \mathcal{B}(x)}$ converges to x in X.

Let W be an arbitrary open subset of X that contains x. By the definition of a local base, there is a $V \in \mathcal{B}(x)$ such that $V \subseteq W$. If $U \in \mathcal{B}(x)$ and $V \preceq U$, then

$$(1.9.8) x_U \in U \subseteq V \subseteq W,$$

as desired.

Corollary 1.9.9 If x is adherent to $E \subseteq X$, then there is a net $\{x_U\}_{U \in \mathcal{B}(x)}$ of elements of E indexed by $\mathcal{B}(x)$ that converges to x. If x is a limit point of E in X, then we may also ask that $x_U \neq x$ for every $U \in \mathcal{B}(x)$.

If x is adherent to E, then we can choose

$$(1.9.10) x_U \in E \cap U$$

for every $U \in \mathcal{B}(x)$. If x is a limit point of E, then we can also choose x_U to be different from x for every $U \in \mathcal{B}(x)$.

1.10 Regular topological spaces

Let (X, τ) be a topological space.

Definition 1.10.1 We say that (X, τ) is regular in the strict sense if for every point $p \in X$ and closed set $E \subseteq X$ with $p \notin E$ there are disjoint open subsets U and V of X such that

$$(1.10.2) p \in U and E \subseteq V.$$

If (X, τ) also satisfies the zeroth separation condition, then (X, τ) is said to be regular in the strong sense.

Sometimes one may say that (X,τ) is regular when X is regular in the strict sense. In this case, one may say that X satisfies the third separation condition, or equivalently that (X,τ) is a T_3 space, when X is regular in the strong sense. However, the opposite convention is sometimes used as well. Alternatively, regularity, the third separation condition, and T_3 spaces may be used to refer to regularity in the strong sense, and regularity in the strict sense may be described in other ways.

1.10.1 Regularity in the strong sense implies Hausdorff

Proposition 1.10.3 If (X, τ) is regular in the strong sense, then (X, τ) is Hausdorff.

To see this, let x and y be distinct elements of X. Thus there is an open set that contains one of x and y and not the other, because (X, τ) satisfies the zeroth separation condition. Equivalently, this means that there is a closed set in X that contains one of x and y, and not the other. It follows that x and y are contained in disjoint open subsets of X, as desired.

1.10.2 Regularity and semimetrics

Proposition 1.10.4 *If the topology on* X *is determined by a semimetric* $d(\cdot, \cdot)$, then X is regular in the strict sense. If $d(\cdot, \cdot)$ is a metric on X, then X is regular in the strong sense.

Of course, the second statement follows from the first.

Lemma 1.10.5 If $p \in X$ and r is a nonnegative real number, then

$$(1.10.6) V(p,r) = \{x \in X : d(p,x) > r\}$$

is an open set in X.

If $x \in V(p,r)$, then t = d(p,x) - r > 0, and one can check that

$$(1.10.7) B(x,t) \subseteq V(p,r),$$

using the triangle inequality.

To prove the first part of Proposition 1.10.4, let $p \in X$ and a closed set $E \subseteq X$ be given, with $p \notin E$. Thus $X \setminus E$ is an open set in X that contains p, so that

$$(1.10.8) B(p,r) \subseteq X \setminus E$$

for some r > 0. It is easy to see that U = B(p, r/2) and V = V(p, r/2) are disjoint open subsets of X that contain p and E, respectively.

1.10.3 Regularity of R with respect to τ_+, τ_-

Proposition 1.10.9 The real line is regular in the strong sense with respect to the topologies τ_+ and τ_- defined in Subsection 1.3.2.

Of course, we already know that \mathbf{R} is Hausdorff with respect to τ_+ and τ_- , and so we only need to verify regularity in the strict sense. We shall do this for τ_+ , the argument for τ_- being analogous. Let $p \in \mathbf{R}$ and a closed set $E \subseteq \mathbf{R}$ with respect to τ_+ be given, with $p \notin E$. This means that $\mathbf{R} \setminus E$ is an open set in \mathbf{R} with respect to τ_+ that contains p, so that there is a $b \in \mathbf{R}$ such that p < b and

$$(1.10.10) [p,b) \subseteq \mathbf{R} \setminus E.$$

Equivalently,

$$(1.10.11) E \subseteq \mathbf{R} \setminus [p, b) = (-\infty, p) \cup [b, +\infty).$$

In this situation, [p,b) and $(-\infty,p) \cup [b,+\infty)$ are disjoint open subsets of **R** with respect to τ_+ that contain p and E, as desired.

Let (X, τ) be a topological space again, and let Y be a subset of X.

1.10.4 Regularity and induced topologies

Proposition 1.10.12 If (X, τ) is regular in the strict sense, then Y is regular in the strict sense, with respect to the induced topology. Similarly, if (X, τ) is regular in the strong sense, then Y is regular in the strong sense, with respect to the induced topology.

The second statement follows from the first, and the analogous fact for the zeroth separation condition. To prove the first statement, let $p \in Y$ and a closed set $A \subseteq Y$ with respect to the induced topology be given, with $p \notin A$. Remember that there is a closed set $A_1 \subseteq X$ such that $A = A_1 \cap Y$, because A is a closed set with respect to the induced topology. Note that $p \notin A_1$, because $p \in Y \setminus A$. It follows that there are disjoint open subsets U and V of X that contain p and A_1 , respectively, because (X, τ) is regular in the strict sense. This implies that $U \cap Y$ and $V \cap Y$ are disjoint relatively open subsets of Y that contain p and A, respectively, as desired.

1.11 Completely Hausdorff spaces

Let (X, τ) be a topological space.

Proposition 1.11.1 The following condition is equivalent to (X, τ) being regular in the strict sense: for every point $p \in X$ and open set $W \subseteq X$ with $p \in W$, there is an open set $U \subseteq X$ such that $p \in U$ and the closure \overline{U} of U in X is contained in W.

Let $p \in X$ and an open set $W \subseteq X$ with $p \in W$ be given, so that $E = X \setminus W$ is a closed set that does not contain p. If (X, τ) is regular in the strict sense, then there are disjoint open subsets U and V of X such that $p \in U$ and $E \subseteq V$. It is easy to see that $\overline{U} \cap V = \emptyset$ in this situation, which means that

$$(1.11.2) \overline{U} \subseteq X \setminus V \subseteq X \setminus E = W,$$

as desired.

Conversely, let $p \in X$ and a closed set $E \subseteq X$ with $p \notin E$ be given. Thus $W = X \setminus E$ is an open set that contains p. If (X, τ) satisfies the condition in the statement of the proposition, then there is an open set $U \subseteq X$ such that $p \in U$ and $\overline{U} \subseteq W$. Put $V = X \setminus \overline{U}$, which is an open set in X, because \overline{U} is a closed set. Clearly $U \cap V = \emptyset$, and

$$(1.11.3) E = X \setminus W \subseteq X \setminus \overline{U} = V,$$

as desired.

1.11.1 The completely Hausdorff condition

Definition 1.11.4 We say that (X, τ) is completely Hausdorff if for every pair x, y of distinct elements of X there are open subsets U and V of X such that $x \in U, y \in V$, and

$$(1.11.5) \overline{U} \cap \overline{V} = \emptyset.$$

In this case we may also say that (X,τ) satisfies separation condition number two and a half, or equivalently that (X,τ) is a $T_{2\frac{1}{2}}$ space.

Of course, completely Hausdorff spaces are Hausdorff in particular.

Proposition 1.11.6 *If* (X, τ) *is regular in the strong sense, then* (X, τ) *is completely Hausdorff.*

Let x, y be distinct elements of X. If (X, τ) is regular in the strong sense, then (X, τ) is Hausdorff, and so there are disjoint open subsets U and V of X with $x \in U$ and $y \in V$. Using Proposition 1.11.1, we get open subsets U_1 and V_1 of X such that $x \in U_1, \overline{U_1} \subseteq U, y \in V_1$, and $\overline{V_1} \subseteq V$. It follows that $\overline{U_1} \cap \overline{V_1} = \emptyset$, as desired.

1.11.2 More on completely Hausdorff spaces

Proposition 1.11.7 Let $\widetilde{\tau}$ be another topology on X such that $\tau \subseteq \widetilde{\tau}$. If (X, τ) is completely Hausdorff, then $(X, \widetilde{\tau})$ is completely Hausdorff too.

If E is any subset of X, then let \overline{E}_{τ} and $\overline{E}_{\widetilde{\tau}}$ be the closures of E with respect to τ and $\widetilde{\tau}$, respectively. It is easy to see that

$$(1.11.8) \overline{E}_{\widetilde{\tau}} \subseteq \overline{E}_{\tau},$$

because $\tau \subseteq \widetilde{\tau}$. Using this, one can verify the proposition directly from the definition of the completely Hausdorff property.

Suppose for the moment that (X, τ) is regular in the strict or strong sense. If $\tilde{\tau}$ is as before, then it is not so clear if $(X, \tilde{\tau})$ should have the same property.

Proposition 1.11.9 If (X, τ) is completely Hausdorff and $Y \subseteq X$, then Y is completely Hausdorff with respect to the induced topology.

If $E \subseteq Y$ and $p \in Y$, then one can check that

$$(1.11.10)$$
 p is adherent to E in X

if and only if

(1.11.11) p is adherent to E in Y, with respect to the induced topology.

This implies that the closure of E in Y, with respect to the induced topology, is the same as the intersection of Y with the closure of E in X. Using this, the proposition can be verified directly from the definitions.

1.12 Separated sets and connectedness

Let (X, τ) be a topological space.

Definition 1.12.1 A pair A, B of subsets of X are said to be separated in X if

$$(1.12.2) \overline{A} \cap B = A \cap \overline{B} = \emptyset.$$

Disjoint closed subsets of X are obviously separated in X. It is easy to see that disjoint open subsets of X are separated in X as well.

Proposition 1.12.3 Let Y be a subset of X, and let A, B be subsets of Y. Under these conditions, A and B are separated in X if and only if A and B are separated in Y, with respect to the induced topology.

This follows from the fact that the closure of a subset E of Y, with respect to the induced topology on Y, is the same as the intersection of Y with the closure of E in X.

Proposition 1.12.4 If A and B are separated subsets of X such that $A \cup B = X$, then A and B are both open and closed in X.

It is easy to see that $\overline{A} = A$ and $\overline{B} = B$ in this case, so that A and B are closed sets. This implies that A and B are open sets in X too, because their complements in X are closed sets.

1.12.1 Connectedness

Definition 1.12.5 A subset E of X is said to be connected in X if E cannot be expressed as the union of two nonempty separated sets in X.

It is well known that a subset E of the real line is connected with respect to the standard topology on $\mathbf R$ if and only if for every $x,y\in E$ with x< y, we have that

$$[x,y] \subseteq E.$$

Proposition 1.12.7 Let Y be a subset of X, and let E be a subset of Y. Under these conditions, E is connected as a subset of X if and only if E is connected as a subset of Y, with respect to the induced topology.

This follows from Proposition 1.12.3.

Proposition 1.12.8 The following three statements are equivalent: (a) X is connected, as a subset of itself; (b) X cannot be expressed as the union of two nonempty disjoint open sets; (c) X cannot be expressed as the union of two nonempty disjoint closed sets.

This follows from Proposition 1.12.4.

If X is any set equipped with the indiscrete topology, then X is connected. If X is any set with at least two elements equipped with the discrete topology, then X is not connected.

1.13 Normality and complete normality

Let (X, τ) be a topological space.

Definition 1.13.1 We say that (X,τ) is normal in the strict sense if for every pair A, B of disjoint closed subsets of X, there are disjoint open subsets U, V of X such that $A \subseteq U$ and $B \subseteq V$. If (X,τ) also satisfies the first separation condition, then (X,τ) is said to be normal in the strong sense.

Sometimes one may say that (X, τ) is normal when X is normal in the strict sense, and that X satisfies the fourth separation condition, or equivalently that X is a T_4 space, when X is normal in the strong sense. The opposite convention may be used sometimes as well. One may also use normality, the fourth separation condition, and T_4 spaces for normality in the strong sense, and refer to normality in the strict sense in other ways.

1.13.1 Normality in the strong sense

Proposition 1.13.2 If (X,τ) is normal in the strong sense, then (X,τ) is Hausdorff, and regular in the strong sense.

One can check that (X, τ) satisfies the first separation condition if and only if for every $p \in X$, $\{p\}$ is a closed set in X. Of course, this is the same as saying that $X \setminus \{p\}$ is an open set in X for every $p \in X$. More precisely, the first separation condition says that every element of $X \setminus \{p\}$ is an element of the interior of $X \setminus \{p\}$. The proposition follows easily from this and the definitions.

1.13.2 Normality in the strict sense

Proposition 1.13.3 The following condition is equivalent to (X, τ) being normal in the strict sense: if $A \subseteq X$ is a closed set, $W \subseteq X$ is an open set, and $A \subseteq W$, then there is an open set $U \subseteq X$ such that $A \subseteq U$ and $\overline{U} \subseteq W$.

Let a closed set $A \subseteq X$ and an open set $W \subseteq X$ with $A \subseteq W$ be given, so that $B = X \setminus W$ is a closed set in X that is disjoint from A. If (X, τ) is normal in the strict sense, then there are disjoint open subsets U, V of X such that $A \subseteq U$ and $B \subseteq V$. In this situation, we have that $\overline{U} \cap V = \emptyset$, so that

$$(1.13.4) \overline{U} \subseteq X \setminus V \subseteq X \setminus B = W,$$

as desired.

Conversely, let A and B be disjoint closed subsets of X, so that $W = X \setminus B$ is an open set that contains A. If (X, τ) satisfies the condition in the statement of the proposition, then there is an open set $U \subseteq X$ such that $A \subseteq U$ and $\overline{U} \subseteq W$. Thus $V = X \setminus \overline{U}$ is an open set in X that is disjoint from U. We also have that

$$(1.13.5) B = X \setminus W \subset X \setminus \overline{U} = V,$$

as desired.

1.13.3 Complete normality

Definition 1.13.6 We say that (X,τ) is completely normal in the strict sense if for every pair A, B of separated subsets of X there are disjoint open sets $U, V \subseteq X$ such that $A \subseteq U$ and $B \subseteq V$. If (X,τ) also satisfies the first separation condition, then (X,τ) is said to be completely normal in the strong sense.

If (X,τ) is completely normal in the strict sense, then (X,τ) is normal in the strict sense, because disjoint closed subsets of X are separated in X. Similarly, if (X,τ) is completely normal in the strong sense, then (X,τ) is normal in the strong sense. Sometimes one may say that (X,τ) is completely normal when (X,τ) is completely normal in the strict sense, and that (X,τ) satisfies the fifth separation condition, or equivalently that (X,τ) is a T_5 space, when (X,τ) is completely normal in the strong sense. As before, the opposite convention may sometimes be used too. One may also use complete normality, the fifth separation condition, and T_5 spaces for complete normality in the strong sense, and refer to complete normality in the strict sense in other ways.

Proposition 1.13.7 If (X, τ) is completely normal in the strict sense and Y is a subset of X, then Y is completely normal in the strict sense, with respect to the induced topology. If (X, τ) is completely normal in the strong sense, then Y is completely normal in the strong sense, with respect to the induced topology.

The first part can be obtained from the definitions, using Proposition 1.12.3. The second part follows from the first part and the fact that Y satisfies the first separation condition when (X, τ) has this property.

1.14 Examples of completely normal spaces

Proposition 1.14.1 If X is a set with a semimetric $d(\cdot, \cdot)$, then X is completely normal in the strict sense with respect to the topology determined by $d(\cdot, \cdot)$. If $d(\cdot, \cdot)$ is a metric on X, then X is completely normal in the strong sense with respect to the topology determined by $d(\cdot, \cdot)$.

The second statement follows immediately from the first. To prove the first statement, let A and B be separated subsets of X. Thus, for each $x \in A$, there is a positive real number r(x) such that

$$(1.14.2) B(x, r(x)) \cap B = \emptyset,$$

because $x \notin \overline{B}$. Similarly, for every $y \in B$, there is a t(y) > 0 such that

$$(1.14.3) B(y, t(y)) \cap A = \emptyset,$$

because $y \notin \overline{A}$. Put

(1.14.4)
$$U = \bigcup_{x \in A} B(x, r(x)/2), \quad V = \bigcup_{y \in B} B(y, t(y)/2).$$

These are open sets in X, because open balls are open sets, and unions of open sets are open sets. We also have that $A \subseteq U$ and $B \subseteq V$, because every $x \in A$ is contained in B(x, r(x)/2), and every $y \in B$ is contained in B(y, t(y)/2).

Suppose for the sake of a contradiction that $U \cap V \neq \emptyset$. This means that there are $x \in A$, $y \in B$, and $w \in X$ such that

$$(1.14.5) w \in B(x, r(x)/2) \cap B(y, t(y)/2).$$

It follows that

$$(1.14.6) d(x,y) \le d(x,w) + d(w,y) < r(x)/2 + t(y)/2.$$

However,

$$(1.14.7) d(x,y) \ge r(x), t(y),$$

by (1.14.2) and (1.14.3). This is a contradiction, as desired.

1.14.1 Complete normality of R with respect to τ_+, τ_-

Proposition 1.14.8 The real line is completely normal in the strong sense with respect to the topologies τ_+ and τ_- defined in Subsection 1.3.2.

We already know that the real line is Hausdorff with respect to τ_+ and τ_- , and so we only have to check complete normality in the strict sense. We shall do this for τ_+ , the argument for τ_- being analogous. Let A and B be subsets of \mathbf{R} that are separated with respect to τ_+ . If $x \in A$, then there is a real number $b_1(x)$ such that $x < b_1(x)$ and

$$(1.14.9) [x, b_1(x)) \cap B = \emptyset,$$

because x is not in the closure of B with respect to τ_+ . Similarly, if $y \in B$, then there is a real number $b_2(y)$ such that $y < b_2(y)$ and

$$(1.14.10) [y, b_2(y)) \cap A = \emptyset,$$

because y is not in the closure of A with respect to τ_+ . Put

(1.14.11)
$$U = \bigcup_{x \in A} [x, b_1(x)), \quad V = \bigcup_{y \in B} [y, b_2(y)),$$

which are open subsets of **R** with respect to τ_+ , because these intervals are open sets with respect to τ_+ . Note that $A \subseteq U$ and $B \subseteq V$, by construction.

If $x \in A$ and $y \in B$, then

$$(1.14.12) [x, b_1(x)) \cap [y, b_2(y)) = \emptyset.$$

More precisely, if $x \leq y$, then $b_1(x) \leq y$, by (1.14.9). Similarly, if $y \leq x$, then $b_2(y) \leq x$, by (1.14.10). It is easy to see that (1.14.12) holds in both cases. This implies that $U \cap V = \emptyset$, as desired.

1.15 More on connectedness

Let (X, τ) be a topological space.

Proposition 1.15.1 If E is a connected subset of X, then the closure \overline{E} of E in X is connected too.

Suppose for the sake of a contradiction that \overline{E} is not connected, so that there are nonempty separated subsets A and B of X such that $\overline{E} = A \cup B$. Put

$$(1.15.2) A_1 = A \cap E, B_1 = B \cap E,$$

and note that $E = A_1 \cup B_1$. It is easy to see that A_1 and B_1 are separated in X, because $A_1 \subseteq A$ and $B_1 \subseteq B$. We would like to check that A_1 and B_1 are nonempty.

Let x be an element of A. Thus $x \notin \overline{B}$, because $A \cap \overline{B} = \emptyset$, by hypothesis. This means that there is an open subset U of X such that $x \in U$ and $B \cap U = \emptyset$. However, $E \cap U \neq \emptyset$, because $x \in A \subseteq \overline{E}$. It follows that $A_1 \neq \emptyset$, because $E \cap U \subseteq A_1$. Similarly, $B_1 \neq \emptyset$. This implies that E is not connected, as desired.

Proposition 1.15.3 Let I be a nonempty set, and suppose that E_j is a connected subset of X for every $j \in I$. If

$$(1.15.4) \qquad \bigcap_{j \in I} E_j \neq \emptyset,$$

then $\bigcup_{i \in I} E_i$ is connected in X.

Suppose for the sake of a contradiction that $\bigcup_{j\in I} E_j$ is not connected, so that there are nonempty separated subsets A and B of X such that $\bigcup_{j\in L} E_j = A \cup B$. Let x be an element of $\bigcap_{j\in I} E_j$. We may as well suppose that $x\in A$, by interchanging the roles of A and B, if necessary. Let y be an element of B, and let j_0 be an element of I such that $y\in E_{j_0}$. Put

$$(1.15.5) A_0 = A \cap E_{j_0}, B_0 = B \cap E_{j_0},$$

and observe that $E_{j_0} = A_0 \cup B_0$. As before, A_0 and B_0 are separated in X, because $A_0 \subseteq A$ and $B_0 \subseteq B$. We also have that $x \in A_0$ and $y \in B_0$, by construction, so that $A_0, B_0 \neq \emptyset$. This implies that E_{j_0} is not connected, which is a contradiction.

1.15.1 A criterion for connectedness

Proposition 1.15.6 Let E be a subset of X. Suppose that for every pair of elements x, y of E there is a connected subset E(x,y) of X such that

$$(1.15.7) x, y \in E(x, y)$$

and

$$(1.15.8) E(x,y) \subseteq E.$$

Under these conditions, E is connected in X.

Suppose for the sake of a contradiction that E is not connected in X, so that there are nonempty separated subsets A and B of X such that $E = A \cup B$. Let x be an element of A, let y be an element of B, and let E(x,y) be as in the statement of the proposition. Put

$$(1.15.9) A_2 = A \cap E(x, y), B_2 = B \cap E(x, y),$$

which satisfy $x \in A_2$, $y \in B_2$, and $A_2 \cup B_2 = E(x, y)$. We also have that A_2 and B_2 are separated in X, because $A_2 \subseteq A$ and $B_2 \subseteq B$. This implies that E(x, y) is not connected, which is a contradiction.

Of course, if E is connected, then E satisfies the condition mentioned in Proposition 1.15.6, with E(x,y)=E for every $x,y\in E$.

Alternatively, Proposition 1.15.6 could be obtained from Proposition 1.15.3. More precisely, Proposition 1.15.6 is trivial when $E = \emptyset$. Otherwise, one can fix $x \in E$, and use the fact that E is the union of E(x, y), $y \in E$.

If E_1 and E_2 are connected subsets of X and $E_1 \cap E_2 \neq \emptyset$, then $E_1 \cup E_2$ is connected in X as well, as in Proposition 1.15.3. One could also prove this directly, and use this to obtain Proposition 1.15.3 from Proposition 1.15.6.

Chapter 2

Cardinality and some more topology

2.1 Cardinal equivalence

Let A and B be sets, and let f be a function defined on A with values in B. This is also known as a mapping from A into B, which may be expressed by $f:A\to B$. As usual, we say that f is one-to-one or injective if for every $x,y\in A$ with $x\neq y$, we have that $f(x)\neq f(y)$. We say that f maps A onto B, or equivalently that f is surjective, if for every $z\in B$ there is an $x\in A$ such that f(x)=z. A one-to-one mapping from A onto B is also known as a bijection, or a one-to-one correspondence.

Let C be another set, and let g be a mapping from B into C. Thus the composition $g \circ f$ is the mapping from A into C defined by

$$(2.1.1) (g \circ f)(x) = g(f(x))$$

for every $x \in A$. If f and g are injections, then it is easy to see that $g \circ f$ is an injection. If f and g are surjections, then $g \circ f$ is a surjection. It follows that if f and g are bijections, then $g \circ f$ is a bijection.

If f is a one-to-one mapping from A onto B, then the inverse mapping is defined on B by $f^{-1}(f(x)) = x$ for every $x \in A$, which is a one-to-one mapping from B onto A. If g is a one-to-one mapping from B onto C, then one can check that

$$(2.1.2) (g \circ f)^{-1} = f^{-1} \circ g^{-1}.$$

If there is a one-to-one mapping from A onto B, then let us express this by

$$(2.1.3) #A = #B.$$

Note that #A = #A, because the identity mapping on A is a one-to-one mapping from A onto itself. It is easy to see that #A = #B implies #B = #A, because the inverse of a bijection is a bijection. Similarly, #A = #B and

#B = #C imply that #A = #C, because the composition of bijections is a bijection.

To say that A is a finite set with exactly n elements for some positive integer n means that

By definition, A is countably infinite when

$$(2.1.5) #A = #Z_{+}.$$

2.1.1 The power set

Let $\mathcal{P}(A)$ be the *power set* of A, which is the set of all subsets of A. Suppose that f is a mapping from A into $\mathcal{P}(A)$, and put

$$(2.1.6) B = \{ a \in A : a \notin f(a) \},\$$

which is a subset of A. If $x \in A$, then it is easy to see that

$$(2.1.7) f(x) \neq B.$$

More precisely, if $x \in f(x)$, then $x \notin B$, so that (2.1.7) holds. Otherwise, if $x \notin f(x)$, then $x \in B$, and (2.1.7) holds. Thus f cannot map A onto $\mathcal{P}(A)$. In particular,

$$(2.1.8) #A \neq #\mathcal{P}(A).$$

A set is said to be *uncountable* if it is neither finite nor countably infinite. Note that $\mathcal{P}(\mathbf{Z}_{+})$ is uncountable, by (2.1.8).

Let \mathcal{B} be the set of infinite sequences $x = \{x_j\}_{j=1}^{\infty}$ such that for each positive integer j, $x_j = 0$ or 1. It is easy to see that

$$(2.1.9) #\mathcal{B} = #\mathcal{P}(\mathbf{Z}_+),$$

using the correspondence between elements of \mathcal{B} and the subsets of \mathbf{Z}_{+} on which they are equal to 1.

2.1.2 The cardinality of [0,1]

It is well known that

$$(2.1.10) #B = #[0,1].$$

More precisely, we can start with the usual mapping that sends $x \in \mathcal{B}$ to $\sum_{j=1}^{\infty} x_j \, 2^{-j}$, which is an element of [0,1]. This mapping is surjective but not injective, and we can modify it on a countable set to get a bijection. If n is a nonnegative integer, then put

(2.1.11)
$$A_{n,0} = \{ x \in \mathcal{B} : x_j = 0 \text{ for every } j > n \}$$

and

$$(2.1.12) A_{n,1} = \{ x \in \mathcal{B} : x_j = 1 \text{ for every } j > n \}.$$

These are finite sets with exactly 2^n elements, and

(2.1.13)
$$A = \bigcup_{n=0}^{\infty} (A_{n,0} \cup A_{n,1})$$

is a countably-infinite subset of \mathcal{B} . Also let E be the subset of [0,1] consisting of nonnegative integer multiples of nonnegative integer powers of 1/2, which is countably infinite as well. The restriction of the mapping from \mathcal{B} onto [0,1] mentioned earlier to $\mathcal{B} \setminus A$ is a one-to-one mapping onto $[0,1] \setminus E$. Because A and E are both countably infinite, one can find a one-to-one mapping from A onto E. One can combine these two mappings to get a one-to-one mapping from \mathcal{B} onto [0,1], as desired.

2.2 Comparing cardinalities

If A and B are sets, and if there is a one-to-one mapping from A into B, then we may express this by

$$(2.2.1) #A \le #B.$$

If #A = #B, then it follows that $\#A \leq \#B$ and $\#B \leq \#A$. In particular, $\#A \leq \#A$ automatically. If A, B, and C are sets such that $\#A \leq \#B$ and $\#B \leq \#C$, then it is easy to see that $\#A \leq \#C$, because the composition of injective mappings is injective.

A famous theorem of Bernstein and Schröder states that if A and B are sets with $\#A \leq \#B$ and $\#B \leq \#A$, then #A = #B. This corresponds to Theorem 7 on p33 of [236].

If A is any set, then

$$(2.2.2) a \mapsto \{a\}$$

defines a one-to-one mapping from A into the power set $\mathcal{P}(A)$ of A. This implies that

Let A be a nonempty set, and let f be a one-to-one mapping from A into a set B. Let a_0 be an element of A, and consider the mapping g from B into A defined as follows. We put

$$(2.2.4) g(f(a)) = a$$

for every $a \in A$, and $g(z) = a_0$ for every $z \in B \setminus f(A)$. Here

$$(2.2.5) f(A) = \{ f(a) : a \in A \}$$

is the image of A under f, as usual. Clearly g maps B onto A.

2.2.1 The axiom of choice

Let I be a nonempty set, and suppose that X_j is a nonempty set for every $j \in I$. Under these conditions, the *axiom of choice* states that there is a function f defined on I with values in $\bigcup_{j \in I} X_j$ such that

$$(2.2.6) f(j) \in X_j$$

for every $j \in I$.

Let g be a mapping from a set B onto a nonempty set A. Thus, if $a \in A$, then

$$(2.2.7) g^{-1}(\{a\}) = \{z \in B : g(z) = a\} \neq \emptyset.$$

Using the axiom of choice, we get a mapping f from A into B such that f(a) is an element of $g^{-1}(\{a\})$ for every $a \in A$. This is the same as saying that g(f(a)) = a for every $a \in A$. In particular, it follows that f is a one-to-one mapping from A into B.

If A and B are any two sets, then there are well-known arguments using the axiom of choice to get that $\#A \leq \#B$ or $\#B \leq \#A$.

Let A be an infinite set, and let C be a set such that

$$(2.2.8) #C \le #A.$$

There are well-known arguments using the axiom of choice to get that

$$(2.2.9) #A = \#(A \cup C).$$

Of course, $\#A \leq \#(A \cup C)$, because the obvious inclusion mapping from A into $A \cup C$ is an injection. Note that we can reduce to the case where $A \cap C = \emptyset$, by replacing C with $C \setminus A$.

If C has only finitely or countably many elements, then we can get (2.2.9) more directly, as follows. Let B be a countably-infinite subset of A, so that $B \cup C$ is countably-infinite as well. In particular, there is a one-to-one mapping from B onto $B \cup C$. If $A \cap C = \emptyset$, then we can combine this with the identity mapping on $A \setminus B$ to get a one-to-one mapping from A onto $A \cup C$, as desired.

If a and b are real numbers with a < b, then we get that

$$(2.2.10) #(a,b) = #[a,b) = #(a,b] = #[a,b].$$

One can use explicit mappings to get that #[a,b] = #[0,1] and $\#\mathbf{R} = \#(-1,1)$. One can also use the argument in the previous paragraph to get that $\#(\mathbf{R} \setminus \mathbf{Q}) = \#\mathbf{R}$.

2.3 Products and exponentials

The Cartesian product of two sets A and B is the set $A \times B$ of all ordered pairs (a, b), with $a \in A$ and $b \in B$. Let \widetilde{A} and \widetilde{B} be sets, let ϕ be a one-to-one

mapping from A onto \widetilde{A} , and let ψ be a one-to-one mapping from B onto \widetilde{B} . Under these conditions,

$$(2.3.1) (a,b) \mapsto (\phi(a), \psi(b))$$

is a one-to-one mapping from $A \times B$ onto $\widetilde{A} \times \widetilde{B}$. This implies that

(2.3.2)
$$\#(A \times B) = \#(\widetilde{A} \times \widetilde{B}).$$

Note that

$$(2.3.3) (a,b) \mapsto (b,a)$$

is a one-to-one mapping from $A \times B$ onto $B \times A$. It follows that

(2.3.4)
$$\#(A \times B) = \#(B \times A).$$

If A is an infinite set, then there are well-known arguments using the axiom of choice to get that

$$(2.3.5) #(A \times A) = #A.$$

Of course, this can be shown more directly when A is countably infinite.

2.3.1 Exponentials

If A and B are any two sets again, then the space of all mappings from A into B may be denoted B^A . Let \widetilde{A} and \widetilde{B} be sets with one-to-one mappings ϕ and ψ from A onto \widetilde{A} and from B onto \widetilde{B} , respectively, again. If f is a mapping from A into B, then $\psi \circ f \circ \phi^{-1}$ is a mapping from \widetilde{A} into \widetilde{B} . It is easy to see that

$$(2.3.6) f \mapsto \psi \circ f \circ \phi^{-1}$$

is a one-to-one mapping from B^A onto $\widetilde{B}^{\widetilde{A}}$. Thus

$$(2.3.7) #B^A = #\widetilde{B}^{\widetilde{A}}$$

under these conditions.

If $B = \{0, 1\}$, then B^A may be denoted 2^A . There is a simple one-to-one correspondence between 2^A and the power set $\mathcal{P}(A)$ of A, where a subset of A corresponds to the function on A that is equal to 1 on the subset, and to 0 on the complement of the subset in A. This means that

$$(2.3.8) #2^A = #\mathcal{P}(A).$$

2.3.2 Exponentials of disjoint unions

If A, B, and C are sets with $A \cap B = \emptyset$, then there is a natural one-to-one correspondence between $C^{A \cup B}$ and $C^A \times C^B$. This is because a mapping from $A \cup B$ into C corresponds exactly to a pair of mappings from A into C and from B into C. It follows that

$$(2.3.9) #C^{A \cup B} = \#(C^A \times C^B).$$

If A and B are countably infinite, then it is well known that $A \cup B$ is countably infinite. Of course, it is easy to find disjoint countably-infinite sets, to get that

$$(2.3.10) #C^{\mathbf{Z}_+} = \#(C^{\mathbf{Z}_+} \times C^{\mathbf{Z}_+}).$$

In particular, we can take $C = \{0, 1\}$, to get that

This implies that $\#([0,1] \times [0,1]) = \#[0,1], \#(\mathbf{R} \times \mathbf{R}) = \#\mathbf{R}.$

2.4 Some more exponentials

Let A, B_1 , and B_2 be sets, and suppose that ψ is a one-to-one mapping from B_1 into B_2 . If f is a mapping from A into B_1 , then $\psi \circ f$ is a mapping from A into B_2 . Thus

$$(2.4.1) f \mapsto \psi \circ f$$

defines a mapping from B_1^A into B_2^A . It is easy to see that (2.4.1) is injective, because ψ is injective, by hypothesis. This shows that

$$(2.4.2) #B_1^A \le #B_2^A$$

when $\#B_1 \leq \#B_2$.

Similarly, let A_1 , A_2 , and B be sets, and suppose that ρ is a mapping from A_2 onto A_1 . If f is a mapping from A_1 into B, then $f \circ \rho$ is a mapping from A_2 into B. This defines a mapping

$$(2.4.3) f \mapsto f \circ \rho$$

from B^{A_1} into B^{A_2} . Observe that (2.4.3) is injective, because $\rho(A_2) = A_1$, by hypothesis. This implies that

when $\#A_1 < \#A_2$.

2.4.1 Exponentials of products

If A, B, and C are sets, then there is a simple one-to-one correspondence between the set $C^{A\times B}$ of mappings from $A\times B$ into C and the set $(C^B)^A$ of all mappings from A into the set C^B of all mappings from B into C. More precisely, let f(a,b) be a mapping from $A\times B$ into C, and put

$$(2.4.5) f_a(b) = f(a,b)$$

for every $a \in A$ and $b \in B$. Thus, for each $a \in A$, this defines f_a as a mapping from B into C. It follows that

$$(2.4.6) a \mapsto f_a$$

defines a mapping from A into C^B . This defines a mapping

$$(2.4.7) f \mapsto (2.4.6)$$

from $C^{A \times B}$ into $(C^B)^A$. Conversely, every mapping from A into C^B leads to a mapping from $A \times B$ into C in this way. In particular, we get that

$$\#C^{A \times B} = \#(C^B)^A.$$

If A is an infinite set, then we can take B=A in (2.4.8) and use (2.3.5) to obtain that

(2.4.9)
$$\#(C^A)^A = \#C^{A \times A} = \#C^A.$$

Of course, this also uses (2.3.7) in the second step. Let E be a set such that

$$(2.4.10) #C \le #E \le #C^A.$$

Using (2.4.2), we get that

$$(2.4.11) #C^A \le #E^A \le #(C^A)^A = #C^A,$$

so that

$$(2.4.12) #C^A = #E^A.$$

In particular, we can take $C = \{0, 1\}$ and E = A, to obtain that

$$(2.4.13) #A^A = #2^A.$$

2.5 Strong limit points and separability

Let (X, τ) be a topological space.

Definition 2.5.1 A point $p \in X$ is said to be a strong limit point of a subset E of X if for every open subset U of X with $p \in U$, there are infinitely many elements of E in U. Similarly, $p \in X$ is said to be a condensation point if for every open set $U \subseteq X$ with $p \in U$, $E \cap U$ is uncountable.

Thus a strong limit point of E is automatically a limit point of E, and a condensation point of E is a strong limit point of E. Of course, there are analogous notions using other conditions on the cardinality of $E \cap U$ as in Definition 2.5.1.

Proposition 2.5.2 If $p \in X$ is a limit point of $E \subseteq X$, and if X satisfies the first separation condition, then p is a strong limit point of E in X.

If p is not a strong limit point of E, then there is an open set $U \subseteq X$ such that $p \in U$ and $E \cap U$ has only finitely many elements. If X satisfies the first separation condition, then one can find a smaller open set $V \subseteq X$ such that $p \in V$ and V does not contain any elements of E, other than p, if $p \in E$. This implies that p is not a limit point of E in X.

2.5.1 Separable topological spaces

Definition 2.5.3 We say that (X, τ) is separable if there is a dense subset E of X such that E has only finitely or countably many elements.

Suppose that (X, τ) is a Hausdorff topological space that satisfies the first countability condition, and that E is a dense subset of X. An infinite sequence of elements of E is the same as a function on the set \mathbf{Z}_+ with values in E, which is an element of $E^{\mathbf{Z}_+}$. Let $\mathcal C$ be the collection of all sequences of elements of E that converge to an element of X, so that $\mathcal C$ may be considered as a subset of $E^{\mathbf{Z}_+}$. Remember that the limit of a convergent sequence is unique in a Hausdorff space. Thus

$$(2.5.4) {x_j}_{j=1}^{\infty} \mapsto \lim_{j \to \infty} x_j$$

defines a function on \mathcal{C} with values in X.

If E is dense in X, and (X, τ) satisfies the first countability condition, then every element of X is the limit of a convergent sequence of elements of E. This means that (2.5.4) maps $\mathcal C$ onto X under these conditions. It follows that

$$(2.5.5) #X \le \#\mathcal{C},$$

using the axiom of choice, as in Subsection 2.2.1. This implies that

$$(2.5.6) #X < #E^{\mathbf{Z}_+},$$

because $\mathcal{C} \subseteq E^{\mathbf{Z}_+}$. If E has only finitely or countably many elements, then we obtain that

(2.5.7)
$$#X \le #\mathbf{Z}_{+}^{\mathbf{Z}_{+}} = #2^{\mathbf{Z}_{+}}.$$

Note that the real line is separable with respect to the standard topology, because \mathbf{Q} is a countable dense subset of \mathbf{R} . We have also seen that \mathbf{R} is Hausdorff and satisfies the first countability condition with respect to the standard topology, and that $\#\mathbf{R} = \#2^{\mathbf{Z}_+}$.

2.6 Bases for topologies

Let (X, τ) be a topological space.

Definition 2.6.1 A collection \mathcal{B} of open subsets of X is said to be a base for the topology τ on X if for every $x \in X$ and open subset V of X with $x \in V$ there is an element U of \mathcal{B} such that $x \in U$ and $U \subseteq V$.

If \mathcal{B} is a base for τ and W is an open subset of X, then

$$(2.6.2) W = \bigcup \{U : U \in \mathcal{B}, U \subseteq W\}.$$

More precisely, the right side of (2.6.2) is automatically contained in W. If \mathcal{B} is a base for τ , then every element of W is contained in the union on the right side of (2.6.2). Conversely, if \mathcal{B} is a collection of open subsets of X, and if every open subset of X can be expressed as a union of elements of \mathcal{B} , then it is easy to see that \mathcal{B} is a base for τ .

2.6.1 Bases and local bases

If \mathcal{B} is a base for τ and $x \in X$, then

$$\mathcal{B}(x) = \{ U \in \mathcal{B} : x \in U \}$$

is a local base for the topology of X at x. Conversely, if \mathcal{B} is a collection of open subsets of X, and for each $x \in X$, (2.6.3) is a local base for the topology of X at x, then \mathcal{B} is a base for τ .

2.6.2 The second countability condition

Definition 2.6.4 If there is a base \mathcal{B} for τ such that \mathcal{B} has only finitely or countably many elements, then we say that (X,τ) satisfies the second countability condition.

Suppose that (X, τ) satisfies the second countability condition, and let \mathcal{B} be a base for τ with only finitely or countably many elements. If $x \in X$, then it follows that (2.6.3) has only finitely or countably many elements as well. This means that (X, τ) satisfies the first countability condition.

2.6.3 Bases and semimetrics

Proposition 2.6.5 Let $d(\cdot, \cdot)$ be a semimetric on X, and suppose that τ is the topology determined on X by $d(\cdot, \cdot)$. Also let E be a dense subset of X, and let \mathcal{B}_E be the collection of open balls in X centered at elements of E with radius 1/j for some positive integer j. Under these conditions, \mathcal{B}_E is a base for τ . If E has only finitely or countably many elements, then \mathcal{B}_E has only finitely or countably many elements too.

Let $x \in X$ and a positive real number r be given, and let j be a positive integer such that 2/j < r. Because E is dense in X, there is a $y \in E$ such that

$$(2.6.6) d(x,y) < 1/j.$$

This implies that $x \in B(y, 1/j)$, and one can check that

$$(2.6.7) B(y,1/j) \subseteq B(x,r),$$

using the triangle inequality. It follows that \mathcal{B}_E is a base for τ , because B(y, 1/j) is an element of \mathcal{B}_E , by construction.

If j is a positive integer, then let $\mathcal{B}_{E,j}$ be the collection of open balls in X centered at elements of E with radius 1/j, so that

(2.6.8)
$$\mathcal{B}_E = \bigcup_{j=1}^{\infty} \mathcal{B}_{E,j}.$$

If E has only finitely or countably many elements, then it is easy to see that $\mathcal{B}_{E,j}$ has only finitely or countably many elements for each $j \geq 1$. This implies that (2.6.8) has only finitely or countably many elements, by standard results about countable sets.

2.6.4 Second countability implies separability

Proposition 2.6.9 If (X, τ) satisfies the second countability condition, then (X, τ) is separable.

Let \mathcal{B} be a base for τ . If $U \in \mathcal{B}$ and $U \neq \emptyset$, then let us choose a point x_U in U, and let E be the set of points chosen in this way. It is easy to see that E is dense in X with respect to τ . If \mathcal{B} has only finitely or countably many elements, then one can check that E has only finitely or countably many elements as well.

A topological space that satisfies the second countability condition is sometimes said to be *completely separable*.

2.6.5 Bases and induced topologies

Proposition 2.6.10 Let Y be a subset of X. If \mathcal{B} is a base for τ , then

$$(2.6.11) \mathcal{B}_Y = \{U \cap Y : U \in \mathcal{B}\}$$

is a base for the induced topology on Y.

In particular, if (X,τ) satisfies the second countability condition, then Y satisfies the second countability condition with respect to the induced topology.

The first part can be verified directly from the definitions. In the second part, if \mathcal{B} has only finitely or countably many elements, then one can verify that \mathcal{B}_Y has only finitely or countably many elements.

2.6.6 Some remarks about bases for τ_+, τ_-

Let τ_+ be the topology defined on the real line as in Subsection 1.3.2, and let \mathcal{B}_+ be a base for τ_+ . If $x \in \mathbf{R}$, then there is an element U(x) of \mathcal{B}_+ such that

$$(2.6.12) x \in U(x) \text{ and } U(x) \subseteq [x, +\infty),$$

because $[x, +\infty)$ is an open set with respect to τ_+ . If $y \in \mathbf{R}$ and $x \neq y$, then it is easy to see that

$$(2.6.13) U(x) \neq U(y).$$

If \mathcal{B}_+ has only finitely or countably many elements, then one could use this to get that the real line has only finitely or countably many elements, which is a contradiction. Thus \mathcal{B}_+ is uncountable, which means that (\mathbf{R}, τ_+) does not satisfy the second countability condition.

Remember that \mathbf{Q} is a countable set which is dense in \mathbf{R} with respect to τ_+ , so that (\mathbf{R}, τ_+) is separable. It follows that there is no metric on \mathbf{R} for which τ_+ is the corresponding topology, by Proposition 2.6.5.

Of course, there are analogous statements for the topology τ_{-} defined on **R** in Subsection 1.3.2.

2.6.7 Some more remarks about bases

Let (X, τ) be any topological space again, and let \mathcal{B} be a base for τ . If V is an open subset of X, then put

$$\phi(V) = \{ U \in \mathcal{B} : U \subseteq V \},$$

which is a subset of \mathcal{B} . This defines ϕ as a mapping from τ into the power set $\mathcal{P}(\mathcal{B})$ of all subsets of \mathcal{B} .

If \mathcal{E} is any subset of \mathcal{B} , then put

(2.6.15)
$$\psi(\mathcal{E}) = \bigcup_{U \in \mathcal{E}} U,$$

which is interpreted as being the empty set when $\mathcal{E} = \emptyset$. Note that (2.6.15) is an open subset of X, because the elements of \mathcal{B} are open subsets of X, by hypothesis. This defines ψ as a mapping from $\mathcal{P}(\mathcal{B})$ into τ .

The composition $\psi \circ \phi$ of ϕ and ψ is the identity mapping on τ , as in (2.6.2). Thus ϕ is an injection, and ψ is a surjection. In particular,

$$(2.6.16)$$
 $\#\tau < \#\mathcal{P}(\mathcal{B}).$

2.7 Continuous mappings

Let X and Y be topological spaces.

Definition 2.7.1 A mapping f from X into Y is said to be continuous at a point $x \in X$ if for every open subset V of Y with $f(x) \in V$ there is an open subset U of X such that $x \in U$ and

$$(2.7.2) f(U) \subseteq V.$$

It is easy to see that this reduces to the usual definition of continuity for mappings between metric spaces when the topologies are determined by metrics.

2.7.1 Continuity and convergent nets

Proposition 2.7.3 Let (A, \preceq) be a nonempty directed system, and let $\{x_a\}_{a \in A}$ be a net of elements of X indexed by A that converges to x in X. If a mapping f from X into Y is continuous at x, then $\{f(x_a)\}_{a \in A}$ converges to f(x), as a net of elements of Y indexed by A.

Let V be an open subset of Y that contains f(x), and let U be an open subset of X that contains x and satisfies (2.7.2). Because $\{x_a\}_{a\in A}$ converges to x in X, there is a $b\in A$ such that $x_a\in U$ for every $a\in A$ with $b\preceq a$. This implies that

$$(2.7.4) f(x_a) \in f(U) \subseteq V$$

for every $a \in A$ with $b \leq a$, as desired.

2.7.2 The converse

Conversely, let f be a mapping from X into Y, and suppose that f is not continuous at $x \in X$. This means that there is an open set $V \subseteq Y$ such that $f(x) \in V$, and for every open set $U \subseteq X$ with $x \in U$, we have that

$$(2.7.5) f(U) \not\subseteq V.$$

Let $\mathcal{B}(x)$ be a local base for the topology of X at x. If $U \in \mathcal{B}(x)$, then let us choose a point $x_U \in U$ such that

$$(2.7.6) f(x_U) \notin V.$$

Remember that $\mathcal{B}(x)$ is a directed system with respect to the partial ordering defined by putting $U_1 \leq U_2$ when $U_1, U_2 \in \mathcal{B}(x)$ satisfies $U_2 \subseteq U_1$. In this situation, we have seen that $\{x_U\}_{U \in \mathcal{B}(x)}$ converges to x, as a net of elements of X indexed by $\mathcal{B}(x)$. However, $\{f(x_U)\}_{U \in \mathcal{B}(x)}$ does not converge to f(x) in Y.

2.7.3 Another version of the converse

Suppose for the moment that there is a local base for the topology of X at x with only finitely or countably many elements. This implies that there is a sequence $U_1(x), U_2(x), U_3(x), \ldots$ of open subsets of X such that $x \in U_j(x)$ for every $j \geq 1$, the collection of $U_j(x)$'s, $j \geq 1$, is a local base for the topology of X at x, and $U_{j+1}(x) \subseteq U_j(x)$ for every $j \geq 1$.

If $V \subseteq Y$ is as in the previous subsection, then, for each positive integer j, we can choose $x_j \in U_j(x)$ such that

$$(2.7.7) f(x_i) \notin V$$

It follows that $\{x_j\}_{j=1}^{\infty}$ converges to x in X, while $\{f(x_j)\}_{j=1}^{\infty}$ does not converge to f(x) in Y.

2.7.4 Continuity on X

Definition 2.7.8 A mapping f from X into Y is said to be continuous on X if f is continuous at every point in X.

Proposition 2.7.9 A mapping f from X into Y is continuous if and only if for every open set $V \subseteq Y$,

$$(2.7.10) f^{-1}(V) = \{x \in X : f(x) \in V\}$$

is an open subset of X. This is also equivalent to the condition that for every closed set $E \subseteq Y$, $f^{-1}(E)$ is a closed set in X.

Suppose that f is continuous, and let V be an open subset of Y. If x is an element of $f^{-1}(V)$, then one can use the continuity of f at x to get that x is

an element of the interior of $f^{-1}(V)$ in X. This implies that $f^{-1}(V)$ is an open set in X, as desired.

Conversely, let $x \in X$ and an open set $V \subseteq Y$ that contains f(x) be given. Observe that $U = f^{-1}(V)$ contains x and satisfies

(2.7.11)
$$f(U) = f(f^{-1}(V)) \subseteq V.$$

If $f^{-1}(V)$ is an open set in X, then the conditions for continuity of f at x are satisfied.

The second part of the proposition can be obtained from the first part, using the fact that

$$(2.7.12) f^{-1}(Y \setminus E) = X \setminus f^{-1}(E)$$

for every $E \subseteq Y$.

2.7.5 Some basic classes of examples

Let us consider a few basic classes of examples of continuous mappings. If X is equipped with the discrete topology, then any mapping f from X into any topological space Y is continuous. Similarly, if Y is equipped with the indiscrete topology, then any mapping f from any topological space X into Y is continuous.

Let X be a set, and let τ , $\widetilde{\tau}$ be topologies on X. It is easy to see that the identity mapping on X is continuous as a mapping from X equipped with $\widetilde{\tau}$ into X equipped with τ if and only if $\tau \subseteq \widetilde{\tau}$.

Let Y be a topological space, and let X be a subset of Y, equipped with the induced topology. One can check that the obvious inclusion mapping from X into Y, which sends every element of X to itself, is continuous as a mapping from X into Y.

Let X be any set equipped with the indiscrete topology, and let Y be a topological space that satisfies the zeroth separation condition. If f is a continuous mapping from X into Y, then f is constant on X.

2.7.6 Locally constant mappings

A mapping f from a topological space X into a set Y is said to be *locally constant* if for every $x \in X$ there is an open subset U(x) of X such that $x \in U(x)$ and f is constant on U(x). It is easy to see that a locally constant mapping from a topological space X into a topological space Y is continuous. If Y is equipped with the discrete topology, then any continuous mapping from a topological space X into Y is locally constant.

2.7.7 Locally constant mappings and connectedness

If f is a locally constant mapping from a connected topological space X into a set Y, then one can check that f is constant on X. Conversely, if X is not connected, then one can find a locally constant mapping on X that is not constant.

2.8 More on continuity

Let X, Y, and Z be topological spaces.

Proposition 2.8.1 Let f be a mapping from X into Y, and let g be a mapping from Y into Z. If f is continuous at $x \in X$, and g is continuous at f(x), then their composition $g \circ f$ is continuous at x, as a mapping from X into Z. In particular, if f is continuous on X, and g is continuous on Y, then $g \circ f$ is continuous as a mapping from X into Z.

The first part can be verified directly from the definitions. The second part follows from the first part, and can also be obtained from the characterization of continuity in Proposition 2.7.9.

2.8.1 Homeomorphisms

Definition 2.8.2 A one-to-one mapping f from X onto Y is said to be a homeomorphism if f is continuous as a mapping from X into Y, and the inverse mapping f^{-1} is continuous as a mapping from Y into X.

The identity mapping on X is a homeomorphism from X onto itself. If f is a homeomorphism from X onto Y, then the inverse mapping f^{-1} is a homeomorphism from Y onto X. In this case, if g is a homeomorphism from Y onto Z, then $g \circ f$ is a homeomorphism from X onto Z.

Consider the mapping f from the real line onto itself defined by f(x) = -x for every $x \in \mathbf{R}$. This is a homeomorphism as a mapping from the real line onto itself with respect to the standard topology. This is also a homeomorphism from the real line equipped with the topology τ_+ defined in Subsection 1.3.2 onto \mathbf{R} equipped with the analogous topology τ_- .

2.8.2 Open mappings

Definition 2.8.3 A mapping f from X into Y is said to be an open mapping if for every open subset U of X, f(U) is an open subset of Y.

If f is a one-to-one mapping from X onto Y, then f is an open mapping if and only if f^{-1} is continuous as a mapping from Y into X.

2.8.3 Restrictions of continuous mappings

Proposition 2.8.4 Let f be a continuous mapping from X into Y, and let X_0 be a subset of X. Under these conditions, the restriction of f to X_0 is continuous as a mapping from X_0 into Y, with respect to the topology induced on X_0 by the topology on X.

This can be verified directly from the definitions. Alternatively, one can use the fact that the restriction of f to X_0 is the same as the composition of f with the inclusion mapping from X_0 into X.

Proposition 2.8.5 Let Y_0 be a subset of Y, and let f be a mapping from X into Y_0 . In this situation, f is continuous as a mapping from X into Y_0 , with respect to the topology induced on Y_0 by the topology on Y, if and only if f is continuous as a mapping from X into Y.

This can be verified directly from the definitions too.

2.8.4 Continuity and connectedness

Proposition 2.8.6 If f is a continuous mapping from X into Y, and E is a connected subset of X, then f(E) is connected as a subset of Y.

It is not difficult to show this directly, using the definition of connectedness in terms of separated sets. Alternatively, one can use the previous two propositions to reduce to the case where E=X and f(X)=Y. This also uses Proposition 1.12.7, concerning connectedness and induced topologies, in each of X and Y. If Y is not connected, then Y can be expressed as the union of two nonempty disjoint open sets V_1 and V_2 . The continuity of f implies that $f^{-1}(V_1)$ and $f^{-1}(V_2)$ are open subsets of X. It is easy to see that $f^{-1}(V_1)$ and $f^{-1}(V_2)$ are disjoint subsets of X whose union is X, because of the corresponding properties of V_1 and V_2 in Y. We also have that $f^{-1}(V_1)$, $f^{-1}(V_2) \neq \emptyset$, because V_1 , $V_2 \neq \emptyset$, and f(X) = Y. This implies that X is not connected, as desired.

2.8.5 Path connectedness

Definition 2.8.7 A subset E of X is said to be path connected in X if for every pair of points $x, w \in E$ there is a continuous mapping p from the closed unit interval [0,1] into X such that p(0) = x, p(1) = y, and

$$(2.8.8) p([0,1]) \subseteq E.$$

More precisely, this uses the topology induced on [0,1] by the standard topology on \mathbf{R} .

Proposition 2.8.9 If E is a path-connected subset of X, then E is connected in X.

Let $x, w \in E$ be given, and let p be as in the definition of path connectedness. Remember that [0,1] is connected as a subset of the real line, with respect to the standard topology on \mathbf{R} . This implies that [0,1] is connected as a subset of itself, with respect to the induced topology. It follows that p([0,1]) is connected as a subset of X. One can use this to get the connectedness of E, as in Proposition 1.15.6.

Proposition 2.8.10 If f is a continuous mapping from X into Y, and E is a path-connected subset of X, then f(E) is path connected in Y.

This follows from the definition of path connectedness, and the fact that compositions of continuous mappings are continuous.

Proposition 2.8.11 Let X_0 be a subset of X, and let E be a subset of X_0 . Under these conditions, E is path connected in X_0 , with respect to the topology induced by the topology on X, if and only if E is path connected in X.

This follows from the definition of path connectedness and Proposition 2.8.5.

2.9 The product topology

Let I be a nonempty set, and let X_j be a set for each $j \in I$. The Cartesian product of the X_j 's, $j \in I$, is the set

$$(2.9.1) \qquad \prod_{j \in I} X_j$$

consisting of all functions f defined on I with values in $\bigcup_{j\in I} X_j$ such that $f(j) \in X_j$ for every $j \in I$.

If $I = \{1, ..., n\}$ for some positive integer n, then the Cartesian product of the X_j 's may be denoted

$$(2.9.2) \qquad \prod_{j=1}^{n} X_j,$$

and its elements identified with n-tuples $x = (x_1, \ldots, x_n)$ such that $x_j \in X_j$ for each $j = 1, \ldots, n$. Similarly, if $I = \mathbf{Z}_+$, then the Cartesian product of the X_j 's may be denoted

$$(2.9.3) \qquad \prod_{j=1}^{\infty} X_j,$$

and its elements identified with sequences $x = \{x_j\}_{j=1}^{\infty}$ with $x_j \in X_j$ for every $j \ge 1$.

Suppose for the moment that U_j and V_j are subsets of X_j for each $j \in I$. Observe that

(2.9.4)
$$\left(\prod_{j \in I} U_j\right) \cap \left(\prod_{j \in I} V_j\right) = \prod_{j \in I} (U_j \cap V_j).$$

2.9.1 Defining the product topology

Suppose now that X_j is a topological space for every $j \in I$. A subset W of $\prod_{j \in I} X_j$ is said to be an open set with respect to the *product topology* if for every $f \in W$ and $j \in I$ there is an open subset U_j of X_j such that

$$(2.9.5) f(j) \in U_j for every j \in I,$$

$$(2.9.6) \qquad \prod_{j \in I} U_j \subseteq W,$$

and

(2.9.7)
$$U_j = X_j$$
 for all but finitely many $j \in I$.

Note that (2.9.5) is the same as saying that

$$(2.9.8) f \in \prod_{j \in I} U_j.$$

Of course, (2.9.7) holds automatically when I has only finitely many elements. One can verify that this defines a topology on $\prod_{j \in I} X_j$.

2.9.2 The strong product topology

Similarly, let us say that a subset W of $\prod_{j\in I} X_j$ is an open set with respect to the *strong product topology* if for every $f\in W$ and $j\in I$ there is an open set $U_j\subseteq X_j$ that satisfies (2.9.5) and (2.9.6). One can check that this defines a topology on $\prod_{j\in I} X_j$ as well. Of course, the strong product topology on $\prod_{j\in I} X_j$ is automatically at least as strong as the product topology. If I has only finitely many elements, then the product topology and the strong product topology on $\prod_{j\in I} X_j$ are the same.

2.9.3 Some basic open sets

If U_j is an open subset of X_j for every $j \in I$, then it is easy to see that

$$(2.9.9) U = \prod_{j \in I} U_j$$

is an open set in $\prod_{j\in I} X_j$ with respect to the strong product topology. The collection of these open sets is a base for the strong product topology on $\prod_{j\in I} X_j$, by construction.

Similarly, if $U_j \subseteq X_j$ is an open set for every $j \in I$, and if $U_j = X_j$ for all but finitely many $j \in I$, then (2.9.9) is an open set with respect to the product topology on $\prod_{j \in I} X_j$. The collection of these open sets is a base for the product topology on $\prod_{j \in I} X_j$, by construction again.

2.9.4 Using the discrete topology on the factors

Suppose for the moment that X_j is equipped with the discrete topology for every $j \in I$. In this case, the strong product topology on $\prod_{j \in I} X_j$ is the same as the discrete topology. Suppose in addition that $X_j \neq \emptyset$ for every $j \in I$, and that X_j has at least two elements for infinitely many $j \in I$. Under these conditions, one can check that the product topology on $\prod_{j \in I} X_j$ is not the discrete topology.

2.9.5 The standard topology on \mathbb{R}^n

If n is a positive integer, then the space \mathbf{R}^n of n-tuples of real numbers is the same as the nth Cartesian power of \mathbf{R} . The product topology on \mathbf{R}^n corresponding to the standard topology on the real line may be considered as the standard topology on \mathbf{R}^n .

2.10 Coordinate mappings

Let I be a nonempty set again, let X_j be a set for each $j \in I$, and let

$$(2.10.1) X = \prod_{j \in I} X_j$$

be the corresponding Cartesian product. If $l \in I$, then let p_l be the lth coordinate mapping from X into X_l , which is defined by

$$(2.10.2) p_l(f) = f(l)$$

for every $f \in X$.

2.10.1 Some properties of p_l

Let W_l be a subset of X_l , and for each $j \in I$, let $\widetilde{W}_{j,l}$ be the subset of X_j defined by

(2.10.3)
$$\widetilde{W}_{j,l} = W_l \text{ when } j = l$$
$$= X_j \text{ when } j \neq l.$$

Observe that

(2.10.4)
$$p_l^{-1}(W_l) = \prod_{j \in I} \widetilde{W}_{j,l}.$$

Similarly, let U_j be a subset of X_j for every $j \in I$, and put $U = \prod_{j \in I} U_j$. If $U_j \neq \emptyset$ for every $j \in I$, then

$$(2.10.5) p_l(U) = U_l$$

for every $l \in I$.

2.10.2 Continuity of the coordinate mappings

Suppose now that X_j is a topological space for each $j \in I$.

Proposition 2.10.6 If $l \in I$, then p_l is a continuous mapping from X into X_l with respect to the product topology on X, and hence with respect to the strong product topology on X. We also have that p_l is an open mapping from X into X_l with respect to the strong product topology on X, and hence with respect to the product topology.

If W_l is an open subset of X_l for some $l \in I$, then (2.10.4) is an open set in X with respect to the product topology. This implies the first part of the proposition. The second part can be obtained from (2.10.5).

2.10.3 Products of closed sets

Corollary 2.10.7 If E_j is a closed set in X_j for each $j \in I$, then $E = \prod_{j \in I} E_j$ is a closed set in X with respect to the product topology, and hence with respect to the strong product topology.

If $l \in I$, then $p_l^{-1}(E_l)$ is a closed set in X with respect to the product topology, by the previous proposition. This implies that

(2.10.8)
$$E = \bigcap_{l \in I} p_l^{-1}(E_l)$$

is a closed set in X with respect to the product topology as well. Alternatively, one can verify that $X \setminus E$ is an open set with respect to the product topology on X.

2.10.4 Convergence of nets and the product topology

Proposition 2.10.9 Let (A, \preceq) be a nonempty directed system, let $\{f_a\}_{a\in A}$ be a net of elements of X indexed by A, and let f be another element of X. Under these conditions, $\{f_a\}_{a\in A}$ converges to f with respect to the product topology on X if and only if for each $l \in I$, $\{f_a(l)\}_{a\in A}$ converges to f(l) as a net of elements of X_l .

The "only if" part of the proposition follows from the fact that p_l is continuous with respect to the product topology on X for every $l \in I$. The "if" part can be verified directly from the definitions.

Of course, if $\{f_a\}_{a\in A}$ converges to f with respect to the strong product topology on X, then $\{f_a\}_{a\in A}$ converges to f with respect to the product topology on X.

2.10.5 Closures of products of sets

Proposition 2.10.10 Let A_j be a subset of X_j for each $j \in I$, and let $\overline{A_j}$ be the closure of A_j in X_j for every $j \in I$, as usual. The closure of $A = \prod_{j \in I} A_j$ in X with respect to the product topology or the strong product topology is equal to $\prod_{j \in I} \overline{A_j}$.

This can be verified directly from the definitions. More precisely, the closure of A with respect to the strong product topology on X is contained in the closure of A with respect to the product topology on X. One can check that the closure of A with respect to the product topology on X is contained in $\prod_{j\in I} \overline{A_j}$, and that $\prod_{j\in I} \overline{A_j}$ is contained in the closure of A with respect to the strong product topology on X.

2.10.6 Some basic separation conditions for the product

Proposition 2.10.11 Let $k \in \{0,1,2,2\frac{1}{2}\}$ be given, and suppose that X_j satisfies the kth separation condition for every $j \in I$. Under these conditions, X satisfies the kth separation condition with respect to the product topology, and hence with respect to the strong product topology.

If f and g are distinct elements of X, then there is an $l \in I$ such that $f(l) \neq g(l)$. In this situation, the kth separation condition for f and g in X can be obtained from the analogous condition for f(l) and g(l) in X_l .

2.10.7 Regularity of the product

Proposition 2.10.12 If X_j is regular in the strict sense for every $j \in I$, then X is regular in the strict sense with respect to the product topology and the strong product topology.

Let $f \in X$ be given, and let W be an open subset of X with respect to the product topology or strong product topology, with $f \in W$. It follows that for every $j \in I$ there is an open set $V_j \subseteq X_j$ such that $f(j) \in V_j$,

$$(2.10.13) V = \prod_{j \in I} V_j \subseteq W,$$

and $V_j = X_j$ for all but finitely many $j \in I$ in the case of the product topology. If $j \in I$, there is an open set $U_j \subseteq X_j$ such that

$$(2.10.14) f(j) \in U_j \text{ and } \overline{U_j} \subseteq V_j,$$

because X_j is regular in the strict sense. We may as well take

(2.10.15)
$$U_j = X_j \text{ when } V_j = X_j,$$

so that $U_j = X_j$ for all but finitely many $j \in I$ in the case of the product topology. Thus $U = \prod_{j \in I} U_j$ is an open set in X with respect to the product topology or strong product topology, as appropriate. Of course, $f \in U$, by construction. The closure of U with respect to the product topology or strong product topology is equal to $\prod_{j \in I} \overline{U_j}$, as before. This is contained in V, by construction, and hence in W, as desired.

Corollary 2.10.16 If X_j is regular in the strong sense for every $j \in I$, then X is regular in the strong sense with respect to the product topology and the strong product topology.

2.11 Bases and finite products

Let X_1, \ldots, X_n be finitely many topological spaces, and let $X = \prod_{j=1}^n X_j$ be their Cartesian product, equipped with the product topology. Also let

$$(2.11.1) x = (x_1, \dots, x_n) \in X$$

be given. Suppose that for each j = 1, ..., n, $\mathcal{B}_j(x_j)$ is a local base for the topology of X_j at x_j . Put

(2.11.2)
$$\mathcal{B}(x) = \left\{ \prod_{j=1}^{n} U_j : U_j \in \mathcal{B}_j(x_j) \text{ for each } j = 1, \dots, n \right\}.$$

It is easy to see that this is a local base for the product topology on X at x.

2.11.1 Countability of $\mathcal{B}(x)$

Suppose that for each j = 1, ..., n, $\mathcal{B}_j(x_j)$ has only finitely or countably many elements. We would like to verify that (2.11.2) has only finitely or countably many elements as well. In this case,

$$(2.11.3) \qquad \qquad \prod_{j=1}^{n} \mathcal{B}_{j}(x_{j})$$

has only finitely or countably many elements, by standard results. There is an obvious mapping from (2.11.3) onto (2.11.2), which sends (U_1, \ldots, U_n) in (2.11.3) to $\prod_{j=1}^n U_j$ in (2.11.2). One can use this to show that (2.11.2) has only finitely or countably many elements, because of the analogous property for (2.11.3).

2.11.2 Another local base for the product topology at x

Alternatively, suppose again that for each $j=1,\ldots,n$, there is a local base for the topology of X_j at x_j with only finitely or countably many elements. This means that for each $j=1,\ldots,n$ there is a sequence $\{U_{j,l}(x_j)\}_{l=1}^{\infty}$ of open subsets of X_j such that $x_j \in U_{j,l}(x_j)$ for every $l \geq 1$, and the collection of $U_{j,l}(x_j)$'s, $l \geq 1$, is a local base for the topology of X_j at x_j .

We may also ask that

$$(2.11.4) U_{j,l+1}(x_j) \subseteq U_{j,l}(x_j)$$

for each j = 1, ..., n and $l \ge 1$, since otherwise we can replace $U_{j,l}(x_j)$ with $\bigcap_{k=1}^{l} U_{j,k}(x_j)$ for every j = 1, ..., n and $l \ge 1$, as before.

(2.11.5)
$$U_l(x) = \prod_{j=1}^n U_{j,l}(x_j)$$

for each $l \in \mathbf{Z}_+$, so that $U_l(x)$ is an open subset of X with respect to the product topology, and $x \in U_l(x)$. One can check that the collection of $U_l(x)$'s, $l \in \mathbf{Z}_+$, is a local base for the product topology on X at x.

2.11.3 Some bases for the product topology on X

Similarly, let \mathcal{B}_j be a base for the topology of X_j for each j = 1, ..., n. It is easy to see that

(2.11.6)
$$\mathcal{B} = \left\{ \prod_{j=1}^{n} U_j : U_j \in \mathcal{B}_j \text{ for each } j = 1, \dots, n \right\}$$

is a base for the product topology on X.

If \mathcal{B}_j has only finitely or countably many elements for each $j = 1, \ldots, n$, then (2.11.6) has only finitely or countably many elements as well. More precisely,

$$(2.11.7) \qquad \qquad \prod_{j=1}^{n} \mathcal{B}_{j}$$

has only finitely or countably many elements in this situation, as before. We can map (2.11.7) onto (2.11.6), by sending (U_1, \ldots, U_n) in (2.11.7) to $\prod_{j=1}^n U_j$ in (2.11.6). One can use this and the fact that (2.11.7) has only finitely or countably many elements to get that (2.11.6) has only finitely or countably many elements, as before.

2.12 Bases and countable products

Let $X_1, X_2, X_3,...$ be an infinite sequence of topological spaces, and let $X = \prod_{j=1}^{\infty} X_j$ be their Cartesian product, equipped with the product topology. Also let $x = \{x_j\}_{j=1}^{\infty} \in X$ be given, and let $\mathcal{B}_j(x_j)$ be a local base for the topology of X_j at x_j for each $j \in \mathbf{Z}_+$.

2.12.1 A local base for the product topology at x

If n is a positive integer, then put

(2.12.1)
$$\mathcal{B}^n(x) = \left\{ \prod_{j=1}^{\infty} U_j : U_j \in \mathcal{B}_j(x_j) \text{ for each } j = 1, \dots, n, \right.$$

and $U_j = X_j \text{ when } j > n \right\}.$

Note that every element of $\mathcal{B}^n(x)$ is an open subset of X with respect to the product topology. One can check that

(2.12.2)
$$\mathcal{B}(x) = \bigcup_{n=1}^{\infty} \mathcal{B}^n(x)$$

is a local base for the product topology on X at x.

2.12.2 Countability of $\mathcal{B}^n(x)$

Suppose that for every $j \in \mathbf{Z}_+$, $\mathcal{B}_j(x_j)$ has only finitely or countably many elements. This implies that

$$(2.12.3) \qquad \qquad \prod_{j=1}^{n} \mathcal{B}_j(x_j)$$

has only finitely or countably many elements for each $n \in \mathbf{Z}_+$, by standard arguments, as before.

Let (U_1, \ldots, U_n) be an element of (2.12.3), and put $U_j = X_j$ when j is larger than n. Under these conditions, $\prod_{j=1}^{\infty} U_j$ is an element of (2.12.1). This defines a mapping from (2.12.3) onto (2.12.1). One can use this and the fact that (2.12.3) has only finitely or countably many elements to obtain that (2.12.1) has only finitely or countably many elements for each $n \in \mathbb{Z}_+$. It follows that (2.12.2) has only finitely or countably many elements as well.

2.12.3 Another countable local base for X at x

Alternatively, suppose again that for each positive integer j, there is a local base for the topology of X_j at x_j with only finitely or countably many elements. This means that for each $j \in \mathbf{Z}_+$ there is a sequence $\{U_{j,l}(x_j)\}_{l=1}^{\infty}$ of open subsets of X_j such that $x_j \in U_{j,l}(x_j)$ for every $l \geq 1$, and the collection of $U_{j,l}(x_j)$'s, $l \geq 1$, is a local base for the topology of X_j at x_j . We may also ask that $U_{j,l+1}(x_j) \subseteq U_{j,l}(x_j)$ for every $j,l \geq 1$, as before.

If $n \in \mathbf{Z}_+$, then put

(2.12.4)
$$V_{j,n}(x_j) = U_{j,n}(x_j) \text{ for } j = 1, ..., n$$

= X_j when $j > n$.

Using this, we put

(2.12.5)
$$V_n(x) = \prod_{j=1}^{\infty} V_{j,n}(x_j).$$

This is an open subset of X with respect to the product topology, and $x \in V_n(x)$. One can check that the collection of $V_n(x)$'s, $n \in \mathbf{Z}_+$, is a local base for the product topology on X at x.

2.12.4 Some bases for X

Now let \mathcal{B}_j be a base for the topology of X_j for every positive integer j. If $n \in \mathbf{Z}_+$, then put

(2.12.6)
$$\mathcal{B}^n = \left\{ \prod_{j=1}^{\infty} U_j : U_j \in \mathcal{B}_j \text{ for each } j = 1, \dots, n \right.$$
 and $U_j = X_j$ when $j > n \right\}.$

Every element of \mathcal{B}^n is an open subset of X with respect to the product topology, by construction. It is not difficult to verify that

(2.12.7)
$$\mathcal{B} = \bigcup_{n=1}^{\infty} \mathcal{B}^n$$

is a base for the product topology on X.

2.12.5 Countable bases for X

Suppose that \mathcal{B}_j has only finitely or countably many elements for each $j \in \mathbf{Z}_+$. This implies that $\prod_{j=1}^n \mathcal{B}_j$ has only finitely or countably many elements for each positive integer n, as usual. One can use this to check that (2.12.6) has only finitely or countably many elements for every $n \in \mathbf{Z}_+$, as before. This implies that (2.12.7) has only finitely or countably many elements too.

2.12.6 Some local bases for the strong product topology

Let I be a nonempty set, let X_j be a topological space for each $j \in I$, and let $X = \prod_{j \in I} X_j$ be the Cartesian product of the X_j 's. Also let $f \in X$ be given, and let $\mathcal{B}_j(f(j))$ be a local base for the topology of X_j at f(j) for each $j \in I$. Under these conditions, one can check that

(2.12.8)
$$\mathcal{B}(f) = \left\{ \prod_{j \in I} U_j : U_j \in \mathcal{B}_j(f(j)) \text{ for each } j \in I \right\}$$

is a local base for the strong product topology on X at f.

If $\mathcal{B}_j(f(j))$ has only finitely many elements for some $j \in I$, then one can reduce to the case where $\mathcal{B}_j(f(j))$ has only one element, because the interesection of finitely many open subsets of X_j is an open set too. Note that

(2.12.9)
$$\prod_{j \in I} \mathcal{B}_j(f(j))$$

is uncountable when $\mathcal{B}_j(f(j))$ has at least two elements for infinitely many $j \in I$.

2.12.7 Some bases for the strong product topology

Similarly, if \mathcal{B}_j is a base for the topology of X_j for each $j \in I$, then one can verify that

(2.12.10)
$$\mathcal{B} = \left\{ \prod_{j \in I} U_j : U_j \in \mathcal{B}_j \text{ for each } j \in I \right\}$$

is a base for the strong product topology on X. Suppose that $X_j \neq \emptyset$ for each $j \in I$, so that $\mathcal{B}_j \neq \emptyset$ for each $j \in I$. If \mathcal{B}_j has at least two elements for infinitely many $j \in I$, then

$$(2.12.11) \qquad \prod_{j \in I} \mathcal{B}_j$$

is uncountable.

2.13 Finite products and semimetrics

Let n be a positive integer, and let X_j be a set with a semimetric $d_j(\cdot,\cdot)$ for each $j=1,\ldots,n$. Put $X=\prod_{j=1}^n X_j$. If $x=(x_1,\ldots,x_n),y=(y_1,\ldots,y_n)\in X$, then one can check that

(2.13.1)
$$d(x,y) = \max_{1 \le j \le n} d_j(x_j, y_j)$$

defines a semimetric on X. If $d_j(\cdot,\cdot)$ is a metric on X_j for every $j=1,\ldots,n$, then (2.13.1) is a metric on X.

2.13.1 Open balls in X with respect to d

Let

$$(2.13.2) B_j(x_j, r) = \{ w_j \in X_j : d_j(x_j, w_j) < r \}$$

be the open ball in X_j centered at $x_j \in X_j$ with radius r > 0 with respect to $d_j(\cdot, \cdot)$ for each $j = 1, \ldots, n$, and let

$$(2.13.3) B(x,r) = \{ w \in X : d(x,w) < r \}$$

be the open ball in X centered at $x \in X$ with radius r > 0 with respect to $d(\cdot, \cdot)$. It is easy to see that

(2.13.4)
$$B(x,r) = \prod_{j=1}^{n} B_j(x_j, r)$$

for every $x \in X$ and r > 0, directly from the definitions. More precisely, $w \in X$ satisfies d(x, w) < r if and only if $d_j(x_j, w_j) < r$ for each $j = 1, \ldots, n$.

2.13.2 Relation with the product topology on X

Let us use the term "product topology" to refer to the product topology on X corresponding to the topologies determined on X_1, \ldots, X_n by the semimetrics d_1, \ldots, d_n , respectively. It is well known and not difficult to show that this is the same as the topology determined on X by the semimetric (2.13.1).

Remember that an open ball is an open set with respect to the topology determined by the corresponding semimetric. This implies that (2.13.4) is an open set in X with respect to the product topology for every $x \in X$ and r > 0. One can use this to check that every open set in X with respect to the topology determined by the semimetric (2.13.1) is also an open set with respect to the product topology.

If $W \subseteq X$ is an open set with respect to the product topology, then one can verify that W is an open set with respect to the topology determined by the semimetric (2.13.1). More precisely, if $x \in W$, then one can find an r > 0 such that (2.13.4) is contained in W.

2.13.3 Another semimetric on X

It is easy to see that

(2.13.5)
$$\widetilde{d}(x,y) = \sum_{j=1}^{n} d_j(x_j, y_j)$$

defines a semimetric on X as well. If $d_j(\cdot,\cdot)$ is a metric on X_j for each $j=1,\ldots,n$, then (2.13.5) is a metric on X, as before. Observe that

$$(2.13.6) d(x,y) \le \widetilde{d}(x,y) \le n d(x,y)$$

for every $x, y \in X$. Using this, one can check that the topologies determined on X by (2.13.1) and (2.13.5) are the same.

2.13.4 An additional semimetric on X

Similarly, put

(2.13.7)
$$\widehat{d}(x,y) = \left(\sum_{j=1}^{n} d_j(x_j, y_j)^2\right)^{1/2}$$

for each $x, y \in X$, using the nonnegative square root on the right side. One can check that this satisfies the triangle inequality on X, using the triangle inequality for the standard Euclidean norm on \mathbf{R}^n . Using this, it is easy to see that (2.13.7) is a semimetric on X, which is a metric on X when $d_j(\cdot, \cdot)$ is a metric on X_j for each $j = 1, \ldots, n$.

One can also verify that

$$(2.13.8) d(x,y) \le \widehat{d}(x,y) \le n^{1/2} d(x,y)$$

for every $x, y \in X$. This implies that the topologies determined on X by (2.13.1) and (2.13.7) are the same, as before.

2.14 Truncating semimetrics

Let X be a set, let d(x, y) be a semimetric on X, and let t be a positive real number. If $x, y \in X$, then put

$$(2.14.1) d_t(x,y) = \min(d(x,y),t).$$

One can check that this defines a semimetric on X. If d(x, y) is a metric on X, then (2.14.1) is a metric on X too.

2.14.1 Open balls in X with respect to d_t

If $x \in X$ and r is a positive real number, then let

$$(2.14.2) B_d(x,r) = \{ w \in X : d(x,w) < r \}$$

be the open ball in X centered at x with radius r with respect to $d(\cdot, \cdot)$. Similarly, let

$$(2.14.3) B_{d_t}(x,r) = \{ w \in X : d_t(x,w) < r \}$$

be the open ball in X centered at x with radius r with respect to (2.14.1). Observe that

(2.14.4)
$$B_{d_t}(x,r) = B_d(x,r) \text{ when } r \le t$$
$$= X \text{ when } r > t.$$

Using this, one can check that the topologies determined on X by $d(\cdot, \cdot)$ and (2.14.1) are the same.

2.15 Countable products and semimetrics

Let $X_1, X_2, X_3, ...$ be a sequence of sets, let $d_j(\cdot, \cdot)$ be a semimetric on X_j for each $j \ge 1$, and put $X = \prod_{j=1}^{\infty} X_j$. Put

$$(2.15.1) d'_i(x_i, y_i) = \min(d_i(x_i, y_i), 1/j)$$

for each $j \in \mathbf{Z}_+$ and $x_j, y_j \in X_j$. As in the previous section, for every $j \in \mathbf{Z}_+$, (2.15.1) defines a semimetric on X_j that determines the same topology on X_j as $d_j(\cdot,\cdot)$. If $d_j(\cdot,\cdot)$ is a metric on X_j , then (2.15.1) is a metric on X_j too, as before.

2.15.1 A semimetric on X

If $x = \{x_j\}_{j=1}^{\infty}, y = \{y_j\}_{j=1}^{\infty} \in X$, then put

(2.15.2)
$$d(x,y) = \max_{j \in \mathbf{Z}_+} d'_j(x_j, y_j).$$

Of course, this is equal to 0 when $d'_j(x_j, y_j) = 0$ for every $j \ge 1$, and in particular when x = y. Otherwise, suppose that there is a positive integer j_0 such that $d'_{j_0}(x_{j_0}, y_{j_0}) > 0$. In this case,

$$(2.15.3) d'_{i}(x_{i}, y_{i}) \le 1/j < d_{i_{0}}(x_{i_{0}}, y_{i_{0}})$$

for all but finitely many $j \in \mathbf{Z}_+$. This means that the right side of (2.15.2) reduces to the maximum of finitely many terms, so that the maximum is attained.

One can check that (2.15.2) defines a semimetric on X. If $d_j(\cdot, \cdot)$ is a metric on X_j for each $j \ge 1$, then (2.15.2) is a metric on X as well.

2.15.2 The corresponding open balls

If $j \in \mathbf{Z}_+$, $x_j \in X_j$, and r is a positive real number, then let

$$(2.15.4) B_j(x_j, r) = \{ w_j \in X_j : d_j(x_j, w_j) < r \}$$

and

$$(2.15.5) B'_i(x_i, r) = \{ w_i \in X_i : d'_i(x_i, w_i) < r \}$$

be the open balls in X_j centered at x_j with radius r with respect to $d_j(\cdot,\cdot)$ and (2.15.1), respectively. Thus

$$(2.15.6) B'_j(x_j, r) = B_j(x_j, r) \text{when } r \le 1/j$$
$$= X_j \text{when } r > 1/j,$$

as in (2.14.4). Also let

$$(2.15.7) B(x,r) = \{ w \in X : d(x,w) < r \}$$

be the open ball in X centered at $x \in X$ with radius r > 0 with respect to (2.15.2).

One can verify that

(2.15.8)
$$B(x,r) = \prod_{j=1}^{\infty} B'_{j}(x_{j},r)$$

for every $x \in X$ and r > 0. This is the same as saying that $w \in X$ satisfies d(x, w) < r if and only if $d'_j(x_j, w_j) < r$ for every $j \in \mathbf{Z}_+$. This uses the fact that the maximum is attained on the right side of (2.15.2).

2.15.3 Relation between d and the product topology on X

Let us use the term "product topology" to refer to the product topology on X corresponding to the topology determined on X_j by $d_j(\cdot,\cdot)$ for each $j\in\mathbf{Z}_+$. Observe that (2.15.8) is an open set in X with respect to the product topology for every $x\in X$ and r>0, because of (2.15.6). One can use this to check that every open set in X with respect to (2.15.2) is an open set with respect to the product topology.

If W is an open subset of X with respect to the product topology and $x \in W$, then it is not too difficult to find an r > 0 such that (2.15.8) is contained in W, so that W is an open set with respect to the topology determined by (2.15.2). Thus the product topology on X is the same as the topology determined on X by (2.15.2).

Chapter 3

Compactness and related topics

3.1 Compact sets

Let X be a topological space, and let K be a subset of X.

Definition 3.1.1 If A is a nonempty set, U_{α} is an open subset of X for each $\alpha \in A$, and

$$(3.1.2) K \subseteq \bigcup_{\alpha \in A} U_{\alpha},$$

then $\{U_{\alpha}\}_{{\alpha}\in A}$ is said to be an open covering of K in X. We say that K is compact in X if every open covering of K can be reduced to a finite subcovering. This means that for every open covering $\{U_{\alpha}\}_{{\alpha}\in A}$ of K in X there are finitely many indices $\alpha_1, \ldots, \alpha_n \in A$ such that

(3.1.3)
$$K \subseteq \bigcup_{j=1}^{n} U_{\alpha_j}.$$

3.1.1 Some examples of compact sets

If K has only finitely many elements, then it is easy to see that K is compact.

If X is equipped with the discrete topology and $K \subseteq X$ is compact, then K has only finitely many elements. This follows by covering K with subsets of X with only one element, which are open sets in this case.

If a, b are real numbers with $a \leq b$, then it is well known that the closed interval [a, b] in the real line is compact, with respect to the standard topology on \mathbf{R} .

If X is any set equipped with the indiscrete topology, then every subset of X is compact.

3.1.2 Another class of examples

Similarly, let X be any set equipped with the cofinite topology. This means that $U \subseteq X$ is an open set when $U = \emptyset$ or $X \setminus U$ has only finitely many elements. We also have that every subset K of X is compact in this situation.

More precisely, let $\{U_{\alpha}\}_{\alpha\in A}$ be any open covering of K in X. We may as well suppose that $K\neq\emptyset$, which implies that there is an $\alpha_0\in A$ such that $U_{\alpha_0}\neq\emptyset$. It follows that U_{α_0} contains all but finitely many elements of X, and in particular that U_{α_0} contains all but finitely many elements of K. The finitely many elements of $K\setminus U_{\alpha_0}$ can easily be covered by finitely many U_{α} 's, so that K can be covered by finitely many U_{α} 's, as desired.

3.1.3 Compactness and induced topologies

Let X be any topological space again.

Proposition 3.1.4 Let Y be a subset of X, and let K be a subset of Y. Under these conditions, K is compact as a subset of X if and only if K is compact as a subset of Y, with respect to the induced topology.

This can be verified directly from the definitions.

3.1.4 Intersections of compact sets and closed sets

Proposition 3.1.5 If K is a compact subset of X, and E is a closed set in X, then $K \cap E$ is compact in X as well.

To see this, let $\{U_{\alpha}\}_{{\alpha}\in A}$ be an arbitrary open covering of $K\cap E$ in X. Note that $X\setminus E$ is an open set in X, because E is a closed set. We also have that

(3.1.6)
$$K = (K \cap E) \cup (K \setminus E) \subseteq \left(\bigcup_{\alpha \in A} U_{\alpha}\right) \cup (X \setminus E).$$

Thus the collection of U_{α} 's, $\alpha \in A$, together with $X \setminus E$, forms an open covering of K in X. Because K is compact in X, there are finitely many indices $\alpha_1, \ldots, \alpha_n \in A$ such that

(3.1.7)
$$K \subseteq \left(\bigcup_{j=1}^{n} U_{\alpha_{j}}\right) \cup (X \setminus E).$$

This implies that

(3.1.8)
$$K \cap E \subseteq \bigcup_{j=1}^{n} U_{\alpha_j},$$

as desired.

3.1.5 An additional class of examples

Proposition 3.1.9 If $\{x_j\}_{j=1}^{\infty}$ is a sequence of elements of X that converges to an element x of X, then

$$(3.1.10) K = \{x_j : j \in \mathbf{Z}_+\} \cup \{x\}$$

is a compact set in X.

This can be verified directly from the definitions.

3.2 Compactness in Hausdorff spaces

Let X be a topological space.

Proposition 3.2.1 If X is Hausdorff, K is a compact subset of X, and x is an element of X not in K, then there are disjoint open subsets U and V of X such that $x \in U$ and $K \subseteq V$.

If $y \in K$, then $x \neq y$, and hence there are disjoint open sets $U(y), V(y) \subseteq X$ such that $x \in U(y)$ and $y \in V(y)$. The collection of open sets V(y) of this type, with $y \in K$, forms an open covering of K in X. Because K is compact, there are finitely many elements y_1, \ldots, y_n of K such that

$$(3.2.2) K \subseteq \bigcup_{j=1}^{n} V(y_j).$$

Put

(3.2.3)
$$U = \bigcap_{j=1}^{n} U(y_j), \quad V = \bigcup_{j=1}^{n} V(y_j),$$

which are open subsets of X. By construction, $x \in U$ and $K \subseteq V$. It is easy to see that

$$(3.2.4) U \cap V = \emptyset,$$

because $U(y_j) \cap V(y_j) = \emptyset$ for each $j = 1, \dots, n$. In particular,

$$(3.2.5) U \subseteq X \setminus K,$$

because $K \subseteq V$.

Corollary 3.2.6 If X is Hausdorff, and K is a compact subset of X, then K is a closed set in X.

Indeed, the previous argument shows that every element of $X \setminus K$ is in the interior of $X \setminus K$.

3.2.1 Disjoint compact sets

Proposition 3.2.7 If X is Hausdorff, and H, K are disjoint compact subsets of X, then there are disjoint open subsets U_1 , V_1 of X such that $H \subseteq U_1$ and $K \subseteq V_1$.

If $x \in H$, then $x \in X \setminus K$, and so there are disjoint open subsets $U_1(x)$, $V_1(x)$ of X such that $x \in U_1(x)$ and $K \subseteq V_1(x)$, as before. The collection of open sets $U_1(x)$ of this type, with $x \in H$, forms an open covering of H in X. Thus there are finitely many elements x_1, \ldots, x_m of H such that

$$(3.2.8) H \subseteq \bigcup_{l=1}^{m} U_1(x_l),$$

because H is compact. Put

(3.2.9)
$$U_1 = \bigcup_{l=1}^m U_1(x_l), \quad V_1 = \bigcap_{l=1}^m V_1(x_l),$$

which are open subsets of X. Observe that $H \subseteq U_1$ and $K \subseteq V_1$, by construction. One can check that

$$(3.2.10) U_1 \cap V_1 = \emptyset,$$

because $U_1(x_l) \cap V_1(x_l) = \emptyset$ for every l = 1, ..., m.

Corollary 3.2.11 If X is Hausdorff and compact as a subset of itself, then X is normal in the strong sense.

This follows from the previous proposition and the fact that closed subsets of X are compact, by Proposition 3.1.5.

3.2.2 Disjoint compact and closed sets

Proposition 3.2.12 If X is regular in the strict sense, K is a compact subset of X, E is a closed set in X, and

$$(3.2.13) K \cap E = \emptyset,$$

then there are disjoint open subsets U_0 , V_0 of X such that $K \subseteq U_0$ and $E \subseteq V_0$.

This can be shown using the same type of argument as for Proposition 3.2.7.

Corollary 3.2.14 If X is regular in the strict sense, and X is compact as a subset of itself, then X is normal in the strict sense.

This uses the fact that closed subsets of X are compact when X is compact, as before.

3.3 Continuity and compactness

Let X and Y be topological spaces.

Theorem 3.3.1 If f is a continuous mapping from X into Y, and K is a compact subset of X, then f(K) is a compact subset of Y.

Let $\{V_{\alpha}\}_{{\alpha}\in A}$ be an arbitrary open covering of f(K) in Y. Note that $f^{-1}(V_{\alpha})$ is an open set in X for each ${\alpha}\in A$, because f is continuous. It is easy to see that

(3.3.2)
$$K \subseteq \bigcup_{\alpha \in A} f^{-1}(V_{\alpha}),$$

because $f(K) \subseteq \bigcup_{\alpha \in A} V_{\alpha}$, by hypothesis. Thus $\{f^{-1}(V_{\alpha})\}_{\alpha \in A}$ is an open covering of K in X. If K is compact, then there are finitely many indices $\alpha_1, \ldots, \alpha_n \in A$ such that

(3.3.3)
$$K \subseteq \bigcup_{j=1}^{n} f^{-1}(V_{\alpha_j}).$$

This implies that

(3.3.4)
$$f(K) \subseteq \bigcup_{j=1}^{n} V_{\alpha_{j}},$$

as desired.

3.3.1 The extreme value theorem

Corollary 3.3.5 (Extreme value theorem) Suppose that f is a continuous real-valued function on X, with respect to the standard topology on \mathbf{R} . If K is a nonempty compact subset of X, then f attains its maximum and minimum on K.

In this case, f(K) is a closed set in \mathbf{R} , as in the previous section, because the real line is Hausdorff with respect to the standard topology. It is well known and not difficult to show that compact subsets of the real line are bounded too. If $K \neq \emptyset$, then $f(K) \neq \emptyset$, and one can use the previous statements to show that f(K) contains its supremum and infimum in \mathbf{R} , as desired.

3.3.2 Compactness and homeomorphisms

Proposition 3.3.6 If X is compact, Y is Hausdorff, and f is a one-to-one continuous mapping from X onto Y, then f is a homeomorphism.

Let $g = f^{-1}$ be the inverse of f, as a mapping from Y onto X. We have seen previously that g is continuous if and only if for every closed set $E \subseteq X$, $g^{-1}(E)$ is a closed set in Y. In this situation, this means that for every closed set $E \subseteq X$, f(E) is a closed set in Y.

If X is compact, and $E \subseteq X$ is a closed set, then it follows that E is compact, as in Proposition 3.1.5. In this case, we get that f(E) is compact in Y, as before. If Y is Hausdorff, then f(E) is a closed set in Y, as in the previous section. This implies that $g = f^{-1}$ is continuous in this situation, as in the preceding paragraph.

3.4 The limit point property

Let X be a topological space.

Definition 3.4.1 A subset K of X is said to have the limit point property if for every subset L of K such that L has infinitely many elements, there is an element x of K that is a limit point of L in X. Similarly, let us say that K has the strong limit point property if for every infinite subset L of K there is an $x \in K$ such that x is a strong limit point of L in X.

If K has the strong limit point property, then K automatically has the limit point property. If K has the limit point property, and if X satisfies the first separation condition, then K has the strong limit point property.

3.4.1 Compactness implies the strong limit point property

Proposition 3.4.2 If K is a compact subset of X, then K has the strong limit point property.

To see this, let an infinite subset L of K be given. Suppose for the sake of a contradiction that L does not have a strong limit point in K. This means that for each $x \in K$ there is an open set $U(x) \subseteq X$ such that $x \in U(x)$ and $U(x) \cap L$ has only finitely many elements. Thus K can be covered by open sets of this type. If K is compact, then K can be covered by finitely many open sets of this type. This implies that L has only finitely many elements, because $L \subseteq K$. This contradicts the hypothesis that L have infinitely many elements, as desired.

3.4.2 A partial converse

Proposition 3.4.3 Suppose that $K \subseteq X$ has the strong limit point property. If V_1, V_2, V_3, \ldots is an infinite sequence of open subsets of X such that

$$(3.4.4) K \subseteq \bigcup_{j=1}^{\infty} V_j,$$

then there is a positive integer n such that

$$(3.4.5) K \subseteq \bigcup_{j=1}^{n} V_j.$$

Suppose for the sake of a contradiction that for each positive integer n,

$$(3.4.6) K \not\subseteq \bigcup_{j=1}^{n} V_{j}.$$

Thus, for every $n \in \mathbf{Z}_+$, we can choose a point

$$(3.4.7) x_n \in K \setminus \Big(\bigcup_{j=1}^n V_j\Big).$$

Let L be the set of points x_n , $n \in \mathbf{Z}_+$, that have been chosen in this way. Let us check that L has infinitely many elements. Otherwise, there is an element y of K such that $y = x_n$ for infinitely many $n \in \mathbf{Z}_+$. Note that $y \in V_{j_0}$ for some $j_0 \in \mathbf{Z}_+$, because $y \in K$. This implies that $x_n \neq y$ when $n \geq j_0$, by the way that x_n was chosen. This contradicts the hypothesis that $y = x_n$ for infinitely many n, as desired.

Thus L has infinitely many elements. If K has the strong limit point property, then it follows that there is an $x \in K$ such that x is a strong limit point of L in X. In particular, $x \in V_{j_1}$ for some $j_1 \in \mathbf{Z}_+$, because $x \in K$. This implies that V_{j_1} contains infinitely many elements of L, because x is a strong limit point of L in X, and V_{j_1} is an open set in X. This means that $x_n \in V_{j_1}$ for infinitely many $n \in \mathbf{Z}_+$, by the way that L was chosen. However, if $n \geq j_1$, then $x_n \notin V_{j_1}$, by construction. This is a contradiction, so that (3.4.5) holds for some $n \in \mathbf{Z}_+$.

3.4.3 Some related results

Proposition 3.4.8 Let K be a subset of a subset Y of X. Under these conditions, K has the limit point property as a subset of X if and only if K has the limit point property as a subset of Y, with respect to the induced topology. Similarly, K has the strong limit point property as a subset of X if and only if K has the strong limit point property in Y, with respect to the induced topology.

This is easy to see, directly from the definitions.

Proposition 3.4.9 *Let* E *be a closed set in* X. *If* $K \subseteq X$ *has the limit point property, then* $K \cap E$ *has the limit point property too. If* K *has the strong limit point property, then* $K \cap E$ *has the strong limit point property as well.*

This follows from the fact that E contains all of its limit points in X.

Proposition 3.4.10 Let f be a continuous mapping from X into another topological space Y. If $K \subseteq X$ has the strong limit point property, then f(K) has the strong limit point property in Y.

Let L be an infinite subset of f(K). If $y \in L$, then choose an element w of K such that y = f(w), and let L_0 be the set of points in K chosen in this way.

Thus L_0 is an infinite subset of K, which has a strong limit point x in K, by hypothesis. One can check that f(x) is a strong limit point of L in Y, using the continuity of f at x. This also uses the fact that f is injective on L_0 , by construction.

3.5 Sequential compactness

Let $\{x_j\}_{j=1}^{\infty}$ be a sequence of elements of some set X. Also let $\{j_l\}_{l=1}^{\infty}$ be a strictly increasing sequence of positive integers, so that $j_l < j_{l+1}$ for every $l \in \mathbf{Z}_+$. Under these conditions, $\{x_{j_l}\}_{l=1}^{\infty}$ is called a *subsequence* of $\{x_j\}_{j=1}^{\infty}$. Now let X be a topological space. If $\{x_j\}_{j=1}^{\infty}$ converges to an element x of X, then it is easy to see that every subsequence $\{x_{j_l}\}_{l=1}^{\infty}$ of $\{x_j\}_{j=1}^{\infty}$ converges to x in X too.

Definition 3.5.1 A subset K of X is said to be sequentially compact if for every sequence $\{x_j\}_{j=1}^{\infty}$ of elements of K there is a subsequence $\{x_{j_l}\}_{l=1}^{\infty}$ of $\{x_j\}_{j=1}^{\infty}$ that converges to an element x of K.

3.5.1 A consequence of sequential compactness

Proposition 3.5.2 If $K \subseteq X$ is sequentially compact, then K has the strong limit point property.

Let L be an infinite subset of K. Because L has infinitely many elements, we can find a sequence $\{x_j\}_{j=1}^{\infty}$ of distinct elements of L. If K is sequentially compact, then there is a subsequence $\{x_{j_l}\}_{l=1}^{\infty}$ of $\{x_j\}_{j=1}^{\infty}$ that converges to an element x of K. If U is an open set in X that contains x, then it follows that $x_{j_l} \in U$ for all but finitely many $l \geq 1$. This implies that x is a strong limit point of L in X, because the terms of the sequence are distinct elements of L.

3.5.2 A criterion for sequential compactness

Proposition 3.5.3 If $K \subseteq X$ has the strong limit point property, and if X satisfies the first countability condition, then K is sequentially compact.

Let $\{x_j\}_{j=1}^{\infty}$ be a sequence of elements of K, and let

$$(3.5.4) L = \{x_i : j \in \mathbf{Z}_+\}$$

be the subset of K consisting of the terms in the sequence. If L has only finitely many elements, then there is an $x \in K$ such that $x_j = x$ for infinitely many $j \in \mathbf{Z}_+$. This means that there is a subsequence $\{x_{j_l}\}_{l=1}^{\infty}$ of $\{x_j\}_{j=1}^{\infty}$ such that $x_{j_l} = x$ for every $l \geq 1$. Of course, $\{x_{j_l}\}_{l=1}^{\infty}$ converges to x in X in this case.

Suppose now that L has infinitely many elements. If K has the strong limit point property, then there is an element x of K that is a strong limit point of

L in X. Thus, if V is an open subset of X that contains x, then V contains infinitely many elements of L. This implies that

$$(3.5.5) x_i \in V$$

for infinitely many positive integers j.

If X satisfies the first countability condition, then there is a local base $\mathcal{B}(x)$ for the topology of X at the point x mentioned in the preceding paragraph such that $\mathcal{B}(x)$ has only finitely or countably many elements. Equivalently, there is a sequence $U_1(x), U_2(x), U_3(x), \ldots$ of open subsets of X that contain x and form a local base for the topology of X at x. We may also suppose that

$$(3.5.6) U_{n+1}(x) \subseteq U_n(x)$$

for every $n \geq 1$, by replacing $U_n(x)$ with $U_1(x) \cap \cdots \cap U_n(x)$ for every n, as usual.

Using (3.5.5) with $V = U_1(x)$, we can get a positive integer j_1 such that $x_{j_1} \in U_1(x)$. Suppose that $j_l \in \mathbf{Z}_+$ has been chosen for some positive integer l. Using (3.5.5) with $V = U_{l+1}(x)$, we get that $x_j \in U_{l+1}(x)$ for infinitely many $j \in \mathbf{Z}_+$. In particular, we can choose $j_{l+1} \in \mathbf{Z}_+$ such that $j_{l+1} > j_l$ and $x_{j_{l+1}} \in U_{l+1}(x)$.

This leads to a subsequence $\{x_{j_l}\}_{l=1}^{\infty}$ of $\{x_j\}_{j=1}^{\infty}$ such that

$$(3.5.7) x_{j_l} \in U_l(x)$$

for every $l \in \mathbf{Z}_+$. It follows that $\{x_{j_l}\}_{l=1}^{\infty}$ converges to x in X, as desired.

3.5.3 Sequential compactness and induced topologies

Proposition 3.5.8 Suppose that $K \subseteq Y \subseteq X$. Under these conditions, K is sequentially compact as a subset of X if and only if K is sequentially compact as a subset of Y, with respect to the induced topology.

This is easy to verify, using the analogous statement for convergence of sequences in Y.

3.5.4 Sequentially closed sets

Definition 3.5.9 A subset E of X is said to be sequentially closed if for every sequence $\{x_j\}_{j=1}^{\infty}$ of elements of E that converges to an element x of X, we have that $x \in E$.

It is easy to see that closed subsets of X are sequentially closed. If X satisfies the first countability condition, then sequentially closed subsets of X are closed sets.

Proposition 3.5.10 *If* $K \subseteq X$ *is sequentially compact and* $E \subseteq X$ *is sequentially closed, then* $K \cap E$ *is sequentially compact in* X.

This can be verified directly from the definitions.

Proposition 3.5.11 If $E \subseteq X$ is sequentially compact, and X is Hausdorff, then E is sequentially closed in X.

Indeed, let $\{x_j\}_{j=1}^{\infty}$ be a sequence of elements of E that converges to an element x of X. Because E is sequentially compact in X, there is a subsequence $\{x_{j_l}\}_{l=1}^{\infty}$ of $\{x_j\}_{j=1}^{\infty}$ that converges to an element x' of E. Note that $\{x_{j_l}\}_{l=1}^{\infty}$ converges to x too, as before. If X is Hausdorff, then x = x', so that $x \in E$, as desired.

3.5.5 Sequential continuity

Let Y be another topological space.

Definition 3.5.12 A mapping f from X into Y is said to be sequentially continuous at a point $x \in X$ if for every sequence $\{x_j\}_{j=1}^{\infty}$ of elements of X that converges to x, $\{f(x_j)\}_{j=1}^{\infty}$ converges to f(x) in Y. If f is sequentially continuous at every point in X, then f is said to be sequentially continuous on X.

If f is continuous at x in the usual sense, then f is sequentially continuous at x. The converse holds when there is a local base for the topology of X at x with only finitely or countably many elements.

Proposition 3.5.13 If f is a sequentially continuous mapping from X into Y, and $K \subseteq X$ is sequentially compact, then f(K) is sequentially compact in Y.

Let $\{y_j\}_{j=1}^{\infty}$ be a sequence of elements of f(K). If j is a positive integer, then let us choose an element x_j of K such that $y_j = f(x_j)$. Because K is sequentially compact, there is a subsequence $\{x_{j_l}\}_{l=1}^{\infty}$ of $\{x_j\}_{j=1}^{\infty}$ that converges to an element x of K. This implies that $\{f(x_{j_l})\}_{l=1}^{\infty}$ converges to f(x) in Y, because f is sequentially continuous at x, by hypothesis. Thus $\{y_{j_l}\}_{l=1}^{\infty} = \{f(x_{j_l})\}_{l=1}^{\infty}$ is a subsequence of $\{y_j\}_{j=1}^{\infty}$ that converges to $y = f(x) \in f(K)$ in Y, as desired.

3.6 Countable compactness

Let X be a topological space, and let K be a subset of X.

Definition 3.6.1 We say that K is countably compact in X if for every sequence U_1, U_2, U_3, \ldots of open subsets of X such that

$$(3.6.2) K \subseteq \bigcup_{j=1}^{\infty} U_j,$$

there is a positive integer n such that

$$(3.6.3) K \subseteq \bigcup_{j=1}^{n} U_j.$$

If K has the strong limit point property, then K is countably compact, as in Section 3.4.

3.6.1 The Lindelöf property

Definition 3.6.4 We say that K has the Lindelöf property in X if for every open covering $\{U_{\alpha}\}_{{\alpha}\in A}$ of K in X there is a subset A_1 of A such that A_1 has only finitely or countably many elements and

$$(3.6.5) K \subseteq \bigcup_{\alpha \in A_1} U_{\alpha}.$$

It is easy to see that K is compact if and only if K is countably compact and K has the Lindelöf property.

3.6.2 Lindelöf's theorem

Theorem 3.6.6 (Lindelöf) Let A be a nonempty set, and let U_{α} be an open subset of X for each $\alpha \in A$. If X satisfies the second countability condition, then there is a subset A_1 of A such that A_1 has only finitely or countably many elements and

$$(3.6.7) \qquad \qquad \bigcup_{\alpha \in A_1} U_{\alpha} = \bigcup_{\alpha \in A} U_{\alpha}.$$

By hypothesis, there is a base \mathcal{B} for the topology of X such that \mathcal{B} has only finitely or countably many elements. If $\alpha \in A$, then put

$$\mathcal{B}_{\alpha} = \{ V \in \mathcal{B} : V \subseteq U_{\alpha} \}.$$

Observe that

(3.6.9)
$$U_{\alpha} = \bigcup \{V : V \in \mathcal{B}_{\alpha}\}$$

for every $\alpha \in A$. More precisely, the union on the right is automatically contained in U_{α} , by the definition of \mathcal{B}_{α} . In order to get that the union is equal to U_{α} , one uses the hypothesis that \mathcal{B} be a base for the topology of X.

(3.6.10)
$$\widetilde{\mathcal{B}} = \bigcup_{\alpha \in A} \mathcal{B}_{\alpha}.$$

Of course, $\widetilde{\mathcal{B}} \subseteq \mathcal{B}$, by construction. It follows that $\widetilde{\mathcal{B}}$ has only finitely or countably many elements, because \mathcal{B} has only finitely or countably many elements, by hypothesis. If $V \in \widetilde{\mathcal{B}}$, then let us choose an element $\alpha(V)$ of A such that $V \in \mathcal{B}_{\alpha(V)}$. Thus

$$(3.6.11) V \subseteq U_{\alpha(V)}$$

for every $V \in \mathcal{B}$, by definition of \mathcal{B}_{α} .

Let A_1 be the set of elements of A of the form $\alpha(V)$ for some $V \in \widetilde{\mathcal{B}}$ that have been chosen in this way. It is easy to see that A_1 has only finitely

or countably many elements, because $\widetilde{\mathcal{B}}$ has only finitely or countably many elements. Observe that

(3.6.12)
$$\bigcup_{\alpha \in A_1} U_{\alpha} = \bigcup_{V \in \widetilde{\mathcal{B}}} U_{\alpha(V)} \supseteq \bigcup_{V \in \widetilde{\mathcal{B}}} V,$$

using (3.6.11) in the second step. We also have that

(3.6.13)
$$\bigcup_{V \in \widetilde{\mathcal{B}}} V = \bigcup_{\alpha \in A} \bigcup_{V \in \mathcal{B}_{\alpha}} V = \bigcup_{\alpha \in A} U_{\alpha}.$$

This uses the definition (3.6.10) of \mathcal{B} in the first step, and (3.6.9) in the second step. Combining (3.6.12) and (3.6.13), we obtain that

$$(3.6.14) \qquad \qquad \bigcup_{\alpha \in A} U_{\alpha} \subseteq \bigcup_{\alpha \in A_1} U_{\alpha}.$$

This implies (3.6.7), because $A_1 \subseteq A$, by construction.

Corollary 3.6.15 If X satisfies the second countability condition, then every subset of X has the Lindelöf property.

3.6.3 Some additional results

Proposition 3.6.16 Suppose that $K \subseteq Y \subseteq X$. Under these conditions, K is countably compact as a subset of X if and only if K is countably compact as a subset of Y, with respect to the induced topology. Similarly, K has the Lindelöf property as a subset of X if and only if K has the Lindelöf property as a subset of Y, with respect to the induced topology.

This can be verified directly from the definitions, as in the case of ordinary compactness.

Proposition 3.6.17 Let E be a closed set in X. If $K \subseteq X$ is countably compact, then $K \cap E$ is countably compact too. If K has the Lindelöf property, then $K \cap E$ has the Lindelöf property as well.

This can be shown in essentially the same way as for ordinary compactness.

Proposition 3.6.18 Let f be a continuous mapping from X into another topological space Y. If $K \subseteq X$ is countably compact, then f(K) is countably compact in Y. If K has the Lindelöf property, then f(K) has the Lindelöf property in Y.

This is similar to the analogous statement for ordinary compactness.

3.7 The finite intersection property

Let X be a set, and let K be a subset of X.

Definition 3.7.1 Let I be a nonempty set, and let E_j be a subset of X for each $j \in I$. We say that $\{E_j\}_{j \in I}$ has the finite intersection property with respect to K if for every finite collection j_1, \ldots, j_n of elements of I, we have that

$$\left(\bigcap_{l=1}^{n} E_{j_{l}}\right) \cap K \neq \emptyset.$$

If $\{E_j\}_{j\in I}$ has the finite intersection property with respect to X, then we may simply say that $\{E_j\}_{j\in I}$ has the finite intersection property.

Of course, (3.7.2) holds when

$$\left(\bigcap_{j\in I} E_j\right) \cap K \neq \emptyset.$$

3.7.1 A characterization of compactness

Suppose now that X is a topological space.

Proposition 3.7.4 A subset K of X is compact if and only if for all nonempty families $\{E_j\}_{j\in I}$ of closed subsets of X with the finite intersection property with respect to K, we have that (3.7.3) holds.

To see this, let a nonempty set I be given again. If $E_j \subseteq X$ is a closed set for some $j \in I$, then

$$(3.7.5) U_i = X \setminus E_i$$

is an open set in X. Similarly, if $U_j \subseteq X$ is an open set for some $j \in I$, then

$$(3.7.6) E_j = X \setminus U_j$$

is a closed set in X. This defines a simple correspondence between families $\{E_j\}_{j\in I}$ of closed subsets of X indexed by I and families $\{U_j\}_{j\in I}$ of open subsets of X indexed by I.

Using this correspondence, we have that (3.7.2) holds for some finite collection of indices $j_1, \ldots, j_n \in I$ if and only if

(3.7.7)
$$K \not\subseteq X \setminus \left(\bigcap_{j=1}^{n} E_{j}\right) = \bigcup_{l=1}^{n} U_{j_{l}}.$$

Thus $\{E_j\}_{j\in I}$ has the finite intersection property with respect to K exactly when K cannot be covered by finitely many U_j 's, $j \in I$. Similarly, (3.7.3) holds if and only if

(3.7.8)
$$K \not\subseteq X \setminus \left(\bigcap_{j \in I} E_j\right) = \bigcup_{j \in I} U_j.$$

This is the same as saying that K is not covered by the U_j 's, $j \in I$. Using this, it is easy to see that the condition mentioned in the proposition is equivalent to compactness.

3.7.2 A characterization of countable compactness

Let us now consider the case where $I = \mathbf{Z}_{+}$.

Proposition 3.7.9 A subset K of X is countably compact if and only if for every sequence E_1, E_2, E_3, \ldots of closed subsets of X that satisfies

$$\left(\bigcap_{j=1}^{n} E_{j}\right) \cap K \neq \emptyset$$

for each positive integer n, we have that

(3.7.11)
$$\left(\bigcap_{j=1}^{\infty} E_j\right) \cap K \neq \emptyset.$$

Let E_j be a subset of X for each positive integer j. Observe that $\{E_j\}_{j\in \mathbf{Z}_+}$ has the finite intersection property with respect to $K\subseteq X$ if and only if (3.7.10) holds for every positive integer n. In this situation, (3.7.11) is the same as (3.7.3) with $I=\mathbf{Z}_+$. Thus the proposition follows from the same type of argument as before.

3.8 The strong limit point property

Let X be a topological space.

Proposition 3.8.1 *If* $K \subseteq X$ *is countably compact, then* K *has the strong limit point property.*

To show this, let an infinite subset L of K be given. As before, we can find an infinite sequence $\{x_j\}_{j=1}^{\infty}$ of distinct elements of L, because L has infinitely many elements. Put

$$(3.8.2) A_l = \{x_j : j \ge l\}$$

for each positive integer l. Note that for each $l, A_l \neq \emptyset$ and

$$(3.8.3) A_{l+1} \subseteq A_l.$$

Let

$$(3.8.4) E_l = \overline{A_l}$$

be the closure of A_l in X for each $l \geq 1$. Thus

$$(3.8.5) A_l \subseteq E_l$$

for every $l \geq 1$, which implies in particular that $E_l \neq \emptyset$. It is easy to see that

$$(3.8.6) E_{l+1} \subseteq E_l$$

for every $l \geq 1$, because of (3.8.3).

Of course, E_l is a closed set in X for each $l \geq 1$, by construction. Observe that

$$(3.8.7) \qquad \qquad \bigcap_{l=1}^{n} E_l = E_n$$

for every positive integer n, by (3.8.6). It follows that

(3.8.8)
$$\left(\bigcap_{l=1}^{n} E_{l}\right) \cap K = E_{n} \cap K$$

for each $n \geq 1$. We also have that

$$(3.8.9) A_n \subseteq E_n \cap K$$

for each $n \ge 1$, because of (3.8.5) and the fact that $A_n \subseteq L \subseteq K$ for every n, by construction. This implies that (3.8.8) is nonempty for every n.

Thus E_1, E_2, E_3, \ldots is a sequence of closed sets in X with the finite intersection property with respect to K. Because K is countably compact, we get that

$$\left(\bigcap_{l=1}^{\infty} E_l\right) \cap K \neq \emptyset,$$

as in the previous section. Let x be an element of the left side of (3.8.10). In particular, $x \in K$, and we would like to check that x is a strong limit point of L in X.

Note that x is adherent to A_l in X for each $l \geq 1$, because $x \in E_l$. Let $U \subseteq X$ be an open set that contains x. It follows that

$$(3.8.11) A_l \cap U \neq \emptyset$$

for each $l \geq 1$, because x is adherent to A_l . This means that for each $l \geq 1$ there is a $j \geq l$ such that

$$(3.8.12) x_j \in U,$$

by the definition (3.8.2) of A_l . Thus (3.8.12) holds for infinitely many positive integers j. In particular, U contains infinitely many elements of L, because the x_j 's are distinct elements of L. This implies that x is a strong limit point of L in X, as desired.

3.9 Compactness and bases

Let X be a topological space, and let \mathcal{B} be a collection of open subsets of X.

Definition 3.9.1 We say that $K \subseteq X$ is compact with respect to \mathcal{B} if every open covering of K by elements of \mathcal{B} can be reduced to a finite subcovering. More precisely, this means that if $\{U_{\alpha}\}_{\alpha \in A}$ is an open covering of K with $U_{\alpha} \in \mathcal{B}$ for every $\alpha \in A$, then there are finitely many indices $\alpha_1, \ldots, \alpha_n \in A$ such that $K \subseteq \bigcup_{j=1}^n U_{\alpha_j}$.

Of course, if K is a compact subset of X, then K is automatically compact with respect to \mathcal{B} . Countable compactness and the Lindelöf property with respect to \mathcal{B} can be defined analogously.

3.9.1 A criterion for compactness

Proposition 3.9.2 Suppose that \mathcal{B} is a base for the topology of X. If $K \subseteq X$ is compact with respect to \mathcal{B} , then K is compact in X in the usual sense.

The analogous statement for the Lindelöf property can be shown similarly. Note that Lindelöf's theorem follows from the version of Proposition 3.9.2 for the Lindelöf property. However, the proofs of Proposition 3.9.2 and its version for the Lindelöf property are also very similar to the proof of Lindelöf's theorem.

3.9.2 The proof of the proposition

To prove the proposition, let $\{U_{\alpha}\}_{{\alpha}\in A}$ be any open covering of K in X. Put

$$(3.9.3) \mathcal{B}_{\alpha} = \{ V \in \mathcal{B} : V \subseteq U_{\alpha} \}$$

for each $\alpha \in A$. Because \mathcal{B} is a base for the topology of X,

$$(3.9.4) U_{\alpha} = \bigcup \{V : V \in \mathcal{B}_{\alpha}\}$$

for every $\alpha \in A$, as before. Put

(3.9.5)
$$\widetilde{\mathcal{B}} = \bigcup_{\alpha \in A} \mathcal{B}_{\alpha}.$$

Observe that

(3.9.6)
$$\bigcup_{V \in \widetilde{\mathcal{B}}} V = \bigcup_{\alpha \in A} \bigcup_{V \in \mathcal{B}_{\alpha}} V = \bigcup_{\alpha \in A} U_{\alpha},$$

using the definition of $\widetilde{\mathcal{B}}$ in the first step, and (3.9.4) in the second step. It follows that

(3.9.7)
$$K \subseteq \bigcup_{\alpha \in A} U_{\alpha} = \bigcup_{V \in \widetilde{\mathcal{B}}} V.$$

If K is compact with respect to \mathcal{B} , then there are finitely many elements V_1, \ldots, V_n of \mathcal{B} such that

$$(3.9.8) K \subseteq \bigcup_{j=1}^{n} V_j.$$

By definition of $\widetilde{\mathcal{B}}$, for each $j=1,\ldots,n$ there is an $\alpha_j \in A$ such that $V_j \in \mathcal{B}_{\alpha_j}$, which means that

$$(3.9.9) V_j \subseteq U_{\alpha_j}.$$

Combining this with (3.9.8), we get that

$$(3.9.10) K \subseteq \bigcup_{j=1}^{n} U_{\alpha_j},$$

as desired.

3.9.3 Subbases

Let \mathcal{B} be any collection of open subsets of X. Consider the collection \mathcal{B}_1 of all subsets of X that can be expressed as the intersection of finitely many elements of \mathcal{B} . Note that the elements of \mathcal{B}_1 are all open subsets of X. If \mathcal{B}_1 is a base for the topology of X, then \mathcal{B} is said to be a *subbase* for the topology of X.

As an example, let \mathcal{B} be the collection of subsets of the real line that are open half-lines, which is to say sets of the form $(a, +\infty)$ and $(-\infty, b)$ for $a, b \in \mathbf{R}$. In this case, \mathcal{B}_1 consists of all open half-lines in \mathbf{R} , as well as all open intervals in \mathbf{R} . It is easy to see that \mathcal{B}_1 is a base for the standard topology on \mathbf{R} , so that \mathcal{B} is a subbase for this topology.

3.9.4 Alexander's subbase theorem

If $K \subseteq X$ is compact with respect to a subbase \mathcal{B} for the topology of X, then Alexander's subbase theorem says that K is compact in X in the usual sense. The proof of this theorem uses the axiom of choice in a significant way. More precisely, this involves Zorn's lemma or the Hausdorff maximality principle, which will be discussed in the next chapter.

3.10 Products of two compact sets

Let X and Y be topological spaces, and consider their Cartesian product $X \times Y$, equipped with the corresponding product topology.

Theorem 3.10.1 (Tychonoff) If $H \subseteq X$ and $K \subseteq Y$ are compact, then their product $H \times K$ is compact in $X \times Y$.

To show this, let \mathcal{B} be the collection of subsets of $X \times Y$ of the form $U \times V$, where $U \subseteq X$ and $V \subseteq Y$ are open sets. This is a base for the product topology on $X \times Y$. Thus it suffices to verify that $H \times K$ is compact with respect to \mathcal{B} , as in the previous section.

Let $\{U_{\alpha} \times V_{\alpha}\}_{{\alpha} \in A}$ be a covering of $H \times K$ by elements of \mathcal{B} , so that $U_{\alpha} \subseteq X$ and $V_{\alpha} \subseteq Y$ are open sets for every $\alpha \in A$, and

$$(3.10.2) H \times K \subseteq \bigcup_{\alpha \in A} (U_{\alpha} \times V_{\alpha}).$$

Let $x \in H$ be given, so that

(3.10.3)
$$\{x\} \times K \subseteq \bigcup_{\alpha \in A} (U_{\alpha} \times V_{\alpha}).$$

because $\{x\} \times K \subseteq H \times K$.

3.10.1 Some open coverings of K in Y

Put

(3.10.4)
$$A(x) = \{ \alpha \in A : x \in U_{\alpha} \}.$$

Observe that

$$(3.10.5) K \subseteq \bigcup_{\alpha \in A(x)} V_{\alpha},$$

by (3.10.3). More precisely, if $y \in K$, then $(x, y) \in \{x\} \times K$, and hence there is an $\alpha \in A$ such that $(x, y) \in U_{\alpha} \times V_{\alpha}$, by (3.10.3). This means that $x \in U_{\alpha}$, so that $\alpha \in A(x)$, and $y \in V_{\alpha}$, as desired. It follows that there is a finite subset $A_1(x)$ of A(x) such that

$$(3.10.6) K \subseteq \bigcup_{\alpha \in A_1(x)} V_{\alpha},$$

because K is compact in Y.

3.10.2 An open covering of H in X

Put

$$(3.10.7) U_1(x) = \bigcap_{\alpha \in A_1(x)} U_{\alpha}.$$

This is an open set in X, because U_{α} is an open set in X for every $\alpha \in A_1(x) \subseteq A$, and $A_1(x)$ has only finitely many elements. Of course, $x \in U_1(x)$, because $x \in U_{\alpha}$ for every $\alpha \in A_1(x) \subseteq A(x)$. We can do this for every $x \in H$, to get an open covering of H in X. This implies that there are finitely many elements x_1, \ldots, x_n of H such that

$$(3.10.8) H \subseteq \bigcup_{j=1}^{n} U_1(x_j),$$

because H is compact in X.

3.10.3 Using these finite subcoverings

Using (3.10.8), we get that

$$(3.10.9) H \times K \subseteq \left(\bigcup_{j=1}^{n} U_1(x_j)\right) \times K = \bigcup_{j=1}^{n} (U_1(x_j) \times K).$$

We also have that

$$(3.10.10) U_1(x_j) \times K \subseteq U_1(x_j) \times \left(\bigcup_{\alpha \in A_1(x_j)} V_{\alpha}\right) = \bigcup_{\alpha \in A_1(x_j)} (U_1(x_j) \times V_{\alpha})$$

$$\subseteq \bigcup_{\alpha \in A_1(x_j)} (U_{\alpha} \times V_{\alpha})$$

for each $j=1,\ldots,n$. This uses (3.10.6) in the first step, and the fact that $U_1(x_j) \subseteq U_\alpha$ when $\alpha \in A_1(x_j)$, by the definition (3.10.7) of $U_1(x_j)$, in the third step. Combining (3.10.9) and (3.10.10), we get that

(3.10.11)
$$H \times K \subseteq \bigcup_{j=1}^{n} \bigcup_{\alpha \in A_{1}(x_{j})} (U_{\alpha} \times V_{\alpha}).$$

Of course, $\bigcup_{j=1}^{n} A_1(x_j)$ is a finite subset of A, because $A_1(x_j)$ is a finite subset of A for each $j = 1, \ldots, n$, as desired.

3.11 Products of more compact sets

Let X_1, \ldots, X_n be finitely many topological spaces for some positive integer n, and consider their Cartesian product

(3.11.1)
$$X = \prod_{j=1}^{n} X_j,$$

equipped with the product topology. Also let K_j be a compact subset of X_j for each j = 1, ..., n, and put

(3.11.2)
$$K = \prod_{j=1}^{n} K_j.$$

Under these conditions, K is a compact subset of X. This is trivial when n=1, and the n=2 case was discussed in the previous section. Otherwise, one can use induction, as follows.

3.11.1 An induction argument

Suppose that $n \geq 2$, and that the analogous statement holds for n-1. Thus $\prod_{j=1}^{n-1} K_j$ is a compact subset of $\prod_{j=1}^{n-1} X_j$, with respect to the product topology. There is a natural identification of X with

(3.11.3)
$$\left(\prod_{j=1}^{n-1} X_j\right) \times X_n,$$

where $x = (x_1, \ldots, x_n) \in X$ is identified with $((x_1, \ldots, x_{n-1}), x_n)$, as an element of (3.11.3). Let us take $\prod_{j=1}^{n-1} X_j$ to be equipped with the product topology, and use the corresponding product topology on (3.11.3). One can check that this corresponds to the product topology on X, using the identification of (3.11.3) with X just mentioned.

As in the previous section,

(3.11.4)
$$\left(\prod_{j=1}^{n-1} K_j\right) \times K_n$$

is a compact subset of (3.11.3), with respect to the product topology. Of course, K corresponds to (3.11.4), with respect to the identification of X with (3.11.3). It follows that K is a compact subset of X, because the identification of X with (3.11.3) is a homeomorphism with respect to the corresponding product topologies, as before.

3.11.2 Some compact subsets of \mathbb{R}^n

Let n be a positive integer again, and \mathbf{R}^n be the usual space of n-tuples of real numbers. This is the same as the Cartesian product of n copies of the real line. The standard topology on \mathbf{R}^n may be defined as the product topology corresponding to the standard topology on \mathbf{R} , as before. This is the same as the topology determined by the standard Euclidean metric on \mathbf{R}^n .

Let a_j and b_j be real numbers with $a_j \leq b_j$ for each $j = 1, \ldots, n$. Remember that the closed interval $[a_j, b_j]$ is a compact subset of the real line for each $j = 1, \ldots, n$, with respect to the standard topology. It follows that their Cartesian product

(3.11.5)
$$\prod_{j=1}^{n} [a_j, b_j]$$

is compact with respect to the standard topology on \mathbb{R}^n .

3.11.3 Compactness of arbitrary products of compact sets

Now let I be a nonempty set, let X_j be a topological space for each $j \in I$, and consider the Cartesian product $X = \prod_{j \in I} X_j$, equipped with the product topology. If $K_j \subseteq X_j$ is compact for each $j \in I$, then another famous theorem of Tychonoff states that $K = \prod_{j \in I} K_j$ is compact in X. Of course, this reduces to the previous statement for (3.11.2) when I has only finitely many elements.

Let \mathcal{B} be the collection of subsets of X of the form $U = \prod_{j \in I} U_j$, where $U_j = X_j$ for all but one $j \in I$, and U_j is an open set in X_j for that j. It is easy to see that \mathcal{B} is a subbase for the product topology on X. To show Tychonoff's theorem, it suffices to get that K is compact with respect to \mathcal{B} , by Alexander's subbase theorem.

3.12 Sequential compactness and finite products

Let X_1, \ldots, X_n be finitely many topological spaces again, and let

$$(3.12.1) X = \prod_{j=1}^{n} X_j$$

be their Cartesian product, equipped with the product topology.

Proposition 3.12.2 If K_j is a sequentially compact subset of X_j for each j = 1, ..., n, then

(3.12.3)
$$K = \prod_{j=1}^{n} K_j$$

is sequentially compact in X.

3.12.1 An initial subsequence

To see this, let $\{x(l)\}_{l=1}^{\infty}$ be a sequence of elements of K. Thus

$$(3.12.4) x(l) = (x_1(l), \dots, x_n(l))$$

for each positive integer l, where $x_j(l) \in K_j$ for j = 1, ..., n. In particular, $\{x_1(l)\}_{l=1}^{\infty}$ is a sequence of elements of K_1 . Because K_1 is sequentially compact, there is a subsequence $\{x_1(l_m)\}_{m=1}^{\infty}$ of $\{x_1(l)\}_{l=1}^{\infty}$ that converges to an element x_1 of K_1 in X_1 . Using the same sequence of indices $\{l_m\}_{m=1}^{\infty}$, we get a subsequence $\{x(l_m)\}_{m=1}^{\infty}$ of $\{x(l)\}_{l=1}^{\infty}$.

3.12.2 A subsequence of the first subsequence when $n \ge 2$

Of course, the statement is trivial when n=1, and so we may as well suppose that $n \geq 2$. As before, $\{x_2(l_m)\}_{m=1}^{\infty}$ is a sequence of elements of K_2 , and so there is a subsequence $\{x_2(l_{m_r})\}_{r=1}^{\infty}$ of $\{x_2(l_m)\}_{m=1}^{\infty}$ that converges to an element x_2 of K_2 , because K_2 is sequentially compact. Using the same sequence $\{m_r\}_{r=1}^{\infty}$ of indices, we get a subsequence $\{x(l_{m_r})\}_{r=1}^{\infty}$ of $\{x(l_m)\}_{m=1}^{\infty}$. In particular, $\{x_1(l_{m_r})\}_{r=1}^{\infty}$ is a subsequence of $\{x_1(l_m)\}_{m=1}^{\infty}$. This implies that $\{x_1(l_{m_r})\}_{r=1}^{\infty}$ converges to x_1 in X_1 , because $\{x_1(l_m)\}_{m=1}^{\infty}$ converges to x_1 in X_1 .

Note that $\{x(l_{m_r})\}_{r=1}^{\infty}$ may be considered as a subsequence of the initial sequence $\{x(l)\}_{l=1}^{\infty}$ as well. If n=2, then we get that $\{x(l_{m_r})\}_{r=1}^{\infty}$ converges to

$$(3.12.5) x = (x_1, x_2) \in K_1 \times K_2 = K$$

in X, as desired. Otherwise, we can repeat the process.

3.12.3 Repeating the process

More precisely, suppose that the jth subsequence of $\{x(l)\}_{l=1}^{\infty}$ has been chosen in this way for some positive integer j < n. We would like to choose the (j+1)th subsequence of $\{x(l)\}_{l=1}^{\infty}$ to be a subsequence of the jth subsequence, using the sequential compactness of K_{j+1} , as before. The (j+1)th subsequence is chosen so that the sequence of (j+1)th coordinates in K_{j+1} of the terms of the (j+1)th subsequence converges to an element x_{j+1} of K_{j+1} in X_{j+1} . Observe that the (j+1)th subsequence is a subsequence of the ith subsequence for each $i=1,\ldots,j$. This implies that the sequence of ith coordinates of the terms of the (j+1)th subsequence converge to x_i in X_i , because the sequence

of ith coordinates of the terms of the ith subsequence converge to x_i in X_i , by construction.

We can continue in this way until the nth subsequence of $\{x(l)\}_{l=1}^{\infty}$ is chosen. As before, for each $i=1,\ldots,n$, the sequence of ith coordinates of the nth subsequence converges to $x_i \in K_i$ in X_i . This implies that the nth subsequence converges to

$$(3.12.6) x = (x_1, \dots, x_n) \in K$$

in X. We also have that the nth subsequence is a subsequence of the initial sequence $\{x(l)\}_{l=1}^{\infty}$, as desired.

3.13 Sequential compactness in countable products

If $\{l_m\}_{m=1}^{\infty}$ is a strictly increasing sequence of positive integers, then it is easy to see that

$$(3.13.1) l_m \ge m$$

for every positive integer m.

Let X_1, X_2, X_3, \ldots be an infinite sequence of topological spaces, and consider their Cartesian product

(3.13.2)
$$X = \prod_{j=1}^{\infty} X_j,$$

equipped with the product topology.

Proposition 3.13.3 If K_j is a sequentially compact subset of X_j for each positive integer j, then

$$(3.13.4) K = \prod_{j=1}^{\infty} K_j$$

is sequentially compact in X.

3.13.1 A sequence of subsequences

To show this, let a sequence $\{x(l)\}_{l=1}^{\infty}$ of elements of K be given. Note that

$$(3.13.5) x(l) = \{x_j(l)\}_{j=1}^{\infty}$$

is a sequence for each l, where $x_j(l) \in K_j$ for every $j \geq 1$. In particular, $\{x_1(l)\}_{l=1}^{\infty}$ is a sequence of elements of K_1 , as before. Hence there is a subsequence $\{x_1(l_m)\}_{m=1}^{\infty}$ of $\{x_1(l)\}_{l=1}^{\infty}$ that converges to an element x_1 of K_1 in X_1 , because K_1 is sequentially compact. We can use the same sequence $\{l_m\}_{m=1}^{\infty}$ of indices to get a subsequence $\{x(l_m)\}_{m=1}^{\infty}$ of $\{x(l)\}_{l=1}^{\infty}$.

We can repeat the process, as in the previous section, to get an infinite sequence of subsequences of $\{x(l)\}_{l=1}^{\infty}$. More precisely, for each positive integer

j, we can get a sequence $\{x(r,j)\}_{r=1}^{\infty}$ of elements of K with the following two poperties. First,

$$(3.13.6)$$
 $\{x(r,1)\}_{r=1}^{\infty}$ is a subsequence of $\{x(l)\}_{l=1}^{\infty}$,

and

(3.13.7)
$$\{x(r,j)\}_{r=1}^{\infty}$$
 is a subsequence of $\{x(m,j-1)\}_{m=1}^{\infty}$

when $j \geq 2$. This implies that

(3.13.8)
$$\{x(r,j)\}_{r=1}^{\infty} \text{ is a subsequence of } \{x(l)\}_{l=1}^{\infty}$$

for every $j \geq 1$, and that

(3.13.9)
$$\{x(r,j)\}_{r=1}^{\infty} \text{ is a subsequence of } \{x(m,i)\}_{m=1}^{\infty}$$

when $1 \leq i \leq j$. The second property is that for every positive integer j there is an element x_j of K_j such that

$$(3.13.10) \{x_j(r,j)\}_{r=1}^{\infty} \text{ converges to } x_j \text{ in } X_j.$$

It follows that

(3.13.11)
$$\{x_i(r,j)\}_{r=1}^{\infty} \text{ converges to } x_i \text{ in } X_i$$

when $1 \le i \le j$, because of (3.13.9). Put

$$(3.13.12) x = \{x_j\}_{j=1}^{\infty},$$

which defines an element of K.

3.13.2 Another subsequence

We would like to find a subsequence of $\{x(l)\}_{l=1}^{\infty}$ that converges to x in X. Put

$$(3.13.13) y(r) = x(r,r)$$

for each positive integer r, which is the rth term of the rth subsequence described in the previous paragraph. One can check that

$$(3.13.14)$$
 $\{y(r)\}_{r=1}^{\infty}$ is a subsequence of $\{x(l)\}_{l=1}^{\infty}$.

Indeed, for each $r \geq 1$, y(r) is one of the terms of $\{x(l)\}_{l=1}^{\infty}$, by construction. One can verify that y(r+1) is chosen among the terms of $\{x(l)\}_{l=1}^{\infty}$ that occurs after the one corresponding to y(r), using (3.13.1). More precisely, y(r+1) is the (r+1)th term of the (r+1)th subsequence, and the (r+1)th subsequence is a subsequence of the rth subsequence, by construction. This implies that y(r+1) occurs in the rth subsequence after y(r), as in (3.13.1). It follows that y(r+1) occurs after y(r) in the previous subsequences, including the initial sequence $\{x(l)\}_{l=1}^{\infty}$.

3.13.3 More on this subsequence

Similarly, one can check that for each positive integer j,

$$(3.13.15)$$
 $\{y(r)\}_{r=j}^{\infty}$ is a subsequence of $\{x(m,j)\}_{m=1}^{\infty}$.

In particular, if $r \geq j$, then y(r) is one of the terms of $\{x(m,j)\}_{m=1}^{\infty}$, because the rth subsequence is a subsequence of the jth subsequence, as in (3.13.9). We also have that y(r+1) is chosen among the terms of $\{x(m,j)\}_{m=1}^{\infty}$ after the one corresponding to y(r) when $r \geq j$, as in the preceding paragraph.

It follows that

(3.13.16)
$$\{y_j(r)\}_{r=j}^{\infty}$$
 converges to x_j in X_j

for each positive integer j, because of (3.13.10) and (3.13.15). This implies that

(3.13.17)
$$\{y_j(r)\}_{r=1}^{\infty}$$
 converges to x_j in X_j

for every $j \geq 1$. This means that $\{y(r)\}_{r=1}^{\infty}$ converges to x with respect to the product topology on X, as desired.

3.14 Compactness and sequential compactness

If a topological space X satisfies the first countability condition and $K\subseteq X$ is compact, then

$$(3.14.1)$$
 K is sequentially compact.

More precisely, compactness implies the strong limit point property, which implies sequential compactness when X satisfies the first countability condition.

Suppose for the moment that X is any set equipped with the cofinite topology, as in Subsection 3.1.2. Remember that every subset of X is compact in this case. One can check that every subset of X is also sequentially compact in this case. However, one can verify that X satisfies the first countability condition only when X has only finitely or countably many elements.

One can get examples of compact Hausdorff topological spaces that do not satisfy the first countability condition using uncountable products of compact Hausdorff spaces and Tychonoff's theorem. One can use uncountable products of finite sets, each of which has at least two elements, and is equipped with the discrete topology, for instance. One could also use uncountable products of closed intervals in the real line, equipped with the topology induced by the standard topology on ${\bf R}$.

Let I be a nonempty set, let X_j be a topological space for each $j \in I$, and let $X = \prod_{j \in I} X_j$ be the Cartesian product of the X_j 's, equipped with the product topology. Suppose that $K_j \subseteq X_j$ is compact for each $j \in I$, so that $K = \prod_{j \in I} K_j$ is a compact subset of X, by Tychonoff's theorem. If I has only finitely or countably many elements, and X_j satisfies the first countability condition for each $j \in I$, then we have seen that X satisfies the first countability condition with respect to the product topology. In this case, the compactness of K implies that K is sequentially compact, as before. Alternatively, K_j is

sequentially compact in X_j for each $j \in I$ under these conditions, which implies that K is sequentially compact, as in the previous two sections.

If K is a sequentially compact subset of a topological space X, then we have seen that K has the strong limit point property, which implies that

$$(3.14.2)$$
 K is countably compact.

If K also satisfies the Lindelöf property, then it follows that K is compact. If X satisfies the second countability condition, then K automatically satisfies the Lindelöf property, by Lindelöf's theorem. If X is the Cartesian product of finitely or countably many topological spaces, each of which satisfies the second countability condition, then we have seen that X satisfies the second countability condition as well, with respect to the product topology.

It is well known that

(3.14.3) sequentially compact subsets of metric spaces are compact.

More precisely, one can reduce to showing that a sequentially compact metric space X is compact. Part of the proof is to show that X is "totally bounded". This implies that X is separable, and thus satisfies the second countability condition.

Suppose that X is the product of finitely or countably many topological spaces, each of which has a metric that determines the topology. We have seen that X has a metric that determines the corresponding product topology in this case. This implies that sequentially compact subsets of X are compact, as in the preceding paragraph.

3.15 Another result of Tychonoff

Let X be a topological space. If X has the Lindelöf property, and X is regular in the strict sense, then a well-known result of Tychonoff states that X is normal in the strict sense.

To see this, let A and B be disjoint closed subsets of X. Note that A and B each have the Lindelöf property in X, because A and B are closed subsets of X and X has the Lindelöf property, as in Proposition 3.6.17. If $x \in A$, then there is an open set $U(x) \subseteq X$ such that

$$(3.15.1) x \in U(x) \text{ and } \overline{U(x)} \subseteq X \setminus B,$$

because X is regular in the strict sense, and $X \setminus B$ is an open set that contains x. Similarly, if $y \in B$, then there is an open set $V(y) \subseteq X$ such that

$$(3.15.2) y \in V(y) \text{ and } \overline{V(y)} \subseteq X \setminus A.$$

Using the Lindelöf property for A and B in X, we can get sequences $\{U_j\}_{j=1}^{\infty}$ and $\{V_j\}_{j=1}^{\infty}$ of open subsets of X such that

(3.15.3)
$$A \subseteq \bigcup_{j=1}^{\infty} U_j, \quad B \subseteq \bigcup_{j=1}^{\infty} V_j,$$

and
$$(3.15.4) \overline{U_i} \subseteq X \setminus B, \quad \overline{V_i} \subseteq X \setminus A$$

for every $j \geq 1$. Put

$$(3.15.5) \widetilde{U}_j = U_j \setminus \bigcup_{l=1}^j \overline{V_l}, \quad \widetilde{V}_j = V_j \setminus \bigcup_{l=1}^j \overline{U_l}$$

for each $j \geq 1$. It is easy to see that \widetilde{U}_j and \widetilde{V}_j are open subsets of X for every $j \geq 1$. One can check that

(3.15.6)
$$A \subseteq \bigcup_{j=1}^{\infty} \widetilde{U}_j, \quad B \subseteq \bigcup_{j=1}^{\infty} \widetilde{V}_j,$$

using (3.15.3) and (3.15.4).

Thus

(3.15.7)
$$\widetilde{U} = \bigcup_{j=1}^{\infty} \widetilde{U}_j, \quad \widetilde{V} = \bigcup_{j=1}^{\infty} \widetilde{V}_j$$

are open subsets of X that contain A and B, respectively. We would like to verify that $\widetilde{U} \cap \widetilde{V} = \emptyset$. To do this, it suffices to check that

$$(3.15.8) \widetilde{U}_j \cap \widetilde{V}_l = \emptyset$$

for every $j, l \geq 1$. If $l \leq j$, then this follows from the definition of \widetilde{U}_j , and the fact that $\widetilde{V}_l \subseteq V_l$. The case where $j \leq l$ can be handled analogously.

Chapter 4

Some more set theory

4.1 Zorn's lemma

Let (A, \preceq) be a partially-ordered set. An element b of A is said to be an upper bound of a subset E of A if for every $a \in E$, we have that

$$(4.1.1) a \leq b.$$

A subset C of A is said to be a *chain* if C is linearly ordered by the restriction of \leq to C. This means that for every $x, y \in C$,

(4.1.2) either
$$x \leq y$$
 or $y \leq x$ (or both, i.e., $x = y$).

An element x of A is said to be maximal in A with respect to \leq if

(4.1.3) for every
$$y \in A$$
 with $x \leq y$, we have that $x = y$.

Let us say that $b \in A$ is a *top element* of A if b is an upper bound for A, so that (4.1.1) holds for every $a \in A$. A top element of A is unique when it exists, and is maximal in A. However, a maximal element of a partially-order set is not necessarily a top element, or unique. A maximal element of a linearly-ordered set is a top element. It is easy to see that a nonempty linearly-ordered set with only finitely many elements has a top element.

If A is a nonempty partially-ordered set with only finitely many elements, then A has a maximal element. To see this, let a_1 be any element of A. If a_1 is maximal in A, then we can stop. Otherwise, there is an element a_2 of A such that

$$(4.1.4)$$
 $a_1 \leq a_2 \text{ and } a_1 \neq a_2.$

We can repeat the process a finite number of times to get a maximal element of A, because A has only finitely many elements.

4.1.1 The statement of Zorn's lemma

Let A be any partially-ordered set again. If

(4.1.5) every chain in A has an upper bound in A,

then Zorn's lemma states that

$$(4.1.6)$$
 A has a maximal element.

Note that the empty set may be considered as a chain in A, so that the hypothesis of Zorn's lemma implies that $A \neq \emptyset$. There are well-known arguments for obtaining Zorn's lemma from the axiom of choice, as in [236], for instance. The converse is much simpler, and we shall return to that later.

4.1.2 Countably-infinite sets A

Suppose that A is countably infinite, and let $\{x_j\}_{j=1}^{\infty}$ be a sequence of elements of A in which every element of A occurs exactly once. We can try to find a maximal element of A using a more precise version of the argument for finite sets, as follows. Put $j_1 = 1$, and suppose that $j_l \in \mathbf{Z}_+$ has been chosen for some positive integer l. If there is an integer $l \in \mathbf{Z}_+$ such that

$$(4.1.7) x_{j_l} \leq x_k,$$

then we take j_{l+1} to be the smallest such integer k, and otherwise we stop. If this process stops after finitely many steps, then we get a maximal element of A. Otherwise, we get an infinite subsequence $\{x_{j_l}\}_{l=1}^{\infty}$ of $\{x_j\}_{j=1}^{\infty}$ such that

$$(4.1.8) x_{j_l} \preceq x_{j_{l+1}}$$

for every $l \geq 1$. In particular,

$$(4.1.9) C = \{x_{j_l} : l \in \mathbf{Z}_+\}$$

is a chain in A. If C has an upper bound in A, then this upper bound is of the form x_n for some positive integer n. In this case, there would have to be a positive integer l_0 such that $j_{l_0} = n$, because of the way that the j_l 's were chosen. This would contradict (4.1.8), because the x_j 's are supposed to be distinct elements of A.

4.2 Hausdorff's maximality principle

Let (A, \preceq) be a partially-ordered set again. Hausdorff's maximality principle states that

(4.2.1) there is a chain in A that is maximal with respect to inclusion.

More precisely, this means that there is a chain C_0 in A such that

(4.2.2) if C is any chain in A with
$$C_0 \subseteq C$$
, then $C_0 = C$.

If A has only finitely many elements, then A has only finitely many subsets, and in particular there are only finitely many chains in A. In this case, one can find a maximal chain in A as in the previous section. Alternatively, one can keep adding points to a chain in A until it is no longer possible to have a chain in A.

4.2.1 Maximal chains in countable sets

Similarly, suppose that A is countably infinite, and let $\{x_j\}_{j=1}^{\infty}$ be a sequence of elements of A in which every element of A occurs exactly once. Of course, $C_1 = \{x_1\}$ is a chain in A. If $x_1 \leq x_2$ or $x_2 \leq x_1$, then put $C_2 = \{x_1, x_2\}$, which is a chain in A, and otherwise put $C_2 = C_1$. If $C_n \subseteq \{x_1, \ldots, x_n\}$ has been chosen in this way for some positive integer n, then we can define C_{n+1} as follows. If $C_n \cup \{x_{n+1}\}$ is a chain in A, then we take it to be C_{n+1} . Otherwise, put $C_{n+1} = C_n$. One can check that C_n is a maximal chain in $\{x_1, \ldots, x_n\}$ for each positive integer n, as in the preceding paragraph. One can also verify that

$$(4.2.3) C = \bigcup_{n=1}^{\infty} C_n$$

is a maximal chain in A.

There are well-known arguments for obtaining Hausdorff's maximality principle from the axiom of choice, as in [236] again.

4.2.2 Zorn's lemma from maximal chains

Let us see how Zorn's lemma can be obtained from Hausdorff's maximality principle. Let (A, \preceq) be a partially-ordered set, and let C_0 be a maximal chain in A, as in Hausdorff's maximality principle. The hypothesis of Zorn's lemma implies that there is a $b \in A$ such that

$$(4.2.4)$$
 b is an upper bound for C_0 .

We would like to check that

$$(4.2.5)$$
 b is a maximal element of A

under these conditions. To do this, suppose that $y \in A$ satisfies $b \leq y$. It is easy to see that

$$(4.2.6) C_0 \cup \{b, y\} \text{ is a chain in } A$$

too in this situation. This implies that

$$(4.2.7) C_0 = C_0 \cup \{b, y\},$$

because C_0 is a maximal chain in A. This means that $b, y \in C_0$. It follows that $y \leq b$, because $y \in C_0$ and b is an upper bound for C_0 . Thus b = y, because $b \leq y$, by hypothesis, as desired.

4.3 Maximal chains from Zorn's lemma

We can also obtain Hausdorff's maximality principle from Zorn's lemma. Let (A, \preceq) be a partially-ordered set again, and let \mathcal{C} be the collection of all chains in A. We may consider \mathcal{C} as a partially-ordered set with respect to inclusion, i.e., using $C_1 \subseteq C_2$ for $C_1, C_2 \in \mathcal{C}$ as the partial ordering on \mathcal{C} . Hausdorff's maximality principle is exactly the statement that

(4.3.1)
$$\mathcal{C}$$
 has a maximal element.

Let us check that C satisfies the hypothesis of Zorn's lemma.

Let \mathcal{E} be a chain in \mathcal{C} . This means that \mathcal{E} is a collection of chains in A such that for every $C_1, C_2 \in \mathcal{E}$, either $C_1 \subseteq C_2$ or $C_2 \subseteq C_1$. We would like to show that \mathcal{E} has an upper bound in \mathcal{C} .

4.3.1 An upper bound for \mathcal{E} in \mathcal{C}

Put

(4.3.2)
$$C(\mathcal{E}) = \bigcup_{C \in \mathcal{E}} C.$$

More precisely, each element C of \mathcal{E} is a subset of A, so that their union is a subset of A as well. If $\mathcal{E} = \emptyset$, then $C(\mathcal{E})$ is interpreted as being the empty set too.

We would like to verify that $C(\mathcal{E})$ is a chain in A. Let $x, y \in C(\mathcal{E})$ be given. By definition of $C(\mathcal{E})$, there are $C_x, C_y \in \mathcal{E}$ such that

$$(4.3.3) x \in C_x \text{ and } y \in C_y.$$

Because \mathcal{E} is a chain in \mathcal{C} , we have that $C_x \subseteq C_y$ or $C_y \subseteq C_x$. It follows that x and y are both contained in C_x , or that x and y are both contained in C_y . In either case, we get that $x \leq y$ or $y \leq x$, because C_x and C_y are chains in A. This shows that $C(\mathcal{E})$ is a chain in A.

Equivalently, this means that $C(\mathcal{E}) \in \mathcal{C}$. Of course, if $C \in \mathcal{E}$, then

$$(4.3.4) C \subseteq C(\mathcal{E}),$$

by construction. Thus $C(\mathcal{E})$ is an upper bound for \mathcal{E} in \mathcal{C} . This implies that \mathcal{C} satisfies the hypothesis of Zorn's lemma. In this situation, the conclusion of Zorn's lemma is that \mathcal{C} has a maximal element, as desired.

4.4 The axiom of choice

Let us show how the axiom of choice can be obtained from Zorn's lemma or Hausdorff's maximality principle. Let I be a nonempty set, and let X_j be a nonempty set for each $j \in I$. We would like to show that there is a mapping f from I into $\bigcup_{j \in I} X_j$ such that

$$(4.4.1) f(j) \in X_j$$

for every $j \in I$.

4.4.1 A partially-ordered set A

Let A be the set of ordered pairs (I_0, f_0) , where I_0 is a subset of I, and f_0 is a mapping from I_0 into $\bigcup_{j \in I_0} X_j$ such that $f_0(j) \in X_j$ for every $j \in I_0$. If $(I_1, f_1), (I_2, f_2) \in A$, then put

$$(4.4.2) (I_1, f_1) \leq (I_2, f_2)$$

when $I_1 \subseteq I_2$ and $f_1 = f_2$ on I_1 . It is easy to see that this defines a partial ordering on A.

4.4.2 Upper bounds for arbitrary chains in A

Let C be a chain in A, and let us check that C has an upper bound in A. Put

$$(4.4.3) I_{\mathcal{C}} = \bigcup_{(I_0, f_0) \in \mathcal{C}} I_0,$$

which is a subset of I. We would like to define a mapping $f_{\mathcal{C}}$ from $I_{\mathcal{C}}$ into $\bigcup_{j \in I_{\mathcal{C}}} X_j$ as follows. If $j \in I_{\mathcal{C}}$, then there is an element (I_1, f_1) of \mathcal{C} such that $j \in I_1$, by definition of $I_{\mathcal{C}}$. In this case, we would like to put

$$(4.4.4) f_C(j) = f_1(j).$$

We need to check that this does not depend on the particular choice of (I_1, f_1) . Suppose that (I_2, f_2) is another element of C such that $j \in I_2$. Note that

$$(4.4.5) (I_1, f_1) \leq (I_2, f_2) \text{ or } (I_2, f_2) \leq (I_1, f_1),$$

because \mathcal{C} is a chain in A. In both cases, we have that $f_1(j) = f_2(j)$, so that $f_{\mathcal{C}}(j)$ is well defined. Of course, $f_{\mathcal{C}}(j) = f_1(j) \in X_j$, so that $(I_{\mathcal{C}}, f_{\mathcal{C}}) \in A$. If (I_0, f_0) is an element of \mathcal{C} , then $I_0 \subseteq I_{\mathcal{C}}$ and $f_0 = f_{\mathcal{C}}$ on I_0 , by construction. This means that

$$(4.4.6) (I_0, f_0) \leq (I_C, f_C),$$

so that $(I_{\mathcal{C}}, f_{\mathcal{C}})$ is an upper bound for \mathcal{C} in A.

This shows that A satisfies the hypothesis of Zorn's lemma, so that Zorn's lemma implies that A has a maximal element. Alternatively, Hausdorff's maximality principle says that A has a maximal chain. The previous argument implies that such a maximal chain has an upper bound in A, which is a maximal element of A, as before.

If (I_0, f_0) is a maximal element of A, then we would like to show that

$$(4.4.7) I_0 = I.$$

Otherwise, there is an element j_1 of I not in I_0 , and we put $I_1 = I_0 \cup \{j_1\}$. Let x_{j_1} be an element of X_{j_1} . Consider the mapping f_1 from I_1 into $\bigcup_{j \in I_1} X_j$ defined by putting $f_1(j) = f_0(j)$ when $j \in I_0$ and $f_1(j_1) = x_{j_1}$. Thus $f_1(j) \in X_j$ for every $j \in I_1$, so that $(I_1, f_1) \in A$. We also have that $(I_0, f_0) \preceq (I_1, f_1)$ and $(I_0, f_0) \neq (I_1, f_1)$, by construction. This contradicts the maximality of (I_0, f_0) in A. It follows that $I_0 = I$, as desired.

4.5 Injective mappings

Let A and B be sets. We would like to show how Zorn's lemma or Hausdorff's maximality principle implies that there is either a one-to-one mapping from A into B, or a one-to-one mapping from B into A.

4.5.1 A partially-ordered set A

Let \mathcal{A} be the collection of ordered triples (A_0, B_0, f_0) , where $A_0 \subseteq A$, $B_0 \subseteq B$, and f_0 is a one-to-one mapping from A_0 onto B_0 . If $(A_1, B_1, f_1), (A_2, B_2, f_2) \in \mathcal{A}$, then put

$$(4.5.1) (A_1, B_1, f_1) \leq (A_2, B_2, f_2)$$

when $A_1 \subseteq A_2$, $B_1 \subseteq B_2$, and $f_1 = f_2$ on A_1 . One can check that this defines a partial ordering on A.

4.5.2 Upper bounds for arbitrary chains in A

Let C be a chain in A, and let us verify that C has an upper bound in A. Put

(4.5.2)
$$A_{\mathcal{C}} = \bigcup_{(A_0, B_0, f_0) \in \mathcal{C}} A_0$$

and

(4.5.3)
$$B_{\mathcal{C}} = \bigcup_{(A_0, B_0, f_0) \in \mathcal{C}} B_0,$$

which are subsets of A and B, respectively. We would like to define a mapping $f_{\mathcal{C}}$ from $A_{\mathcal{C}}$ into $B_{\mathcal{C}}$ as follows. If $x \in A_{\mathcal{C}}$, then there is an element (A_1, B_1, f_1) of \mathcal{C} such that $x \in A_1$, and we would like to put

$$(4.5.4) f_{\mathcal{C}}(x) = f_1(x).$$

One can check that this does not depend on the particular choice of (A_1, B_1, f_1) , as in the previous section. Note that (4.5.4) is an element of $B_1 \subseteq B_{\mathcal{C}}$, so that $f_{\mathcal{C}}$ is a well-defined mapping from $A_{\mathcal{C}}$ into $B_{\mathcal{C}}$. We would like to verify that $f_{\mathcal{C}}$ is a one-to-one mapping from $A_{\mathcal{C}}$ onto $B_{\mathcal{C}}$, so that $(A_{\mathcal{C}}, B_{\mathcal{C}}, f_{\mathcal{C}}) \in \mathcal{A}$.

Let x_1 and x_2 be distinct elements of $A_{\mathcal{C}}$. It follows that there are elements (A_1, B_1, f_1) and (A_2, B_2, f_2) of \mathcal{C} such that $x_1 \in A_1$ and $x_2 \in A_2$, by the definition (4.5.2) of $A_{\mathcal{C}}$. Because \mathcal{C} is a chain in \mathcal{A} , we have that

(4.5.5) either
$$(A_1, B_1, f_1) \leq (A_2, B_2, f_2)$$
 or $(A_2, B_2, f_2) \leq (A_1, B_1, f_1)$.

In the first case, we get that $A_1 \subseteq A_2$, $x_1, x_2 \in A_2$, and hence $f_{\mathcal{C}}(x_1) = f_2(x_1)$, $f_{\mathcal{C}}(x_2) = f_2(x_2)$. Of course, f_2 is injective on A_2 , by hypothesis, so that

$$(4.5.6) f_2(x_1) \neq f_2(x_2)$$

This means that

(4.5.7)
$$f_{\mathcal{C}}(x_1) \neq f_{\mathcal{C}}(x_2)$$

in this case, and and the other case can be handled in the same way. This shows that $f_{\mathcal{C}}$ is injective on $A_{\mathcal{C}}$.

Now let $y \in \mathcal{B}_{\mathcal{C}}$ be given. By the definition (4.5.3) of $B_{\mathcal{C}}$, there is an element (A_0, B_0, f_0) of \mathcal{C} such that $y \in B_0$. This implies that there is an $x \in A_0$ such that

$$(4.5.8) f_0(x) = y,$$

because f_0 maps A_0 onto B_0 , by hypothesis. It follows that $x \in A_{\mathcal{C}}$, and that

$$(4.5.9) f_C(x) = y,$$

by the definitions of $A_{\mathcal{C}}$ and $f_{\mathcal{C}}$. Thus $f_{\mathcal{C}}$ maps $A_{\mathcal{C}}$ onto $B_{\mathcal{C}}$.

This shows that $(A_{\mathcal{C}}, B_{\mathcal{C}}, f_{\mathcal{C}}) \in \mathcal{A}$. It is easy to see that $(A_{\mathcal{C}}, B_{\mathcal{C}}, f_{\mathcal{C}})$ is an upper bound for \mathcal{C} in \mathcal{A} , by construction. This means that \mathcal{A} satisfies the hypothesis of Zorn's lemma, so that Zorn's lemma implies that \mathcal{A} has a maximal element. Alternatively, Hausdorff's maximality principle says that \mathcal{A} has a maximal chain, and the upper bound in \mathcal{A} for such a maximal chain obtained as before is a maximal element of \mathcal{A} .

If (A_0, B_0, f_0) is a maximal element of \mathcal{A} , then we would like to show that

$$(4.5.10) A_0 = A \text{ or } B_0 = B.$$

Otherwise, there is an element a_1 of A not in A_0 , and an element b_1 of B not in B_0 . Put $A_1 = A_0 \cup \{a_1\}$ and $B_1 = B_0 \cup \{b_1\}$, so that $A_1 \subseteq A$ and $B_1 \subseteq B$. Let f_1 be the mapping from A_1 into B_1 defined by putting $f_1(x) = f_0(x)$ when $x \in A_0$ and $f_1(a_1) = b_1$. Thus f_1 is a one-to-one mapping from A_1 onto B_1 , so that $(A_1, B_1, f_1) \in \mathcal{A}$. By construction, $(A_0, B_0, f_0) \preceq (A_1, B_1, f_1)$ and $(A_0, B_0, f_0) \neq (A_1, B_1, f_1)$, contradicting the maximality of (A_0, B_0, f_0) in \mathcal{A} . This shows that $A_0 = A$ or $B_0 = B$.

If $A_0 = A$, then f_0 is a one-to-one mapping from A into B. If $B_0 = B$, then the inverse of f_0 is a one-to-one mapping from B into A.

4.6 Well-ordered sets

Let (A, \preceq) be a linearly-ordered set. Thus, for each $x, y \in A$, we have that $x \preceq y$ or $y \preceq x$. We say that A is well ordered by \preceq if

(4.6.1) every nonempty subset of A has a smallest element.

More precisely, this means that if E is a nonempty subset of A, then

(4.6.2) there is an
$$x \in E$$
 such that $x \leq y$ for every $y \in E$.

Note that the smallest element in ${\cal E}$ is automatically unique.

It is easy to see that a linearly-ordered set with only finitely many elements is well ordered. The set \mathbf{Z}_+ of positive integers is well ordered by its standard ordering.

4.6.1 The well-ordering principle

Zermelo's well-ordering principle states that every set can be well ordered. This can be obtained from the axiom of choice, as discussed in [236], for instance.

It is easy to see how the axiom of choice can be obtained from the well-ordering principle. Let I be a nonempty set, and suppose that X_j is a nonempty set for each $j \in I$. The well-ordering principle implies that $\bigcup_{j \in I} X_j$ can be well ordered. If $j \in I$, then let f(j) be the smallest element of X_j with respect to this ordering. This defines a mapping f from I into $\bigcup_{j \in I} X_j$ such that $f(j) \in X_j$ for each $j \in I$.

4.6.2 Ideals and segments in partially-ordered sets

Let (A, \preceq) be a partially-ordered set, and let B be a subset of A. It is easy to see that the restriction of \preceq to B is a partial ordering on B. If A is linearly ordered by \preceq , then B is linearly ordered by the restriction of \preceq to B. If A is well ordered by \preceq , then B is well ordered by \preceq too.

Let (A, \preceq) be a partially-ordered set again. A subset B of A is said to be an ideal in A if

(4.6.3) for every $x \in A$ and $y \in B$ with $x \leq y$ we have that $x \in B$.

If $a \in A$, then the *segment* in A associated to a is defined by

$$(4.6.4) S(a) = S_A(a) = \{x \in A : x \leq a, x \neq a\}.$$

It is easy to see that segments in A are ideals in A. Of course, A is an ideal in itself

Suppose that (A, \preceq) is a linearly-ordered set. If $a \in A$, then it is easy to see that

$$(4.6.5) A \setminus S(a) = \{x \in A : a \leq x\}.$$

4.6.3 Ideals in well-ordered sets

Suppose now that (A, \leq) is a well-ordered set, and that $B \subseteq A$ is an ideal in A. If $B \neq A$, then there is a smallest element a_0 of $A \setminus B$. One can check that

$$(4.6.6) B = S(a_0)$$

under these conditions. More precisely,

$$(4.6.7) S(a_0) \subseteq B$$

automatically, because a_0 is the smallest element of $A \setminus B$. To show that

$$(4.6.8) B \subseteq S(a_0),$$

suppose for the sake of a contradiction that $x \in B$ and $x \notin S(a_0)$. This implies that $a_0 \leq x$, because A is linearly ordered by \leq , as in (4.6.5). It follows that $a_0 \in B$, because B is an ideal in A. This contradicts the fact that $a_0 \in A \setminus B$, as desired.

4.6.4 Checking that a linear ordering is a well ordering

Let (A, \preceq) be a linearly-ordered set. If

(4.6.9) every segment in A is well ordered by
$$\leq$$
,

then A is well ordered by \leq . To see this, let E be a nonempty subset of A, and let us show that E has a smallest element. Let a be an element of E. If a is already the smallest element of E, then we can stop. Otherwise,

$$(4.6.10) E \cap S(a) \neq \emptyset.$$

In this case, $E \cap S(a)$ has a smallest element, because S(a) is well ordered, by hypothesis. It is easy to see that the smallest element of $E \cap S(a)$ is also the smallest element of E, using (4.6.5), as desired.

4.7 The well-ordering principle

The well-ordering principle can be obtained from Zorn's lemma or Hausdorff's maximality principle as follows. Let A be a set, on which we would like to find a well ordering.

4.7.1 The collection of well-ordered subsets of A

Let \mathcal{A} be the collection of ordered pairs (A_0, \leq_0) , where A_0 is a subset of A well ordered by \leq_0 . If $(A_1, \leq_1), (A_2, \leq_2) \in \mathcal{A}$, then put

$$(4.7.1) (A_1, \preceq_1) \preceq_{\mathcal{A}} (A_2, \preceq_2)$$

when $A_1 \subseteq A_2$, the restriction of \leq_2 to A_1 is the same as \leq_1 , and A_1 is an ideal in A_2 with respect to \leq_2 . One can check that this defines a partial ordering on A.

4.7.2 Upper bounded for chains of well-ordered subsets

Let C be a chain in A. We would like to show that C has an upper bound in A, as usual. Put

$$(4.7.2) A_{\mathcal{C}} = \bigcup_{(A_0, \preceq_0) \in \mathcal{C}} A_0,$$

which is a subset of A. We first need to define an ordering $\leq_{\mathcal{C}}$ on $A_{\mathcal{C}}$.

Let $x, y \in A_{\mathcal{C}}$ be given, so that there are $(A_1, \preceq_1), (A_2, \preceq_2) \in \mathcal{C}$ such that $x \in A_1$ and $y \in A_2$. Because \mathcal{C} is a chain in \mathcal{A} , we have that

$$(4.7.3) (A_1, \preceq_1) \preceq_{\mathcal{A}} (A_2, \preceq_2) \text{ or } (A_2, \preceq_2) \preceq_{\mathcal{A}} (A_1, \preceq_1).$$

In particular, this means that $A_1 \subseteq A_2$ or $A_2 \subseteq A_1$. It follows that $x, y \in A_1$ or $x, y \in A_2$.

Let (A_0, \leq_0) be any element of \mathcal{C} such that $x, y \in A_0$, the existence of which follows from the remarks in the preceding paragraph. We would like to put

$$(4.7.4) x \leq_{\mathcal{C}} y$$

when $x \leq_0 y$. One can check that this does not depend on the particular element (A_0, \leq_0) of \mathcal{C} with $x, y \in A_0$, because \mathcal{C} is a chain in \mathcal{A} .

It is not difficult to show that $A_{\mathcal{C}}$ is linearly ordered by $\preceq_{\mathcal{C}}$, using the fact that the elements of \mathcal{C} are linearly-ordered sets. More precisely, to check that $\preceq_{\mathcal{C}}$ is transitive on $A_{\mathcal{C}}$, let $x,y,z\in A_{\mathcal{C}}$ be given. One can verify that there is an element (A_0, \prec_0) of \mathcal{C} such that $x,y,z\in A_0$, using the fact that \mathcal{C} is a chain in \mathcal{A} , as before. If $x\preceq_{\mathcal{C}} y$ and $y\preceq_{\mathcal{C}} z$, then one can get that $x\preceq_{\mathcal{C}} z$ using transitivity of \preceq_0 on A_0 .

4.7.3 Elements of \mathcal{C} correspond to ideals in $A_{\mathcal{C}}$

Let (A_1, \leq_1) be any element of \mathcal{C} . By construction, $A_1 \subseteq A_{\mathcal{C}}$, and $\leq_{\mathcal{C}}$ agrees with \leq_1 on A_1 . Let us check that A_1 is an ideal in $A_{\mathcal{C}}$ with respect to $\leq_{\mathcal{C}}$. Let $x \in A_{\mathcal{C}}$ and $y \in A_1$ be given, with $x \leq_{\mathcal{C}} y$. By definition of $A_{\mathcal{C}}$, there is an element (A_2, \leq_2) of \mathcal{C} such that $x \in A_2$. We also have that (4.7.3) holds, because \mathcal{C} is a chain in \mathcal{A} . In the first case, $x, y \in A_2$ and $x \leq_2 y$, which implies that $x \in A_1$, because $y \in A_1$ and A_1 is an ideal in A_2 . In the second case, $x \in A_2 \subseteq A_1$. Thus $x \in A_1$ in both cases, as desired.

4.7.4 Checking that $A_{\mathcal{C}}$ is well ordered by $\leq_{\mathcal{C}}$

Let us check that $A_{\mathcal{C}}$ is well ordered by $\leq_{\mathcal{C}}$. Let $a \in A_{\mathcal{C}}$ be given, and let us verify that the corresponding segment $S_{A_{\mathcal{C}}}(a)$ in $A_{\mathcal{C}}$ is well ordered by $\leq_{\mathcal{C}}$. By construction, there is an $(A_0, \leq_0) \in \mathcal{C}$ such that $a \in A_0$. We also have that

$$(4.7.5) S_{A_c}(a) \subseteq A_0,$$

because A_0 is an ideal in $A_{\mathcal{C}}$, as in the preceding paragraph. It follows that $S_{A_{\mathcal{C}}}(a)$ is well ordered by $\leq_{\mathcal{C}}$, because \leq_0 and $\leq_{\mathcal{C}}$ are the same on A_0 , and A_0 is well ordered by \leq_0 .

This shows that $(A_{\mathcal{C}}, \preceq_{\mathcal{C}})$ is an element of \mathcal{A} . If $(A_0, \preceq_0) \in \mathcal{C}$, then it is easy to see that $(A_0, \preceq_0) \preceq_{\mathcal{A}} (A_{\mathcal{C}}, \preceq_{\mathcal{C}})$, by the previous remarks. This means that $(A_{\mathcal{C}}, \preceq_{\mathcal{C}})$ is an upper bound for \mathcal{C} in \mathcal{A} . It follows that \mathcal{A} has a maximal element, by Zorn's lemma. Alternatively, Hausdorff's maximality principle implies that \mathcal{A} has a maximal chain, and an upper bound for such a chain is a maximal element of \mathcal{A} , as usual.

Let (A_1, \preceq_1) be a maximal element of \mathcal{A} . We would like to show that $A_1 = A$. Otherwise, there is an $a_2 \in A$ such that $a_2 \notin A_1$. Put $A_2 = A_1 \cup \{a_2\}$, and let us define \preceq_2 on A_2 as follows. We take \preceq_2 to be the same as \preceq_1 on A_1 , and we put $x \preceq_2 a_2$ for every $x \in A_2$. It is easy to see that A_2 is well ordered by \preceq_2 , because A_1 is well ordered by \preceq_1 . Clearly A_1 is an ideal in A_2 , by construction. Thus $(A_2, \preceq_2) \in \mathcal{A}$ and $(A_1, \preceq_1) \preceq_{\mathcal{A}} (A_2, \preceq_2)$. This contradicts the maximality of (A_1, \preceq_1) in \mathcal{A} , because $A_1 \neq A_2$.

4.8 Order isomorphisms

Let (A_1, \leq_1) and (A_2, \leq_2) be partially-ordered sets. A one-to-one mapping f from A_1 onto A_2 is said to be an *order isomorphism* if for every $x, y \in A_1$,

(4.8.1)
$$x \leq_1 y$$
 if and only if $f(x) \leq_2 f(y)$.

In this case, the inverse mapping f^{-1} is an order isomorphism from A_2 onto A_1 . Let (A_3, \leq_3) be another partially-ordered set, and suppose that g is an order isomorphism from A_2 onto A_3 . Under these conditions, $g \circ f$ is an order isomorphism from A_1 onto A_3 .

An order isomorphism from a partially-ordered set onto itself may be called an *order automorphism*. Of course, the identity mapping on any partially-ordered set is an order automorphism.

Now let (A_1, \leq_1) and (A_2, \leq_2) be well-ordered sets, and let f be an order isomorphism from A_1 onto A_2 . If E_1 is a nonempty subset of A_1 , then f maps the smallest element of E_1 to the smallest element of $f(E_1)$. In particular, if $A_1 \neq \emptyset$, then f maps the smallest element of A_1 to the smallest element of A_2 .

4.8.1 Automorphisms of well-ordered sets

Let (A, \preceq) be a well-ordered set, and let f be an order automorphism on A. We would like to check that

$$(4.8.2)$$
 f is the identity mapping on A.

Otherwise, let a be the smallest element of A such that $f(a) \neq a$. Thus f is the identity mapping on the segment S(a), and in particular f maps S(a) onto itself. This implies that f maps the complement of S(a) in A onto itself. However, a is the smallest element of the complement of S(a) in A. It follows that f(a) = a, which is a contradiction.

4.8.2 Uniqueness of isomorphisms between well-ordered sets

Let (A_1, \leq_1) and (A_2, \leq_2) be well-ordered sets, and suppose that f and g are order isomorphisms from A_1 onto A_2 . This implies that $g^{-1} \circ f$ is an order automorphism on A_1 . It follows that $g^{-1} \circ f$ is the identity mapping on A_1 , as in the preceding paragraph. Of course, this means that

$$(4.8.3) g = f,$$

so that an order isomorphism between well-ordered sets is unique when it exists.

4.8.3 Isomorphisms onto ideals in well-ordered sets

Let (A, \preceq) be a well-ordered set again, and let B be an ideal in A. As before, B is well ordered by the restriction of \preceq to B. Suppose that f is an order

isomorphism from A onto B. Under these conditions, f is the identity mapping on A, so that

$$(4.8.4) B = f(A) = A.$$

This is an extension of the earlier statement for order automorphisms on A, which can be shown in essentially the same way. More precisely, if f is not the identity mapping on A, then there is a smallest element a of A such that $f(a) \neq a$. This means that f is equal to the identity mapping on S(a), so that f(S(a)) = S(a). It follows that $f(a) \notin S(a)$, because $a \notin S(a)$ and f is injective, which is to say that $a \leq f(a)$. This implies that $a \in B$, because $f(a) \in f(A) = B$ and B is an ideal in A. Note that

$$(4.8.5) f(A \setminus S(a)) = f(A) \setminus f(S(a)) = f(A) \setminus S(a).$$

Of course, a is the smallest element of $A \setminus S(a)$, and a is the smallest element of (4.8.5) too. Thus f(a) = a, which is a contradiction.

Alternatively, one can show more directly that a well-ordered set cannot be order-isomorphic to any of its segments, as in Theorem 20 on p51 of [236].

4.8.4 Uniqueness of isomorphic segments in well-ordered sets

Let (A_1, \leq_1) and (A_2, \leq_2) be partially-ordered sets again, and let f be an order isomorphism from A_1 onto A_2 . If B_1 is an ideal in A_1 , then $f(B_1)$ is an ideal in A_2 . If $a_1 \in A_1$, then f maps the corresponding segment $S_{A_1}(a_1)$ in A_1 onto the segment $S_{A_2}(f(a_1))$ associated to $f(a_1)$ in A_2 .

Suppose that A_1 and A_2 are well ordered by \leq_1 and \leq_2 , respectively. If $a_1 \in A_1$, then there is at most one $a_2 \in A_2$ such that $S_{A_1}(a_1)$ is order isomorphic to $S_{A_2}(a_2)$, with respect to the restrictions of \leq_1 and \leq_2 to $S_{A_1}(a_1)$ and $S_{A_2}(a_2)$, respectively. Otherwise, there are elements a_2' and a_2'' of A_2 such that $a_2' \neq a_2''$ and $S_{A_2}(a_2')$ is order isomorphic to $S_{A_2}(a_2'')$. We may as well suppose that $a_2' \leq_2 a_2''$, because A_2 is linearly ordered by \leq_2 , and otherwise we could interchange the roles of a_2' and a_2'' . This means that $a_2' \in S_{A_2}(a_2'')$, because $a_2' \neq a_2''$. Thus $S_{A_2}(a_2')$ may be considered as a segment in $S_{A_2}(a_2'')$. This contradicts the fact that a well-ordered set cannot be order isomorphic to any of its segments, as before.

This can also be used to prove the uniqueness of order isomorphisms between well-ordered sets, by considering the images of segments under the order isomorphisms.

4.9 Order isomorphisms, continued

Let (A, \preceq_A) and (B, \preceq_B) be well-ordered sets. Under these conditions, Theorem 21 on p51 of [236] states that

(4.9.1) A is order-isomorphic to an ideal in B, or B is order-isomorphic to an ideal in A.

More precisely, this means that

(4.9.2) either A is order-isomorphic to B, A is order-isomorphic to a segment in B,
or B is order-isomorphic to a segment in A.

It is easy to see that only one of these three possibilities can occur, because a well-ordered set cannot be order-isomorphic to any of its segments, as in the previous section.

4.9.1 A subset I_A of A

Let I_A be the set of $a \in A$ for which there is a $b \in B$ such that the segment $S_A(a)$ corresponding to a in A is order-isomorphic to the segment $S_B(b)$ corresponding to b in B. One can check that b is uniquely determined by this property, because a well-ordered set cannot be order-isomorphic to any of its segments. If $a \in I_A$, then let f(a) be the element of B such that $S_A(a)$ is order-isomorphic to $S_B(f(a))$. This defines a mapping from I_A into B.

4.9.2 More on I_A

Let $a \in I_A$ be given, so that there is an order isomorphism ϕ_a from $S_A(a)$ onto $S_B(f(a))$. More precisely, this uses the restrictions of \leq_A and \leq_B to $S_A(a)$ and $S_B(f(a))$, respectively.

If $a_0 \in S_A(a)$, then $S_A(a_0)$ is the same as the segment corresponding to a_0 in $S_A(a)$. Similarly, $\phi_a(a_0) \in S_B(f(a))$, and $S_B(\phi_a(a_0))$ is the same as the segment corresponding to $\phi_a(a_0)$ in $S_B(f(a))$. It follows that

$$\phi_a(S_A(a_0)) = S_B(\phi_a(a_0)),$$

as in the previous section.

Note that the restriction of ϕ_a to $S_A(a_0)$ is an order isomorphism onto $S_B(\phi_a(a_0))$. This means that $a_0 \in I_A$, with

$$(4.9.4) f(a_0) = \phi_a(a_0).$$

Thus

$$(4.9.5) S_A(a) \subseteq I_A,$$

and

(4.9.6)
$$f(S_A(a)) = \phi_a(S_A(a)) = S_B(f(a)).$$

In particular, I_A is an ideal in A.

4.9.3 A subset I_B of B

Similarly, let I_B be the set of $b \in B$ such that $S_B(b)$ is order-isomorphic to $S_A(a)$ for some $a \in A$. This means that $a \in I_A$ and f(a) = b, and in fact

$$(4.9.7) f(I_A) = I_B.$$

It is easy to see that I_B is an ideal in B, using (4.9.6). One can check that f is an order isomorphism from I_A onto I_B , with respect to the restrictions of \leq_A and \leq_B to I_A and I_B , respectively.

4.9.4 Either $I_A = A$ **or** $I_B = B$

Suppose for the sake of a contradiction that $I_A \neq A$ and $I_B \neq B$. This implies that there are $x \in A$ and $y \in B$ such that $I_A = S_A(x)$ and $I_B = S_B(y)$, as in Subsection 4.6.3. It follows that $x \in I_A$ and $y \in I_B$, because f is an order isomorphism from I_A onto I_B . This is a contradiction, and so we get that $I_A = A$ or $I_B = B$, as desired.

Chapter 5

Some additional notions in topology

5.1 Continuous real-valued functions

Let X be a nonempty topological space, and let C(X) be the space of continuous real-valued functions on X. This uses the standard topology on the real line, as the range of the functions on X. Of course, constant functions on X are continuous.

5.1.1 Sums and products

If
$$f, g \in C(X)$$
, then
$$(5.1.1) f + g, f g \in C(X)$$

too. This can be shown using the same type of arguments as for continuous real-valued functions on the real line, or on a metric space.

Alternatively, it is easy to see that

$$(5.1.2) x \mapsto (f(x), g(x))$$

is continuous as a mapping from X into \mathbb{R}^2 , using the product topology on \mathbb{R}^2 corresponding to the standard topology on \mathbb{R} . One can also show that

(5.1.3) addition and multiplication on
$$\mathbf{R}$$
 are continuous as mappings from \mathbf{R}^2 into \mathbf{R} ,

using standard arguments. To get the continuity of f + g and f g, one can consider these functions as the compositions of (5.1.2) with the mappings from \mathbf{R}^2 into \mathbf{R} that correspond to addition and multiplication of real numbers, respectively.

5.1.2 Reciprocals of continuous functions

Similarly, if $f \in C(X)$ and $f(x) \neq 0$ for each $x \in X$, then

$$(5.1.4) 1/f \in C(X).$$

This can be obtained using the same type of arguments as for functions on the real line or a metric space, or by considering 1/f as the composition of f with the mapping $t \mapsto 1/t$ from $\mathbb{R} \setminus \{0\}$ into itself.

5.1.3 Urysohn spaces

Let us say that C(X) separates points in X if for every $x, y \in X$ with $x \neq y$ there is an $f \in C(X)$ such that

$$(5.1.5) f(x) \neq f(y).$$

In this case, X is said to be a *Urysohn space*. One can check that

This uses the fact that the real line is completely Hausdorff with respect to the standard topology.

Let X be a set, and suppose that τ and $\widetilde{\tau}$ are topologies on X, with $\tau \subseteq \widetilde{\tau}$. If a real-valued function f on X is continuous with respect to τ , then

(5.1.7)
$$f$$
 is continuous with respect to $\tilde{\tau}$

too. If (X,τ) is a Urysohn space, then it follows that

(5.1.8)
$$(X, \tilde{\tau})$$
 is a Urysohn space

as well.

5.1.4 Continuous functions from semimetrics

Now let X be a set with a semimetric d(x,y). If $p \in X$, then one can verify that

(5.1.9)
$$f_p(x) = d(p, x)$$

is continuous on X, with respect to the topology determined by $d(\cdot, \cdot)$. If $d(\cdot, \cdot)$ is a metric on X, then it follows that X is a Urysohn space with respect to this topology.

5.1.5 Maxima and minima of continuous real-valued func-

If f and g are continuous real-valued functions on a topological space X again, then it is not difficult to show that

(5.1.10)
$$\max(f(x), g(x)) \quad \text{and} \quad \min(f(x), g(x))$$

are continuous on X as well, directly from the definitions. Alternatively, one can check that

(5.1.11) the maximum and minimum of two real numbers define continuous mappings from \mathbb{R}^2 into \mathbb{R} .

The continuity of the maximum and minimum of f and g can be obtained from this and the continuity of (5.1.2) as a mapping from X into \mathbb{R}^2 , as before.

In particular, if a and b are real numbers, then

(5.1.12)
$$\max(f(x), a) \text{ and } \min(f(x), b)$$

are continuous on X. This can also be seen using the continuity of

(5.1.13)
$$\max(t, a) \text{ and } \min(t, b)$$

as functions of $t \in \mathbf{R}$, by composing f with these functions.

5.2 Urysohn functions

Let X be a topological space, and let A and B be disjoint subsets of X. A continuous real-valued function f on X is said to be a *Urysohn function* for A and B if

$$(5.2.1) f(x) = 0 for every x \in A,$$

$$(5.2.2) f(y) = 1 for every y \in B,$$

and

$$(5.2.3) 0 \le f(w) \le 1 \text{for every } w \in X.$$

The third condition can always be arranged by replacing f with

$$(5.2.4)$$
 min(max(f, 0), 1),

as in the previous section.

If f is a Urysohn function on X for A and B, then

$$(5.2.5)$$
 $1-f$ is a Urysohn function on X for B and A.

It is easy to see that X is a Urysohn space if and only if every pair A, B of disjoint subsets of X with only one element each has a Urysohn function.

5.2.1 Urysohn's lemma

If X is normal in the strict sense, and A, B are disjoint closed subsets of X, then Urysohn's lemma states that

(5.2.6) there is a Urysohn function on
$$X$$
 for A and B .

One can check that this property implies that X is normal in the strict sense. If X is normal in the strong sense, then it follows that X is a Urysohn space.

5.2.2 Urysohn's metrization theorem

Urysohn's metrization theorem states that if X is second countable and normal in the strong sense, then there is a metric on X that determines the same topology. Tychonoff extended this to second countable spaces that are regular in the strong sense.

5.2.3 Some properties of Urysohn spaces

Suppose that X is a Urysohn space, and let x, y be distinct elements of X. Thus there is a Urysohn function f on X for $\{x\}$ and $\{y\}$, as before. Under these conditions,

$$(5.2.7) 2 \min(f, 1/2)$$

is a Urysohn function on X for $\{x\}$ and

$$\{z \in X : f(z) \ge 1/2\}.$$

Note that

$$(5.2.9) {z \in X : f(z) > 1/2}$$

is an open subset of X that contains y and is contained in (5.2.8).

Let B be a compact subset of X, and suppose that $x \in X \setminus B$. One can get

(5.2.10) a Urysohn function on
$$X$$
 for $\{x\}$ and B ,

as follows. If $y \in B$, then there is a Urysohn function on X for $\{x\}$ and a neighborhood of y in X, as in the preceding paragraph. Because B is compact, B can be covered by finitely many such neighborhoods of its elements. The maximum of the corresponding Urysohn functions for $\{x\}$ and these finitely many neighborhoods of elements of B is a Urysohn function for $\{x\}$ and B.

If A and B are disjoint compact subsets of X, then one can get

$$(5.2.11)$$
 a Urysohn function on X for A and B

using analogous arguments. More precisely, for each $x \in A$, one can get a Urysohn function for $\{x\}$ and B, as before. One can use this to get a Urysohn function on X for a neighborhood of x and B. Because A is compact, A can be covered by finitely many such neighborhoods of its elements. The minimum of the corresponding Urysohn functions is a Urysohn function on X for A and B.

5.3 Complete regularity

Let us say that a topological space X is completely regular in the strict sense if for every $x \in X$ and closed set $E \subseteq X$ with $x \notin E$ there is an $f \in C(X)$ such that $f(x) \neq 0$ and

$$(5.3.1) f(y) = 0 for every y \in E.$$

In this case, it is easy to modify f a bit, if necessary, to get that

(5.3.2)
$$f$$
 is a Urysohn function for E and $\{x\}$.

If X also satisfies the zeroth separation condition, then we say that X is completely regular in the strong sense. In this case, X is said to be a Tychonoff space as well.

As usual, completely regular spaces in the strict sense are sometimes said to be completely regular, and completely regular spaces in the strong sense may be said to satisfy separation condition number three and a half, or equivalently be $T_{3\frac{1}{2}}$ spaces, but these terms may be used the other way. Alternatively, complete regularity, separation condition number three and a half, and $T_{3\frac{1}{2}}$ spaces may be used for complete regularity in the strong sense, and one may refer to complete regularity in the strict sense in other ways.

5.3.1 Relation with other separation conditions

If X is completely regular in the strict sense, then it is easy to see that

$$(5.3.3)$$
 X is regular in the strict sense,

because the real line is Hausdorff with respect to the standard topology. If X is completely regular in the strong sense, then it follows that

$$(5.3.4)$$
 X is regular in the strong sense.

We also have that

(5.3.5) X is a Urysohn space

in this situation. If X is normal in the strong sense, then

$$(5.3.6)$$
 X is completely regular in the strong sense,

by Urysohn's lemma.

If X is regular in the strict sense, and X is normal in the strict sense, then

$$(5.3.7)$$
 X is completely regular in the strict sense.

To see this, let $x \in X$ and a closed set $E \subseteq X$ be given, with $x \notin E$. One can use regularity in the strict sense to get an open set $U \subseteq X$ such that $x \in U$ and $\overline{U} \subseteq X \setminus E$. In particular, E and \overline{U} are disjoint closed subsets of X, and $x \in \overline{U}$. Thus one can use Urysohn's lemma to get a suitable element of C(X).

5.3.2 Relation with induced topologies

Let X be a topological space again, and let Y be a subset of X, equipped with the induced topology. If f is a continuous real-valued function on X, then

$$(5.3.8)$$
 the restriction of f to Y is continuous.

If X is a Urysohn space, then it is easy to see that

$$(5.3.9)$$
 Y is a Urysohn space.

Similarly, if X is completely regular in the strict sense, then one can check that

(5.3.10) Y is completely regular in the strict sense

too, with respect to the induced topology. In particular, if X is normal in the strong sense, then X is completely regular in the strong sense, by Urysohn's lemma, and hence

(5.3.11) Y is completely regular in the strong sense.

5.3.3 More on completely regular spaces

Suppose that X is completely regular in the strict sense, and let $x \in X$ and a closed set $E \subseteq X$ be given, with $x \notin E$. Thus there is a Urysohn function on X for E and $\{x\}$, as before. One can use this to get

(5.3.12) a Urysohn function on X for E and a neighborhood of x in X,

as in the previous section.

Suppose now that $E\subseteq X$ is a closed set, $K\subseteq X$ is compact, and $E\cap K=\emptyset$. If $x\in K$, then there is a Urysohn function on X for E and a neighborhood of x in X, as in the preceding paragraph. Because K is compact, K can be covered by finitely many such neighborhoods of its elements. The maximum of the corresponding Urysohn functions for E and these finitely many neighborhoods of elements of K

(5.3.13) is a Urysohn function for E and K.

5.4 Local compactness and manifolds

A topological space X is said to be *locally compact* if for every $x \in X$ there is an open set $U \subseteq X$ and a compact set $K \subseteq X$ such that

$$(5.4.1) x \in U \text{ and } U \subseteq K.$$

If X is Hausdorff, then K is a closed set in X, and it follows that the closure \overline{U} of U in X is contained in K. This implies that

(5.4.2)
$$\overline{U}$$
 is compact in X ,

because \overline{U} is a closed set contained in a compact set. Sometimes local compactness is defined by asking that \overline{U} be compact.

Note that \mathbb{R}^n is locally compact with respect to the standard topology for each positive integer n.

5.4.1 Locally compact Hausdorff spaces

Suppose that X is a locally compact Hausdorff space. It is not too difficult to show that

(5.4.3) X is regular as a topological space.

More precisely, one can show that

(5.4.4) X is completely regular.

This can be reduced to Urysohn's lemma, or obtained using similar arguments.

5.4.2 Locally Euclidean of dimension n

Let X be a topological space, and let n be a positive integer. We say that X is locally Euclidean of dimension n if for every $x \in X$ there is an open set $U \subseteq X$ such that $x \in U$ and

(5.4.5) U is homeomorphic to an open subset W of \mathbb{R}^n .

More precisely, this uses the standard topology on \mathbb{R}^n , and the appropriate induced topologies on U and W.

Suppose that X is locally Euclidean of dimension n, and let us check that

(5.4.6) X satisfies the first separation condition.

Let x and y be distinct elements of X, so that we would like to find an open subset of X that contains x and not y. By hypothesis, there is an open set $U \subseteq X$ such that $x \in U$ and U is homeomorphic to an open subset of \mathbf{R}^n . If $y \notin U$, then we can take U to be the open set that we want. Otherwise, if $y \in U$, we can use the fact that \mathbf{R}^n satisfies the first separation condition to find an open set that contains x and not y.

Note that

(5.4.7) X satisfies the first countability condition,

because of the analogous property of \mathbf{R}^n .

Let $x \in X$ be given again, and let $U \subseteq X$ be an open set such that $x \in U$ and U is homeomorphic to an open subset W of \mathbf{R}^n . If $K \subseteq \mathbf{R}^n$ is compact and $K \subseteq W$, then

(5.4.8) K is compact as a subset of W,

with respect to the induced topology. This means that

(5.4.9) K corresponds to a compact subset of U,

with respect to the induced topology. It follows that

(5.4.10) the subset of U corresponding to K is compact as a subset of X

as well. In particular, one can use this to check that X is locally compact.

However, X may not be Hausdorff, and this is often included as an additional condition. In this case, one can use the regularity of \mathbb{R}^n to get that X is regular, a bit more directly than for arbitrary locally compact Hausdorff spaces. Similarly, complete regularity of X can be obtained more directly from the complete regularity of \mathbb{R}^n .

5.4.3 Topological manifolds

An n-dimensional topological manifold is often defined as

(5.4.11) a Hausdorff topological space X that is locally Euclidean of dimension n and satisfies the second countability condition.

There is also a notion of an n-dimensional topological manifold with boundary, although we shall not get into the details of this here. A basic example is a closed ball of positive radius in \mathbf{R}^n , with respect to the standard Euclidean metric, equipped with the induced topology. Another example is the set of $x \in \mathbf{R}^n$ with first coordinate $x_1 \geq 0$, equipped with the topology induced by the standard topology on \mathbf{R}^n .

5.4.4 Baire's theorem

Let X be any topological space again. If U_1, \ldots, U_n are finitely many dense open subsets of X, then one can check that their intersection $\bigcap_{j=1}^n U_j$ is a dense open set in X too. If U_1, U_2, U_3, \ldots is an infinite sequence of dense open sets in X, then a famous theorem of Baire gives conditions under which

(5.4.12)
$$\bigcap_{j=1}^{\infty} U_j \text{ is dense in } X.$$

One condition for this to hold is that there be a metric $d(\cdot, \cdot)$ on X such that X is equipped with the topology determined by $d(\cdot, \cdot)$, and that

(5.4.13) X is complete with respect to
$$d(\cdot, \cdot)$$
,

which is to say that every Cauchy sequence in X with respect to $d(\cdot, \cdot)$ converges to an element of X. Another condition for this to hold is that X be a locally compact Hausdorff space.

5.5 σ -Compactness

Let X be a topological space. It is easy to see that

(5.5.1) the union of finitely many compact subsets of X is compact as well.

Similarly,

(5.5.2) the union of finitely or countably many subsets of X with the Lindelöf property has the Lindelöf property too.

A subset E of X is said to be σ -compact if there is a sequence K_1, K_2, K_3, \ldots of compact subsets of X such that

$$(5.5.3) E = \bigcup_{j=1}^{\infty} K_j.$$

This implies that E has the Lindelöf property, as before.

5.5.1 Some criteria for σ -compactness

Suppose that X is locally compact, so that X can be covered by open sets that are contained in compact sets. If X also has the Lindelöf property, then it follows that X can be covered by finitely or countably many open sets, each of which is contained in a compact set. In particular, this means that

(5.5.4)
$$X$$
 is σ -compact.

Remember that X has the Lindelöf property when X satisfies the second countability condition, by Lindelöf's theorem. It follows that

(5.5.5) topological manifolds are
$$\sigma$$
-compact,

for instance.

5.5.2 Some bases from open coverings

Let $\{U_{\alpha}\}_{{\alpha}\in A}$ be an open covering of X, and suppose that for each ${\alpha}\in A$, \mathcal{B}_{α} is a base for the topology induced on U_{α} by the topology on X. Note that the elements of \mathcal{B}_{α} are open subsets of X for each ${\alpha}\in A$, because the U_{α} 's are open subsets of X. Thus

$$(5.5.6) \mathcal{B} = \bigcup_{\alpha \in A} \mathcal{B}_{\alpha}$$

is a collection of open subsets of X. Under these conditions, one can check that

(5.5.7)
$$\mathcal{B}$$
 is a base for the topology of X .

Of course, if A has only finitely or countably many elements, and if \mathcal{B}_{α} has only finitely or countably many elements for each $\alpha \in A$, then

(5.5.8) \mathcal{B} has only finitely or countably many elements.

If X has the Lindelöf property, then one can automatically reduce to the case where A has only finitely or countably many elements.

Suppose that X is locally Euclidean of dimension n for some positive integer n. Thus X can be covered by open sets that are homeomorphic to open subsets of \mathbf{R}^n . If X has the Lindelöf property, then it follows that X can be covered by finitely or countably many open sets that are homeomorphic to open subsets of \mathbf{R}^n . Each of these open sets has a countable base for its topology, because of the analogous property for open subsets of \mathbf{R}^n . This leads to

$$(5.5.9)$$
 a countable base for the topology of X ,

as in the preceding paragraph.

5.5.3 An example in R

One can show that the set $\mathbf{R} \setminus \mathbf{Q}$ of irrational numbers is not σ -compact with respect to the standard topology on \mathbf{R} , using the Baire category theorem. Note that every subset of the real line has the Lindelöf property, by Lindelöf's theorem.

5.6 Quotient spaces

Let X be a set. A binary relation \sim on X is said to be an equivalence relation on X if it is reflexive, symmetric, and transitive on X, as usual. In this case, if $x \in X$, then

$$[x] = \{ y \in X : x \sim y \}$$

is called an *equivalence class* in X. This is a subset of X that contains x, by reflexivity. If $x, w \in X$, then [x] = [w] if and only if $x \sim w$, and otherwise [x] and [w] are disjoint.

Conversely, let \mathcal{P} be a partition of X, which is to say a collection of pairwise-disjoint nonempty subsets of X whose union is equal to X. If $x, y \in X$, then put

(5.6.2) $x \sim_{\mathcal{P}} y$ when x and y lie in the same element of \mathcal{P} .

This defines an equivalence relation on X, for which the corresponding equivalence classes are the elements of \mathcal{P} .

If \sim is any equivalence relation on X, then let

$$(5.6.3) X/\sim$$

be the corresponding quotient set, which is the set of equivalence classes in X. In this situation, there is a natural quotient mapping from X onto X/\sim , which sends $x\in X$ to $[x]\in X/\sim$.

Let Y be another set, and let f be a mapping from X onto Y. If $x, w \in X$, then put

(5.6.4)
$$x \sim_f w \text{ when } f(x) = f(w).$$

This defines an equivalence relation on X, for which the corresponding equivalence classes are the sets of the form $f^{-1}(\{y\})$, $y \in Y$.

5.6.1 The quotient or identification topology

Suppose now that X is a topological space. Let us say that

$$(5.6.5) V \subseteq Y \text{ is an open set}$$

exactly when

$$(5.6.6)$$
 $f^{-1}(V)$ is an open set in X .

One can check that this defines a topology on Y, which is called the *quotient* topology or identification topology associated to f. Equivalently, this means that

(5.6.7)
$$E \subseteq Y$$
 is a closed set if and only if $f^{-1}(E)$ is a closed set in X .

Note that f is automatically continuous with respect to the quotient topology on Y.

Remember that Y satisfies the first separation condition if and only if for each $y \in Y$, $\{y\}$ is a closed set in Y. Thus Y satisfies the first separation condition with respect to the quotient topology associated to f if and only if

(5.6.8) for each
$$y \in Y$$
, $f^{-1}(\{y\})$ is a closed set in X .

Let Z be another topological space, and let g be a mapping from Y into Z. One can check that g is continuous with respect to the quotient topology on Y associated to f if and only if $g \circ f$ is continuous as a mapping from X into Z.

5.7 Local connectedness

Let X be a topological space, and let x be an element of X. Let us say that X is *locally connected at* x if for every open subset W of X that contains x, there is an open subset U_0 of X such that $x \in U_0$, $U_0 \subseteq W$, and

(5.7.1) every element of U_0 can be connected to x in W.

More precisely, this means that

(5.7.2) for every
$$y \in U_0$$
 there is a connected set $E(x,y) \subseteq X$ such that $x,y \in E(x,y)$ and $E(x,y) \subseteq W$.

If there is a connected open subset U of X such that $x \in U$ and $U \subseteq W$, then we can take $U_0 = U$.

Alternatively, let W be an open subset of X that contains x, and let U_1 be

(5.7.3) the union of all the connected subsets of W that contain x.

Thus $x \in U_1$, because $\{x\}$ is a connected set that contains $x, U_1 \subseteq W$ by construction, and U_1 is connected, by Proposition 1.15.3. In this situation, local connectedness of X at x says exactly that

(5.7.4) x is an element of the interior of U_1 .

If X is locally connected at every point in X, then X is said to be *locally connected* as a topological space. Let W be an open subset of X that contains x again, and let U_1 be as in the preceding paragraph. If X is locally connected, then one can check that

$$(5.7.5)$$
 U_1 is an open set in X .

5.7.1 Local path connectedness

Similarly, X is said to be *locally path connected* at x if for every open subset W of X that contains x, there is an open subset V_0 of X such that $x \in V_0$, $V_0 \subseteq W$, and

(5.7.6) every element of
$$V_0$$
 can be connected to x by a continuous path in W .

If there is a path-connected open subset V of X such that $x \in V$ and $V \subseteq W$, then we can take $V_0 = V$. Note that local path connectedness at x implies local connectedness at x.

Let W be an open subset of X that contains x, and let V_1 be the set of points in W that can be connected to x by a continuous path in W. Note that V_1 contains the images of these paths, so that V_1 is a path-connected subset of W. Local path connectedness of X at x says exactly that

$$(5.7.7)$$
 x is in the interior of V_1 ,

as before.

If X is locally path connected at every point, then X is said to be *locally path connected* as a topological space. Thus local path connectedness implies local connectedness. Let W be an open subset of X that contains x, and let V_1 be as in the previous paragraph. If X is locally path connected, then one can verify that

$$(5.7.8)$$
 V_1 is an open set in X .

5.7.2 Some examples and basic properties

It is easy to see that

(5.7.9)
$$\mathbf{R}^n$$
 is locally path connected

for every positive integer n, with respect to the standard topology. Similarly, locally Euclidean spaces of dimension n are locally path connected.

If X_0 is an open subset of X, and if X is locally connected or path connected, then X_0 has the same property, with respect to the induced topology.

If X is connected and locally path connected, then one can show that X is path connected. In particular, connected open subsets of \mathbf{R}^n are path connected.

5.8 Connected and path-connected components

Let X be a topological space, and consider the binary relation \sim_c defined on X by putting $x \sim_c y$ when $x, y \in X$ and there is a connected subset E(x, y) of X such that

$$(5.8.1) x, y \in E(x, y).$$

One can check that this defines an equivalence relation on X, using Proposition 1.15.3 to get that \sim_c is transitive.

One can also verify that

(5.8.2) the equivalence classes in X corresponding to \sim_c are connected subsets of X,

using Proposition 1.15.3 or 1.15.6. These equivalence classes are called the *connected components* of X. It is easy to see that the connected components of X are the maximal connected subsets of X, with respect to inclusion.

Observe that

(5.8.3) the connected components of X are closed sets in X,

by Proposition 1.15.1. If X is locally connected, then

(5.8.4) the connected components of X are open sets in X.

5.8.1 Another equivalence relation on X

Similarly, consider the binary relation \sim_p defined on X by putting $x \sim_p y$ when $x, y \in X$ are connected by a continuous path in X. This means that there are real numbers a, b with a < b and a continuous mapping p from [a, b] into X such that

(5.8.5)
$$p(a) = x, \ p(b) = y.$$

This uses the topology induced on [a, b] by the standard topology on **R**. Note that one can always reduce to the case where a = 0 and b = 1, by reparameterizing the path.

It is not too difficult to show that \sim_p defines an equivalence relation on X. The equivalence classes in X corresponding to \sim_p are path-connected subsets of X, by construction. These equivalence classes are called the path-connected components of X. As before, these are the maximal path-connected subsets of X, with respect to inclusion.

If
$$x, y \in X$$
, then

$$(5.8.6) x \sim_p y \text{ implies } x \sim_c y.$$

Indeed, if $a, b \in \mathbf{R}$ and a < b, then [a, b] is connected as a subset of \mathbf{R} with respect to the standard topology. This implies that [a, b] is connected as a subset of itself, with respect to the induced topology. If p is a continuous mapping from [a, b] into X, then it follows that p([a, b]) is a connected set in X.

If X is locally path connected, then

(5.8.7) the path-connected components of X are open subsets of X.

If X is also connected, then one can check that X is path connected.

5.9 One-point compactifications

Let X be a Hausdorff topological space that is not compact. The *one-point* compactification of X can be defined initially as a set by

$$(5.9.1) X^* = X \cup \{p_*\},$$

where $p_* \notin X$. One often refers to p_* as the *point at infinity* in X^* . Let us say that $U \subseteq X^*$ is an open set if either

$$(5.9.2)$$
 U is an open subset of X,

or if

(5.9.3)
$$p_* \in U$$
 and $X^* \setminus U$ is a compact subset of X .

Remember that compact subsets of X are closed sets in X, because X is Hausdorff, by hypothesis. If $U \subseteq X^*$ is an open set and $p_* \in U$, then it follows that $X^* \setminus U$ is a closed set in X. Note that

$$(5.9.4) X \setminus (U \cap X) = X \setminus U = X^* \setminus U,$$

because $p_* \in U$. Thus

$$(5.9.5)$$
 $U \cap X$ is an open set in X

in this case.

5.9.1 Checking that this defines a topology on X^*

Of course, the empty set is an open subset of X^* , because it is an open subset of X. We also have that X^* is an open set in itself, because it contains p_* and its complement in itself is the empty set, which is a compact subset of X.

If U_1, \ldots, U_n are finitely many open sets in X^* , then we would like to check that their intersection is an open set in X^* as well. If $U_l \subseteq X$ for some l, then

$$(5.9.6) \qquad \bigcap_{j=1}^{n} U_j \subseteq X,$$

and it suffices to verify that $\bigcap_{j=1}^n U_j$ is an open set in X. This follows from the fact that $U_i \cap X$ is an open set in X for each $j = 1, \ldots, n$, so that

(5.9.7)
$$\bigcap_{j=1}^{n} U_{j} = \bigcap_{j=1}^{n} (U_{j} \cap X)$$

is an open set in X. Otherwise, if $p_* \in U_j$ for each j = 1, ..., n, then we should verify that

(5.9.8)
$$X^* \setminus \left(\bigcap_{j=1}^n U_j\right) = \bigcup_{j=1}^n (X^* \setminus U_j)$$

is a compact subset of X. In fact, it is easy to see that the union of finitely many compact subsets of any topological space is compact, as mentioned in Section 5.5.

Let $\{U_{\alpha}\}_{\alpha\in A}$ be a family of open subsets of X^* , and let us check that their union is an open set in X^* . If $U_{\alpha}\subseteq X$ for every $\alpha\in A$, then U_{α} is an open set in X for each $\alpha\in A$. This implies that $\bigcup_{\alpha\in A}U_{\alpha}$ is an open set in X, and hence in X^* . Otherwise, suppose that $p_*\in U_{\alpha_0}$ for some $\alpha_0\in A$, so that $p_*\in \bigcup_{\alpha\in A}U_{\alpha}$. In this case, we would like to verify that

(5.9.9)
$$X^* \setminus \left(\bigcup_{\alpha \in A} U_{\alpha}\right)$$
 is a compact subset of X .

Observe that

(5.9.10)
$$X^* \setminus \left(\bigcup_{\alpha \in A} U_{\alpha}\right) = X \setminus \left(\bigcup_{\alpha \in A} U_{\alpha}\right) = \bigcap_{\alpha \in A} (X \setminus U_{\alpha}).$$

This is a closed set in X, because $X \setminus U_{\alpha}$ is a closed set in X for every $\alpha \in A$. We also have that $X \setminus U_{\alpha_0} = X^* \setminus U_{\alpha_0}$ is compact in X, because $p_* \in U_{\alpha_0}$. It follows that (5.9.9) holds, because it is a closed set that is contained in the compact set $X \setminus U_{\alpha_0}$.

This shows that the collection of open subsets of X^* defined earlier is a topology on X^* . By construction, X is an open subset of X^* , and the topology induced on X by this topology on X^* is the same as the given topology on X. Note that

(5.9.11)
$$p_*$$
 is a limit point of X in X^* ,

because X is not compact, by hypothesis.

5.9.2 Compactness of X^*

Let us check that

$$(5.9.12) X^* ext{ is compact}$$

with respect to this topology. Let $\{U_{\alpha}\}_{{\alpha}\in A}$ be an arbitrary open covering of X^* in itself. In particular, there is an $\alpha_0\in A$ such that $p_*\in U_{\alpha_0}$. Thus

(5.9.13)
$$X \setminus U_{\alpha_0} = X^* \setminus U_{\alpha_0}$$
 is a compact subset of X .

We may consider $\{U_{\alpha} \cap X\}_{\alpha \in A}$ as an open covering of $X \setminus U_{\alpha_0}$ in X, because $U_{\alpha} \cap X$ is an open set in X for each $\alpha \in A$, and the union of these open sets is X. Because $X \setminus U_{\alpha_0}$ is compact in X, there are finitely many indices $\alpha_1, \ldots, \alpha_n \in A$ such that

(5.9.14)
$$X^* \setminus U_{\alpha_0} = X \setminus U_{\alpha_0} \subseteq \bigcup_{j=1}^n (U_{\alpha_j} \cap X).$$

This implies that

$$(5.9.15) X^* \subseteq \bigcup_{j=0}^n U_{\alpha_j},$$

as desired.

5.9.3 When X^* is Hausdorff

Observe that X^* is Hausdorff if and only if for every $x \in X$ there are disjoint open subsets U and V of X^* such that $x \in U$ and $p_* \in V$, because X is Hausdorff by hypothesis. One can verify that this happens exactly when

$$(5.9.16)$$
 X is locally compact.

The one-point compactification of a topological space X can also be defined a bit more carefully when X is not necessarily Hausdorff.

5.10 Homotopic mappings

Let X and Y be topological spaces, and let f, g be continuous mappings from X into Y. A homotopy between f and g is a mapping H from $X \times [0,1]$ into Y that satisfies the following two conditions. First, H should be continuous with respect to the product topology on $X \times [0,1]$. Of course, this uses the topology induced on [0,1] by the standard topology on \mathbf{R} . Second,

$$(5.10.1) H(x,0) = f(x),$$

$$(5.10.2) H(x,1) = g(x)$$

for every $x \in X$. In this case, one may say that f is homotopic to g. It is well known and not too difficult to show that this defines an equivalence relation on the space C(X,Y) of continuous mappings from X into Y.

5.10.1 Additional conditions on f, g

In many cases, one may wish to include additional conditions on f and g, and the same conditions on H(x,t) for each $t \in [0,1]$. Suppose for instance that X = [0,1], and let $p,q \in Y$ be given. Let f,g be continuous mappings from [0,1] into Y such that

(5.10.3)
$$f(0) = g(0) = p, \ f(1) = g(1) = q.$$

If a homotopy H(x,t) between f and g also satisfies

(5.10.4)
$$H(0,t) = p, \ H(1,t) = q$$

for every $t \in [0,1]$, then one says that f and g are homotopic with endpoints fixed. One can show that this defines an equivalence relation on the space of continuous paths in Y from p to q.

5.10.2 Convex sets Y in \mathbb{R}^n

Suppose now that Y is a convex subset of \mathbb{R}^n for some positive integer n. This means that for every $y, z \in Y$, the line segment in \mathbb{R}^n connecting y and z is contained in Y too. We consider Y as a topological space, using the topology induced by the standard topology on \mathbb{R}^n . Let X be any topological space again, and let f, g be continuous mappings from X into Y. One can check that

$$(5.10.5) H(x,t) = (1-t) f(x) + t g(x)$$

defines a homotopy between f and g, as continuous mappings from X into Y.

Let $p, q \in Y$ be given again, and suppose that f, g are continuous mappings from [0,1] into Y that satisfy (5.10.3). Note that (5.10.5) satisfies (5.10.4) in this case.

Let Z be another topological space, and let ϕ be a continuous mapping from Y into Z. If f, g are homotopic continuous mappings from X into Y, with homotopy H, then $\phi \circ H$ is a homotopy between $\phi \circ f$ and $\phi \circ g$, as continuous mappings from X into Z. There is an analogous statement for homotopic paths with endpoints fixed, as usual.

5.11 Countable unions and intersections

A subset E of a topological space X is said to be an F_{σ} set if

(5.11.1) E can be expressed as the union of a sequence of closed subsets of X.

Similarly, a subset A of X is said to be a G_{δ} set if

(5.11.2) A can be expressed as the intersection of a sequence of open subsets of X.

It is easy to see that

(5.11.3) $E \subseteq X$ is an F_{σ} set if and only if $X \setminus E$ is a G_{δ} set.

Of course, closed sets are F_{σ} sets, and open sets are G_{δ} sets.

Suppose for the moment that X satisfies the first separation condition. Let x be an element of X, and let $\mathcal{B}(x)$ be a local base for the topology of X at x. Under these conditions, the intersection of the elements of $\mathcal{B}(x)$ is equal to $\{x\}$. If $\mathcal{B}(x)$ has only finitely many elements, then $\{x\}$ is an open set in X. If $\mathcal{B}(x)$ is countable, then $\{x\}$ is a G_{δ} set in X.

5.11.1 Some subsets of semimetric spaces

Suppose now that d(x,y) is a semimetric on X, and that X is equipped with the topology determined by $d(\cdot,\cdot)$. If A is any subset of X and r is a positive

real number, then put

$$(5.11.4) A_r = \bigcup_{x \in A} B(x, r).$$

This is an open subset of X that contains A. One can check that

$$(5.11.5) \overline{A} = \bigcap_{j=1}^{\infty} A_{1/j}.$$

This implies that every closed set in X is a G_{δ} set, and hence that every open set in X is an F_{σ} set.

5.11.2 Some remarks about inverse images

Let X and Y be arbitrary topological spaces, and let f be a continuous mapping from X into Y. If $E \subseteq Y$ is an F_{σ} set, then

(5.11.6)
$$f^{-1}(E) \text{ is an } F_{\sigma} \text{ set in } X.$$

Similarly, if $A \subseteq Y$ is a G_{δ} set, then

(5.11.7)
$$f^{-1}(A) \text{ is a } G_{\delta} \text{ set in } X.$$

In particular, suppose that f is a continuous real-valued function on X, with respect to the standard topology on \mathbf{R} . If $t \in \mathbf{R}$, then $\{t\}$ is a G_{δ} set in \mathbf{R} , so that

(5.11.8)
$$f^{-1}(\{t\})$$
 is a G_{δ} set in X .

5.11.3 Perfectly normal spaces

Let us say that X is perfectly normal in the strict sense if X is normal in the strict sense, and every closed set in X is a G_{δ} set. If X also satisfies the first separation condition, then we say that X is perfectly normal in the strong sense. One may say that X is perfectly normal when X is perfectly normal in the strict sense, and perfectly T_4 when X is perfectly normal in the strong sense, but the opposite convention is used sometimes too.

5.11.4 Nowhere dense sets and category

A subset E of X is said to be *nowhere dense* if the interior of the closure \overline{E} of E in X is empty. Equivalently, this means that $X \setminus \overline{E}$ is dense in X. This is the same as saying that the interior of $X \setminus E$ is dense in X. Remember that the intersection of finitely many dense open subsets of X is a dense open set in X, as in Subsection 5.4.4. One can use this to check that the union of finitely many nowhere dense subsets of X is nowhere dense in X.

A subset A of X is said to be of the first category or meager in X if A can be expressed as the union of a sequence of nowhere dense sets. Otherwise, A is said to be of the second category or non-meager in X. The theorem of Baire

mentioned in Subsection 5.4.4 is equivalent to saying that if A is of the first category in X, then

(5.11.9)
$$X \setminus A$$
 is dense in X ,

under the same conditions as before. This is the same as saying that the interior of A in X is empty. In particular, it follows that X is of second category in this case.

5.12 Distance functions

Let X be a set with a semimetric d(x,y), and let A be a nonempty subset of X. If $x \in X$, then the distance from x to A with respect to $d(\cdot,\cdot)$ is defined by

(5.12.1)
$$\operatorname{dist}(x, A) = \inf\{d(x, a) : a \in A\},\$$

which is a nonnegative real number. One can check that $\operatorname{dist}(x, A) = 0$ if and only if x is an element of the closure \overline{A} of A with respect to the topology determined on X by $d(\cdot, \cdot)$.

5.12.1 Continuity of distance functions

Observe that

(5.12.2)
$$\operatorname{dist}(x, A) < d(x, a) < d(x, y) + d(y, a)$$

for every $x, y \in X$ and $a \in A$. Using this, one can check that

$$(5.12.3) dist(x, A) \le d(x, y) + dist(y, A)$$

for every $x, y \in X$. One can use this to verify that dist(x, A) is continuous as a real-valued function on X, with respect to the standard topology on \mathbf{R} .

5.12.2 Separated sets and open sets

Suppose that A and B are nonempty separated subsets of X. Put

(5.12.4)
$$U = \{x \in X : dist(x, A) < dist(x, B)\}\$$

and

$$(5.12.5) V = \{x \in X : dist(x, B) < dist(x, A)\}.$$

It is easy to see that these are open subsets of X. We also have that $A \subseteq U$, $B \subseteq V$, and $U \cap V = \emptyset$. This is another way to look at the complete normality of X in the strict sense, with respect to the topology determined by $d(\cdot, \cdot)$.

5.12.3 Urysohn's lemma for semimetric spaces

Suppose now that A and B are disjoint nonempty closed subsets of X. Note that

(5.12.6)
$$dist(x, A) + dist(x, B) > 0$$

for every $x \in X$. Thus

(5.12.7)
$$\frac{\operatorname{dist}(x,A)}{\operatorname{dist}(x,A) + \operatorname{dist}(x,B)}$$

defines a nonnegative real-valued function on X, which is less than or equal to 1 for every $x \in X$. This function is continuous on X, equal to 0 on A, and equal to 1 on B. Of course, this corresponds to Urysohn's lemma in this case.

5.13 Local finiteness

Let X be a topological space, let A be a nonempty set, and let E_{α} be a subset of X for every $\alpha \in A$. Let us say that $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite at a point $x\in X$ if there is an open subset U of X such that $x\in U$ and

$$(5.13.1) E_{\alpha} \cap U = \emptyset$$

for all but finitely many $\alpha \in A$. If $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite at every $x\in X$, then $\{E_{\alpha}\}_{{\alpha}\in A}$ is said to be *locally finite in* X.

5.13.1 Local finiteness and adherent points

Of course, if $x \in X$ is adherent to E_{β} for some $\beta \in A$, then x is adherent to $\bigcup_{\alpha \in A} E_{\alpha}$.

Let $x \in X$ be given, and suppose that for each $\alpha \in A$, x is not adherent to E_{α} . Suppose also that $\{E_{\alpha}\}_{\alpha \in A}$ is locally finite at x, so that there is an open set $U \subseteq X$ such that $x \in U$ and (5.13.1) holds for all but finitely many $\alpha \in A$. Let $\alpha_1, \ldots, \alpha_n$ be a list of the finitely many $\alpha \in A$ such that E_{α} intersects U, if there are any. By hypothesis, for each $j = 1, \ldots, n$, x is not adherent to E_{α_j} , and so there is an open subset V_j of X such that $x \in V_j$ and $E_{\alpha_j} \cap V_j = \emptyset$.

It follows that

(5.13.2)
$$W = U \cap \left(\bigcap_{j=1}^{n} V_{j}\right)$$

is an open subset of X such that $x \in W$ and

$$(5.13.3) W \cap \left(\bigcup_{\alpha \in A} E_{\alpha}\right) = \emptyset.$$

This means that x is not adherent to $\bigcup_{\alpha \in A} E_{\alpha}$. Equivalently, if $x \in X$ is adherent to $\bigcup_{\alpha \in A} E_{\alpha}$, and if $\{E_{\alpha}\}_{\alpha \in A}$ is locally finite at x, then x is adherent to E_{α} for some $\alpha \in A$.

5.13.2 Local finiteness and closures

If $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite in X, then it follows that

(5.13.4)
$$\overline{\left(\bigcup_{\alpha \in A} E_{\alpha}\right)} = \bigcup_{\alpha \in A} \overline{E_{\alpha}}.$$

Of course, the right side is automatically contained in the left side, as before. If $\alpha \in A$ satisfies (5.13.1) for some open set $U \subseteq X$, then

$$(5.13.5) \overline{E_{\alpha}} \cap U = \emptyset.$$

If $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite at x, then we get that $\{\overline{E_{\alpha}}\}_{{\alpha}\in A}$ is locally finite at x too. If $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite in X, then $\{\overline{E_{\alpha}}\}_{{\alpha}\in A}$ is locally finite in X as well.

5.13.3 Local finiteness and induced topologies

Let X_0 be a subset of X, equipped with the induced topology. If $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite at $x\in X_0$, then $\{E_{\alpha}\cap X_0\}_{{\alpha}\in A}$ is locally finite at x, as a family of subsets of X_0 . Similarly, if $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite in X, then $\{E_{\alpha}\cap X_0\}_{{\alpha}\in A}$ is locally finite in X_0 .

Suppose for the moment that $\{E_{\alpha}\}_{{\alpha}\in A}$ is a family of subsets of X_0 . If $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite at $x\in X_0$, as a family of subsets of X_0 , then it is easy to see that $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite at x, as a family of subsets of X.

5.13.4 Local finiteness and compactness

Let $\{E_{\alpha}\}_{{\alpha}\in A}$ be a family of subsets of X again, and let K be a subset of X. Suppose that $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite at every $x\in K$. Thus, for each $x\in K$, there is an open subset U(x) of X such that $x\in U(x)$ and $E_{\alpha}\cap U(x)=\emptyset$ for all but finitely many $\alpha\in A$. If K is compact, then there are finitely many elements x_1,\ldots,x_n of K such that

(5.13.6)
$$K \subseteq \bigcup_{j=1}^{n} U(x_j).$$

We also have that

(5.13.7)
$$E_{\alpha} \cap \left(\bigcup_{j=1}^{n} U(x_{j})\right) = \emptyset$$

for all but finitely many $\alpha \in A$ in this situation.

In particular, suppose that $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite in X, and that X is compact. Under these conditions, we get that $E_{\alpha}=\emptyset$ for all but finitely many $\alpha\in A$.

5.14 Sums of real-valued functions

Let X be a set, let I be a nonempty set, and let ϕ_j be a real-valued function on X for each $j \in I$. If $x \in X$, then put

$$(5.14.1) I(x) = \{ j \in I : \phi_j(x) \neq 0 \}.$$

If I(x) has only finitely many elements, then the sum

(5.14.2)
$$\Phi(x) = \sum_{j \in I} \phi_j(x)$$

can be defined as a real number, by reducing to a finite sum. More precisely, (5.14.2) is the same as the sum over any nonempty finite subset of I that contains I(x).

Put

$$(5.14.3) E_i = \{x \in X : \phi_i(x) \neq 0\}$$

for each $j \in I$. If $x \in X$, then I(x) is the same as the set of $j \in I$ such that $x \in E_j$. If U_0 is a subset of X, then put

(5.14.4)
$$I(U_0) = \bigcup_{x \in U_0} I(x) = \{ j \in I : \phi_j(x) \neq 0 \text{ for some } x \in U_0 \}.$$

This is the same as the set of $j \in I$ such that $E_j \cap U_0 \neq \emptyset$.

5.14.1 Local finiteness and continuity

Suppose now that X is a topological space. Observe that $\{E_j\}_{j\in I}$ is locally finite at a point $x_0 \in X$ if and only if there is an open set $U_0 \subseteq X$ such that $x_0 \in U_0$ and $I(U_0)$ has only finitely many elements. In particular, this implies that for each $x \in U_0$, I(x) has only finitely many elements.

Suppose that ϕ_j is continuous on X for each $j \in I$, with respect to the standard topology on \mathbf{R} . If $\{E_j\}_{j\in I}$ is locally finite in X, then (5.14.2) defines a real-valued function on X. It is easy to see that this function is continuous at every point in X, under these conditions.

5.14.2 Some partitions of unity

Of course, if $\phi_j(x) \geq 0$ for every $j \in I$ and $x \in X$, then $\Phi(x) \geq 0$ for every $x \in X$. If we also have that for each $x \in X$ there is a $j \in I$ such that $\phi_j(x) > 0$, then we get that $\Phi(x) > 0$ for every $x \in X$. Note that $1/\Phi(x)$ is continuous on X in this case. Put

$$(5.14.5) \psi_l(x) = \phi_l(x)/\Phi(x)$$

for every $l \in I$ and $x \in X$, which defines a continuous real-valued function on X. By construction, $\psi_l(x) \ge 0$ for every $l \in I$ and $x \in X$, and

$$\{x \in X : \psi_l(x) > 0\} = E_l.$$

Thus, for each $x \in X$, $\psi_l(x) > 0$ for only finitely many $l \in I$, and

(5.14.7)
$$\sum_{l \in I} \psi_l(x) = \left(\sum_{l \in I} \phi_l(x) / \Phi(x) \right) = \Phi(x) / \Phi(x) = 1.$$

The family of functions ψ_l , $l \in I$, is said to be a partition of unity on X.

5.15 Local compactness and σ -compactness

Let X be a locally compact topological space. If K is a compact subset of X, then there is an open set $U\subseteq X$ such that $K\subseteq U$ and U is contained in a compact subset of X. Indeed, every element of K is contained in an open set that is contained in a compact subset of X, because X is locally compact. It follows that K is contained in the union of finitely many of these open sets, because K is compact. By construction, the union of these finitely many open sets is contained in the union of finitely many compact subsets of X, which is compact as well.

If X is Hausdorff, then every element of X is contained in an open set whose closure in X is compact, as in Section 5.4. Similarly, if $K \subseteq X$ is compact, then there is an open set $U \subseteq X$ such that $K \subseteq U$ and \overline{U} is compact.

5.15.1 Some sequences of compact and open sets

Suppose now that X is σ -compact, so that there is a sequence K_1, K_2, K_3, \ldots of compact subsets of X with

(5.15.1)
$$X = \bigcup_{j=1}^{\infty} K_j.$$

If X is locally compact too, then there is an open set $U_1 \subseteq X$ and a compact set $E_1 \subseteq X$ such that $K_1 \subseteq U_1$ and $U_1 \subseteq E_1$, as before.

Similarly, for each $j \geq 1$, we can choose an open set $U_j \subseteq X$ and a compact set $E_j \subseteq X$ such that

$$(5.15.2) \qquad \qquad \bigcup_{l=1}^{j} K_{l} \subseteq U_{j} \subseteq E_{j},$$

and so that

$$(5.15.3) E_{i-1} \subseteq U_i$$

when $j \geq 2$. More precisely, suppose that U_j and E_j have been chosen in this way for some j, and let us see how we can choose U_{j+1} and E_{j+1} .

Note that $E_j \cup K_{j+1}$ is compact in X, because E_j and K_{j+1} are compact. Thus there is an open set $U_{j+1} \subseteq X$ such that $E_j \cup K_{j+1} \subseteq U_{j+1}$ and U_{j+1} is contained in a compact set $E_{j+1} \subseteq X$, as before. It follows that $\bigcup_{l=1}^{j+1} K_l \subseteq U_{j+1}$, because of (5.15.2). If X is Hausdorff, then we can take $E_j = \overline{U_j}$ for each j.

5.15.2 A locally finite sequence of compact sets

Of course,

(5.15.4)
$$X = \bigcup_{j=1}^{\infty} U_j = \bigcup_{j=1}^{\infty} E_j,$$

by (5.15.1) and (5.15.2). Put

(5.15.5)
$$A_1 = E_1 \text{ and } A_j = E_j \setminus U_{j-1} \text{ when } j \ge 2.$$

It is easy to see that A_j is compact in X for every j, because $A_j = E_j \cap (X \setminus U_{j-1})$ is the intersection of a compact set and a closed set when $j \geq 2$. One can check that

$$(5.15.6) X = \bigcup_{j=1}^{\infty} A_j,$$

using (5.15.4).

More precisely, if $x \in X$ and j_0 is the smallest positive integer such that $x \in E_{j_0}$, then $x \in A_{j_0}$. In this case, $x \notin A_j$ when $j < j_0$, and when $j > j_0 + 1$. By construction,

$$(5.15.7) A_i \cap U_l = \emptyset$$

when j > l, which implies that the family of A_j 's is locally finite in X.

5.15.3 A locally finite sequence of open sets

Suppose now that X is Hausdorff, so that we can take $E_j = \overline{U}_j$ for each j, as before. Clearly $A_j \subseteq U_{j+1}$ for every j, by (5.15.3), and in fact

$$(5.15.8) A_j \subseteq U_{j+1} \setminus \overline{U_{j-2}}$$

when $j \geq 3$. Put

$$(5.15.9) V_2 = U_3 \quad \text{and} \quad V_j = U_{j+1} \setminus \overline{U_{j-2}} \text{ when } j \ge 3,$$

so that V_j is an open subset of X for every $j \geq 2$. It is easy to see that

(5.15.10)
$$X = \bigcup_{j=2}^{\infty} V_j,$$

using (5.15.3) and (5.15.4).

Observe that

$$(5.15.11) V_i \cap V_l = \emptyset$$

when $j \geq l + 3$, and in particular that the family of V_j 's is locally finite in X.

Chapter 6

Some spaces of mappings

6.1 Bounded sets and mappings

Let Y be a set with a semimetric $d_Y(\cdot,\cdot)$. A subset A of Y is said to be bounded with respect to $d_Y(\cdot,\cdot)$ if A is contained in a ball in Y with respect to $d_Y(\cdot,\cdot)$. It is convenient to consider $A = \emptyset$ as a bounded subset of Y even when $Y = \emptyset$.

6.1.1 Countable compactness implies boundedness

If $A \subseteq Y$ is countably compact with respect to the topology determined on Y by $d_Y(\cdot,\cdot)$, then it is easy to see that A is bounded in Y. More precisely, this can be obtained by covering A by open balls in Y centered at a fixed point in Y and with positive integer radii.

6.1.2 Bounded mappings

A mapping f from a set X into Y is said to be bounded if f(X) is a bounded set in Y.

Suppose that X is a topological space, and that f is a continuous mapping from X into Y, with respect to the topology determined on Y by $d_Y(\cdot,\cdot)$. If $K \subseteq X$ is countably compact, then f(K) is countably compact in Y, and hence f(K) is bounded in Y. In particular, if X is countably compact, then f is bounded.

6.1.3 Some related real-valued functions

Let p be an element of Y, let f be a mapping from X into Y, and let E be a subset of X. One can check that f(E) is bounded in Y if and only if

$$(6.1.1) d_Y(p, f(x))$$

is bounded as a real-valued function on E.

If X is a topological space and f is continuous with respect to the topology determined on Y by $d_Y(\cdot, \cdot)$, then (6.1.1) is continuous as a real-valued function on X, with respect to the standard topology on **R**. Indeed, (6.1.1) is the same as the composition of f with the real-valued function on Y defined by $d_Y(p, \cdot)$, which is continuous with respect to the topology determined on Y by $d_Y(\cdot, \cdot)$.

6.1.4 Supremum semimetrics

Let X be a set, and let E be a nonempty subset of X. Consider the space $\mathcal{B}_E(X,Y)$ of mappings f from X into Y such that f(E) is a bounded set in Y. If $f,g \in \mathcal{B}_E(X,Y)$, then one can check that $d_Y(f(x),g(x))$ is bounded as a real-valued function on E. In this case, we put

(6.1.2)
$$\theta_E(f,g) = \sup\{d_Y(f(x),g(x)) : x \in E\}.$$

One can verify that this defines a semimetric on $\mathcal{B}_E(X,Y)$, which is called the supremum semimetric on $\mathcal{B}_E(X,Y)$ associated to $d_Y(\cdot,\cdot)$.

If E = X, then we may use $\mathcal{B}(X,Y)$ instead of $\mathcal{B}_X(X,Y)$, which is the space of all bounded mappings from X into Y. Similarly, we may use $\theta(f,g)$ instead of $\theta_X(f,g)$. Note that this is a metric on $\mathcal{B}(X,Y)$ when $d_Y(\cdot,\cdot)$ is a metric on Y.

6.1.5 Some supremum semimetrics on C(X,Y)

Let X be a topological space, and let C(X,Y) be the space of continuous mappings from X into Y, as before. Also let K be a nonempty countably compact subset of X, so that

$$(6.1.3) C(X,Y) \subset \mathcal{B}_K(X,Y),$$

as before. Thus the restriction of $\theta_K(f,g)$ to $f,g\in C(X,Y)$ defines a semimetric on C(X,Y).

6.1.6 Existence of the maximum

If $f,g\in C(X,Y)$, then one can check that

$$(6.1.4) d_Y(f(x), g(x))$$

is continuous as a real-valued function on X, with respect to the standard topology on \mathbf{R} . If K is a nonempty compact subset of X, then the maximum of (6.1.4) on K is attained, by the extreme value theorem.

This also works when K is countably compact, by analogous arguments. More precisely, one can use the fact that countably compact subsets of the real line are compact, because there is a countable base for the standard topology on \mathbf{R} . One can also show more directly that countably compact subsets of \mathbf{R} are closed and bounded.

6.1.7 Bounded continuous mappings

Let

$$(6.1.5) C_b(X,Y) = C(X,Y) \cap \mathcal{B}(X,Y)$$

be the space of mappings from X into Y that are both bounded and continuous. If X is countably compact, then this is the same as C(X,Y), as before.

If X is any topological space, then $C_b(X,Y)$ is a closed set in $\mathcal{B}(X,Y)$ with respect to the supremum semimetric $\theta(\cdot,\cdot)$, as in Subsection 6.2.8.

6.2 Uniform convergence

Let X be a nonempty set, and let Y be a topological space. Also let (A, \preceq) be a nonempty directed system, let $\{f_a\}_{a\in A}$ be a net of mappings from X into Y indexed by A, and let f be another mapping from X into Y. We say that $\{f_a\}_{a\in A}$ converges to f pointwise on X if for every $x\in X$, $\{f_a(x)\}_{a\in A}$ converges to f(x) as a net of elements of Y.

6.2.1 Relation with the product topology

The space Y^X of mappings from X into Y may be considered as the Cartesian product of copies of Y indexed by X. Thus the product and strong product topologies can be defined on Y^X in the usual way, using the topology determined on Y by $d_Y(\cdot,\cdot)$.

Note that pointwise convergence of a net of mappings from X into Y is the same as convergence with respect to the product topology. The product topology on Y^X is sometimes also called the *topology of pointwise convergence*.

6.2.2 Using a semimetric on Y

Now let $d_Y(\cdot,\cdot)$ be a semimetric on Y, and suppose that Y is equipped with the topology determined by $d_Y(\cdot,\cdot)$. A net $\{f_a\}_{a\in A}$ of mappings from X into Y is said to converge *uniformly* on X to a mapping f from X into Y with respect to $d_Y(\cdot,\cdot)$ on Y if for every $\epsilon > 0$ there is a $b \in A$ such that

$$(6.2.1) d_Y(f_a(x), f(x)) < \epsilon$$

for every $x \in X$ and $a \in A$ with $b \leq a$. Clearly uniform convergence implies pointwise convergence, and the converse holds when X has only finitely many elements.

6.2.3 Some sets $N_{\epsilon}(f)$ and $\widetilde{N}_{\epsilon}(f)$

Let f be a mapping from X into Y, and let ϵ be a positive real number. Consider the set $N_{\epsilon}(f)$ of all mappings g from X into Y such that

(6.2.2)
$$d_Y(f(x), g(x)) < \epsilon \text{ for every } x \in X.$$

Similarly, let $\widetilde{N}_{\epsilon}(f)$ be the set of all mappings g from X into Y such that $d_Y(f(x), g(x))$ is bounded as a real-valued function on X, with

$$(6.2.3) \qquad \sup\{d_Y(f(x), g(x)) : x \in X\} < \epsilon.$$

Clearly

(6.2.4)
$$\widetilde{N}_{\epsilon}(f) \subseteq N_{\epsilon}(f),$$

because (6.2.3) implies (6.2.2). If η is any positive real number greater than ϵ , then

(6.2.5)
$$N_{\epsilon}(f) \subseteq \widetilde{N}_{\eta}(f).$$

This is because (6.2.2) implies that

$$(6.2.6) \qquad \sup\{d_Y(f(x), g(x)) : x \in X\} \le \epsilon.$$

Of course, if the supremum on the left is attained, then (6.2.2) implies (6.2.3).

6.2.4 The topology of uniform convergence

Let us say that $W \subseteq Y^X$ is an open set with respect to the topology of uniform convergence if for every $f \in W$ there is an $\epsilon > 0$ such that

$$(6.2.7) N_{\epsilon}(f) \subseteq W.$$

This is equivalent to saying that for every $f \in W$ there is an $\eta > 0$ such that

(6.2.8)
$$\widetilde{N}_{\eta}(f) \subseteq W,$$

by the remarks in the previous paragraph. One can check that this defines a topology on Y^X .

It is easy to see that this topology is at least as strong as the product topology on Y^X . The strong product topology on Y^X is at least as strong as the topology of uniform convergence on Y^X .

6.2.5 More on $\widetilde{N}_{\eta}(f)$

Let f, g, and h be mappings from X into Y, and suppose that $d_Y(f(x), g(x))$ and $d_Y(g(x), h(x))$ are bounded as real-valued functions on X. Observe that

(6.2.9)
$$d_Y(f(x), h(x)) \le d_Y(f(x), g(x)) + d_Y(g(x), h(x))$$

 $\le \sup_{w \in X} d_Y(f(w), g(w)) + \sup_{w \in X} d_Y(g(w), h(w))$

for every $x \in X$, using the triangle inequality in the first step. This implies that $d_Y(f(x), h(x))$ is bounded as a real-valued function on X, with

$$(6.2.10) \sup_{x \in X} d_Y(f(x), h(x)) \le \sup_{x \in X} d_Y(f(x), g(x)) + \sup_{x \in X} d_Y(g(x), h(x)).$$

Suppose that $g \in \widetilde{N}_{\eta}(f)$ for some $\eta > 0$, so that

(6.2.11)
$$\delta = \eta - \sup_{x \in X} d_Y(f(x), g(x)) > 0.$$

Under these conditions, one can check that

(6.2.12)
$$\widetilde{N}_{\delta}(g) \subseteq \widetilde{N}_{\eta}(f),$$

using (6.2.10). This implies that $\widetilde{N}_{\eta}(f)$ is an open set in Y^X , with respect to the topology of uniform convergence.

6.2.6 Uniformly convergent nets

Let $\{f_a\}_{a\in A}$ be a net of mappings from X into Y indexed by A again, and let f be another mapping from X into Y. One can check that $\{f_a\}_{a\in A}$ converges to f uniformly on X with respect to $d_Y(\cdot,\cdot)$ if and only if $\{f_a\}_{a\in A}$ converges to f with respect to the topology of uniform convergence on Y^X .

6.2.7 Relation with the supremum semimetric on $\mathcal{B}(X,Y)$

Let $\theta(f,g)$ be the supremum semimetric on the space $\mathcal{B}(X,Y)$ of bounded mappings from X into Y with respect to $d_Y(\cdot,\cdot)$, as in the previous section. One can verify that the topology determined on $\mathcal{B}(X,Y)$ by $\theta(\cdot,\cdot)$ is the same as the topology induced on $\mathcal{B}(X,Y)$ by the topology of uniform convergence on Y^X .

One can also check that $\mathcal{B}(X,Y)$ is a closed set in Y^X with respect to the topology of uniform convergence. Equivalently, if a net of elements of $\mathcal{B}(X,Y)$ converges uniformly to a mapping f from X into Y, then f is bounded on X too.

6.2.8 Uniform convergence and continuity

Suppose that X is a topological space, and let $x_0 \in X$ be given. Also let $\{f_a\}_{a \in A}$ be a net of mappings from X into Y indexed by A that converges uniformly to a mapping f from X into Y. If, for each $a \in A$, f_a is continuous at x_0 , then it is well known and not too difficult to show that f is continuous at x_0 . Similarly, C(X,Y) is a closed set in Y^X with respect to the topology of uniform convergence.

6.3 Infinite series of functions

Let X be a set, and let $a_1(x), a_2(x), a_3(x), \ldots$ be an infinite sequence of real-valued functions on X. Also let A_1, A_2, A_3, \ldots be an infinite sequence of non-negative real numbers such that

$$(6.3.1) |a_j(x)| \le A_j$$

for every $j \geq 1$ and $x \in X$. Suppose that $\sum_{j=1}^{\infty} A_j$ converges as an infinite series of nonnegative real numbers, so that $\sum_{j=1}^{\infty} a_j(x)$ converges absolutely for every $x \in X$, by the comparison test.

Under these conditions, it is not too difficult to show that the sequence of partial sums $\sum_{j=1}^{n} a_j(x)$ converges to $\sum_{j=1}^{\infty} a_j(x)$ uniformly on X. This is a well-known criterion of Weierstrass.

Suppose now that X is a topological space. If $a_j(x)$ is continuous on X for each $j \geq 1$, with respect to the standard topology on \mathbf{R} , in addition to the conditions mentioned in the preceding paragraph, then it follows that $\sum_{j=1}^{\infty} a_j(x)$ is continuous on X as well.

6.3.1 Some continuous functions defined using infinite series

Suppose that X is normal in the strict sense, and let $E \subseteq X$ be a closed G_{δ} set. Thus there is an infinite sequence U_1, U_2, U_3, \ldots of open subsets of X such that $E = \bigcap_{j=1}^{\infty} U_j$. Urysohn's lemma implies that for each $j \geq 1$ there is a continuous real-valued function f_j on X such that $0 \leq f_j(x) \leq 1$ for every $j \geq 1$ and $x \in X$, $f_j(x) = 0$ for every $j \geq 1$ and $x \in X \setminus U_j$.

Put

(6.3.2)
$$f(x) = \sum_{j=1}^{\infty} 2^{-j} f_j(x)$$

for every $x \in X$, where the series on the right converges by comparison with the convergent geometric series $\sum_{j=1}^{\infty} 2^{-j} = 1$. Note that f(x) = 0 for every $x \in E$, and that $0 \le f(x) \le 1$ for every $x \in X$, by construction. If $x \in X \setminus E$, then $x \in X \setminus U_j$ for some $j \ge 1$, which means that $f_j(x) = 1$. This implies that f(x) > 0 for every $x \in X \setminus E$. We also have that f is continuous on X, as before.

6.3.2 Perfect normality implies complete normality

Suppose that X is perfectly normal in the strict sense, and let A and B be separated subsets of X. The closures \overline{A} , \overline{B} of A, B in X are closed sets in X, and G_{δ} sets in X too, by hypothesis. It follows that there are nonnegative continuous real-valued functions f_A , f_B on X such that $f_A(x) = 0$ if and only if $x \in A$, and $f_B(x) = 0$ if and only if $x \in B$, as in the preceding paragraph.

Put

(6.3.3)
$$U = \{x \in X : f_A(x) < f_B(x)\}\$$

and

(6.3.4)
$$V = \{x \in X : f_B(x) < f_A(x)\}.$$

These are open subsets of X, because f_A and f_B are continuous on X. It is easy to see that $A \subseteq U$, $B \subseteq V$, and $U \cap V = \emptyset$, by construction. This shows that X is completely normal in the strict sense.

6.4 Tietze's extension theorem

Let X be a topological space that is normal in the strict sense, and let E be a closed set in X. Also let f be a continuous real-valued function on E, with respect to the induced topology on E and the standard topology on \mathbf{R} , that satisfies $|f(x)| \leq 1$ for every $x \in E$. Under these conditions, Tietze's extension theorem implies that f can be extended to a continuous real-valued function on X that takes values in the interval [-1,1] as well.

6.4.1 An initial approximation

To see this, observe that

(6.4.1)
$$A_0 = \{x \in E : f(x) \le -1/3\}$$

and

$$(6.4.2) B_0 = \{x \in E : f(x) \ge 1/3\}$$

are disjoint closed subsets of X. More precisely, A_0 and B_0 are closed subsets of E, with respect to the induced topology, because f is continuous on E. This implies that A_0 and B_0 are closed subsets of X, because E is a closed set in X.

Using Urysohn's lemma, one can find a continuous real-valued function g_0 on X such that $g_0 = -1/3$ on A_0 , $g_0 = 1/3$ on B_0 , and $|g_0| \leq 1/3$ on X. If $x \in E$, then one can check that

$$|f(x) - g_0(x)| \le 2/3.$$

6.4.2 Some more precise approximations

Of course, $f-g_0$ defines a continuous real-valued function on E. One can repeat the process, to get a sequence $\{g_j\}_{j=0}^{\infty}$ of continuous real-valued functions on X such that $|g_j| \leq (1/3) \, (2/3)^j$ on X for each $j \geq 0$, and

(6.4.4)
$$\left| f - \sum_{j=0}^{n} g_j \right| \le (2/3)^{n+1}$$

on E for every $n \geq 0$. In particular, $\sum_{j=0}^{\infty} g_j(x)$ converges as an infinite series of nonnegative real numbers for every $x \in X$, by comparison with the convergent series $\sum_{j=0}^{\infty} (1/3) (2/3)^j$.

More precisely, the sequence of partial sums $\sum_{j=0}^{n} g_j$ converges uniformly on X, by Weierstrass' criterion. This implies that $\sum_{j=0}^{\infty} g_j$ is continuous on X, as before. Note that

(6.4.5)
$$\left| \sum_{j=0}^{\infty} g_j(x) \right| \le \sum_{j=0}^{\infty} |g_j(x)| \le \sum_{j=0}^{\infty} (1/3) (2/3)^j = 1$$

for every $x \in X$. If $x \in E$, then $\sum_{j=0}^{\infty} g_j(x) = f(x)$, by (6.4.4).

6.4.3 A refinement of this argument

If |f(x)| < 1 for every $x \in E$, then one might wish to have an extension of f to a continuous mapping from X into (-1,1). This can obtained by multiplying the previous extension by a suitable continuous real-valued function on X. More precisely, note that the set where the previous extension is ± 1 is a closed set in X. This set is disjoint from E, because |f| < 1 on E. Thus one can use Urysohn's lemma to find a continuous mapping from X into [0,1] that is equal to 1 on E and to 0 on the set where the previous extension is equal to ± 1 .

6.4.4 Another version of Tietze's theorem

Now let f be any continuous real-valued function on E, with respect to the induced topology on E, and the standard topology on \mathbf{R} . Also let ϕ be a homeomorphism from the real line onto (-1,1), with respect to the standard topology on \mathbf{R} , and the induced topology on (-1,1). It follows that $\phi \circ f$ is a continuous mapping from E into (-1,1), which can be extended to a continuous mapping from X into (-1,1), as in the preceding paragraph. The composition of this extension with the inverse of ϕ is a continuous extension of f on X. This corresponds to Exercise 6 on p78 of [156].

6.5 Uniform convergence on compact sets

Let X and Y be sets, and let $d_Y(\cdot,\cdot)$ be a semimetric on Y. Also let E be a nonempty subset of X, let f be a mapping from X into Y, and let ϵ be a positive real number. Consider the set $N_{E,\epsilon}(f)$ of all mappings g from X into Y such that

(6.5.1)
$$d_Y(f(x), g(x)) < \epsilon \text{ for every } x \in E.$$

As before, let $N_{E,\epsilon}(f)$ be the set of all mappings g from X into Y such that $d_Y(f(x), g(x))$ is bounded on E, with

$$\sup_{x \in E} d_Y(f(x), g(x)) < \epsilon.$$
 Thus

(6.5.3)
$$\widetilde{N}_{E,\epsilon}(f) \subseteq N_{E,\epsilon}(f),$$

and $(6.5.4) N_{E,\epsilon}(f) \subseteq \widetilde{N}_{E,\eta}(f)$

for every $\eta > \epsilon$, as in Subsection 6.2.2.

6.5.1 More on $N_{E,\epsilon}(f)$ and $\widetilde{N}_{E,\epsilon}(f)$

If f, g, and h are mappings from X into Y such that $d_Y(f(x), g(x))$ and $d_Y(g(x), h(x))$ are bounded on E, then $d_Y(f(x), h(x))$ is bounded on E, and

(6.5.5)
$$\sup_{x \in E} d_Y(f(x), h(x)) \le \sup_{x \in E} d_Y(f(x), g(x)) + \sup_{x \in E} d_Y(g(x), h(x)),$$

as in Subsection 6.2.5. If $g \in \widetilde{N}_{E,\eta}(f)$ for some $\eta > 0$, then

(6.5.6)
$$\delta = \eta - \sup_{x \in E} d_Y(f(x), g(x)) > 0,$$

and

(6.5.7)
$$\widetilde{N}_{E,\delta}(g) \subseteq \widetilde{N}_{E,n}(f),$$

as before.

Let E_0 be a nonempty subset of X such that

$$(6.5.8) E_0 \subseteq E.$$

If f is a mapping from X into Y, then

$$(6.5.9) N_{E,\epsilon}(f) \subseteq N_{E_0,\epsilon}(f)$$

and

$$(6.5.10) \widetilde{N}_{E,\epsilon}(f) \subseteq \widetilde{N}_{E_0,\epsilon}(f)$$

for every $\epsilon > 0$.

6.5.2 The topology of uniform convergence on compact sets

Suppose now that X is a nonempty topological space. A subset W of the space Y^X of all mappings from X into Y is said to be an open set with respect to the topology of uniform convergence on compact subsets of X if for every $f \in W$ there are a nonempty compact subset K of X and an $\epsilon > 0$ such that

$$(6.5.11) N_{K,\epsilon}(f) \subseteq W.$$

This is equivalent to saying that for every $f \in W$ there are a nonempty compact set $K \subseteq X$ and an $\eta > 0$ such that

$$(6.5.12) \widetilde{N}_{K,\epsilon}(f) \subseteq W.$$

One can check that this defines a topology on Y^X . More precisely, this uses the fact that the union of finitely many compact subsets of X is compact, to get that the intersection of finitely many open subsets of Y^X is an open set with respect to this topology.

6.5.3 Some properties of this topology

Of course, finite subsets of X are automatically compact. This implies that the topology of uniform convergence on compact subsets of X is at least as strong as the topology of pointwise convergence. If X is equipped with the discrete topology, then every compact subset of X is a finite set, and these two topologies on Y^X are the same.

The topology of uniform convergence on Y^X is at least as strong as the topology of uniform convergence on compact subsets of X. If X is compact, then these two topologies on Y^X are the same.

6.5.4 Convergent nets with respect to this topology

If K is a nonempty compact subset of X, f is a mapping from X into Y, and η is a positive real number, then $\widetilde{N}_{K,\eta}(f)$ is an open set in Y^X with respect to the topology of uniform convergence on compact subsets of X. This follows from (6.5.7).

Let (A, \preceq) be a nonempty directed system, let $\{f_a\}_{a\in A}$ be a net of mappings from X into Y indexed by A, and let f be another mapping from X into Y. One can check that $\{f_a\}_{a\in A}$ converges to f with respect to the topology of uniform convergence on compact subsets of X if and only if for every compact subset K of X, the restriction of f_a to K converges uniformly to the restriction of f to K, as a net of mappings from K into Y indexed by A.

6.6 The compact-open topology

Let X and Y be nonempty topological spaces. If K is a subset of X and V is a subset of Y, then let N(K,V) be the collection of mappings f from X into Y such that

$$(6.6.1) f(K) \subseteq V.$$

A subset W of the space Y^X of all mappings from X into Y is said to be an open set with respect to the *compact-open topology* if there are finitely many compact subsets K_1, \ldots, K_n of X and open sets V_1, \ldots, V_n of Y such that

$$(6.6.2) f(K_i) \subseteq V_i$$

for each $j = 1, \ldots, n$, and

(6.6.3)
$$\bigcap_{j=1}^{n} N(K_j, V_j) \subseteq W.$$

One can verify that this defines a topology on Y^X .

If K is a compact subset of X, and V is an open subset of Y, then N(K,V) is an open subset of Y^X with respect to the compact-open topology. The collection of these open sets forms a subbase for the compact-open topology on Y^X .

6.6.1 Some properties of the compact-open topology

Remember that Y^X is the same as the Cartesian product of copies of Y indexed by X. It is easy to see that the compact-open topology on Y^X is at least as strong as the product topology, because finite subsets of X are compact. If X is equipped with the discrete topology, then one can check that the compact-open topology on Y^X is the same as the product topology, because compact subsets of X are finite. The strong product topology on Y^X is at least as strong as the compact-open topology.

6.6.2 The compact-open topology on C(X,Y)

Sometimes one may be particularly concerned with the topology induced on the space C(X,Y) of continuous mappings from X into Y by the compact-open topology on Y^X . This may be called the *compact-open topology on* C(X,Y). If K is a compact subset of X and Y is an open subset of Y, then

(6.6.4)
$$N^{C}(K, V) = N(K, V) \cap C(X, Y)$$

is relatively open in C(X,Y), and the collection of these relatively open subsets of C(X,Y) is a subbase for the compact-open topology on C(X,Y).

6.6.3 Using a semimetric on Y again

Let $d_Y(\cdot,\cdot)$ be a semimetric on Y, and suppose from now on in this section that Y is equipped with the topology determined by $d_Y(\cdot,\cdot)$. Under these conditions, it is well known that the compact-open topology on C(X,Y) is the same as the topology induced by the topology of uniform convergence on compact subsets of X. Before showing this, let us mention a property of compact subsets of Y.

6.6.4 A property of compact subsets of Y

Let A be a compact subset of Y, and let V be an open subset of Y that contains K. It is well known that there is a positive real number r such that

(6.6.5)
$$\bigcup_{a \in A} B_Y(a, r) \subseteq V,$$

where $B_Y(a,r)$ is the open ball in Y centered at a with radius r with respect to $d_Y(\cdot,\cdot)$.

Indeed, if $a \in A$, then $a \in V$, and there is a positive real number r(a) such that $B_Y(a, r(a)) \subseteq V$. The collection of open balls of the form $B_Y(a, r(a)/2)$, $a \in A$, is an open covering of A in Y, and so there are finitely many elements a_1, \ldots, a_n of A such that

(6.6.6)
$$A \subseteq \bigcup_{j=1}^{n} B_Y(a_j, r(a_j)/2),$$

because A is compact. One can check that (6.6.5) holds with

(6.6.7)
$$r = \min_{1 \le j \le n} r(a_j)/2.$$

6.6.5 Some remarks about continuous mappings and compact sets

Let f be a continuous mapping from X into Y, let K be a nonempty compact subset of X, and let V be an open subset of Y such that $f(K) \subseteq V$. Because

f(K) is a compact subset of Y, there is a positive real number r such that

(6.6.8)
$$\bigcup_{y \in f(K)} B_Y(y, r) \subseteq V,$$

as in the previous subsection. If g is a continuous mapping from X into Y such that

$$(6.6.9) d_Y(f(x), g(x)) < r for every x \in K,$$

then it follows that $g(K) \subseteq V$ too.

This shows that $N^{C}(K, V)$ is a open subset of C(X, Y) with respect to the topology induced by the topology of uniform convergence on compact subsets of X. Using this, it is easy to see that the topology induced on C(X, Y) by the topology of uniform convergence on compact subsets of X is at least as strong as the compact-open topology.

6.6.6 More on continuous mappings and compact sets

Let f be a continuous mapping from X into Y again, let K be a nonempty compact subset of X, and let $\epsilon > 0$ be given. If $x \in K$, then there is an open subset U(x) of X such that $x \in U(x)$ and

$$(6.6.10) f(U(x)) \subseteq B_Y(f(x), \epsilon/3),$$

because f is continuous at x. The collection of open sets of this type, with $x \in K$, forms an open covering of K in X, and so there are finitely many elements x_1, \ldots, x_n of K such that

(6.6.11)
$$K \subseteq \bigcup_{j=1}^{n} U(x_j),$$

because K is compact.

6.6.7 Finitely many smaller compact sets

Put

(6.6.12)
$$E_j = \{ w \in X : d_Y(f(w), f(x_j)) \le \epsilon/3 \}$$

for each $j=1,\ldots,n$. One can check that this is a closed set in X for every $j=1,\ldots,n$, because f is continuous. This implies that $K\cap E_j$ is a compact subset of X for each $j=1,\ldots,n$. Note that $U(x_j)\subseteq E_j$ for every $j=1,\ldots,n$, by (6.6.10). It follows that

(6.6.13)
$$K = \bigcup_{j=1}^{n} (K \cap E_j),$$

because of (6.6.11).

6.6.8 Using the compact-open topology

By construction,

$$(6.6.14) f(K \cap E_i) \subseteq f(E_i) \subseteq B_Y(f(x_i), \epsilon/2)$$

for every $j = 1, \dots, n$. Let q be any mapping from X into Y such that

$$(6.6.15) g(K \cap E_j) \subseteq B_Y(f(x_j), \epsilon/2)$$

for each $j=1,\ldots,n$. Let $w\in K$ be given, so that $w\in U(x_{j_0})$ for some $j_0\in\{1,\ldots,n\}$, by (6.6.11). It follows that

$$(6.6.16) d_Y(f(w), g(w)) \le d_Y(f(w), f(x_{j_0})) + d_Y(f(x_{j_0}), g(w)) < \epsilon/3 + \epsilon/2,$$

using (6.6.10) and (6.6.15) in the second step. This implies that the compactopen topology on C(X,Y) is at least as strong as the topology induced by the topology of uniform convergence on compact subsets of X.

6.7 Continuity on compact sets

Let X and Y be topological spaces, and let f be a mapping from X into Y. If f is continuous on X, then the restriction of f to any subset X_0 of X is continuous as a mapping from X_0 into Y, with respect to the topology induced on X_0 by the topology on X. Similarly, if f is continuous at a point $x_0 \in X$, as a mapping on X, then the restriction of f to any subset X_0 of X that contains x_0 is continuous at x_0 too, with respect to the induced topology on X_0 .

In some situations, one may be concerned with mappings on X whose restrictions to some subsets of X have some continuity properties, such as restrictions to compact subsets of X. One may also wish to obtain continuity properties on X from continuity properties of these restrictions to some subsets of X, under suitable conditions.

6.7.1 Local compactness and continuity on compact sets

Let x_0 be an element of X, let U_0 be an open subset of X that contains x_0 , and let K_0 be a compact subset of X that contains x_0 . If f is a mapping from X into Y, and the restriction of f to K_0 is continuous at x_0 with respect to the induced topology on K_0 , then it is easy to see that f is continuous at x_0 . In particular, if X is locally compact, and if the restriction of f to any compact subset K of X is continuous with respect to the induced topology on K, then f is continuous on X.

6.7.2 Some remarks about sequential continuity

Let $\{x_j\}_{j=1}^{\infty}$ be a sequence of elements of X that converges to an element x of X. Remember that $K = \{x_j : j \in \mathbf{Z}_+\} \cup \{x\}$ is a compact subset of X in this

case. Note that $\{x_j\}_{j=1}^{\infty}$ converges to x with respect to the induced topology on K.

If the restriction of f to K is sequentially continuous at x, with respect to the induced topology on K, then $\{f(x_j)\}_{j=1}^{\infty}$ converges to f(x) in Y. If the restriction of f to any compact subset of X that contains x is sequentially continuous at x, with respect to the induced topology, then it follows that f is sequentially continuous at x.

6.7.3 Using the first countability condition

If there is a local base for the topology of X at x with only finitely or countably many elements, then sequential continuity at x implies continuity at x. In this situation, we get that f is continuous at x when the restriction of f to any compact subset of X that contains x is continuous at x, with respect to the induced topology. If X satisfies the first countability condition, and if the restriction of f to any compact subset of X is continuous, with respect to the induced topology, then f is continuous on X.

6.7.4 Uniform convergence and continuity conditions

Let $d_Y(\cdot,\cdot)$ be a semimetric on Y, and suppose that Y is equipped with the topology determined by $d_Y(\cdot,\cdot)$. Also let (A, \preceq) be a nonempty directed system, and let $\{f_a\}_{a\in A}$ be a net of mappings from X into Y indexed by A that converges to a mapping f from X into Y uniformly on compact subsets of X. Suppose that for each $a\in A$, f_a is continuous at a point x_0 in X. If K_0 is a compact subset of X that contains x_0 , then the restriction of f_a to K_0 is continuous at x_0 for every $a\in A$, with respect to the induced topology on K_0 . This implies that the restriction of f to K_0 is continuous at x_0 as well, with respect to the induced topology on K_0 .

6.8 Supports of real-valued functions

Let X be a topological space, and let f be a real-valued function on X. The support of f in X is defined by

(6.8.1)
$$\operatorname{supp} f = \overline{\{x \in X : f(x) \neq 0\}},$$

which is the closure in X of the set where f is not 0.

If g is another real-valued function on X, then it is easy to see that

$$(6.8.2) supp(f+g) \subseteq (supp f) \cup (supp g)$$

and

$$(6.8.3) supp(f g) \subseteq (supp f) \cap (supp g).$$

6.8.1 Compactly-supported functions

In some situations, one may be concerned with continuous real-valued functions on X whose support is a compact subset of X. If f is a continuous real-valued function on X, then

$$\{x \in X : f(x) \neq 0\}$$

is an open set in X. If, for every $x \in X$, there is a continuous real-valued function f on X such that f has compact support in X and $f(x) \neq 0$, then X has to be locally compact.

6.8.2 Another version of Urysohn's lemma

Suppose that X is a locally compact Hausdorff space. If $K \subseteq X$ is compact, $U \subseteq X$ is an open set, and $K \subseteq U$, then a version of Urysohn's lemma implies that there is a continuous real-valued function f on X with compact support contained in U such that

$$(6.8.5) f \equiv 1 \text{ on } K \text{ and } 0 \le f \le 1 \text{ on } X.$$

6.9 Functions with finite support

Let X be a nonempty set, and let c(X) be the space of all real-valued functions on X. Let us say that $f \in c(X)$ has finite support in X if f(x) = 0 for all but finitely many $x \in X$. If we take X to be equipped with the discrete topology, then the support of $f \in c(X)$ in X as defined in the previous section is the same as the set of $x \in X$ such that $f(x) \neq 0$.

Let $c_{00}(X)$ be the subset of c(X) consisting of functions on X with finite support. Note that $c_{00}(X) = c(X)$ exactly when X has finitely many elements.

6.9.1 Relation with Cartesian products

Equivalently, c(X) is the same as the Cartesian product of copies of **R** indexed by X. Using the standard topology on the real line, we get the corresponding product topology and strong product topology on c(X).

It is easy to see that $c_{00}(X)$ is dense in c(X) with respect to the product topology. It is not too difficult to show that $c_{00}(X)$ is a closed set in c(X) with respect to the strong product topology.

6.9.2 Using positive weights on X

Let a be a positive real-valued function on X. If $f, g \in c_{00}(X)$, then put

(6.9.1)
$$d_a(f,g) = \max\{a(x) | f(x) - g(x) | : x \in X\}.$$

Note that the maximum is automatically attained, because it reduces to the maximum of finitely many nonnegative real numbers. One can check that this defines a metric on $c_{00}(X)$.

133

It is easy to see that the topology determined on $c_{00}(X)$ by (6.9.1) is at least as strong as the topology induced on $c_{00}(X)$ by the product topology on c(X). One can also verify that the topology induced on $c_{00}(X)$ by the strong product topology on c(X) is at least as strong as the topology determined on $c_{00}(X)$ by (6.9.1).

If b is another positive real-valued function on X such that

(6.9.2)
$$a(x) \le b(x)$$
 for every $x \in X$,

then

$$(6.9.3) d_a(f,g) \le d_b(f,g)$$

for every $f, g \in c_{00}(X)$.

6.9.3 Another metric associated to a

Similarly, if $f, g \in c_{00}(X)$, then put

(6.9.4)
$$\widetilde{d}_a(f,g) = \sum_{x \in X} a(x) |f(x) - g(x)|,$$

where the sum on the right reduces to a finite sum of nonnegative real numbers. It is easy to see that this defines a metric on $c_{00}(X)$ as well.

Observe that

$$(6.9.5) d_a(f,g) \le \widetilde{d}_a(f,g)$$

for every $f, g \in c_{00}(X)$. If b is another positive real-valued function on X that satisfies (6.9.2), then

(6.9.6)
$$\widetilde{d}_a(f,g) \le \widetilde{d}_b(f,g)$$

for every $f, g \in c_{00}(X)$.

6.9.4 The case where $X = \mathbf{Z}_{+}$

If $X = \mathbf{Z}_+$, and $\sum_{j=1}^{\infty} a(j)/b(j)$ converges as an infinite series of positive real numbers, then

(6.9.7)
$$\widetilde{d}_a(f,g) \le \left(\sum_{j=1}^{\infty} a(j)/b(j)\right) d_b(f,g)$$

for every $f, g \in c_{00}(\mathbf{Z}_+)$.

6.10 Some subspaces of c(X)

Let X be a nonempty set again, and let E be a subset of X. Put

(6.10.1)
$$c^{E}(X) = \{ f \in c(X) : f(x) = 0 \text{ for every } x \in X \setminus E \}$$

and

(6.10.2)
$$c_{00}^{E}(X) = c_{00}(X) \cap c^{E}(X).$$

Of course, if E has only finitely many elements, then $c^{E}(X) \subseteq c_{00}(X)$, and $c^{E}(X) = c_{00}^{E}(X)$.

6.10.1 Relation with c(E) and $c_{00}(E)$

Suppose that $E \neq \emptyset$, so that c(E) and $c_{00}(E)$ can be defined as in the previous section. Of course, if $f \in c(X)$, then the restriction of f to E defines an element of c(E). Similarly, if $f \in c_{00}(X)$, then the restriction of f to E defines an element of $c_{00}(E)$. Any real-valued function on E can be extended to a real-valued function on E, by setting it equal to E0 on E1. This defines a one-to-one mapping from E2 onto E3, which maps E4 onto E6 onto E6.

6.10.2 Relation with Cartesian products again

Remember that c(E) and c(X) are the same as Cartesian products of copies of \mathbf{R} indexed by E and X, respectively. This leads to the corresponding product topologies and strong product topologies on c(E) and c(X), using the standard topology on \mathbf{R} .

Note that $c^E(X)$ is a closed set in c(X) with respect to the product topology. One can check that the product topology on c(E) corresponds exactly to the topology induced on $c^E(X)$ by the product topology on c(X), with respect to the one-to-one mapping from c(E) onto $c^E(X)$ mentioned in the preceding paragraph.

Similarly, the strong product topology on c(E) corresponds exactly to the topology induced on $c^E(X)$ by the strong product topology on c(X), with respect to this mapping.

In particular, if E has only finitely many elements, then the product and strong product topologies on c(E) are the same. In this case, the topologies induced on $c^{E}(X)$ by the product and strong product topologies on c(X) are the same.

6.10.3 Using positive weights again

Let a be a positive real-valued function on X, and let $d_a(f,g)$ and $\widetilde{d}_a(f,g)$ be the metrics defined on $c_{00}(X)$ in (6.9.1) and (6.9.4), respectively. Suppose that E has only finitely many elements, and put

(6.10.3)
$$C(a, E) = \sum_{x \in E} a(x).$$

If $f, g \in c^E(X)$, then it is easy to see that

(6.10.4)
$$\widetilde{d}_a(f,g) \le C(a,E) d_a(f,g).$$

In this situation, the topologies determined on $c^E(X)$ by the restrictions of $d_a(f,g)$ and d(f,g) to $f,g \in c^E(X)$ are the same as the topology induced on $c^E(X)$ by the product topology on c(X).

6.11 Some mappings into products

Let A be a nonempty set, let Y_{α} be a set for each $\alpha \in A$, and put $Y = \prod_{\alpha \in A} Y_{\alpha}$. If $\beta \in A$, then let p_{β} be the corresponding coordinate mapping from Y into Y_{β} , so that $p_{\beta}(f) = f(\beta)$ for every $f \in Y$.

Let X be another set, and let ϕ be a mapping from X into Y. Thus, for each $\beta \in A$, $p_{\beta} \circ \phi$ is a mapping from X into Y_{β} . Conversely, given a mapping from X into Y_{β} for each $\beta \in A$, there is a unique mapping from X into Y whose composition with p_{β} is the given mapping from X into Y_{β} for every $\beta \in A$.

6.11.1 Continuous mappings into Y

Suppose now that Y_{α} is a topological space for every $\alpha \in A$, and that X is a topological space as well. If ϕ is a continuous mapping from X into Y, with respect to the corresponding product topology on Y, then $p_{\beta} \circ \phi$ is continuous as a mapping from X into Y_{β} for every $\beta \in A$. Conversely, if $p_{\beta} \circ \phi$ is continuous as a mapping from X into Y_{β} for every $\beta \in A$, then one can check that ϕ is continuous as a mapping from X into Y, with respect to the product topology on Y.

6.11.2 Mappings into $c_{00}(A)$

Let us take Y_{α} to be the real line for every $\alpha \in A$, so that Y is the same as the space c(A) of all real-valued functions on A. Remember that $c_{00}(A)$ is the subset of c(A) consisting of functions on A with finite support, as in Section 6.9. A mapping ϕ from X into c(A) takes values in $c_{00}(A)$ if and only if for every $x \in X$,

(6.11.1)
$$p_{\beta}(\phi(x)) = 0$$
 for all but finitely many $\beta \in A$.

6.11.3 A local finiteness condition

Suppose that for every $x \in X$ there is an open set $U(x) \subseteq X$ and a finite subset A(x) of A such that $x \in U(x)$ and

(6.11.2)
$$p_{\beta}(\phi(y)) = 0$$
 for every $y \in U(x)$ and $\beta \in A \setminus A(x)$.

This is the same as saying that the family of subsets of X where $p_{\beta} \circ \phi \neq 0$ is locally finite in X, as in Section 5.13. In particular, this condition implies that ϕ maps X into $c_{00}(A)$.

Suppose also that $p_{\beta} \circ \phi$ is continuous as a real-valued function on X for every $\beta \in A$, with respect to the standard topology on \mathbf{R} . Under these conditions, one can verify that ϕ is continuous as a mapping from X into $c_{00}(A)$, with respect to the topology induced on $c_{00}(A)$ by the corresponding strong product topology on c(A), as in the previous sections.

6.11.4 Continuity of ϕ

More precisely, if B is a subset of A, then let $c^B(A)$ be the space of $f \in c(A)$ such that $f(\alpha) = 0$ for every $\alpha \in A \setminus B$, as in the previous section. If B has only finitely many elements, then the topologies induced on $c^B(A)$ by the product and strong product topologies on c(A) are the same, as before. Using this notation, (6.11.2) says that $\phi(y)$ is an element of $c^{A(x)}(A)$ for every $y \in U(x)$.

Let a be a positive real-valued function on A, and remember that

(6.11.3)
$$\widetilde{d}_a(f,g) = \sum_{\alpha \in A} a(\alpha) |f(\alpha) - g(\alpha)|$$

defines a metric on $c_{00}(A)$, as in Subsection 6.9.3. If $B \subseteq A$ has only finitely many elements, then the topology determined on $c^B(A)$ by the restriction of (6.11.3) to $f, g \in c^B(A)$ is the same as the topology induced on $c^B(A)$ by the product topology on c(A), as in the previous section.

Suppose as before that for every $x \in X$ there is an open set $U(x) \subseteq X$ and a finite set $A(x) \subseteq A$ such that (6.11.2) holds, and that $p_{\beta} \circ \phi$ is continuous as a real-valued function on X for every $\beta \in A$. Under these conditions, ϕ is continuous as a mapping from X into $c_{00}(A)$, with respect to the topology determined on $c_{00}(A)$ by (6.11.3).

Chapter 7

Filters and ultrafilters

7.1 Filters

Let X be a nonempty set. A nonempty collection \mathcal{F} of nonempty subsets of X is said to be a *filter* on X if it satisfies the following two conditions. First,

(7.1.1) if
$$A, B \in \mathcal{F}$$
, then $A \cap B \in \mathcal{F}$.

Second,

(7.1.2) if
$$A \in \mathcal{F}$$
, $E \subseteq X$, and $A \subseteq E$, then $E \in \mathcal{F}$.

7.1.1 Some basic examples of filters

If X_0 is a nonempty subset of X, then the collection of all subsets of X that contain X_0 is a filter on X.

If X has infinitely many elements, then one can check that the collection of all subsets A of X such that $X \setminus A$ has only finitely many elements is a filter on X.

If X is a topological space and $x \in X$, then

(7.1.3)
$$\{A \subseteq X : \text{there is an open set } U \subseteq X \text{ such that } x \in U \text{ and } U \subseteq A\}$$

is a filter on X. In this situation, a filter \mathcal{F} on X is said to *converge* to x if for every open subset U of X with $x \in U$, we have that

$$(7.1.4) U \in \mathcal{F}.$$

7.1.2 Pre-filters or filter bases

Let \mathcal{B} be a nonempty collection of nonempty subsets of a nonempty set X. We say that \mathcal{B} is a *pre-filter* or a *filter base* on X if

(7.1.5) for every $A, B \in \mathcal{B}$ there is a $C \in \mathcal{B}$ such that $C \subseteq A \cap B$.

In this case, one can check that

(7.1.6)
$$\mathcal{F} = \{ E \subseteq X : \text{there is an } A \in \mathcal{B} \text{ such that } A \subseteq E \}$$

is a filter on X.

Conversely, if there is a filter \mathcal{F} on X such that (7.1.6) holds, then it is easy to see that \mathcal{B} is a filter base on X, and \mathcal{B} is said to be a *base* for \mathcal{F} .

If X is a topological space and $x \in X$, then a local base for the topology of X at x is the same as a base for the filter (7.1.3).

7.1.3 Filter bases from nets

Let (A, \preceq) be a nonempty directed system, and let $\{x_a\}_{a \in A}$ be a net of elements of X indexed by A. If $a \in A$, then put

$$(7.1.7) E_a = \{x_b : b \in A, \ a \leq b\}.$$

It is easy to see that the collection of these subsets E_a , $a \in A$, of X form a filter base on X, and we let \mathcal{F} be the corresponding filter on X.

Suppose that X is a topological space, and let x be an element of X. One can check that $\{x_a\}_{a\in A}$ converges to x as a net of elements of X if and only if the corresponding filter \mathcal{F} converges to x in X.

7.1.4 Some nets related to filters

Let \mathcal{F} be a filter on a nonempty set X, and let \mathcal{B} be a base for \mathcal{F} . If $A, B \in \mathcal{B}$, then put $A \leq B$ when $B \subseteq A$. This defines a partial ordering on \mathcal{B} , which makes \mathcal{B} a directed system, because of (7.1.5).

Suppose that X is a topological space again, and that $x \in X$. If \mathcal{F} converges to x on X, and if $\{x_A\}_{A \in \mathcal{B}}$ is a net of elements of X indexed by \mathcal{B} such that

$$(7.1.8) x_A \in A$$

for every $A \in \mathcal{B}$, then $\{x_A\}_{A \in \mathcal{B}}$ converges to x in X.

Conversely, if \mathcal{F} does not converge to x in X, then there is an open set $U \subseteq X$ such that $x \in U$ and $U \notin \mathcal{F}$. This implies that for each $A \in \mathcal{B}$, $A \not\subseteq U$. If $x_A \in A \setminus U$ for every $A \in \mathcal{B}$, then $\{x_A\}_{A \in \mathcal{B}}$ does not converge to x, as a net of elements of X.

7.1.5 Uniqueness of the limit in Hausdorff spaces again

If X is a Hausdorff topological space, then a filter \mathcal{F} on X can converge to at most one element of X. Indeed, suppose for the sake of a contradiction that \mathcal{F} converges to distinct elements x and y of X. Because X is Hausdorff, there are disjoint open subsets U and V of X such that $x \in U$ and $y \in V$. It follows that $U, v \in \mathcal{F}$, so that $\emptyset = U \cap V \in \mathcal{F}$, which is a contradiction.

Conversely, if X is not Hausdorff, then there are distinct elements x, y of X such that for all open subsets U and V of X with $x \in U$ and $y \in V$, we have

that $U \cap V \neq \emptyset$. Observe that the collection of subsets of X of the form $U \cap V$, where $U, V \subseteq X$ are open sets with $x \in U$ and $y \in V$, is a filter base on X. It is easy to see that the corresponding filter on X converges to both x and y.

7.1.6 Adherent points and filters

Let X be a topological space, let \mathcal{F} be a filter on X that converges to x, and let E be an element of \mathcal{F} . If U is an open subset of X that contains x, then $U \in \mathcal{F}$, and hence

$$(7.1.9) E \cap U \in \mathcal{F}.$$

This implies that $E \cap U \neq \emptyset$, so that

$$(7.1.10) x \in \overline{E}.$$

Conversely, suppose that $x \in X$ is adherent to $E \subseteq X$, and let $\mathcal{B}(x)$ be a local base for the topology of X at x. One can check that

$$(7.1.11) {E \cap U : U \in \mathcal{B}(x)}$$

is a filter base on X, for which the corresponding filter on X converges to x and contains E as an element.

7.1.7 Filters and induced topologies

Let \mathcal{F} be a filter on a nonempty set X. If $Y \in \mathcal{F}$, then

(7.1.12)
$$\mathcal{F}_Y = \{ A \in \mathcal{F} : A \subseteq Y \}$$

may be considered as a filter on Y. Note that any filter on a nonempty subset of X may be considered as a filter base on X.

Suppose now that X is a topological space, $Y \in \mathcal{F}$, and $x \in Y$. Under these conditions, one can check that \mathcal{F} converges to x if and only if (7.1.12) converges to x with respect to the induced topology on Y.

7.2 Continuity and refinements

Let X and Y be nonempty sets, and let f be a mapping from X into Y. If \mathcal{F} is a filter on X, then one can check that

$$(7.2.1) {f(A): A \in \mathcal{F}}$$

is a filter base on Y. More precisely, this uses the fact that if $A, B \in \mathcal{F}$, then $A \cap B \in \mathcal{F}$, and

$$(7.2.2) f(A \cap B) \subseteq f(A) \cap f(B).$$

7.2.1 Pushing filters forward

Let $f_*(\mathcal{F})$ be the filter on Y obtained from (7.2.1) as in the previous section. Equivalently, one can verify that

$$(7.2.3) f_*(\mathcal{F}) = \{ E \subseteq Y : f^{-1}(E) \in \mathcal{F} \}.$$

Suppose now that X and Y are topological spaces. If \mathcal{F} converges to $x \in X$, and if f is continuous at x, then one can check that $f_*(\mathcal{F})$ converges to f(x) in Y.

Conversely, if \mathcal{F} is the filter (7.1.3) obtained from the open sets in X that contain x, and if $f_*(\mathcal{F})$ converges to f(x) in Y, then it is easy to see that f is continuous at x.

7.2.2 Refinements of filters

Let X be a nonempty set again, and let \mathcal{F} be a filter on X. A filter \mathcal{F}' on X is said to be a *refinement* of \mathcal{F} if

$$(7.2.4) \mathcal{F} \subseteq \mathcal{F}',$$

as collections of subsets of X.

If X is a topological space, $x \in X$, and \mathcal{F} is a filter on X that converges to x, then every filter on X that is a refinement of \mathcal{F} converges to x too. Note that a filter \mathcal{F} on X converges to x exactly when \mathcal{F} is a refinement of the filter (7.1.3) obtained from the open subsets of X that contain x.

7.2.3 Convergent refinements

Suppose that a filter \mathcal{F} on X has a refinement \mathcal{F}' that converges to $x \in X$. If $E \in \mathcal{F}'$, then $x \in \overline{E}$, as in the previous section. In particular, this holds for $E \in \mathcal{F}$, so that

$$(7.2.5) x \in \bigcap_{E \in \mathcal{F}} \overline{E}.$$

Conversely, suppose that (7.2.5) holds, and let $\mathcal{B}(x)$ be a local base for the topology of X at x. One can verify that

$$(7.2.6) \{E \cap U : E \in \mathcal{F}, U \in \mathcal{B}(x)\}\$$

is a filter base on X, for which the corresponding filter is a refinement of \mathcal{F} that converges to x.

7.2.4 Filters and compact sets

Let K be a compact subset of X, and let \mathcal{F} be a filter on X such that $K \in \mathcal{F}$. If E_1, \ldots, E_n are finitely many elements of \mathcal{F} , then

(7.2.7)
$$\left(\bigcap_{j=1}^{n} E_{j}\right) \cap K \in \mathcal{F},$$

so that $\left(\bigcap_{j=1}^n E_j\right) \cap K \neq \emptyset$. In particular,

(7.2.8)
$$\left(\bigcap_{j=1}^{n} \overline{E_{j}}\right) \cap K \neq \emptyset,$$

which means that the collection of closed sets \overline{E} , $E \in \mathcal{F}$, has the finite intersection property with respect to K. It follows that

(7.2.9)
$$\left(\bigcap_{E\in\mathcal{F}}\overline{E}\right)\cap K\neq\emptyset,$$

because K is compact. This implies that \mathcal{F} has a refinement that converges to an element of K in X, as in the preceding paragraph.

7.2.5 Filter bases and the finite intersection property

Let X be a set, let K be a subset of X, and let $\{E_j\}_{j\in I}$ be a nonempty family of subsets of X with the finite intersection property with respect to K. Consider the collection of subsets of X of the form

$$\left(\bigcap_{l=1}^{n} E_{j_{l}}\right) \cap K,$$

where j_1, \ldots, j_n are finitely many elements of I. It is easy to see that this collection is a filter base on X.

Suppose that X is a topological space, and that $K \subseteq X$ has the property that every filter on X that contains K as an element has a refinement that converges to an element of K. In order to show that K is compact, let $\{E_j\}_{j\in I}$ be any nonempty family of closed subsets of X with the finite intersection property with respect to K. Consider the filter \mathcal{F} on X for which the collection of sets of the form (7.2.10) is a base. By construction, $K \in \mathcal{F}$, so that \mathcal{F} has a refinement that converges to an element x of K. This implies that

$$(7.2.11) x \in \left(\bigcap_{j \in I} E_j\right) \cap K,$$

by (7.2.5), so that K is compact, as desired.

7.3 Ultrafilters

A filter \mathcal{F} on a nonempty set X is said to be an *ultrafilter* if \mathcal{F} is maximal with respect to refinement. This means that if \mathcal{F}' is a filter on X that is a refinement of \mathcal{F} , then $\mathcal{F} = \mathcal{F}'$.

7.3.1 Some ultrafilters

If p is any element of X, then it is easy to see that the collection of all subsets of X that contain p as an element is an ultrafilter on X.

If \mathcal{F} is any filter on X, then one can use Zorn's lemma or Hausdorff's maximality principle to show that \mathcal{F} has a refinement which is an ultrafilter.

7.3.2 Compactness and ultrafilters

Let X be a topological space. If $K \subseteq X$ is compact and \mathcal{F} is a filter on X that contains K as an element, then there is a refinement of \mathcal{F} that converges to an element of K, as in the previous section. If \mathcal{F} is an ultrafilter, then it follows that \mathcal{F} converges to an element of K.

Conversely, suppose that $K \subseteq X$ has the property that every ultrafilter on X that contains K as an element converges to an element of K. If \mathcal{F} is any filter on X that contains K, then \mathcal{F} has a refinement that is an ultrafilter, as before. This refinement contains K, and thus converges to an element of K, by hypothesis. This implies that K is compact, as in the previous section.

7.3.3 Some refinements of filters

Let \mathcal{F} be a filter on a nonempty set X, and suppose that $E \subseteq X$ has the property that $A \cap E \neq \emptyset$ for every $A \in \mathcal{F}$. It is easy to see that

$$(7.3.1) {A \cap E : A \in \mathcal{F}}$$

is a filter base on X.

The filter on X for which (7.3.1) is a base is a refinement of \mathcal{F} that contains E as an element. If \mathcal{F} is an ultrafilter on X, then it follows that $E \in \mathcal{F}$.

7.3.4 A characterization of ultrafilters

If \mathcal{F} is an ultrafilter on X and $E \subseteq X$, then either E or $X \setminus E$ is an element of \mathcal{F} . More precisely, if $X \setminus E$ is not an element of \mathcal{F} , then no element A of \mathcal{F} is contained in $X \setminus E$. This means that every element A of \mathcal{F} intersects E, so that E is an element of \mathcal{F} , as in the preceding paragraph.

Conversely, if \mathcal{F} is a filter on X, and for every $E \subseteq X$, either E or $X \setminus E$ is an element of \mathcal{F} , then one can verify that \mathcal{F} is an ultrafilter on X.

7.3.5 Pushing ultrafilters forward

Let f be a mapping from X into a nonempty set Y. If \mathcal{F} is an ultrafilter on X, then the corresponding filter $f_*(\mathcal{F})$ on Y defined in the previous section is an ultrafilter. To see this, let $E \subseteq Y$ be given, and let us check that E or $Y \setminus E$ is an element of $f_*(\mathcal{F})$. This is the same as saying that $f^{-1}(E)$ or $f^{-1}(Y \setminus E)$ is an element of \mathcal{F} , by (7.2.3). This holds because $f^{-1}(Y \setminus E) = X \setminus f^{-1}(E)$, as in the previous subsection.

7.3.6 Filters and the product topology

Let I be a nonempty set, and let X_j be a nonempty topological space for every $j \in I$. Also let \mathcal{F} be a filter on the Cartesian product $\prod_{j \in I} X_j$, and let x be an element of $\prod_{j \in I} X_j$. One can check that \mathcal{F} converges to x with respect to the product topology on $\prod_{j \in I} X_j$ if and only if $(p_l)_*(\mathcal{F})$ converges to $p_l(x)$ on X_l for every $l \in I$, where p_l is the usual coordinate mapping from $\prod_{j \in I} X_j$ into X_l . Of course, the "only if" part follows from the continuity of p_l .

7.3.7 Tychonoff's theorem using ultrafilters

If $K_j \subseteq X_j$ is compact for every $j \in I$, then Tychonoff's theorem says that $\prod_{j \in I} K_j$ is compact in $\prod_{j \in I} X_j$ with respect to the product topology. To show this, let \mathcal{F} be an ultrafilter on $\prod_{j \in I} X_j$ that contains $\prod_{j \in I} K_j$ as an element. If $l \in I$, then K_l is an element of $(p_l)_*(\mathcal{F})$. This implies that $(p_l)_*(\mathcal{F})$ converges to an element of K_l , because $(p_l)_*(\mathcal{F})$ is an ultrafilter on K_l . One can use this to get that \mathcal{F} converges to an element of $\prod_{j \in I} K_j$.

7.4 Filters and relations

Let I be a nonempty set, and let X_j be a set for each $j \in I$. Also let \mathcal{F} be a filter on I. If $f, g \in \prod_{i \in I} X_j$, then put

$$(7.4.1) f \sim_{\mathcal{F}} g$$

when

$$\{j \in I : f(j) = g(j)\} \in \mathcal{F}.$$

7.4.1 An equivalence relation associated to the filter \mathcal{F}

Note that $f \sim_{\mathcal{F}} f$ automatically, because $I \in \mathcal{F}$. Clearly (7.4.1) is symmetric in f and g. If $h \in \prod_{j \in I} X_j$ too, then

$$(7.4.3) \{j \in I : f(j) = g(j)\} \cap \{j \in I : g(j) = h(j)\} \subseteq \{j \in I : f(j) = h(j)\}.$$

If (7.4.1) holds and $g \sim_{\mathcal{F}} h$, then it follows that $f \sim_{\mathcal{F}} h$.

7.4.2 The corresponding quotient space

Thus (7.4.1) defines an equivalence relation on $\prod_{j \in I} X_j$. Let

(7.4.4)
$$\prod_{j \in I} X_j / \sim_{\mathcal{F}}$$

be the corresponding quotient space of equivalence classes in $\prod_{j\in I} X_j$ with respect to (7.4.1), as in Section 5.6. If $f\in \prod_{j\in I} X_j$, then the equivalence class containing f may be denoted $[f]_{\mathcal{F}}$.

7.4.3 The case where \mathcal{F} is an ultrafilter

If $f, g \in \prod_{i \in I} X_i$, then

$$\{j \in I : f(j) = g(j)\} \cup \{j \in I : f(j) \neq g(j)\} = I.$$

If \mathcal{F} is an ultrafilter on I, then either (7.4.1) holds, or

$$\{j \in I : f(j) \neq g(j)\} \in \mathcal{F}.$$

7.4.4 Some related partial orderings

Let \leq_j be a partial ordering on X_j for every $j \in I$. If $f, g \in \prod_{i \in I} X_j$, then put

$$(7.4.7) f \leq_{\mathcal{F}} g$$

when

$$\{j \in I : f(j) \prec_i g(j)\} \in \mathcal{F}.$$

Of course,

$$(7.4.9) \{ j \in I : f(j) = g(j) \} = \{ j \in I : f(j) \leq_j g(j) \} \cap \{ j \in I : g(j) \leq_j f(j) \},$$

because \leq_j is a partial ordering on X_j for each $j \in I$. This implies that (7.4.1) holds exactly when (7.4.7) and $g \leq_{\mathcal{F}} f$ hold. In particular, $f \leq_{\mathcal{F}} f$ holds automatically.

If $h \in \prod_{j \in I} X_j$ as well, then

(7.4.10)
$$\{j \in I : f(j) \leq_j g(j)\} \cap \{j \in I : g(j) \leq_j h(j)\}$$
$$\subseteq \{j \in I : f(j) \leq_j h(j)\},$$

because \leq_j is transitive on X_j for each $j \in I$. If (7.4.7) holds and $g \leq_{\mathcal{F}} h$, then we get that $f \leq_{\mathcal{F}} h$. This means that $\leq_{\mathcal{F}}$ is transitive as a binary relation on $\prod_{j \in I} X_j$, so that it defines a pre-order on $\prod_{j \in I} X_j$.

Suppose that $\widetilde{f}, \widetilde{g} \in \prod_{j \in I} X_j$ satisfy

$$(7.4.11) f \sim_{\mathcal{F}} \widetilde{f}, \quad q \sim_{\mathcal{F}} \widetilde{q}.$$

Under these conditions, it is easy to see that (7.4.7) holds if and only if $\widetilde{f} \preceq_{\mathcal{F}} \widetilde{g}$. If (7.4.7) holds, then put

$$(7.4.12) [f]_{\mathcal{F}} \preceq_{\mathcal{F}} [g]_{\mathcal{F}}.$$

This does not depend on the particular choices f and g of the equivalence classes, by the previous remark. This defines $\leq_{\mathcal{F}}$ as a binary relation on the quotient space (7.4.4). We use the same notation as for the binary relation (7.4.7) on $\prod_{j \in I} X_j$, for simplicity.

Observe that (7.4.12) is reflexive and transitive on the quotient space (7.4.4), because of the analogous properties of (7.4.7) on $\prod_{j \in I} X_j$.

If (7.4.12) and $[g]_{\mathcal{F}} \preceq_{\mathcal{F}} [f]_{\mathcal{F}}$ hold, then (7.4.7) and $g \preceq_{\mathcal{F}} f$ hold. This implies that (7.4.1) holds, as before, so that $[f]_{\mathcal{F}} = [g]_{\mathcal{F}}$. Thus (7.4.12) defines a partial ordering on the quotient space (7.4.4).

Suppose that X_j is linearly ordered by \leq_j for each $j \in I$. If $f, g \in \prod_{j \in I} X_j$, then it follows that

$$(7.4.13) \{j \in I : f(j) \leq_j g(j)\} \cup \{j \in I : g(j) \leq_j f(j)\} = I.$$

If \mathcal{F} is an ultrafilter on I, then either (7.4.7) holds, or $g \leq_{\mathcal{F}} f$. This means that either (7.4.12) holds, or $[g]_{\mathcal{F}} \leq_{\mathcal{F}} [f]_{\mathcal{F}}$. This implies that the quotient space (7.4.4) is linearly ordered by (7.4.12) in this case.

Chapter 8

Families and coverings

8.1 Coverings and refinements

Let X be a set, and let E be a subset of X. Also let A be a nonempty set, and let U_{α} be a subset of X for every $\alpha \in A$. As usual, $\{U_{\alpha}\}_{{\alpha}\in A}$ is said to be a covering of E in X if $E\subseteq \bigcup_{{\alpha}\in A}U_{\alpha}$. If A_0 is a subset of A and

(8.1.1)
$$E \subseteq \bigcup_{\alpha \in A_0} U_{\alpha},$$

then $\{U_{\alpha}\}_{{\alpha}\in A_0}$ is said to be a *subcovering* of E from $\{U_{\alpha}\}_{{\alpha}\in A}$.

8.1.1 Refinements of families of sets

Let B be another nonempty set, and let $\{V_{\beta}\}_{{\beta}\in B}$ be a family of subsets of X indexed by B. We say that $\{V_{\beta}\}_{{\beta}\in B}$ is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$ if for every ${\beta}\in B$ there is an ${\alpha}\in A$ such that

$$(8.1.2) V_{\beta} \subseteq U_{\alpha}.$$

Of course, this implies that

(8.1.3)
$$\bigcup_{\beta \in B} V_{\beta} \subseteq \bigcup_{\alpha \in A} U_{\alpha}.$$

If $\{U_{\alpha}\}_{{\alpha}\in A}$ covers E, then we may be particularly interested in refinements that cover E too.

If $A_0 \subseteq A$, then $\{U_\alpha\}_{\alpha \in A_0}$ may be considered as a refinement of $\{U_\alpha\}_{\alpha \in A}$.

8.1.2 Refinements of refinements

Let C be another nonempty set, and let $\{W_{\gamma}\}_{{\gamma}\in C}$ be a family of subsets of X indexed by C. If $\{V_{\beta}\}_{{\beta}\in B}$ is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$, and $\{W_{\gamma}\}_{{\gamma}\in C}$ is a refinement of $\{V_{\beta}\}_{{\beta}\in B}$, then it is easy to see that $\{W_{\gamma}\}_{{\gamma}\in C}$ is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$.

8.1.3 Some subfamilies associated to refinements

Suppose that $\{V_{\beta}\}_{{\beta}\in B}$ is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$ again. Thus, for each ${\beta}\in B$, we can choose an ${\alpha}({\beta})\in A$ such that $V_{\beta}\subseteq U_{{\alpha}({\beta})}$. Let

$$(8.1.4) A_1 = \{\alpha(\beta) : \beta \in B\}$$

be the set of indices in A that have been chosen in this way. By construction, $A_1 \subseteq A$, and

(8.1.5)
$$\bigcup_{\beta \in B} V_{\beta} \subseteq \bigcup_{\beta \in B} U_{\alpha(\beta)} = \bigcup_{\alpha \in A_1} U_{\alpha}.$$

If $\{V_{\beta}\}_{{\beta}\in B}$ covers E, then $\{U_{\alpha}\}_{{\alpha}\in A}$ covers E, and $\{U_{\alpha}\}_{{\alpha}\in A_1}$ is a subcovering of E from $\{U_{\alpha}\}_{{\alpha}\in A}$.

We also have that

$$(8.1.6) #A_1 \le #B.$$

In particular, if B has only finitely many elements, or only finitely or countably many elements, then A_1 has the same property.

8.1.4 Some common refinements

Let $\{U_{\alpha}\}_{{\alpha}\in A}$ and $\{V_{\beta}\}_{{\beta}\in B}$ be arbitrary families of subsets of X, not necessarily related by refinement. It is easy to see that

$$(8.1.7) \{U_{\alpha} \cap V_{\beta}\}_{(\alpha,\beta) \in A \times B}$$

is a refinement of both $\{U_{\alpha}\}_{{\alpha}\in A}$ and $\{V_{\beta}\}_{{\beta}\in B}$. Observe that

(8.1.8)
$$\left(\bigcup_{\alpha \in A} U_{\alpha}\right) \cap \left(\bigcup_{\beta \in B} V_{\beta}\right) = \bigcup_{\alpha \in A} \bigcup_{\beta \in B} (U_{\alpha} \cap V_{\beta}).$$

If $\{U_{\alpha}\}_{{\alpha}\in A}$ and $\{V_{\beta}\}_{{\beta}\in B}$ cover E, then it follows that (8.1.7) covers E too.

8.2 Open coverings

Let X be a topological space, and let \mathcal{B} be a base for the topology of X. Also let $\{U_{\alpha}\}_{{\alpha}\in A}$ be a family of open subsets of X. If ${\alpha}\in A$, then put

(8.2.1)
$$\mathcal{B}_{\alpha} = \{ V \in \mathcal{B} : V \subseteq U_{\alpha} \}.$$

Thus

(8.2.2)
$$U_{\alpha} = \bigcup_{V \in \mathcal{B}_{\alpha}} V,$$

because \mathcal{B} is a base for the topology of X.

Put

(8.2.3)
$$\widetilde{\mathcal{B}} = \bigcup_{\alpha \in A} \mathcal{B}_{\alpha},$$

so that

(8.2.4)
$$\bigcup_{V \in \widetilde{\mathcal{B}}} V = \bigcup_{\alpha \in A} \bigcup_{V \in \mathcal{B}_{\alpha}} V = \bigcup_{\alpha \in A} U_{\alpha}.$$

Note that \widetilde{B} is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$, by construction. If $\{U_{\alpha}\}_{{\alpha}\in A}$ is an open covering of a subset E of X, then $\widetilde{\mathcal{B}}$ is an open covering of E in X too.

8.2.1 Another look at Lindelöf's theorem

This type of refinement was already used in the proof of Lindelöf's theorem, Theorem 3.6.6. More precisely, if $\{U_{\alpha}\}_{{\alpha}\in A}$ is any nonempty family of open subsets of X, then $\{U_{\alpha}\}_{{\alpha}\in A}$ may be considered as a covering of its union $E = \bigcup_{{\alpha}\in A} U_{\alpha}$.

As in the previous section, there is a subset A_1 of A that satisfies (8.1.5) and (8.1.6). In this situation, (8.1.5) implies that

(8.2.5)
$$\bigcup_{\alpha \in A_1} U_{\alpha} = \bigcup_{\alpha \in A} U_{\alpha},$$

because the left side is automatically contained in the right side. If \mathcal{B} has only finitely or countably many elements, then (8.1.6) implies that A_1 has only finitely or countably many elements, as before.

8.2.2 Another look at compactness with respect to a base

This type of refinement was used in the proof of Proposition 3.9.2 as well. In that case, compactness of $E \subseteq X$ with respect to \mathcal{B} implies that there is a finite subcover of E from $\widetilde{\mathcal{B}}$. This finite subcovering from $\widetilde{\mathcal{B}}$ may be considered as a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$, which leads to a finite subcovering of E from $\{U_{\alpha}\}_{{\alpha}\in A}$, as before.

8.2.3 A theorem of Lebesgue

Let d(x,y) be a semimetric on X, and suppose now that X is equipped with the topology determined by $d(\cdot,\cdot)$. Suppose that $E\subseteq X$ is compact, and let $\{U_{\alpha}\}_{{\alpha}\in A}$ be an open covering of E in X.

A well-known result going back to Lebesgue implies that there is a positive real number r such that for each $x \in E$ there is an $\alpha \in A$ with

$$(8.2.6) B(x,r) \subseteq U_{\alpha}.$$

This means that the covering of E by the open balls B(x,r) in X centered at elements of E with radius r is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$.

8.2.4 The proof of Lebesgue's theorem

To see this, let $x \in E$ be given, so that there is an index $\alpha_0(x) \in A$ such that $x \in U_{\alpha_0(x)}$. This implies that there is a positive real number $r_0(x)$ such that

$$(8.2.7) B(x, r_0(x)) \subseteq U_{\alpha_0(x)},$$

because $U_{\alpha_0(x)}$ is an open set in X.

Consider the covering of E by open balls of the form $B(x, r_0(x)/2)$, where $x \in E$ and $r_0(x) > 0$ is as before. If E is compact, then there are finitely many elements x_1, \ldots, x_n of E such that

(8.2.8)
$$E \subseteq \bigcup_{j=1}^{n} B(x_j, r_0(x_j)/2).$$

Put

(8.2.9)
$$r = \min_{1 \le j \le n} r_0(x_j)/2.$$

If $x \in E$, then $x \in B(x_j, r_0(x_j)/2)$ for some $j = 1, \ldots, n$, by (8.2.8). This implies that

$$(8.2.10) B(x,r) \subseteq B(x_j, r_0(x_j)/2 + r) \subseteq B(x_j, r_0(x_j)) \subseteq U_{\alpha_0(x_j)},$$

using the triangle inequality in the first step.

8.3 Paracompactness

Let X be a topological space.

8.3.1 Paracompactness in the strict sense

Let us say that X is paracompact in the strict sense if for every open covering $\{U_{\alpha}\}_{{\alpha}\in A}$ of X there is a refinement $\{V_{\beta}\}_{{\beta}\in B}$ of $\{U_{\alpha}\}_{{\alpha}\in A}$ such that $\{V_{\beta}\}_{{\beta}\in B}$ is a locally finite open covering of X.

If X is compact, then X is paracompact in the strict sense.

8.3.2 Paracompactness in the strong sense

Let us say that X is paracompact in the strong sense if X is paracompact in the strict sense and X is Hausdorff. One may simply say that X is paracompact in this case, but this term is sometimes used for paracompactness in the strict sense. If X is equipped with the discrete topology, then it is easy to see that X is paracompact in the strong sense.

8.3.3 Another version of paracompactness

Paracompactness of X is defined on p156 of [240] to mean that X is both paracompact in the strict sense and regular in the strict sense. If X is paracompact in the strong sense, then it is well known that X is regular, as we shall see in a moment.

8.3.4 A paracompactness property of closed sets in X

Let E be a closed set in X, and let $\{U_{\alpha}\}_{{\alpha}\in A}$ be an open covering of E in X. Thus $X\setminus E$ is an open set in X, and the collection of open subsets of X consisting of the U_{α} 's, ${\alpha}\in A$, together with $X\setminus E$, is an open covering of X.

If X is paracompact in the strict sense, then there is an open covering $\{V_{\beta}\}_{{\beta}\in B}$ of X that is a refinement of the collection of U_{α} 's, $\alpha\in A$, and $X\setminus E$. Put

$$(8.3.1) B_0 = \{ \beta \in B : V_\beta \cap E \neq \emptyset \},$$

and observe that $\{V_{\beta}\}_{\beta \in B_0}$ is an open covering of E in X. It is easy to see that $\{V_{\beta}\}_{\beta \in B_0}$ is locally finite in X, because $\{V_{\beta}\}_{\beta \in B}$ if locally finite in X.

If $\beta \in B$, then either $V_{\beta} \subseteq X \setminus E$, or there is an $\alpha \in A$ such that $V_{\beta} \subseteq U_{\alpha}$, because $\{V_{\beta}\}_{\beta \in B}$ is a refinement of the collection of U_{α} 's, $\alpha \in A$, and $X \setminus E$. If $\beta \in B_0$, then it follows that there is an $\alpha \in A$ such that $V_{\beta} \subseteq U_{\alpha}$, because $V_{\beta} \cap E \neq \emptyset$. This means that $\{V_{\beta}\}_{\beta \in B_0}$ is a refinement of $\{U_{\alpha}\}_{\alpha \in A}$.

8.3.5 Paracompactness and induced topologies

In particular, one can use this to check that E is paracompact, with respect to the induced topology. More precisely, any open covering of E by relatively open subsets of E can be obtained from an open covering of E in X, by taking intersections of open subsets of X with E to get relatively open subsets of E.

The remarks in the preceding subsection imply that an open covering of E in X has a locally finite refinement which is an open covering of E in X too. One can take the intersections of the open subsets of X in this refinement with E, to get a refinement of the initial covering of E by relatively open subsets of E that is locally finite in E and a covering of E by relatively open subsets of itself.

8.3.6 Paracompactness in the strong sense implies regularity

Suppose that X is paracompact in the strong sense, and let us verify that X is regular. Let E be a closed set in X, and let x be an element of X not in E.

If $y \in E$, then there is an open subset U_y of X such that $y \in U_y$ and $x \notin \overline{U_y}$, because X is Hausdorff. Thus the collection of open subsets of X that do not contain x in their closures is an open covering of E in X.

It follows that there is a locally finite open covering $\{W_{\gamma}\}_{{\gamma}\in C}$ of E in X that is a refinement of the previous open covering of E, because X is paracompact in the strict sense, as before.

If $\gamma \in C$, then $x \notin \overline{W_{\gamma}}$, because W_{γ} is contained in a subset of X whose closure does not contain x. This implies that

$$(8.3.2) x \notin \overline{\left(\bigcup_{\gamma \in C} W_{\gamma}\right)},$$

as in Subsection 5.13.2. This is equivalent to what we want for regularity, because $\bigcup_{\gamma \in C} W_{\gamma}$ is an open subset of X that contains E.

8.3.7 Paracompactness and normality

Similarly, if X is paracompact in the strict sense and regular in the strict sense, then it is well known that X is normal in the strict sense. To see this, let E be a closed set in X, and let W be an open subset of X that contains E.

Every element of E is contained in an open subset of X whose closure is contained in W, because X is regular in the strict sense. This means that the collection of open subsets of X whose closures are contained in W is an open covering of E in X.

If X is paracompact in the strict sense, then there is locally finite refinement $\{V_{\beta}\}_{{\beta}\in B}$ of the open covering of E just mentioned that is an open covering of E too, as before.

If $\beta \in B$, then $\overline{V_{\beta}} \subseteq W$, because V_{β} is contained in a subset of X whose closure is contained in W. This implies that

(8.3.3)
$$\overline{\left(\bigcup_{\beta \in B} V_{\beta}\right)} = \bigcup_{\beta \in B} \overline{V_{\beta}} \subseteq W,$$

where the first step is as in Subsection 5.13.2. Thus $\bigcup_{\beta \in B} V_{\beta}$ is an open subset of X that contains E and whose closure is contained in W, as desired.

8.3.8 Some criteria for paracompactness

It is well known and not too difficult to show that if a locally compact Hausdorff space X is σ -compact, then X is paracompact. However, more precise results are known, and in particular X is paracompact in the strict sense when X is regular in the strict sense and X has the Lindelöf property. It is also well known that X is paracompact in the strict sense when the topology on X is determined by a semimetric.

8.4 Closed refinements

Let X be a topological space again.

8.4.1 A variant of paracompactness

Consider the following condition:

(8.4.1) if $\{U_{\alpha}\}_{{\alpha}\in A}$ is an open covering of X, then there is a locally finite covering $\{E_{\beta}\}_{{\beta}\in B}$ of X that is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$.

Paracompactness in the strict sense is the same as saying that (8.4.1) holds, with E_{β} an open subset of X for every $\beta \in B$. In particular, paracompactness in the strict sense implies (8.4.1) automatically.

If X is regular in the strict sense, then it is well known that (8.4.1) implies that X is paracompact in the strict sense.

8.4.2 Another variant of paracompactness

A family $\{E_{\beta}\}_{{\beta}\in B}$ of subsets of X is said to be *closed* if E_{β} is a closed set in X for every ${\beta}\in B$. Consider the following variant of (8.4.1):

(8.4.2) if $\{U_{\alpha}\}_{{\alpha}\in A}$ is an open covering of X, then there is a locally finite closed covering $\{E_{\beta}\}_{{\beta}\in B}$ of X that is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$.

This condition obviously implies (8.4.1).

8.4.3 Using the first variant

Suppose that X is regular in the strict sense and satisfies (8.4.1), and let us check that (8.4.2) holds. Let $\{U_{\alpha}\}_{{\alpha}\in A}$ be an open covering of X. Let $x\in X$ be given, so that $x\in U_{\alpha}$ for some $\alpha\in A$. Because X is regular in the strict sense, there is an open subset W of X such that $x\in W$ and $\overline{W}\subseteq U_{\alpha}$. Thus the collection of open subsets W of X such that $\overline{W}\subseteq U_{\alpha}$ for some $\alpha\in A$ is an open covering of X.

By hypothesis, there is a locally finite covering $\{E_{\beta}\}_{\beta \in B}$ of X that is a refinement of the covering of X just mentioned. This implies that for each $\beta \in B$ there is an open subset W of X and an $\alpha \in A$ such that $E_{\beta} \subseteq W$ and $\overline{W} \subseteq U_{\alpha}$. Of course, this means that $\overline{E_{\beta}} \subseteq \overline{W} \subseteq U_{\alpha}$, so that $\{\overline{E_{\beta}}\}_{\beta \in B}$ is a refinement of $\{U_{\alpha}\}_{\alpha \in A}$. We also have that $\{\overline{E_{\beta}}\}_{\beta \in B}$ is locally finite in X, because $\{E_{\beta}\}_{\beta \in B}$ is locally finite in X, as in Subsection 5.13.2.

8.4.4 Using the second variant

Let $\{U_{\alpha}\}_{{\alpha}\in A}$ be an open covering of X again, and suppose that $\{E_{\beta}\}_{{\beta}\in B}$ is a locally finite closed covering of X that is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$. Thus, for each ${\beta}\in B$, we can choose ${\alpha}({\beta})\in A$ such that

$$(8.4.3) E_{\beta} \subseteq U_{\alpha(\beta)}.$$

Let $x \in X$ be given, so that there is an open subset W(x) of X such that $x \in W(x)$ and $W(x) \cap E_{\beta} = \emptyset$ for all but finitely many $\beta \in B$, because $\{E_{\beta}\}_{\beta \in B}$ is locally finite at x, by hyothesis. Put

$$(8.4.4) B(x) = \{ \beta \in B : x \in E_{\beta} \}$$

and

$$(8.4.5) B_1(x) = \{ \beta \in B : W(x) \cap E_\beta \neq \emptyset \},$$

so that $B_1(x)$ is a finite subset of B that contains B(x).

$$(8.4.6) W_1(x) = W(x) \cap \Big(\bigcap_{\beta \in B(x)} U_{\alpha(\beta)}\Big) \cap \Big(\bigcap_{\beta \in B_1(x) \setminus B(x)} (X \setminus E_\beta)\Big),$$

where the intersection over $\beta \in B_1(x) \setminus B(x)$ can be omitted when $B(x) = B_1(x)$. Of course, $B(x) \neq \emptyset$, because $\{E_\beta\}_{\beta \in B}$ covers X.

Note that $x \in W_1(x)$, by construction. Clearly $W_1(x)$ is an open set in X, because it is the intersection of finitely many open sets. If $\beta \in B \setminus B(x)$, then

$$(8.4.7) W_1(x) \cap E_{\beta} = \emptyset,$$

because β is an element of $B \setminus B_1(x)$ or $B_1(x) \setminus B(x)$.

8.5 Star refinements

Let X be a set, let C be a nonempty set, and let W_{γ} be a subset of X for each $\gamma \in C$. Let $x \in X$ be given, and put

(8.5.1)
$$C(x) = \{ \gamma \in C : x \in W_{\gamma} \}.$$

8.5.1 The star of x with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$

The star of x with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$ is defined to be

(8.5.2)
$$\bigcup_{\gamma \in C(x)} W_{\gamma}.$$

This union is interpreted as being the empty set when $C(x) = \emptyset$. If $\{W_{\gamma}\}_{{\gamma} \in C}$ covers X, then $C(x) \neq \emptyset$ for every $x \in X$.

8.5.2 Star refinements of a family of subsets of X

Let $\{U_{\alpha}\}_{{\alpha}\in A}$ be another nonempty family of subsets of X.

We say that $\{W_\gamma\}_{\gamma\in C}$ is a star refinement of $\{U_\alpha\}_{\alpha\in A}$ if for each $x\in X$ there is an $\alpha\in A$ such that the star of x with respect to $\{W_\gamma\}_{\gamma\in C}$ is contained in U_α . Equivalently, this means that the collection of subsets of X that are stars of elements of X with respect to $\{W_\gamma\}_{\gamma\in C}$ is a refinement of $\{U_\alpha\}_{\alpha\in A}$. It is easy to see that $\{W_\gamma\}_{\gamma\in C}$ is a refinement of $\{U_\alpha\}_{\alpha\in A}$ in this case, because W_γ is contained in the star of each of its elements.

8.5.3 Fully normal spaces in the strict sense

A topological space X is said to be fully normal in the strict sense if for every open covering $\{U_{\alpha}\}_{{\alpha}\in A}$ of X there is an open covering $\{W_{\gamma}\}_{{\gamma}\in C}$ that is a star refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$.

8.5.4 A criterion for full normality in the strict sense

If X satisfies (8.4.2), then it is well known that X is fully normal in the strict sense. More precisely, let $\{U_{\alpha}\}_{\alpha\in A}$ be an open covering of X, and let $\{E_{\beta}\}_{\beta\in B}$ be a locally finite closed covering of X that is a refinement of $\{U_{\alpha}\}_{\alpha\in A}$. If $x\in X$, then let B(x) and $W_1(x)$ be as in the previous section. The family of sets $W_1(x)$, $x\in X$, is an open covering of X, and we would like to check that it is a star refinement of $\{U_{\alpha}\}_{\alpha\in A}$.

8.5.5 Checking that $W_1(x)$, $x \in X$, is a star refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$

Let $\beta_0 \in B$ be given. If $x \in X$ and $W_1(x) \cap E_{\beta_0} \neq \emptyset$, then $\beta_0 \in B(x)$, as before. This implies that

$$(8.5.3) W_1(x) \subseteq U_{\alpha(\beta_0)},$$

by the definition of $W_1(x)$.

Let $y \in X$ be given, and let β_0 be an element of B such that $y \in E_{\beta_0}$. If $x \in X$ and $y \in W_1(x)$, then $W_1(x) \cap E_{\beta_0} \neq \emptyset$, so that (8.5.3) holds. It follows that the star of y with respect to the family of sets $W_1(x)$, $x \in X$, is contained in $U_{\alpha(\beta_0)}$, as desired.

8.5.6 Normality from full normality

Suppose that X is fully normal in the strict sense, and let us check that X is normal in the strict sense. Let A and B be disjoint closed subsets of X, so that their complements $X \setminus A$, $X \setminus B$ form an open covering of X.

By hypothesis, there is an open covering $\{W_{\gamma}\}_{{\gamma}\in C}$ of X that is a star refinement of $\{X\setminus A, X\setminus B\}$. Put

(8.5.4)
$$U = \bigcup \{W_{\gamma} : \gamma \in C, \ W_{\gamma} \cap A \neq \emptyset\}$$

and

(8.5.5)
$$V = \bigcup \{W_{\gamma} : \gamma \in C, \ W_{\gamma} \cap B \neq \emptyset\}.$$

These are open subsets of X, with $A \subseteq U$ and $B \subseteq V$.

Suppose for the sake of a contradiction that there is a point x in $U \cap V$. This means that there are $\gamma_1, \gamma_2 \in C$ such that $x \in W_{\gamma_1} \cap W_{\gamma_2}$, $W_{\gamma_1} \cap A \neq \emptyset$, and $W_{\gamma_2} \cap B \neq \emptyset$. Thus W_{γ_1} and W_{γ_2} are both contained in the star of x with respect to $\{W_{\gamma}\}_{{\gamma} \in C}$.

It follows that the star of x with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$ is not contained in either $X\setminus A$ or $X\setminus B$. This contradicts the hypothesis that $\{W_{\gamma}\}_{{\gamma}\in C}$ be a star refinement of $\{X\setminus A, X\setminus B\}$.

8.5.7 Fully normal spaces in the strong sense

Let us say that X is fully normal in the strong sense if X is fully normal in the strict sense and X satisfies the first separation condition. One may say that that X is fully normal when X is fully normal in the strict sense, and that X is fully T_4 when T_4 is if fully normal in the strong sense, but the opposite convention may be used as well.

8.6 Using another refinement

Let X be a set, and let $\{W_{\gamma}\}_{{\gamma}\in C}$ be a nonempty family of subsets of X. If E is a subset of X, then put

(8.6.1)
$$C(E) = \{ \gamma \in C : W_{\gamma} \cap E \neq \emptyset \}.$$

8.6.1 The star of E with respect to $\{W_{\gamma}\}_{\gamma inC}$

The star of E with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$ is defined to be

(8.6.2)
$$\bigcup_{\gamma \in C(E)} W_{\gamma},$$

which is interpreted as being the empty set when $C(E) = \emptyset$. Of course, if $\{W_{\gamma}\}_{{\gamma} \in C}$ covers X, then $C(E) \neq \emptyset$ when $E \neq \emptyset$. In this case, E is contained in its star with respect to $\{W_{\gamma}\}_{{\gamma} \in C}$.

If $x \in X$, then $C(\{x\})$ is the same as C(x) in the previous section, and the star of $\{x\}$ with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$ is the same as the star of x. Observe that C(E) is the same as the union of C(x) over $x \in E$, so that the star of E with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$ is the same as the union of the stars of the elements of E.

8.6.2 The star of the star of $x \in X$ with respect to $\{W_{\gamma}\}_{{\gamma} \in C}$

The star of $x \in X$ with respect to $\{W_{\gamma}\}_{{\gamma} \in C}$ is the set of $y \in X$ for which there is a ${\gamma} \in C$ such that $x, y \in W_{\gamma}$. This condition is symmetric in x and y, so that y is in the star of x if and only if x is in the star of y.

The star of the star of $x \in X$ with respect to $\{W_{\gamma}\}_{{\gamma} \in C}$ is the union of the stars of the elements y of the star of x. This is the same as the union of the stars of $y \in X$ such that x is in the star of y, as in the preceding paragraph.

8.6.3 Star refinements of star refinements

Let $\{U_{\alpha}\}_{\alpha\in A}$, $\{V_{\beta}\}_{\beta\in B}$, and $\{W_{\gamma}\}_{\gamma\in C}$ be nonempty families of subsets of X. Suppose that $\{V_{\beta}\}_{\beta\in B}$ is a star refinement of $\{U_{\alpha}\}_{\alpha\in A}$, and that $\{W_{\gamma}\}_{\gamma\in C}$ is a star refinement of $\{V_{\beta}\}_{\beta\in B}$.

Let $x \in X$ be given, and suppose that $y \in X$ is an element of the star of x with respect to $\{W_{\gamma}\}_{{\gamma} \in C}$. Equivalently, this means that x is an element of the star of y with respect to $\{W_{\gamma}\}_{{\gamma} \in C}$, as before.

in this situation.

Because $\{W_{\gamma}\}_{{\gamma}\in C}$ is a star refinement of $\{V_{\beta}\}_{{\beta}\in B}$, for each such y there is a ${\beta}(y)\in B$ such that

(8.6.3) $V_{\beta(y)}$ contains the star of y with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$.

It follows that $x \in V_{\beta(y)}$ for these points y, so that

(8.6.4) $V_{\beta(y)}$ is contained in the star of x with respect to $\{V_{\beta}\}_{{\beta}\in B}$.

Because $\{V_{\beta}\}_{{\beta}\in B}$ is a star refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$, there is an ${\alpha}\in A$ such that U_{α} contains the star of x with respect to $\{V_{\beta}\}_{{\beta}\in B}$. This means that

(8.6.5) U_{α} contains the star of the star of x with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$

8.6.4 Using full normality in the strict sense

Now let X be a topological space, and let $\{E_{\beta}\}_{{\beta}\in B}$ be a locally finite family of subsets of X. Thus, for each $x\in X$, there is an open subset V(x) of X such that $x\in V(x)$ and $V(x)\cap E_{\beta}=\emptyset$ for all but finitely many $\beta\in B$. Of course, the family of such open sets V(x) is an open covering of X.

Suppose that X is fully normal in the strict sense, so that this open covering has a star refinement that is an open covering of X. Using full normality again, we can get an open covering $\{W_{\gamma}\}_{{\gamma}\in C}$ of X that is a star refinement of the previous open covering of X that is a star refinement of the family of V(x)'s, $x\in X$.

8.6.5 The star of E_{β} with resect to $\{W_{\gamma}\}_{{\gamma}\in C}$

If $\beta \in B$, then let Z_{β} be the star of E_{β} with respect to $\{W_{\gamma}\}_{\gamma \in C}$. Note that

$$(8.6.6) E_{\beta} \subseteq Z_{\beta},$$

and that Z_{β} is an open set in X.

Let $x_0 \in X$ be given, and let γ_0 be an element of C such that W_{γ_0} contains x_0 . Suppose that

$$(8.6.7) W_{\gamma_0} \cap Z_{\beta} \neq \emptyset,$$

and let y be an element of $W_{\gamma_0} \cap Z_{\beta}$. By definition of Z_{β} , there is a $\gamma \in C$ such that $W_{\gamma} \cap E_{\beta} \neq \emptyset$ and $y \in W_{\gamma}$.

Thus y is in the star of x_0 with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$, and E_{β} intersects the star of the star of x_0 with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$.

8.6.6 Local finiteness of $\{Z_{\beta}\}_{{\beta}\in B}$

Because $\{W_{\gamma}\}_{{\gamma}\in C}$ is a star refinement of a star refinement of the family of V(x)'s, $x\in X$, there is an $x_1\in X$ such that the star of the star of x_0 with respect to $\{W_{\gamma}\}_{{\gamma}\in C}$ is contained in $V(x_1)$, as before.

Remember that $V(x_1) \cap E_{\beta} \neq \emptyset$ for all but finitely many $\beta \in B$, so that the star of the star of x_0 with respect to $\{W_{\gamma}\}_{{\gamma} \in C}$ intersects E_{β} for only finitely many $\beta \in B$. This implies that (8.6.7) holds for only finitely many $\beta \in B$. This means that $\{Z_{\beta}\}_{{\beta} \in B}$ is locally finite in X as well.

8.6.7 Staring with an open covering of X

Let $\{U_{\alpha}\}_{{\alpha}\in A}$ be an open covering of X, and suppose that $\{E_{\beta}\}_{{\beta}\in B}$ is also a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$. Thus, for each ${\beta}\in B$, we can choose ${\alpha}({\beta})\in A$ such that $E_{\beta}\subseteq U_{{\alpha}({\beta})}$. If Z_{β} is as before, then

$$(8.6.8) Z_{\beta} \cap U_{\alpha(\beta)}$$

is an open subset of X that contains E_{β} and is contained in $U_{\alpha(\beta)}$.

Of course, the family of sets (8.6.8), with $\beta \in B$, is locally finite in X, because $\{Z_{\beta}\}_{{\beta}\in B}$ is locally finite in X. If $\{E_{\beta}\}_{{\beta}\in B}$ covers X, then the family of sets (8.6.8), with $\beta \in B$, covers X as well.

8.6.8 Using this to get paracompactness in the strict sense

Suppose that every open covering $\{U_{\alpha}\}_{\alpha\in A}$ of X has a locally finite refinement $\{E_{\beta}\}_{\beta\in B}$ that covers X, as in (8.4.1). If X is fully normal in the strict sense, then we get a locally finite open covering of X that is a refinement of $\{U_{\alpha}\}_{\alpha\in A}$, as in the previous subsection. This means that X is paracompact in the strict sense.

If every open covering of X has a locally finite closed refinement that covers X, as in (8.4.2), then X satisfies (8.4.1) and is fully normal in the strict sense, as in the previous sections.

If X is regular in the strict sense and satisfies (8.4.1), then we have seen that X satisfies (8.4.2), so that X is fully normal in the strict sense, and paracompact in the strict sense.

8.7 σ -Local finiteness

A family $\{E_{\beta}\}_{{\beta}\in B}$ of subsets of a topological space X is said to be σ -locally finite in X if there is a sequence B_1, B_2, B_3, \ldots of subsets of B such that

$$(8.7.1) B = \bigcup_{j=1}^{\infty} B_j$$

and $\{E_{\beta}\}_{{\beta}\in B_j}$ is locally finite in X for every $j\geq 1$. Of course, if $\{E_{\beta}\}_{{\beta}\in B}$ is locally finite in X, then $\{E_{\beta}\}_{{\beta}\in B}$ is σ -locally finite in X.

8.7.1 σ -Locally finite refinements

Consider the following condition:

(8.7.2) if $\{U_{\alpha}\}_{{\alpha}\in A}$ is an open covering of X, then there is a σ -locally finite open covering $\{V_{\beta}\}_{{\beta}\in B}$ of X that is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$.

If X is paracompact in the strict sense, then X satisfies (8.7.2), because local finiteness implies σ -local finiteness. If X satisfies (8.7.2), then it is well known that X satisfies (8.4.1), which is to say that every open covering of X has a locally finite refinement that covers X.

8.7.2 Using σ -locally finite refinements

To see this, let $\{U_{\alpha}\}_{{\alpha}\in A}$ be an open covering of X, so that there is a σ -locally finite open covering $\{V_{\beta}\}_{{\beta}\in B}$ of X that is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$. Let B_1, B_2, B_3, \ldots be a sequence of subsets of B whose union is B such that $\{V_{\beta}\}_{{\beta}\in B_j}$ is locally finite in X for each $j\geq 1$. We may as well suppose that the B_j 's are pairwise-disjoint, by replacing B_j with $B_j\setminus \left(\bigcup_{l=1}^{j-1}B_l\right)$ when $j\geq 2$.

8.7.3 A subset E_{β} of V_{β}

If $\beta \in B$, then put $E_{\beta} = V_{\beta}$ when $\beta \in B_1$, and

(8.7.3)
$$E_{\beta} = V_{\beta} \setminus \left(\bigcup_{l=1}^{j-1} \bigcup_{\gamma \in B_l} V_{\gamma} \right)$$

when $\beta \in B_j$ for some $j \geq 2$. Thus

$$(8.7.4) E_{\beta} \subseteq V_{\beta}$$

for every $\beta \in B$. This implies that $\{E_{\beta}\}_{{\beta}\in B}$ is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$, because $\{V_{\beta}\}_{{\beta}\in B}$ is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$, by hypothesis.

8.7.4 The E_{β} 's, $\beta \in B$, cover X

Let $x \in X$ be given, so that $x \in \bigcup_{\beta \in B} V_{\beta} = \bigcup_{j=1}^{\infty} \bigcup_{\beta \in B_{j}} V_{\beta}$. Let $j_{0}(x)$ be the smallest positive integer j such that $x \in \bigcup_{\beta \in B_{j}} V_{\beta}$, and let $\beta_{0}(x)$ be an element of $B_{j_{0}(x)}$ such that $x \in V_{\beta_{0}(x)}$. Observe that

$$(8.7.5) x \in E_{\beta_0(x)},$$

because $j_0(x)$ is as small as possible. Thus $\{E_\beta\}_{\beta\in B}$ covers X. If $\beta\in B_l$ for some $l\geq j_0(x)+1$, then

$$(8.7.6) E_{\beta} \cap V_{\beta_0(x)} = \emptyset,$$

by the definition of E_{β} .

8.7.5 Local finiteness of $\{E_{\beta}\}_{{\beta}\in B}$

If $1 \leq l \leq j_0(x)$, then there is an open subset W_l of X such that $x \in W_l$ and $W_l \cap V_\beta = \emptyset$ for all but finitely many $\beta \in B_l$, because $\{V_\beta\}_{\beta \in B_l}$ is locally finite in X. Put

(8.7.7)
$$W = \left(\bigcap_{l=1}^{j_0(x)} W_l\right) \cap V_{\beta_0(x)},$$

so that W is an open subset of X that contains x. It is easy to see that $W \cap E_{\beta} = \emptyset$ for all but finitely many $\beta \in B$. This means that $\{E_{\beta}\}_{\beta \in B}$ is locally finite in X, as desired.

If X satisfies (8.7.2) and X is regular in the strict sense, then it follows that X is paracompact in the strict sense, as in the previous section.

8.8 Semimetrics and full normality

Let X be a set with a semimetric d(x, y), and let us show that X is fully normal in the strict sense with respect to the topology determined by $d(\cdot, \cdot)$.

8.8.1 Some collections of open balls in X

Let $\{U_{\alpha}\}_{{\alpha}\in A}$ be an arbitrary open covering of X. Consider the collection of open balls in X of the form B(w,r), where $w\in X, 0< r\leq 1$, and

$$(8.8.1) B(w,5r) \subseteq U_{\alpha}$$

for some $\alpha \in A$. It is easy to see that the collection of these open balls is an open covering of X, and we would like to verify that it is a star refinement of $\{U_{\alpha}\}_{\alpha \in A}$.

Let $x \in X$ be given, so that we would like to check that the star of x with respect to this collection of open balls is contained in U_{α} for some $\alpha \in A$. In this situation, the star of x is the union of the open balls of the form B(w,r), where $w \in X$, $0 < r \le 1$, (8.8.1) holds for some $\alpha \in A$, and $x \in B(w,r)$. In particular, there are $w_0 \in X$ and $r_0 \in (0,1]$ with these properties, and so that r_0 is strictly larger that one-half the supremum of the set of $r \in (0,1]$ that occur in this way.

Suppose that $w \in X$ and $r \in (0,1]$ have the properties mentioned in the preceding paragraph. Thus

$$(8.8.2) r < 2 r_0,$$

by construction. Observe that

$$(8.8.3) d(w_0, w) \le d(w_0, x) + d(x, w) < r_0 + r < 3r_0,$$

because x is an element of $B(w_0, r_0)$ and B(w, r). This implies that

$$(8.8.4) B(w,r) \subseteq B(w_0, 3r_0 + r) \subseteq B(w_0, 5r_0).$$

It follows that the star of x with respect to this collection of open balls is contained in $B(w_0, 5r_0)$. By construction, $B(w_0, 5r_0)$ is contained in U_{α_0} for some $\alpha_0 \in A$. This means that this collection of open balls is a star refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$, as desired.

8.9 Discrete families of sets

Let $\{E_{\beta}\}_{{\beta}\in B}$ be a family of subsets of a topological space X. Let us say that $\{E_{\beta}\}_{{\beta}\in B}$ is discrete at a point $x\in X$ if there is an open subset V of X such that $x\in V$ and

$$(8.9.1) E_{\beta} \cap V = \emptyset$$

for all but at most one $\beta \in B$. Of course, this implies that $\{E_{\beta}\}_{{\beta} \in B}$ is locally finite at x

If $\{E_{\beta}\}_{{\beta}\in B}$ is discrete at every $x\in X$, then $\{E_{\beta}\}_{{\beta}\in B}$ is said to be discrete in X. This implies that $\{E_{\beta}\}_{{\beta}\in B}$ is locally finite in X, and that the E_{β} 's are pairwise disjoint.

Note that (8.9.1) implies that

$$(8.9.2) \overline{E_{\beta}} \cap V = \emptyset,$$

because V is an open set. If $\{E_{\beta}\}_{{\beta}\in B}$ is discrete at $x\in X$, then $\{\overline{E_{\beta}}\}_{{\beta}\in B}$ is discrete at x as well. If $\{E_{\beta}\}_{{\beta}\in B}$ is discrete in X, then $\{\overline{E_{\beta}}\}_{{\beta}\in B}$ is discrete in X too.

8.9.1 Discrete families and induced topologies

Let X_0 be a subset of X. If $\{E_\beta\}_{\beta\in B}$ is discrete at a point $x\in X_0$, then $\{E_\beta\cap X_0\}_{\beta\in B}$ is discrete at x as a family of subsets of X_0 , with respect to the induced topology. If $\{E_\beta\}_{\beta\in B}$ is discrete in X, then $\{E_\beta\cap X_0\}_{\beta\in B}$ is discrete in X_0 . If a family of subsets of X_0 is discrete at $x\in X_0$, with respect to the induced topology on X_0 , then this family is discrete at x as a family of subsets of X.

8.9.2 Discrete families and full normality

Suppose that $\{E_{\beta}\}_{{\beta}\in B}$ is discrete in X, and that X is fully normal in the strict sense. Under these conditions, one can find an open subset Z_{β} of X for each ${\beta}\in B$ such that $E_{\beta}\subseteq Z_{\beta}$, and $\{Z_{\beta}\}_{{\beta}\in B}$ is discrete in X too. This uses essentially the same argument as in Section 8.6.

Suppose that $\{E_{\beta}\}_{\beta\in B}$ is also a refinement of an open covering $\{U_{\alpha}\}_{\alpha\in A}$ of X. If $\beta\in B$, then we can choose $\alpha(\beta)\in A$ such that $E_{\beta}\subseteq U_{\alpha(\beta)}$. This implies that $Z_{\beta}\cap U_{\alpha(\beta)}$ is an open subset of X that contains E_{β} and is contained in $U_{\alpha(\beta)}$. Clearly $\{Z_{\beta}\cap U_{\alpha(\beta)}\}_{\beta\in B}$ is discrete in X, because $\{Z_{\beta}\}_{\beta\in B}$ is discrete in X.

8.9.3 σ -Discrete families of sets

A family $\{E_{\beta}\}_{{\beta}\in B}$ of subsets of a topological space X is said to be σ -discrete in X if there is a sequence B_1, B_2, B_3, \ldots of subsets of B such that $B = \bigcup_{j=1}^{\infty} B_j$ and $\{E_{\beta}\}_{{\beta}\in B_j}$ is discrete in X for every $j \geq 1$. In this case, $\{E_{\beta}\}_{{\beta}\in B}$ is σ -locally finite in X.

Suppose that $\{E_{\beta}\}_{{\beta}\in B}$ is a σ -discrete family of subsets of X that is a refinement of an open convering $\{U_{\alpha}\}_{{\alpha}\in A}$ of X. If X is fully normal in the strict sense, then for each ${\beta}\in B$ we can find an open subset of X that contains E_{β} , so that the resulting family of open sets is a σ -discrete refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$. This follows from the previous argument for discrete families.

8.10 Sequences of star refinements

Let X be a topological space, and suppose that X is fully normal in the strict sense. If \mathcal{B}_0 is an open covering of X, then there is an open covering \mathcal{B}_1 of X that is a star refinement of an open covering of X that is a star refinement of \mathcal{B}_0 .

Continuing in this way, we get a sequence of open coverings \mathcal{B}_j of X such that \mathcal{B}_{j+1} is a star refinement of an open covering of X that is a star refinement of \mathcal{B}_j for each $j \geq 0$. If $x \in X$ and $j \in \mathbf{Z}_+$, then

(8.10.1) the star of the star of x with respect to \mathcal{B}_j is contained in an element of \mathcal{B}_{j-1} ,

as in Section 8.6.

8.10.1 The star of $E \in \mathcal{B}_j$ with respect to \mathcal{B}_j

If $E \in \mathcal{B}_j$ for some $j \geq 1$, and $x \in E$, then E is contained in the star of x with respect to \mathcal{B}_j . This implies that the star of E with respect to \mathcal{B}_j is contained in the star of the star of x with respect to x. It follows that

(8.10.2) the star of E with respect to \mathcal{B}_j is contained in an element of \mathcal{B}_{j-1} ,

by (8.10.1).

8.10.2 The star of the star of E with respect to \mathcal{B}_{j+1}

Let E be a subset of X, and let j be a positive integer. Remember that the star of E with respect to \mathcal{B}_j is the same as the union of the stars of the elements of E. The star of the star of E with respect to \mathcal{B}_j is the same as the union of the stars of the stars of the elements of E with respect to \mathcal{B}_j . Using (8.10.1), we get that

(8.10.3) the star of the star of E with respect to \mathcal{B}_{j+1} is contained in the star of E with respect to \mathcal{B}_{j} .

8.10.3 Some more collections of subsets of *X*

Put $C_1 = \mathcal{B}_1$. If C_j has been defined for some $j \in \mathbf{Z}_+$, then let C_{j+1} be the collection of subsets of X obtained by taking the star with respect to \mathcal{B}_{j+1} of an element of C_j . Similarly, let \widetilde{C}_{j+1} be the collection of subsets of X obtained by taking the star with respect to \mathcal{B}_{j+1} of an element of C_{j+1} .

Equivalently, the elements of $\widetilde{\mathcal{C}}_{j+1}$ can be obtained by taking the star of the star of an element of \mathcal{C}_j , with respect to \mathcal{B}_{j+1} . Thus

(8.10.4) every element of \widetilde{C}_{j+1} is contained in the star with respect to \mathcal{B}_j of an element of \mathcal{C}_j ,

by (8.10.3).

If $j \geq 2$, then the elements of C_j can be obtained as the star with respect to \mathcal{B}_j of an element of C_{j-1} . This implies that every element of \widetilde{C}_{j+1} is contained in the star of the star of an element of C_{j-1} , with respect to \mathcal{B}_j . This means that every element of \widetilde{C}_{j+1} is contained in an element of \widetilde{C}_j , so that

(8.10.5)
$$\widetilde{\mathcal{C}}_{j+1}$$
 is a refinement of $\widetilde{\mathcal{C}}_{j}$.

If $l \geq 2$, then we get that

(8.10.6)
$$\widetilde{\mathcal{C}}_l$$
 is a refinement of $\widetilde{\mathcal{C}}_2$.

Every element of $\widetilde{\mathcal{C}}_2$ is contained in the star with respect to \mathcal{B}_1 of an element of $\mathcal{B}_1 = \mathcal{C}_1$, as in (8.10.4). The star with respect to \mathcal{B}_1 of an element of \mathcal{B}_1 is contained in an element of \mathcal{B}_0 , as in (8.10.3). Thus every element of $\widetilde{\mathcal{C}}_2$ is contained in an element of \mathcal{B}_0 , which is to say that

(8.10.7)
$$\widetilde{\mathcal{C}}_2$$
 is a refinement of \mathcal{B}_0 .

It follows that

(8.10.8) $\widetilde{\mathcal{C}}_l$ is a refinement of \mathcal{B}_0

for every $l \geq 2$.

Note that

(8.10.9) the star of
$$E \subseteq X$$
 with respect to \mathcal{B}_j contains E

for every $j \geq 0$, because \mathcal{B}_j covers X, by hypothesis. This implies that \mathcal{C}_l is a refinement of $\widetilde{\mathcal{C}}_l$ for every $l \geq 2$. Thus

(8.10.10)
$$\mathcal{C}_l$$
 is a refinement of \mathcal{B}_0

for every $l \geq 2$. Of course, this also holds when l = 1, because $\mathcal{B}_1 = \mathcal{C}_1$ is a refinement of \mathcal{B}_0 , by construction.

Similarly, C_j is a refinement of C_{j+1} for every $j \geq 1$, by (8.10.9). This implies that $\mathcal{B}_1 = C_1$ is a refinement of C_j for each $j \geq 1$. Thus C_j covers X for every $j \geq 1$, because \mathcal{B}_1 covers X. If $E \subseteq X$ and $j \geq 0$, then it is easy to see that

(8.10.11) the star of E with respect to \mathcal{B}_i is an open set,

because the elements of \mathcal{B}_j are open subsets of X. This means that \mathcal{C}_l is an open covering of X for each $l \geq 1$.

8.11 σ -Discreteness and full normality

Let X be a topological space that is fully normal in the strict sense again. If $\{U_{\alpha}\}_{{\alpha}\in A}$ is an open covering of X, then it is well known that there is a σ -discrete open covering of X that is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$. It suffices to show that there is a σ -discrete covering of X that is a refinement of $\{U_{\alpha}\}_{{\alpha}\in A}$, as in Section 8.9.

Let \mathcal{B}_0 be an open covering of X that is a star refinement of $\{U_{\alpha}\}_{\alpha\in A}$. As in the previous section, we can find an open covering \mathcal{B}_j of X for each positive integer j, so that \mathcal{B}_j is a star refinement of an open covering of X that is a star refinement of \mathcal{B}_{j-1} . Let \mathcal{C}_j be as in the previous section for each $j \in \mathbf{Z}_+$, which is an open covering of X. Remember that \mathcal{C}_j is a refinement of \mathcal{B}_0 for every $j \geq 1$. This implies that \mathcal{C}_j is a star refinement of $\{U_{\alpha}\}_{\alpha \in A}$.

If $x \in X$ and $l \in \mathbf{Z}_+$, then let $W_l(x)$ be the star of x with respect to \mathcal{C}_l . This is the union of the elements of \mathcal{C}_l that contain x. The star of each of these elements of \mathcal{C}_l with respect to \mathcal{B}_{l+1} is an element of \mathcal{C}_{l+1} that contains x. This implies that

(8.11.1) the star of $W_l(x)$ with respect to \mathcal{B}_{l+1} is contained in $W_{l+1}(x)$.

8.11.1 Using a well-ordering on X to get some more sets in X

Let \leq be a well-ordering on X. Put

$$(8.11.2) \qquad \widetilde{W}_l(x) = W_l(x) \setminus \bigcup \{W_{l+1}(y) : y \in X, \ y \leq x, \ y \neq x\}.$$

We would like to check that the collection of sets $\widetilde{W}_l(x)$, $x \in X$, is discrete in X for each $l \in \mathbf{Z}_+$.

Let $l \ge 1$ be given, and let x, y be distinct elements of X. We would like to verify that

(8.11.3) the star of $\widetilde{W}_l(y)$ with respect to \mathcal{B}_{l+1} is disjoint from $\widetilde{W}_l(x)$.

This is the same as saying that

(8.11.4) no element of \mathcal{B}_{l+1} intersects both $\widetilde{W}_l(x)$ and $\widetilde{W}_l(y)$.

Of course, (8.11.4) is symmetric in x and y, so that (8.11.3) is equivalent to the star of $\widetilde{W}_l(x)$ with respect to \mathcal{B}_{l+1} being disjoint from $\widetilde{W}_l(y)$.

Note that the star of $\widetilde{W}_l(y)$ with respect to \mathcal{B}_{l+1} is contained in the star of $W_l(y)$ with respect to \mathcal{B}_{l+1} , because $\widetilde{W}_l(y) \subseteq W_l(y)$. This implies that the star of $\widetilde{W}_l(y)$ with respect to \mathcal{B}_{l+1} is contained in $W_{l+1}(y)$, by (8.11.1). If $y \leq x$, then $W_{l+1}(y)$ is disjoint from $\widetilde{W}_l(x)$, by construction. It follows that (8.11.3) holds when $y \leq x$.

Similarly, if $x \leq y$, then the star of $\widetilde{W}_l(x)$ with respect to \mathcal{B}_{l+1} is disjoint from $\widetilde{W}_l(y)$. This implies (8.11.3), as before. Thus (8.11.3) holds, which means that (8.11.4) holds. This implies that the collection of sets $\widetilde{W}_l(x)$, $x \in X$, is discrete in X, because \mathcal{B}_{l+1} is an open covering of X.

Let $y \in X$ be given, and note that $y \in W_l(y)$ for every $l \ge 1$. Let z be the smallest element of X, with respect to \preceq , such that $y \in W_l(z)$ for some $l \ge 1$. It is easy to see that $y \in \widetilde{W}_l(z)$ in this case. This shows that the collection of sets $\widetilde{W}_l(x)$, with $x \in X$ and $l \in \mathbf{Z}_+$, covers X.

If $x \in X$ and $l \in \mathbf{Z}_+$, then there is an $\alpha_0 \in A$ such that $W_l(x) \subseteq U_{\alpha_0}$, because \mathcal{C}_l is a star refinement of $\{U_\alpha\}_{\alpha \in A}$. This implies that $\widetilde{W}_l(x) \subseteq U_{\alpha_0}$. Thus the collection of sets $\widetilde{W}_l(x)$, with $x \in X$ and $l \in \mathbf{Z}_+$, is a refinement of $\{U_\alpha\}_{\alpha \in A}$. This collection is σ -discrete in X, because the collection of sets $\widetilde{W}_l(x)$, $x \in X$, is discrete in X for each $l \geq 1$, as before.

8.12 Point finiteness

A family $\{E_{\alpha}\}_{{\alpha}\in A}$ of subsets of a set X is said to be *point finite* in X if for every $x\in X$ there are only finitely many $\alpha\in A$ such that $x\in E_{\alpha}$. Suppose now that X is a topological space. If $\{E_{\alpha}\}_{{\alpha}\in A}$ is locally finite in X, then $\{E_{\alpha}\}_{{\alpha}\in A}$ is point finite in X.

8.12.1 Metacompactness

If every open covering of X has a refinement that is a point finite open covering of X, then X is said to be metacompact. If X is paracompact in the strict sense, then X is metacompact.

8.12.2 Using point finiteness

Suppose that X is normal in the strict sense, and that $\{V_{\alpha}\}_{\alpha\in A}$ is a point finite open covering of X. Under these conditions, one can choose an open subset U_{α} of X for each $\alpha\in A$ such that $\overline{U_{\alpha}}\subseteq V_{\alpha}$ and $\{U_{\alpha}\}_{\alpha\in A}$ is an open covering of X, as in Problem V (a) on p171 of [240]. More precisely, this can be obtained using Zorn's lemma or Hausdorff's maximality principle, as follows.

Let A_0 be a subset of A, and let ϕ_0 be a mapping from A_0 into the set of all

open subsets of X. Let us say that (A_0, ϕ_0) is admissible if

$$(8.12.1) \overline{\phi_0(\alpha)} \subseteq V_\alpha$$

for every $\alpha \in A_0$, and

(8.12.2)
$$X = \left(\bigcup_{\alpha \in A_0} \phi_0(\alpha)\right) \cup \left(\bigcup_{\alpha \in A \setminus A_0} V_\alpha\right).$$

Let \mathcal{A} be the collection of admissible pairs (A_0, ϕ_0) . If $(A_1, \phi_1), (A_2, \phi_2) \in \mathcal{A}$, then put

$$(8.12.3) (A_1, \phi_1) \preceq (A_2, \phi_2)$$

when $A_1 \subseteq A_2$ and $\phi_1 = \phi_2$ on A_1 . This defines a partial ordering on \mathcal{A} . Let \mathcal{C} be a chain in \mathcal{A} , and put

$$(8.12.4) A_{\mathcal{C}} = \bigcup_{(A_0, \phi_0) \in \mathcal{C}} A_0,$$

which is a subset of A. If $\alpha \in A_{\mathcal{C}}$, then there is an element (A_0, ϕ_0) in \mathcal{C} such that $\alpha \in A_0$, and we would like to put

(8.12.5)
$$\phi_{\mathcal{C}}(\alpha) = \phi_0(\alpha).$$

One can check that this does not depend on the particular element (A_0, ϕ_0) of \mathcal{C} with $\alpha \in A_0$. Thus $\phi_{\mathcal{C}}$ is a well-defined mapping from $A_{\mathcal{C}}$ into the set of all open subsets of X. Note that

$$(8.12.6) \overline{\phi_{\mathcal{C}}(\alpha)} \subseteq V_{\alpha}$$

for every $\alpha \in A_{\mathcal{C}}$, because of the analogous property (8.12.1) of every (A_0, ϕ_0) in \mathcal{C} .

We would like to verify that

(8.12.7)
$$X = \left(\bigcup_{\alpha \in A_{\mathcal{C}}} \phi_{\mathcal{C}}(\alpha)\right) \cup \left(\bigcup_{\alpha \in A \setminus A_{\mathcal{C}}} V_{\alpha}\right).$$

Let $x \in X$ be given, and let us check that x is an element of the right side of (8.12.7). Of course, $x \in V_{\alpha}$ for some $\alpha \in A$, because $\{V_{\alpha}\}_{{\alpha} \in A}$ covers X, by hypothesis. If $x \in V_{\alpha}$ for some $\alpha \in A \setminus A_{\mathcal{C}}$, then x is an element of the right side of (8.12.7). Suppose now that for each $\alpha \in A \setminus A_{\mathcal{C}}$, we have that $x \notin V_{\alpha}$.

Remember that there are only finitely many $\alpha \in A$ such that $x \in V_{\alpha}$, because $\{V_{\alpha}\}_{\alpha \in A}$ is point finite in X, by hypothesis. In this situation, $\mathcal{A}_{\mathcal{C}}$ contains all of the $\alpha \in A$ with $x \in V_{\alpha}$, by construction. This means that each of these α 's is an element of A_0 for some $(A_0, \phi_0) \in \mathcal{C}$, by definition of $A_{\mathcal{C}}$. More precisely, there is an $(A_0, \phi_0) \in \mathcal{C}$ such that A_0 contains every $\alpha \in A$ with $x \in A_{\alpha}$. This uses the facts that there are only finitely many of these α 's, and that \mathcal{C} is a chain in \mathcal{A} .

Of course, (A_0, ϕ_0) satisfies (8.12.2), because $(A_0, \phi_0) \in \mathcal{A}$. If $\alpha \in A \setminus A_0$, then $x \notin V_{\alpha}$, as in the preceding paragraph. Using (8.12.2), we get that there is an $\alpha \in A_0$ such that $x \in \phi_0(\alpha)$. This α is contained in $A_{\mathcal{C}}$ and satisfies $x \in \phi_{\mathcal{C}}(\alpha)$, by definition of $(A_{\mathcal{C}}, \phi_{\mathcal{C}})$. It follows that x is an element of the right side of (8.12.7).

Thus (8.12.7) holds, so that $(A_{\mathcal{C}}, \phi_{\mathcal{C}})$ is an element of \mathcal{A} . Using Zorn's lemma or Hausdorff's maximality principle, one can get a maximal element (A_1, ϕ_1) of \mathcal{A} . We would like to verify that $A_1 = A$. Suppose for the sake of a contradiction that $A_1 \neq A$, and let α_2 be an element of $A \setminus A_1$.

Put $A_2 = A_1 \cup \{\alpha_2\}$ and

(8.12.8)
$$E_{\alpha_2} = X \setminus \left(\left(\bigcup_{\alpha \in A_1} \phi_1(\alpha) \right) \cup \left(\bigcup_{\alpha \in A \setminus A_2} V_{\alpha} \right) \right).$$

Note that E_{α_2} is a closed set in X, because $\phi_1(\alpha)$ is an open set for every $\alpha \in A_1$, and V_{α} is an open set for every $\alpha \in A$. We also have that

$$(8.12.9) E_{\alpha_2} \subseteq V_{\alpha_2},$$

because of the analogue of (8.12.2) for (A_1, ϕ_1) . Thus there is an open subset W_{α_2} of X such that

(8.12.10)
$$E_{\alpha_2} \subseteq W_{\alpha_2} \text{ and } \overline{W_{\alpha_2}} \subseteq V_{\alpha_2},$$

because X is normal in the strict sense, by hypothesis.

Put $\phi_2(\alpha) = \phi_1(\alpha)$ for every $\alpha \in A_1$, and $\phi_2(\alpha_2) = W_{\alpha_2}$. This defines ϕ_2 as a mapping from A_2 into the set of all open subsets of X. Observe that $\phi_2(\alpha) \subseteq V_{\alpha}$ for every $\alpha \in A_2$, because of the analogous property of ϕ_1 on A_1 , and the second part of (8.12.10). Using the first part of (8.12.10), we get that

(8.12.11)
$$X = \left(\bigcup_{\alpha \in A_2} \phi_2(\alpha)\right) \cup \left(\bigcup_{\alpha \in A \setminus A_2} V_\alpha\right).$$

This means that (A_2, ϕ_2) is admissible, and hence an element of \mathcal{A} . By construction, (8.12.3) holds, and $(A_1, \phi_1) \neq (A_2, \phi_2)$. This contradicts the maximality of (A_1, ϕ_1) , so that $A_1 = A$, as desired.

8.13 Partitions of unity

Let X be a topological space, and suppose that X is normal in the strict sense. Also let $\{V_{\alpha}\}_{{\alpha}\in A}$ be a locally finite open covering of X, so that $\{V_{\alpha}\}_{{\alpha}\in A}$ is point finite in X in particular. Thus, for each ${\alpha}\in A$, we can choose an open subset U_{α} of X such that $\overline{U_{\alpha}}\subseteq V_{\alpha}$ and $\{U_{\alpha}\}_{{\alpha}\in A}$ is an open covering of X, as in the previous section.

If $\alpha \in A$, then we can use Urysohn's lemma to get a continuous real-valued function ϕ_{α} on X such that $\phi_{\alpha}(x) = 1$ for every $x \in \overline{U_{\alpha}}$, $\phi_{\alpha}(x) = 0$ for every $x \in X \setminus V_{\alpha}$, and $0 \le \phi_{\alpha}(x) \le 1$ for every $x \in X$. In this case,

$$\{x \in X : \phi_{\alpha}(x) \neq 0\}$$

is contained in V_{α} for every $\alpha \in A$, so that the family of these subsets of X is locally finite too.

It follows that

(8.13.2)
$$\Phi(x) = \sum_{\alpha \in A} \phi_{\alpha}(x)$$

defines a continuous real-valued function on X, as in Subsection 5.14.1. If $x \in \overline{U_{\alpha_0}}$ for some $\alpha_0 \in A$, then

(8.13.3)
$$\Phi(x) \ge \phi_{\alpha_0}(x) \ge 1,$$

because $\phi_{\alpha}(x) \geq 0$ for every $\alpha \in A$, by construction. This means that $\Phi(x) \geq 1$ for every $x \in X$, because X is covered by $\{U_{\alpha}\}_{{\alpha} \in A}$.

If we put

(8.13.4)
$$\psi_{\beta}(x) = \phi_{\beta}(x)/\Phi(x)$$

for every $\beta \in A$ and $x \in X$, then we get a partition of unity on X, as in Subsection 5.14.2. This corresponds to Problem W on p171 of [240].

8.14 Minimal coverings

Let X be a set, and let $\{E_{\alpha}\}_{{\alpha}\in A}$ be a point finite covering of X. Under these conditions, there is a minimal subcovering of X from $\{E_{\alpha}\}_{{\alpha}\in A}$, as in Problem V (b) on p171 of [240]. This is also mentioned on p23 of [408], for topological spaces, although the topology does not play a role for this part.

More precisely, let \mathcal{A} be the collection of subsets B of A such that $\{E_{\alpha}\}_{{\alpha}\in B}$ covers X. We would like to show that \mathcal{A} has a minimal element, with respect to inclusion. Let \preceq be the partial ordering on \mathcal{A} defined by putting $A_1 \preceq A_2$ when $A_1, A_2 \in \mathcal{A}$ satisfy $A_2 \subseteq A_1$. We would like to show that \mathcal{A} has a maximal element with respect to \preceq , using Zorn's lemma or Hausdorff's maximality principle.

Let \mathcal{C} be a chain in \mathcal{A} with respect to \leq , which is the same as saying that \mathcal{C} is a chain in \mathcal{A} with respect to inclusion. Put

$$(8.14.1) A_{\mathcal{C}} = \bigcap_{A_0 \in \mathcal{C}} A_0,$$

which is a subset of A. We would like to check that

(8.14.2)
$$\bigcup_{\alpha \in A_{\mathcal{C}}} E_{\alpha} = X,$$

so that $A_{\mathcal{C}} \in \mathcal{A}$.

Let $x \in X$ be given, and remember that $x \in E_{\alpha}$ for only finitely many $\alpha \in A$, by hypothesis. If x is not an element of the union on the left side of (8.14.2), then $A_{\mathcal{C}}$ does not contain any $\alpha \in A$ such that $x \in E_{\alpha}$. This means that for each $\alpha \in A$ with $x \in E_{\alpha}$, there is an element of \mathcal{C} that does not contain α . Because \mathcal{C} is a chain, and there are only finitely many of these α 's, there is an

 $A_0 \in \mathcal{C}$ such that A_0 does not contain any $\alpha \in A$ such that $x \in E_\alpha$. This implies that x is not an element of $\bigcup_{\alpha \in A_0} E_\alpha$, contradicting the fact that $A_0 \in \mathcal{A}$.

Thus (8.14.2) holds, so that $A_{\mathcal{C}} \in \mathcal{A}$. One can use this and Zorn's lemma or Hausdorff's maximality principle to get a maximal element of \mathcal{A} with respect to \preceq . This is the same as a minimal element of \mathcal{A} with respect to inclusion, as desired.

8.14.1 Using metacompactness

Now let X be a topological space, and suppose that X is metacompact, as in Section 8.12. If X is also countably compact, then X is compact, as on p23f of [408]. This corresponds to Problem V (c) on p171 of [240], where X is asked to satisfy the first separation condition too. However, this additional hypothesis appears to be included because of the way that some other results were stated, and is not really needed. In fact, the argument indicated in [240] is the same as the one in [408].

Let $\{U_{\alpha}\}_{\alpha\in A}$ be an arbitrary open covering of X. Because X is metacompact, there is a point finite refinement $\{V_{\beta}\}_{\beta\in B}$ of $\{U_{\alpha}\}_{\alpha\in A}$ that is an open covering of X. As before, there is a minimal subset B_0 of B such that $\{V_{\beta}\}_{\beta\in B_0}$ covers X. If B_0 has only finitely many elements, then we can get a finite subcovering of X from $\{U_{\alpha}\}_{\alpha\in A}$.

If $\beta \in B_0$, then there is a point $x_\beta \in V_\beta$ that is not in V_γ for any other $\gamma \in B_0$, because B_0 is minimal. Let us choose such a point x_β for every $\beta \in B_0$, and let L be the set of points x_β , $\beta \in B_0$, that have been chosen in this way. Note that these points are distinct, by construction. Suppose that B_0 has infinitely many elements, so that L has infinitely many elements too. Remember that countable compactness is equivalent to the strong limit point property, which implies that there is a point $x \in X$ that is a strong limit point of L. Because $\{V_\beta\}_{\beta \in B_0}$ covers X, there is a $\beta_0 \in B_0$ such that $x \in V_{\beta_0}$. It follows that V_{β_0} contains infinitely many elements of L, because x is a strong limit point of L, and L is an open set. This contradicts the fact that L is the only element of L in L in L in L is an open set.

Bibliography

- [1] A. Abian, The Theory of Sets and Transfinite Arithmetic, Saunders, 1965.
- [2] A. Abian, On the independence of the set theoretical axioms, American Mathematical Monthly **76** (1969), 787–790.
- [3] A. Abian, On the similarity of partially ordered sets, American Mathematical Monthly 77 (1970), 1092–1094.
- [4] C. Adams, Why knot? An Introduction to the Mathematical Theory of Knots, Key College Publishing, 2004.
- [5] C. Adams, The Knot Book: An Elementary Introduction to the Mathematical Theory of Knots, revised reprint of the 1994 original, American Mathematical Society, 2004.
- [6] C. Adams, Turning knots into flowers and related undergraduate research, American Mathematical Monthly **124** (2017), 791–806.
- [7] C. Adams, A. Henrich, K. Kearney, N. Scoville, Knots related by knotoids, American Mathematical Monthly 126 (2019), 483–490.
- [8] J. Adams, Classroom notes: A note on homotopy theory, American Mathematical Monthly **74** (1967), 581.
- [9] S. Ahmad, The Kuratowski closure axioms, Mathematics Magazine 37 (1964), 296–297.
- [10] S. Ahmad, Classroom notes: Absolutely independent axioms for the derived set operator, American Mathematical Monthly 73 (1966), 390–392.
- [11] J. Alexander, Classroom notes: On the connected sum of projective planes, tori, and Klein bottles, American Mathematical Monthly 78 (1971), 185–187.
- [12] P. Amster, *The power of the half power*, American Mathematical Monthly **128** (2021), 655–657.
- [13] R. Anderson, Award for distinguished dervice to Professor R. H. Bing, American Mathematical Monthly 81 (1974), 111-113.

[14] P. Andersson, *The color invariant for knots and links*, American Mathematical Monthly **102** (1995), 442–448.

- [15] P. Andrews, *The classification of surfaces*, American Mathematical Monthly 95 (1988), 861–867.
- [16] T. Archibald, Connectivity and smoke-rings: Green's second identity in its first fifty years, Mathematics Magazine 62 (1989), 219–232.
- [17] R. Arens, Note on convergence in topology, Mathematics Magazine 23 (1950), 229–234.
- [18] R. Arens, Advanced algebra: operations without numbers, Mathematics Magazine 24 (1951), 253–264.
- [19] M. Armstrong, *Basic Topology*, corrected reprint of the 1979 original, Springer-Verlag, 1983.
- [20] M. Atiyah, Bakerian lecture, 1975: Global geometry, Proceedings of the Royal Society A347 (1976), 291–299; American Mathematical Monthly 111 (2004), 716–723.
- [21] M. Atiyah, *Mathematics in the 20th century*, American Mathematical Monthly **108** (2001), 654–666; Bulletin of the London Mathematical Society **34** (2002), 1–15.
- [22] G. Atneosen, An application of simplicial homology theory, American Mathematical Monthly 77, 381–384.
- [23] D. Auckly, Folklore, the Borromean rings, the icosahedron, and three dimensions, Mathematics Magazine 93 (2020), 330–342.
- [24] C. Aull and R. Lowen, editiors, Handbook of the History of General Topology, Volume 1, Kluwer, 1997.
- [25] C. Aull and R. Lowen, editors, Handbook of the History of General Topology, Volume 2, Papers from the Special Session of the American Mathematical Society Meeting held in San Antonio, TX, 1993, Kuwer, 1998.
- [26] C. Aull and R. Lowen, editors, *Handbook of the History of General Topology*, Volume 3, Kluwer, 2001.
- [27] W. Ayres, Some elementary aspects of topology, American Mathematical Monthly 45 (1938), 88–92.
- [28] J. Baker, J. Lawrence, C. Ng, and F. Zorzitto, Classroom notes: Sequence topologies on the real line, American Mathematical Monthly 85 (1978), 667–668.
- [29] B. Ball, Arcwise connectedness and the persistence of errors, American Mathematical Monthly **91** (1984), 431–433.

[30] R. Bartle, Nets and filters in topology, American Mathematical Monthly **62** (1955), 551–557; correction, ibid., **70** (1963), 52–53.

- [31] J. Baum, Elements of point set topology, Dover, 1991.
- [32] E. Bell, The place of rigor in mathematics, American Mathematical Monthly 41 (1934), 599–607.
- [33] M. Bell and J. Ginsburg, A space which will not go where its proper subspaces want it to go, American Mathematical Monthly 89 (1982), 749–751.
- [34] Y. Benyamini and J. Lindenstrauss, Geometric Nonlinear Functional Analysis, American Mathematical Society, 2000.
- [35] C. Berge, *Topological Spaces*, translated from the French original by E. Patterson, Dover, 1997.
- [36] P. Bernays, *Axiomatic Set Theory*, with a historical introduction by Abraham A. Fraenkel, Dover, 1991.
- [37] S. Beslin, A note on topological continuity, Mathematics Magazine 65 (1992), 257–258.
- [38] G. Bezhanishvili, P. Morandi, and B. Olberding, A new approach to the Katetov-Tong theorem, American Mathematical Monthly 129 (2022), 66–74.
- [39] G. Bezhanishvili and D. Pengelley, What does "less than or equal" really mean?, American Mathematical Monthly 122 (2015), 983–989.
- [40] R. Bing, An equilateral distance, American Mathematical Monthly 58 (1951), 380–383.
- [41] R. Bing, *Elementary point set topology*, American Mathematical Monthly **67** (1960), no. 7, part II.
- [42] R. Bing, Spheres in E^3 , American Mathematical Monthly **71** (1964), 353–364.
- [43] R. Bing, *Retractions onto spheres*, American Mathematical Monthly **71** (1964), 481–484.
- [44] R. Bing, *Challenging conjectures*, American Mathematical Monthly **74** (1967), no. 1, part II, 56–64.
- [45] R. Bing, The elusive fixed point property, American Mathematical Monthly **76** (1969), 119–132.
- [46] R. Bing, Award for distinguished service to Professor Raymond L. Wilder, American Mathematical Monthly 80 (1973), 117–119.

[47] R. Bing, *The Geometric Topology of 3-Manifolds*, American Mathematical Society, 1983.

- [48] R. Blair, M. Campisi, J. Johnson, S. Taylor, and M. Tomova, *Neighbors of knots in the Gordian graph*, American Mathematical Monthly **124** (2017), 4–23.
- [49] A. Błaszczyk, A simple proof of Sierpiński's theorem, American Mathematical Monthly 126 (2019), 464–466.
- [50] E. Bloch, A First Course in Geometric Topology and Differential Geometry, Birkhäuser, 1997.
- [51] E. Bloch, A simple proof of a generalized no retraction theorem, American Mathematical Monthly **116** (2009), 342–350.
- [52] E. Bloch, Proofs and Fundamentals: A First Course in Abstract Mathematics, 2nd edition, Springer, 2011.
- [53] F. Bloore and H. Morton, *Advice on hanging pictures*, American Mathematical Monthly **92** (1985), 309–321.
- [54] M. Brill and M. Stueben, *Demystifying the projective plane*, Mathematics Magazine **63** (1990), 48–51.
- [55] J. Britt, The anatomy of low-dimensional stable singularities, American Mathematical Monthly **92** (1985), 183–201.
- [56] A. Browder, Topology in the complex plane, American Mathematical Monthly 107 (2000), 393–401.
- [57] A. Browder, Complex numbers and the ham sandwich theorem, American Mathematical Monthly 113 (2006), 935–937.
- [58] R. Brown, Elementary consequences of the noncontractibility of the circle, American Mathematical Monthly 81 (1974), 247–252.
- [59] R. Brown, The fixed point property and Cartesian products, American Mathematical Monthly 89 (1982), 654–678.
- [60] R. Brown, A good question won't go away: an example of mathematical research, American Mathematical Monthly 128 (2021), 62–68.
- [61] R. Brown and J. Girolo, Isolating fixed points, American Mathematical Monthly 109 (2002), 595–611.
- [62] R. Brown and R. Greene, An interior fixed point property of the disc, American Mathematical Monthly 101 (1994), 39–47.
- [63] S. Brown, F. Cabrera, R. Evans, G. Gibbs, A. Henrich, J. Kreinbihl, *The region unknotting game*, Mathematics Magazine **90** (2017), 323–337.

[64] J. Bruce and P. Giblin, Generic geometry, American Mathematical Monthly 90 (1983), 529–545.

- [65] J. Bruce, P. Giblin, and C. Gibson, On caustics of plane curves, American Mathematical Monthly 88 (1981), 651–667.
- [66] B. Brunson, *Miscellanea: Ode to a* ϕ ?, American Mathematical Monthly **92** (1985), 304.
- [67] J. Bryant, J. Kuzmanovich, and A. Pavlichenkov, Functions with compact preimages of compact sets, Mathematics Magazine 70 (1997), 362–364.
- [68] C. Burgess, *Classification of surfaces*, American Mathematical Monthly **92** (1985), 349–354.
- [69] C. Burgess, Continuous functions and connected graphs, American Mathematical Monthly 97 (1990), 337–339.
- [70] H. Burgiel, D. Franzblau, and K. Gutschera, *The mystery of the linked triangles*, Mathematics Magazine **69** (1996), 94–102.
- [71] S. Cairns, Deformations of plane rectilinear complexes, American Mathematical Monthly **51** (1944), 247–252.
- [72] F. Callahan and S. Kneale, A note on the Schroeder–Bernstein theorem, American Mathematical Monthly **64** (1957), 423–424.
- [73] J. Callahan, Singularities and plane maps, American Mathematical Monthly 81 (1974), 211–240.
- [74] J. Callahan, Singularities and plane maps II: Sketching catastrophes, American Matheatical Monthly 84 (1977), 765–803.
- [75] D. Cameron, The mini-max property of the Tychonoff product topology, American Mathematical Monthly 80 (1973), 925–927.
- [76] D. Cameron, On subspaces of separable spaces, Mathematics Magazine 48 (1975), 288.
- [77] P. Cameron, J. Hocking, and S. Naimpally, Nearness a better approach to continuity and limits, American Mathematical Monthly 81 (1974), 739– 745.
- [78] J. Cannon, *Topology as Fluid Geometry*, Two-Dimensional Spaces, Volume 2, American Mathematical Society, 2017.
- [79] G. Cantrell, Maximal homeomorphisms, American Mathematical Monthly **78** (1971), 998–1000.
- [80] D. Cao Labora, When is a continuous bijection a homeomorphism?, American Mathematical Monthly 127 (2020), 547–553.

[81] S. Cappell and J. Shaneson, *Linear algebra and topology*, Bulletin of the American Mathematical Society (N.S.) 1 (1979), 685–687.

- [82] S. Cappell and J. Shaneson, *Nonlinear similarity of matrices*, Bulletin of the American Mathematical Society (N.S.) **1** (1979), 899–902.
- [83] J. Carter, How Surfaces Intersect in Space: An Introduction to Topology, 2nd edition, World Scientific, 1995.
- [84] J. Carter, D. Silver, and S. Williams, *Three dimensions of knot coloring*, American Mathematical Monthly **121** (2014), 506–514.
- [85] D. Cenzer, J. Larson, C. Porter, and J. Zapletal, Set Theory and Foundations of Mathematics An Introduction to Mathematical Logic, Volume 1, Set Theory, Volume II, Foundations of Mathematics. World Scientific, 2020, 2022.
- [86] R. Chandler, New compactifications from old, American Mathematical Monthly 79 (1972), 501–503.
- [87] A. Charlesworth, A note on Urysohn's metrization theorem, American Mathematical Monthly 83 (1976), 718–720.
- [88] J. Cheeger and J. Kister, Counting topological manifolds, Topology 9 (1970), 149–151.
- [89] S.-S. Chern, From triangles to manifolds, American Mathematical Monthly 86 (1979), 339–349.
- [90] S.-S. Chern, What is geometry?, American Mathematical Monthly 97 (1990), 679–686.
- [91] P. Chernoff, A simple proof of Tychonoff's theorem via nets American Mathematical Monthly 99 (1992), 932–934.
- [92] J. Chew, Regularity as a relaxation of paracompactness, American Mathematical Monthly 79 (1972), 630–632. See also Complements and comments, ibid., 79 (1972), 1100–1103.
- [93] J. Chew, Contagious properties, Mathematics Magazine 47 (1974), 84–87.
- [94] J. Chew, *Invertibility of function spaces*, American Mathematical Monthly **82** (1975), 618–621.
- [95] J. Chew, A note on closure continuity, American Mathematical Monthly **95** (1988), 744-745.
- [96] J. Chew and J. Tong, *Some remarks on weak continuity*, American Mathematical Monthly **98** (1991), 931–934.
- [97] K. Ciesielski, Set Theory for the Working Mathematician, Cambridge University Press, 1997.

[98] K. Ciesielski, Sierpiński's topological characterization of Q, Mathematics Magazine 93 (2020), 136–138.

- [99] K. Ciesielski, M. Martinez-Gómez, and J. Seoane-Sepúlveda, "Big" continuous restrictions of arbitrary functions, American Mathematical Monthly 126 (2019), 547–552.
- [100] J. Cobb and W. Voxman, Dispersion points and fixed points, American Mathematical Monthly 87 (1980), 278–281.
- [101] P. Cohen, Set Theory and the Continuum Hypothesis, Benjamin, 1966.
- [102] J. Conway, A Course in Point Set Topology, Springer, 2014.
- [103] R. Cox, A proof of the Schoeder-Bernstein theorem, American Mathematical Monthly 75 (1968), 508.
- [104] S. Cullinane, *Miscellanea* ("Triangles are square"), American Mathematical Monthly **91** (1984), 382.
- [105] D. Cunningham, Set Theory: A First Course, Cambridge University Press, 2016.
- [106] A. Dasgupta, Set Theory: With an Introduction to Real Point Sets, Birkhäuser / Springer, 2014.
- [107] F. Dashiell, Jr., Countable metric spaces without isolated points, American Mathematical Monthly 128 (2021), 265-267.
- [108] R. Daverman, Decompositions of Manifolds, AMS Chelsea, 2007.
- [109] R. Daverman and G. Venema, *Embeddings in Manifolds*, American Mathematical Society, 2009.
- [110] A. Davis, Indexed systems of neighborhoods for general topological spaces, American Mathematical Monthly 68 (1961), 886–893.
- [111] A. Davis, Mathematical notes: Total boundedness, Gál's theorem, and completion for regular topological spaces, American Mathematical Monthly **70** (1963), 1075–1079.
- [112] W. Derrick, A condition under which a mapping is a homeomorphism, American Mathematical Monthly 80 (1973), 554–555.
- [113] A. Dewdney, *Embedding graphs in Euclidean 3-space*, American Mathematical Monthly **84** (1977), 372–373.
- [114] K. Devlin, The Joy of Sets: Fundamentals of Contemporary Set Theory, 2nd edition, Springer-Verlag, 1993.
- [115] J. Dijkstra, On homeomorphism groups and the compact-open topology, American Mathematical Monthly 112 (2005), 910–912.

[116] A. Dold, A simple proof of the Jordan–Alexander complement theorem, American Mathematical Monthly **100** (1993), 856–857.

- [117] S. Dolecki and F. Mynard, Convergence Foundations of Topology, World Scientific, 2016.
- [118] S. Donaldson and D. Sullivan, *Quasiconformal 4-manifolds*, Acta Mathematica **163** (1989), 181–252.
- [119] P. Doyle, Symmetry in geometric complexes, American Mathematical Monthly 73 (1966), 625–628.
- [120] P. Doyle and J. Hocking, *Invertible spaces*, American Mathematical Monthly 68 (1961), 959–965.
- [121] F. Drake and D. Singh, Intermediate Set Theory, Wiley, 1996.
- [122] V. Drobot and J. Sawka, The teaching of mathematics: Why the product topology?, American Mathematical Monthly **91** (1984), 137–138.
- [123] W. Dunham, A historical gem from Vito Volterra, Mathematics Magazine **63** (1990), 234–237.
- [124] C. Eberhart, Some remarks on the irrational and rational numbers, American Mathematical Monthly 84 (1977), 32–35.
- [125] A. Einstein, Miscellanea, American Mathematical Monthly 90 (1983), 653.
- [126] M. Eisenberg, Axiomatic Theory of Sets and Classes, Holt, Rinehart, and Winston, 1971.
- [127] H. Enderton, Elements of Set Theory, Academic Press, 1977.
- [128] M. Espelie and J. Joseph, A characterization of continuous closed real functions, Mathematics Magazine 45 (1972), 200–201.
- [129] M. Espelie and J. Joseph, Compact subsets of the Sorgenfrey line, Mathematics Magazine 49 (1976), 250–251.
- [130] B. Faires, Yueh-Gin Gong and Dr. Charles Y. Hu Award for 2010 to Kenneth A. Ross for distinguished service to mathematics, American Mathematical Monthly 117 (2010), 195–197.
- [131] D. Farmer and T. Stanford, *Knots and Surfaces: A Guide to Discovering Mathematics*, American Mathematical Society, 1996.
- [132] T. Farrell and Y. Su, Introductory Lectures on Manifold Topology: Signposts, International Press, 2014.
- [133] I. Fáry and E. Isenberg, On a converse of the Jordan Curve theorem, American Mathematical Monthly 81 (1974), 636–639.

[134] S. Fefferman, *Does mathematics need new axioms?*, American Mathematical Monthly **106** (1999), 99-111.

- [135] O. Feichtinger, More on lower semi-continuity, American Mathematical Monthly 83 (1975), 39.
- [136] R. Fenn, What is the geometry of a surface?, American Mathematical Monthly **90** (1983), 87–98.
- [137] H. Ferguson, Two theorems, two sculptures, two posters, American Mathematical Monthly 97 (1990), 589–610.
- [138] J. Fernández-Sánchez and W. Trutschnig, Nested square roots of 2 revisited, American Mathematical Monthly 127 (2020), 344–351.
- [139] J. Fife, The Kuratowski closure-complement problem, Mathematics Magazine 64 (1991), 180–182.
- [140] J. Fisher, Miscellanea: Do as I say, not as I do or how to state a rule in a statement that breaks the rule, American Mathematical Monthly 93 (1986), 664.
- [141] E. Flapan, When Topology Meets Chemistry: A Topological Look at Molecular Chirality, Cambridge University Press, Mathematical Association of America, 2000.
- [142] E. Flapan, Knots, Molecules, and the Universe: An Introduction to Topology, American Mathematical Society, 2016.
- [143] G. Folland, A tale of topology, American Mathematical Monthly 117 (2010), 663–672.
- [144] M. Fort, Jr., Some properties of continuous functions, American Mathematical Monthly 59 (1952), 372–375.
- [145] M. Fort, Jr., Essential mappings, American Mathematical Monthly 63 (1956), 238–241.
- [146] A. Fraenkel, Abstract Set Theory, 3rd revised edition, North-Holland, 1966.
- [147] A. Fraenkel, Set Theory and Logic, Addison-Wesley, 1966.
- [148] A. Fraenkel, Y. Bar-Hillel, and A. Levy, Foundations of Set Theory, second revised edition, with the collaboration of D. val Dalen, North-Holland, 1973.
- [149] G. Francis, Drawing Seifert surfaces that fiber the figure-8 knot complement in S^3 over S^1 , American Mathematical Monthly **90** (1983), 589–599.
- [150] G. Francis, A Topological Picturebook, Springer, 2007.

[151] G. Francis and J. Weeks, Conway's ZIP proof, American Mathematical Monthly 106 (1999), 393–399.

- [152] M. Freedman and F. Quinn, Topology of 4-Manifolds, Princeton University Press, 1990.
- [153] H. Gabai, Classroom notes: The exterior operator and the boundary operator, American Mathematical Monthly **71** (1964), 1029–1031.
- [154] I. Gál, Advanced problems and solutions: Solutions: 4891 (Compact topological space), American Mathematical Monthly 68 (1961), 300–301.
- [155] D. Gale, The teaching of mathematics: The classification of 1-manifolds: A take-home exam, American Mathematical Monthly **94** (1987), 170–175.
- [156] T. Gamelin and R. Greene, Introduction to Topology, 2nd edition, Dover, 1999.
- [157] S. Gass, R. Buck, and R. Boas, *Miscellanea* ("Continuity: A do-it-yourself editorial"), American Mathematical Monthly **90** (1983), 296–297.
- [158] D. Gauld, Topological properties of manifolds, American Mathematical Monthly 81 (1974), 633-636.
- [159] D. Gauld, Did the young Volterra know about Cantor?, Mathematics Magazine **66** (1993), 246–247.
- [160] M. Gemignani, Elementary Topology, corrected reprint of the 2nd edition, Dover, 1990.
- [161] P. Giblin and S. Brassett, *Local symmetry of plane curves*, American Mathematical Monthly **92** (1985), 689–707.
- [162] P. Giblin and D. O'Shea, *The bitangent sphere problem*, American Mathematical Monthly **97** (1990), 5–23.
- [163] P. Giblin and F. Warder, Evolving evolutoids, American Mathematical Monthly 121 (2014), 871–889.
- [164] L. Gillman, Miscellanea, American Mathematical Monthly 89 (1982), 375.
- [165] L. Gillman, Two classical surprises concerning the axiom of choice and the continuum hypothesis, American Mathematical Monthly 109 (2002), 544-553.
- [166] L. Gillman and M. Jerison, Rings of Continuous Functions, Springer-Verlag, 1976.
- [167] J. Ginsburg and B. Sands, *Minimal infinite topological spaces*, American Mathematical Monthly 86 (1979), 574–576.

[168] A. Gleason, Yueh-Gin Gung and Dr. Charles Y. Hu Award for distinguished dervice to Paul R. Halmos, American Mathematical Monthly 107 (2000), 193–194.

- [169] K. Gödel, What is Cantor's continuum problem?, American Mathematical Monthly 54 (1947), 515–525.
- [170] R. Goldstein, Miscellanea, American Mathematical Monthly 93 (1986), 275
- [171] F. Gouvêa, Was Cantor surprised?, American Mathematical Monthly 118 (2011), 198–209.
- [172] J. Gray, *Miscellanea: Clear theory*, American Mathemtical Monthly **92** (1985), 158–159.
- [173] W. Gray, On the metrizability of invertible spaces, American Mathematical Monthly 71 (1964), 533–534.
- [174] J. Green, The theory of functions of a real variable, Mathematics Magazine **24** (1951), 209–217.
- [175] J. Gross, Classroom notes: A third definition of local compactness, American Mathematical Monthly **74** (1967), 1120–1122.
- [176] K. Gross and P. Hilton, Award for distinguished service to Professor Gail S. Young, American Mathematical Monthly 94 (1987), 233–236.
- [177] F. Grove, Miscellanea, American Mathematical Monthly 91 (1984), 412.
- [178] M. Guay, J. Hocking, and H. Kronk. Local near-homogeneity, American Mathematical Monthly 70 (1963), 827–833.
- [179] H. Guggenheimer, The Jordan and Schoenflies theorems in axiomatic geometry, American Mathematical Monthly 85 (1978), 753–756.
- [180] A. Haddock, Fixed point theorems, American Mathematical Monthly 71 (1964), 1095–1099.
- [181] A. Hajnal and P. Hamburger, Set Theory, translated from the 1983 Hungarian original by A. Máté, Cambridge University Press, 1999.
- [182] L. Halbeisen, Combinatorial Set Theory: With a Gentle Invoduction to Forcing, 2nd edition, Springer, 2017.
- [183] T. Hales, The Jordan curve theorem, formally and informally, American Mathematical Monthly 114 (2007), 882–894.
- [184] P. Halmos, Naive Set Theory, Springer-Verlag, 1974.
- [185] R. Hansell and E. Wolk, On continuous functions and convergence of nets, American Mathematical Monthly 83 (1976), 365–366.

[186] F. Harary, On the algebraic structure of knots, American Mathematical Monthly **56** (1949), 466–468. This is related to [415].

- [187] D. Hartig, Classroom notes: An important functor in analysis and topology, American Mathematical Monthly 85 (1978), 41–43.
- [188] F. Harvey, Classroom notes: The derived set operator, American Mathematical Monthly 70 (1963), 1085–1086.
- [189] A. Hatcher, Algebraic Topology, Cambridge University Press, 2002.
- [190] F. Hausdorff, Set Theory, 2nd edition, translated from the German by J. Aumann et al., Chelsea, 1962.
- [191] F. Hausdorff, Hausdorff on Ordered Sets, translated from the German, edited, and with commentary by J. Plotkin, American Mathematical Society, London Mathematical Society, 2005.
- [192] M. Hellmann, A short proof of an equivalent form of the Schroeder– Bernstein theorem, American Mathematical Monthly **68** (1961), 770.
- [193] J. Henle, An Outline of Set Theory, Dover, 2007.
- [194] A. Henrich, *Henry Segerman: visualizing topology*, Mathematics Magazine **93** (2020), 301–305.
- [195] A. Henrich and L. Kauffman, *Unknotting unknots*, American Mathematical Monthly 121 (2014), 379–390.
- [196] C. Hermite, *Miscellanea* (The good old days?), American Mathematical Monthly **86** (1979), 512.
- [197] C. Hermite, Miscellanea, American Mathematical Monthly 89 (1982), 795.
- [198] C. Hermite, *Miscellanea*, American Mathematical Monthly **90** (1983), 501.
- [199] H. Herrlich, Axiom of Choice, Lecture Notes in Mathematics 1876, Springer-Verlag, 2006.
- [200] D. Higgs, *Iterating the derived set function*, American Mathematical Monthly **90** (1983), 693–697.
- [201] S. Hildebrand and R. Poe, *The separation axioms for invertible spaces*, American Mathematical Monthly **75** (1968), 391–392.
- [202] E. Hille, Mathematics and mathematicians from Abel to Zermelo, Mathematics Magazine **26** (1953), 127–146.
- [203] P. Hilton, A brief, subjective history of homology and homotopy theory in this century, Mathematics Magazine **61** (1988), 282–291.

[204] C. Himmelberg, On the product of quotient spaces, American Mathematical Monthly 72 (1965), 1103–1106.

- [205] C. Himmelberg, Products of closed maps, American Mathematical Monthly 74 (1967), 39–41.
- [206] N. Ho, J. Godzik, J. Jones, T. Mattman, and D. Sours, *Invisible knots and rainbow rings: knots not determined by their determinants*, Mathematics Magazine 93 (2020), 4–18.
- [207] J. Hocking, *Invert sets in polyhedra*, American Mathematical Monthly **75** (1968), 357–362.
- [208] J. Hocking and G. Young, Topology, 2nd edition, Dover, 1988.
- [209] K. Hoffman and J. Leitzel, Award for distinguished service to Dr. Lynn Arthur Steen, American Mathematical Monthly 99 (1992), 99–100.
- [210] D. Hong, Generalized invertible spaces, American Mathematical Monthly **73** (1966), 150–154.
- [211] D. Hong, Some theorems on generalized invertible spaces, American Mathematical Monthly **75** (1968), 378–382.
- [212] H. Howards, *Brunnian spheres*, American Mathematical Monthly **115** (2008), 114–124.
- [213] K. Hrbacek and T. Jech, Introduction to Set Theory, 3rd edition, Dekker, 1999.
- [214] S. Huggett and D. Jordan, A Topological Apertif, revised edition, with a foreward by R. Penrose, Springer-Verlag, 2009.
- [215] A. Insel, A note on the Hausdorff separation property in first countable spaces, American Mathematical Monthly 72 (1965). 289–290.
- [216] G. Isaacs, Real Numbers: A Development of the Real Numbers in an Axiomatic Set Theory, McGraw-Hill, 1968.
- [217] S. Jackson, Classroom notes: A development of the Jordan curve theorem and the Schoenflies theorem for polygons, American Mathematical Monthly 75 (1968), 989–998.
- [218] I. James, editor, History of Topology, North-Holland, 1999.
- [219] R. James, Combinatorial topology of surfaces, Mathematics Magazine 29 (1955), 1–39.
- [220] D. Janković, M. Ganster, and I. Reilly, A characterization of Baire spaces, American Mathematical Monthly 98 (1991), 624–625.

[221] D. Janković and T. Hamlett, New topologies from old via ideals, American Mathematical Monthly 97 (1990), 295–310.

- [222] T. Jech, The Axiom of Choice, North-Holland, 1973.
- [223] G. Jeffery, Miscellanea, American Mathematical Monthly 89 (1982), 640.
- [224] P. Johnson, Miscellanea, American Mathematical Monthly 85 (1978), 569.
- [225] R. Johnson, A compact non-metrizable space such that every closed subset is a G-delta, American Mathematical Monthly 77 (1970), 172–176.
- [226] R. Johnsonbaugh, Classroom notes: Two definitions of simple connectivity, American Mathematical Monthly 74 (1967), 117–1120.
- [227] R. Johnsonbaugh, Compact and connected spaces, Mathematics Magazine **50** (1977), 24-25.
- [228] P. Johnstone, *Notes on Logic and Set Theory*, Cambridge University Press, 1987.
- [229] D. Jordan and H. Porteous, A map of sources, sinks and saddles, American Mathematical Monthly **79** (1972), 587–596.
- [230] J. Joseph, Continuous functions and spaces in which compact sets are closed, American Mathematical Monthly **76** (1969), 1125–1126.
- [231] J. Joseph, On a characterization of compactness for T_1 spaces, American Mathematical Monthly 83 (1976), 728–729.
- [232] J. Joseph, Spaces in which compact sets are closed, Mathematics Magazine 49 (1976), 90.
- [233] A. Jurišić, *The Mercedes knot problem*, American Mathematical Monthly **103** (1996), 756–770.
- [234] M. Kac, Miscellanea, American Mathematical Monthly 90 (1983), 516.
- [235] D. Kahn, Topology: An Introduction to the Point-Set and Algebraic Areas, Dover, 1995.
- [236] I. Kaplansky, Set Theory and Metric Spaces, 2nd edition, Chelsea, 1977.
- [237] R. Kasriel, Undergraduate Topology, Dover, 2009.
- [238] L. Kauffman, New invariants in the theory of knots, American Mathematical Monthly 95 (1988), 195–242.
- [239] L. Kauffman, D. Prasad, and C. Zhu, Uncolorable Brunnian links are linked, Mathematics Magazine 95 (2022), 437–451.
- [240] J. Kelley, General Topology, Springer-Verlag, 1975.

[241] M. Kim, A compact graph theorem, Mathematics Magazine 47 (1974), 99.

- [242] R. Kirby, *The Topology of 4-Manifolds*, Lecture Notes in Mathematics **1374**, Springer-Verlag, 1989.
- [243] R. Kirby and L. Siebenmann, Foundational Essays on Topological Manifolds, Smoothings, and Triangulations, with notes by J. Milnor and M. Atiyah, Annals of Mathematics Studies 88, Princeton University Press, 1977.
- [244] M. Kirch, A class of spaces in which compact sets are finite, American Mathematical Monthly **76** (1969), 42.
- [245] J. Kline, What is the Jordan curve theorem?, American Mathematical Monthly 49 (1942), 281–286.
- [246] S. Kolay, Knot colorings: coloring and Goeritz matrices, American Methamtical Monthly 130 (2023), 335–354.
- [247] I. Kolodner, A simple proof of the Schröder-Bernstein theorem, American Mathematical Monthly **74** (1967), 995–996.
- [248] I. Kolodner, The compact graph theorem, American Mathematical Monthly **75** (1968), 167.
- [249] P. Komjáth and V. Totik, Problems and Theorems in Classical Set Theory, Springer, 2006.
- [250] H. Konig, Notes: A simple example of non-sequentialness in topological spaces, American Mathematical Monthly 100 (1993), 674–675.
- [251] S. Krantz, A Guide to Topology, Mathematical Association of America, 2009.
- [252] S. Krantz, Essentials of Topology and Applications, CRC Press, 2010.
- [253] V. Krishnamurthy, On the number of topologies on a finite set, American Mathematical Monthly 73 (1966), 154–157.
- [254] J.-L. Krivine, Introduction to Axiomatic Set Theory, translated from the French by D. Miller, Reidel, Humanities Press, 1971.
- [255] H. Kronk, A note on continuously near-homogeneous spaces, American Mathematical Monthly **75** (1968), 508–509.
- [256] K. Kuratowski, Introduction to Set Theory and Topology, containing a supplement, "Elements of algebraic topology", by R. Engelking, translated from the Polish by L. Boroń, PWN — Polish Scientific Publishers, Pergamon Press, 1977.

[257] K. Kuratowski and A. Mostowski, Set Theory: With an Introduction to Descriptive Set Theory, translated from the 1966 Polish original, second, completely revised, edition, North-Holland, PWN — Polish Scientific Publishers, 1976.

- [258] R. Laatsch, Classroom notes: Compact spaces and locally compact subspaces, American Mathematical Monthly 72 (1965), 537.
- [259] H. Lebesgue, Miscellanea, American Mathematical Monthly 89 (1982), 330.
- [260] J. Lee, Introduction to Topological Manifolds, 2nd edition, Springer, 2011.
- [261] N. Levine, A note on functions continuous almost everywhere, American Mathematical Monthly 66 (1959), 791–792.
- [262] N. Levine, Remarks on uniform continuity in metric spaces, American Mathematical Monthly 67 (1960), 562–563.
- [263] N. Levine, A note on convergence in topological spaces, American Mathematical Monthly 67 (1960), 667–668.
- [264] N. Levine, A decomposition of continuity in topological spaces, American Mathematical Monthly 68 (1961), 44-46.
- [265] N. Levine, On the commutativity of the closure and interior operators in topological spaces, American Mathematical Monthly 68 (1961), 474–477.
- [266] N. Levine, Semi-open sets and semi-continuity in topological spaces, American Mathematical Monthly **70** (1963), 36–41.
- [267] N. Levine, Some remarks on invertible spaces, American Mathematical Monthly 70 (1963), 181–183.
- [268] N. Levine, Largest subtopologies with special properties, American Mathematical Monthly 70 (1963), 284–285.
- [269] N. Levine, Inherited properties of functions in quotient spaces, American Mathematical Monthly **70** (1963), 969–973.
- [270] N. Levine, Simple extensions of topologies, American Mathematical Monthly 71 (1964), 22–25.
- [271] N. Levine, Spaces in which compact sets are never closed, American Mathematical Monthly **72** (1965), 635–638.
- [272] N. Levine, Strongly connected sets in topology, American Mathematical Monthly 72 (1965), 1098-1101.
- [273] N. Levine, On families of topologies for a set, American Mathematical Monthly 73 (1966), 358–361.

[274] N. Levine, On the equivalence of compactness and finiteness in topology, American Mathematical Monthly **75** (1968), 178–180.

- [275] N. Levine, Dense topologies, American Mathematical Monthly 75 (1968), 847–852.
- [276] A. Levy, Basic Set Theory, Dover, 2002.
- [277] J. Lewin, A simple proof of Zorn's lemma, American Mathematical Monthly 98 (1991), 353–354.
- [278] W. Lickorish and K. Millet, The new polynomial invariants of knots and links, Mathematics Magazine **61** (1988), 3–23.
- [279] E. Lima, The Jordan–Brouwer separation theorem for smooth hypersurfaces, American Mathematical Monthly 95 (1988), 39–42.
- [280] B. Lindström and H.-O. Zetterström, Borromean circles are impossible, American Mathematical Monthly 98 (1991), 340–341.
- [281] C. Livingston, Knot Theory, Mathematical Association of America, 1993.
- [282] C. Livingston, *Enhanced linking numbers*, American Mathematical Monthly **110** (2003), 361–385.
- [283] P. Loeb, A new proof of the Tychonoff theorem, American Mathematical Monthly **72** (1965), 711–717.
- [284] J. London, Miscellanea, American Mathematical Monthly 91 (1984), 403.
- [285] F. Lucas, Miscellanea, American Mathematical Monthly 91 (1984), 664.
- [286] J. Luukkainen, Extensions of spaces, maps, and metrics in Lipschitz topology, Annales Academiæ Scientiarum Fennicæ Series A I Mathematica Dissertationes 17 (1978).
- [287] J. Luukkainen and J. Väisälä, Elements of Lipschitz topology, Annales Academiæ Scientiarum Fennicæ Series A I Mathematica 3 (1977), 85– 122.
- [288] R. Mabry and L. McCormick, Branko's M2 is Brunnian, Mathematics Magazine 93 (2020), 323–329.
- [289] D. MacHale, *Notes: Any questions?*, American Mathematical Monthly **90** (1983), 42-43.
- [290] M. Machover, Set Theory, Logic, and their Limitations, Cambridge University Press, 1996.
- [291] P. Maddy, Defending the Axioms: On the Philosophical Foundations of Set Theory, Oxford University Press, 2011.

[292] H. Maehara, Why is P² not embeddable in R³?, American Mathematical Monthly 100 (1993), 862–864.

- [293] R. Maehara, The Jordan curve theorem via the Brouwer fixed point theorem, American Mathematical Monthly 91 (1984), 641–643.
- [294] K. Magill, Jr., Semigroups of continuous functions, American Mathematical Monthly 71 (1964), 984–988.
- [295] K. Magill, Jr., N-Point compactifications, American Mathematical Monthly 72 (1965), 1075–1081.
- [296] K. Magill, Jr., Binary operations on families of continuous functions, American Mathematical Monthly 82 (1975), 637–639.
- [297] K. Magill, Jr., Universal topological spaces, American Mathematical Monthly 95 (1988), 942–946; Errata, ibid., 97 (1990), 412.
- [298] J. Martino, Miscellanea, American Mathematical Monthly 90 (1983), 280.
- [299] W. Massey, A Basic Course in Algebraic Topology, Springer-Verlag, 1991.
- [300] E. Matheron, *Three proofs of Tychonoff's theorem*, American Mathematical Monthly **127** (2020), 437–443.
- [301] B. McAllister, Cyclic elements in topology, a history, American Mathematical Monthly **73** (1966), 337–350.
- [302] B. McAllister, Classroom notes: On the intersection of a nested sequence of compact sets, American Mathematical Monthly **74** (1967), 841–842.
- [303] G. McCarty, Topology: An Introduction, with Application to Topological Groups, 2nd edition, Dover, 1988.
- [304] R. McCoy, Second countable and separable function spaces, American Mathematical Monthly 85 (1978), 487–489.
- [305] P. McGrath, On the smooth Jordan Brouwer separation theorem, American Mathematical Monthly 123 (2016), 292–295.
- [306] E. McShane, Partial orderings and Moore–Smith limits, American Mathematical Monthly **59** (1952), 1-11.
- [307] (No author listed.) Award for distinguished service to Professor Edward James McShane, American Mathematical Monthly 71 (1964), 1–2.
- [308] P. McSwiggen and K. Meyer, *Conjugate phase portraits of linear systems*, American Mathematical Monthly **115** (2008), 596–614.
- [309] B. Mendelson, Introduction to Topology, 3rd edition, Dover, 1990.

[310] P. Meyer, A compactness condition weaker than T_2 , American Mathematical Monthly **75** (1968), 758–759.

- [311] P. Meyer, A short proof of the Urysohn metrization theorem, Mathematics Magazine 43 (1970), 268–269.
- [312] J. Milnor, Topology through the centuries: Low dimensional manifolds, Bulletin of the American Mathematical Society (N.S.) 52 (2015), 545– 584.
- [313] E. Moise, The use of induction in existence proofs, American Mathematical Monthly 61 (1954), 192–193.
- [314] E. Moise, Geometric Topology in Dimensions 2 and 3, Springer-Verlag, 1977.
- [315] E. Moise, Introductory Problem Courses in Analysis and Topology, Springer-Verlag, 1982.
- [316] D. Monk, Introduction to Set Theory, McGraw-Hill, 1969.
- [317] D. Montgomery, Remarks on groups of homeomorphisms, American Mathematical Monthly 48 (1941), 310–317.
- [318] G. Moore, Zermelo's Axiom of Choice: Its Origins, Development, and Influence, Springer-Verlag, 1982.
- [319] F. Morley, Questions and discussions: Discussions: A note on knots, American Mathematical Monthly **31** (1924), 237–239.
- [320] S. Morris, An elementary proof that the Hilbert cube is compact, American Mathematical Monthly **91** (1984), 563–564.
- [321] M. Morse, What is analysis in the large?, American Mathematical Monthly 49 (1942), 358–364.
- [322] M. Morse, *Topology and equilibria*, American Mathematical Monthly **114** (2007), 819–834.
- [323] Y. Moschovakis, Notes on Set Theory, 2nd edition, Springer, 2006.
- [324] F. Mynard, Distinguishing the plane from the punctured plane without homotopy, Mathematics Magazine **91** (2018), 16–19.
- [325] S. Nadler, Jr., Mathematical notes: A note on projections and compactness, American Mathematical Monthly 73 (1966), 275–276.
- [326] S. Naimpally, Function spaces of invertible spaces, American Mathematical Monthly **73** (1966), 513–515.

[327] Y. Nakandakari and S. Tsukuda, An elementary proof that the third finite subset space of the circle is the 3-sphere, American Mathematical Monthly 127 (2020), 789–805.

- [328] O. Nanyes, An elementary proof that the Borromean rings are nonsplittable, American Mathematical Monthly **100** (1993), 786–797.
- [329] P. Nanzetta and G. Strecker, Set Theory and Topology, Bogden & Quigley, 1971.
- [330] M. Newman, Elements of the Topology of Plane Sets of Points, reprint of the 2nd edition, Dover, 1992.
- [331] R. Nielsen and C. Sloyer, On embedding in quasi-cubes, American Mathematical Monthly **75** (1968), 514–515. See also [375].
- [332] F. Norwood, *Miscellanea* (concerning Dehn's lemma), American Mathematical Monthly **89** (1982), 309.
- [333] K. O'Brien, *Miscellanea: The Bernoullis*, American Mathematical Monthly **86** (1979), 482.
- [334] K. O'Brien, Miscellanea: Three haiku: What is mathematics?, American Mathematical Monthly 88 (1981), 626.
- [335] K. O'Brien, *Miscellanea: Bilateral convolution*, American Mathematical Monthly **93** (1986), 399.
- [336] M. Ozawa and S. Taylor, Two more proofs that the Kinoshita graph is knotted, American Mathematical Monthly 126 (2019), 352–357.
- [337] R. Parris, Classroom notes: A three-dimentionsal system with knotted trajectories, American Mathematical Monthly 84 (1977), 468–469.
- [338] T. Parsons, Ancestors, cardinals, and representatives, American Mathematical Monthly **93** (1986), 197–200.
- [339] R. Pederson, The Jordan curve theorem for piecewise smooth curves, American Mathematical Monthly **76** (1969), 605–610.
- [340] D. Peifer, An introduction to combinatorial group theory and the word problem, Mathematics Magazine **70** (1997), 3–10.
- [341] D. Peifer, Max Dehn and the origins of topology and infinite group theory, American Mathematical Monthly 122 (2015), 217–233.
- [342] C. Pinter, Set Theory, Addison-Wesley, 1971.
- [343] H. Poincaré, Miscellanea, American Mathematical Monthly 84 (1977), 749.

[344] H. Poincaré, *Miscellanea* ("The clouded crystal ball"), American Mathematical Monthly **85** (1978), 262.

- [345] H. Poincaré, Miscellanea, American Mathematical Monthly 87 (1980), 283
- [346] M. Potter, Set Theory and its Philosophy: A Critical Introduction, Oxford University Press, 2004.
- [347] R. Prather and R. Wilson, *Miscellanea: Our Favorite Things*, American Mathematical Monthly **90** (1983), 481.
- [348] W. Preece, Miscellanea, American Mathematical Monthly 91 (1984), 435.
- [349] W. Priestley, Classroom notes: Nets and sequences, an example, American Mathematical Monthly 75 (1968), 1098–1099.
- [350] T. Radó, On semi-continuity, American Mathematical Monthly 49 (1942), 446–450.
- [351] M. Raman-Sundström, A pedagogical history of compactness, American Mathematical Monthly 122 (2015), 619–635.
- [352] A. Ranicki, Algebraic L-Theory and Topological Manifolds, Cambridge University Press, 1992.
- [353] A. Ranicki, A. Casson, D. Sullivan, M. Armstrong, C. Rourke, and G. Cooke, *The Hauptvermutung Book*, A collection of papers of the topology of manifolds, Kluwer, 1996.
- [354] H. Reiter and A. Reiter, The space of closed subsets of a convergent sequence, Mathematics Magazine **69** (1996), 217–221.
- [355] P. Rice, Classroom notes: A topological characterization of the real numbers, American Mathematical Monthly **76** (1969), 184–185.
- [356] M. Roberts, Miscellanea, American Mathematical Monthly 90 (1983), 675.
- [357] L. Robertson and S. Franklin, O-Sequences and O-nets, American Mathematical Monthly **72** (1965), 506–510.
- [358] J. Roitman, Introduction to Modern Set Theory, Wiley, 1990.
- [359] I. Rosenholtz, Another proof that any compact metric space is the continuous image of the Cantor set, American Mathematical Monthly 83 (1976), 646–647.
- [360] K. Ross, Yueh-Gin Gung and Dr. Charles Y. Hu Award for distinguished service to Leonard Gillman, American Mathematical Monthly 106 (1999), 97–98.

[361] K. Ross and A. Stone, *Products of separable spaces*, American Mathematical Monthly **71** (1964), 398–403.

- [362] B. Rotman and G. Kneebone, *The Theory of Sets and Transfinite Numbers*, Oldbourne, 1966.
- [363] J. Rubin, Set Theory for the Mathematician, Holden-Day, 1967.
- [364] J. Rubin, Finite sets, Mathematics Magazine 46 (1973), 183-192.
- [365] M. E. Rudin, A subset of the countable ordinals, American Mathematical Monthly 64 (1957), 351.
- [366] M. E. Rudin, Souslin's conjecture, American Mathematical Monthly 76 (1969), 1113–1119.
- [367] M. E. Rudin, Lectures on Set Theoretic Topology, American Mathematical Society, 1975.
- [368] Y. Rudyak, Piecewise Linear Structures on Topological Manifolds, World Scientific, 2016.
- [369] T. Rushing, On countably compact nonlocally compact spaces, American Mathematical Monthly **74** (1967), 280–283.
- [370] T. Rushing, Topological Embeddings, Academic Press, 1973.
- [371] B. Russell and A. Whitehead, *Miscellanea: Russell and Whitehead*, American Mathematical Monthly **90** (1983), 220.
- [372] N. Rutt, On derived sets, National Mathematics Magazine 18 (1943), 53–63.
- [373] D. Ryeburn, Finite additivity, invertibility, and simple extensions of topologies, American Mathematical Monthly 74 (1967), 148–152.
- [374] J. Sack, *Miscellanea* ("Bolzano-Weierstrass for ornithologists", suggested by H. Boas), American Mathematical Monthly **90** (1983), 244.
- [375] S. Salbany amd G. Brümmer, Pathology of upper Stone-Čech compactifications, American Mathematical Monthly 78 (1971), 187–188.
- [376] H. Samelson, *Progress reports: Hauptvermutung*, American Mathematical Monthly **85** (1978), 567–569.
- [377] D. Sanderson, Advanced plane topology from an elementary standpoint, Mathematics Magazine **53** (1980), 81–89.
- [378] D. Sanderson and B. Sims, A characterization and generalization of semimetrizability, American Mathematical Monthly **73** (1966), 361–365.

[379] E. Schimmerling, A Course on Set Theory, Cambridge University Press, 2011.

- [380] R. Schindler, Set Theory: Exploring Independence and Truth, Springer, 2014.
- [381] P. Schnare, Classroom notes: Two definitions of local compactness, American Mathematical Monthly 72 (1965), 764–765.
- [382] P. Schnare, The maximal T₀ (respectively, T₁) subspace lemma is equivalent to the axiom of choice, American Mathematical Monthly 75 (1968), 7611.
- [383] R. Schultz, Some recent results on topological manifolds, American Mathematical Monthly 78 (1971), 941–952.
- [384] B. Scott, A "more topological" proof of the Tietze Urysohn theorem, American Mathematical Monthly 85 (1978), 192–193.
- [385] B. Scott and Z. Robinson, *The boundary topology of a space*, American Mathematical Monthly **89** (1982), 307–309.
- [386] H. Segerman, Visualizing Mathematics with 3D Printing, Johns Hopkins University Press, 2016.
- [387] H. Sharp, Denseness and completeness in certain function spaces, American Mathematical Monthly **74** (1967), 266–271.
- [388] D. Shimamoto and C. Vandervaart, Spaces of polygons in the plane and Morse theory, American Mathematical Monthly 112 (2005), 289–310.
- [389] L. Siebenmann and D. Sullivan, One complexes that are Lipschitz manifolds, in Geometric Topology (Proceedings of the Georgia Topology Conference, Athens, Georgia, 1977), 503–525, Academic Press, 1979.
- [390] W. Sierpinski, *General Topology*, translated by C. Krieger, University of Toronto Press, 1952.
- [391] W. Sierpiński, Hypothèse du Continu, 2nd edition, Chelsea, 1956.
- [392] W. Sierpinski, *Introduction to the Theory of Sets and Topology*, translated from a French manuscript by A. Sharma, The Indian Press, 1962.
- [393] W. Sierpiński, Cardinal and Ordinal Numbers, second revised edition, Państwowe Wydawnictwo Naukowe (PWN), 1965.
- [394] L. Sigler, Exercises in Set Theory, Springer-Verlag, 1976.
- [395] S. Smale, What is global analysis?, American Mathematical Monthly **76** (1969), 4-9.

[396] M. Smiley, Filters and equivalent nets, American Mathematical Monthly **64** (1957), 336–338.

- [397] A. Smith, *Miscellanea: The value of research!*, American Mathematical Monthly **93** (1986), 821.
- [398] D. Smith, Miscellanea, American Mathematical Monthly 91 (1984), 514.
- [399] R. Smithson, A fixed point theorem for connected multi-valued functions, American Mathematical Monthly **73** (1966), 351–355.
- [400] R. Smithson, A characterization of lower semi-continuity, American Mathematical Monthly **75** (1968), 505.
- [401] R. Smithson, A counterexample to a lemma on the existence of simple refinements, American Mathematical Monthly 77 (1970), 293–294.
- [402] R. Smullyan, Miscellanea, American Mathematical Monthly 90 (1983), 390.
- [403] R. Smullyan and M. Fitting, Set Theory and the Continuum Problem, Oxford University Press, 1996.
- [404] R. Spira, An open discontinuous function, American Mathematical Monthly **69** (1962), 128–129.
- [405] R. St. Andre, A classification of ultrafilters, American Mathematical Monthly 78 (1971), 1126–1127.
- [406] M. Starbird and F. Su, Topology through Inquiry, AMS / MAA Textbooks 58, MAA Press, 2019.
- [407] L. Steen, Conjectures and counterexamples in metrization theory, American Mathematical Monthly **79** (1972), 113-132.
- [408] L. Steen and J. Seebach, Jr., Counterexamples in Topology, 2nd edition, Dover, 1995.
- [409] Y. Sternfeld, *Homeomorphisms of cubes into Euclidean spaces*, American Mathematical Monthly 88 (1981), 436–439.
- [410] J. Stillwell, *The continuum problem*, American Mathematical Monthly **109** (2002), 286–297.
- [411] R. Stoll, Set Theory and Logic, Dover, 1979.
- [412] L. Stout, Two discrete forms of the Jordan curve theorem, American Mathematical Monthly 95 (1988), 332–336.
- [413] P. Strzelecki, *The Poincaré conjecture?*, translated from DELTA 2004, no. 1, 1–3, by A. Shenitzer, American Mathematical Monthly **113** (2006), 75–78.

[414] D. Sullivan, Hyperbolic geometry and homeomorphisms, in Geometric Topology (Proceedings of the Georgia Topology Conference, Athens, Georgia, 1977), 543–555, Academic Press, 1979.

- [415] M. Sullivan, Knot factoring, American Mathematical Monthly 107 (2000), 297–315. This is related to [186].
- [416] P. Suppes. Axiomatic Set Theory, Dover, 1972.
- [417] L. Taalman, Visualizing Mathematics with 3D Printing by Henry Segerman and Pasta by Design by George L. Legendre, American Mathematical Monthly 125 (2018), 379–384.
- [418] G. Takeuti and W. Zaring, *Introduction to Axiomatic Set Theory*, 2nd edition, Springer-Verlag, 1982.
- [419] A. Tarski, *Miscellanea: Definition*, American Mathematical Monthly **92** (1985), 358.
- [420] K. Teixeira, On the topology of punctured spaces, American Mathematical Monthly 130 (2023), 182–185.
- [421] R. Thom, Structural Stability and Morphogenesis: An Outline of a General Theory of Models, translated from the French by D. Fowler, with a foreward by C. Waddington, Addison-Wesley, 1989.
- [422] J. Thomas, Classroom notes: A regular space, not completely regular, American Mathematical Monthly **76** (1969), 181–182.
- [423] J. P. Thomas, Maximal non- T_i spaces, i = 0, 1, 2, 3, American Mathematical Monthly 74 (1967), 169–171.
- [424] J. P. Thomas, The long line as a subset of $\mathcal{P}(\mathbf{R})$, American Mathematical Monthly **76** (1969), 675–677.
- [425] C. Thomassen, A link between the Jordan curve theorem and the Kuratowski planarity criterion, American Mathematical Monthly 97 (1990), 216–218.
- [426] C. Thomassen, The Jordan-Schönflies theorem and the classification of surfaces, American Mathematical Monthly 99 (1992), 116–130.
- [427] M. Tiles, The Philosophy of Set Theory: An Historical Introduction to Cantor's Paradise, Dover, 2004.
- [428] G. Tourlakis, Lectures in Logic and Set Theory, Volume 2, Set Theory, Cambridge University Press, 2003.
- [429] J. Tucker, Concerning consecutive derived sets, American Mathematical Monthly **74** (1967), 555–556.

[430] P. Tukia and J. Väisälä, Quasisymmetric embeddings of metric spaces, Annales Academiæ Scientiarum Fennicæ Series A I Mathematica 5 (1980), 97–114.

- [431] R. Vaidyanathaswamy, Set Topology, 2nd edition, Dover, 1999.
- [432] J. Väisälä, Quasisymmetric embeddings in Euclidean spaces, Transactions of the American Mathematical Society **264** (1981), 191–204.
- [433] R. Vakil, *The mathematics of doodling*, American Mathematical Monthly **118** (2011), 116–129.
- [434] O. Varsavsky, *The reciprocal iterated limit theorem*, American Mathematical Monthly **68** (1961), 356.
- [435] H. Vaughan, On two theorems of plane topology, American Mathematical Monthly **60** (1953), 462–468.
- [436] R. Vaught, Set Theory: An Introduction, 2nd edition, Birkhäuser, 1995.
- [437] Webster's Collegiate Dictionary (3rd edition, 1926), *Miscellania*, American Mathematical Monthly 88 (1981), 563.
- [438] J. Weeks, The Shape of Space, 2nd edition, Dekker, 2002.
- [439] H. Weyl, Miscellanea, American Mathematical Monthly 85 (1978), 127.
- [440] G. Whyburn, What is a curve?, American Mathematical Monthly 49 (1942), 493–497.
- [441] G. Whyburn, Sense and orientation on the disk, American Mathematical Monthly **64** (1957), no. 8, part II, 103–106.
- [442] G. Whyburn, Topological Analysis, 2nd revised edition, Princeton University Press, 1964.
- [443] G. Whyburn, *Dynamic topology*, American Mathematical Monthly **77** (1970), 556–570.
- [444] G. Whyburn and E. Duda, *Dynamic Topology*, with a foreword by J. Kelley, Springer-Verlag, 1979.
- [445] A. Wilansky, Between T_1 and T_2 , American Mathematical Monthly **74** (1967), 261–266. See also Brief versions: Editorial note on "Between T_1 and T_2 ", ibid., **74** (1967), no. 10, 1241.
- [446] A. Wilansky, Life without T_2 , American Mathematical Monthly 77 (1970, 157–161; correction, ibid., 77 (1970), 728.
- [447] R. Wilder, Some Unsolved Problems of Topology, American Mathematical Monthly 44 (1937), no. 2, 61–70.

[448] R. Wilder, The nature of mathematical proof, American Mathematical Monthly 51 (1944), 309–323. Some excepts appear in Comments, ibid., 100 (1993), 434.

- [449] R. Wilder, The role of the axiomatic method, American Mathematical Monthly 74 (1967), 115–127.
- [450] R. Wilder, Evolution of the topological concept of "connected", American Mathematical Monthly 85 (1978), 720–726; correction and addendum, ibid., 87 (1980), 31–32.
- [451] R. Wilder, Topology of Manifolds, reprint of the 1963 edition, American Mathematical Society, 1979.
- [452] S. Willard, General Topology, Dover, 2004.
- [453] H. Wilson, A canonical form for linear transformations of E_n under non-linear substitutions, American Mathematical Monthly **71** (1964), 988–993.
- [454] H. Wilson, *Positive limit sets of unbounded trajectories*, American Mathematical Monthy **78** (1971), 155–164.
- [455] H. Wiser, Near-homogeneity on 2-manifolds, American Mathematical Monthly 74 (1967), 423–426.
- [456] E. Wolk, On the inadequacy of subnets and transfinite sequences, American Mathematical Monthly 89 (1982), 310–311.
- [457] Y.-M. Wong, Some remarks on invertible spaces, American Mathematical Monthly 73 (1966), 835–841.
- [458] C. Yang, Miscellanea, American Mathematical Monthly 90 (1983), 668.
- [459] G. Young, A condition for the absolute homotopy extension property, American Mathematical Monthly **71** (1964), 896–897.
- [460] W. Young, Miscellanea, American Mathematical Monthly 87 (1980), 182.

Index

$2^A, 30$	completely normal topological spaces,
absolute value, 2 adherent point, 7 Alexander's subbase theorem, 70 algebraic topology, iii axiom of choice, 29	completely regular in the strict sense, 97 completely regular in the strong sense, 98 completely regular topological spaces,
B(x,r), 3 B^A , 30 $\mathcal{B}_E(X,Y)$, 119 Baire's theorem, 101 bases for filters, 138 bases for topologies, 33 bijections, 26 bounded mappings, 118 bounded sets, 118	98 completely separable spaces, 35 compositions of mappings, 26 condensation points, 32 connected sets, 20 connected components, 106 continuous mappings, 36, 37 convergent filters, 137 convergent nets, 13
C(X), 94 c(X), 132 C(X,Y), 109, 119 $c_{00}(X)$, 132 $C_b(X,Y)$, 120 Cartesian products, 29, 41	convergent sequences, 9 convex sets, 110 coordinate mappings, 43 countably compact sets, 63 countably infinite sets, 27 coverings of sets, 146
chains, 80 closed families, 152 closed sets, 6 closure, 7 compact sets, 54 compact-open topology, 127 on $C(X,Y)$, 128	dense sets, 5 directed systems, 13 discrete families, 160 discrete metric, 2 discrete topology, 1 distance to a set, 112
compactness with respect to \mathcal{B} , 68 complement of a set, 6 completely Hausdorff, 18 completely normal in the strict sense, 21 completely normal in the strong sense,	equivalence classes, 103 equivalence relations, 103 extended real numbers, 4 extreme value theorem, 58 F_{σ} sets, 110
21	fifth separation condition, 21

INDEX 197

filter bases, 137 filters, 137 finite intersection property, 66 finite support, 132 first category sets, 111 first countability condition, 12 first separation condition, 2 fourth separation condition, 20 fully normal spaces, 154, 155

 G_{δ} sets, 110

Hausdorff topological spaces, 2 Hausdorff's maximality principle, 81 homeomorphisms, 39 homotopies, 109 with endpoints fixed, 109

ideals in partially-ordered sets, 87 identification topology, 104 indiscrete topology, 1 induced topology, 8 injections, 26 interior, 7 intervals, 4

Lebesgue's theorem, 148 limit points, 7 limit point property, 59 Lindelöf property, 64 Lindelöf's theorem, 64 linear orderings, 13 local bases, 10 local connectedness, 104, 105 local finiteness, 113 local path connectedness, 105 locally compact topological spaces, 99 locally constant mappings, 38 locally Euclidean spaces, 100

manifolds with boundary, 101 mappings, 26 maximal elements, 80 meager sets, 111 metacompactness, 164 metrics, 3

nets, 13 non-meager sets, 111 normal in the strong sense, 20 normal in the strict sense, 20 normal topological spaces, 20 nowhere dense sets, 111

one-point compactifications, 107 one-to-one correspondences, 26 one-to-one mappings, 26 open balls, 3 open coverings, 54 open mappings, 39 open sets, 1, 3 order automorphism, 90 order isomorphism, 90

 $\mathcal{P}(A)$, 27 paracompactness, 149 partial orderings, 12 partitions, 103 partitions of unity, 116, 167 path connectedness, 40 path-connected components, 106 perfectly normal spaces, 111 point at infinity, 107 point finiteness, 164 pointwise convergence, 120 power sets, 27 pre-filters, 137 pre-orderings, 13 product topology, 41 pseudometrics, 3

Q, 5 quotient mappings, 103 quotient sets, 103 quotient topology, 104

R, 2 refinements of families, 146 refinements of filters, 140 reflexivity, 12 regular in the strict sense, 15 regular in the strong sense, 15

regular topological spaces, 16

198 INDEX

relatively open sets, 7

second category sets, 111 second countability condition, 34 second separation condition, 2 segments in partially ordered sets, 87 semimetrics, 3 separable topological spaces, 33 separated sets, 19 separating points, 95 separation condition $2\frac{1}{2}$, 18 separation condition $3\frac{1}{2}$, 98 sequential compactness, 61 sequential continuity, 63 sequentially closed sets, 62 σ -compactness, 102 σ -discreteness, 161 σ -local finiteness, 157 standard metric on \mathbf{R} , 2 standard topology on \mathbf{R} , 2 standard topology on \mathbb{R}^n , 42 star of a point, 153 star of a set, 155 star refinements, 153 strong limit point property, 59 strong limit points, 32 strong product topology, 42 stronger topologies, 5 subbase for a topology, 70 subcoverings of sets, 146 subsequences, 61 supports of functions, 131 supremum semimetrics, 119 surjections, 26

 T_0 spaces, 2 T_1 spaces, 2 $T_{2\frac{1}{2}}$ spaces, 18 T_2 spaces, 2 $T_{3\frac{1}{2}}$ spaces, 98 T_3 spaces, 16 T_4 spaces, 20, 111, 155 T_5 spaces, 21 τ_+ , 5 τ_- , 5 third separation condition, 16 Tietze's extension theorem, 124
top elements, 80
topological algebra, iii
topological manifolds, 101
topological spaces, 1
topology, 1
topology of pointwise convergence, 120
topology of uniform convergence, 121
on compact sets, 126
total orderings, 13
transitivity, 13
triangle inequality, 3
Tychonoff spaces, 98
Tychonoff's result, 78
Tychonoff's theorem, 70, 73, 143

ultrafilters, 141 uncountable sets, 27 uniform convergence, 120 upper bounds, 80 Urysohn spaces, 95 Urysohn functions, 96 Urysohn's lemma, 96, 113, 132 Urysohn's metrization theorem, 97

Weierstrass' criterion, 123 well-ordered sets, 86 well-ordering principle, 87

 $\mathbf{Z}_{+},\,13$ zeroth separation condition, 2 Zorn's lemma, 81